

Legislative Research Commission

# Compendium of State Education Rankings 2008

Research Report No. 362

Prepared by

Marcia Ford Seiler, Director; Brenda Landy; Ken Chilton, Ph.D.; Al Alexander; Deborah Nelson, Ph.D.; Sabrina Olds; and Pam Young

# Compendium of State Education Rankings 2008

# **Project Staff**

Marcia Ford Seiler, Director Brenda Landy Ken Chilton, Ph.D. Al Alexander Deborah Nelson, Ph.D. Sabrina Olds Pam Young

# **Research Report No. 362**

Legislative Research Commission Frankfort, Kentucky lrc.ky.gov

Accepted August 12, 2008, by Education Assessment and Accountability Review Subcommittee

Paid for with state funds. Available in alternative form by request.

# Foreword

In November 2007, the Education Assessment and Accountability Review Subcommittee approved this second annual edition of the *Compendium of State Education Rankings* as part of the 2008 research agenda for the Office of Education Accountability.

This publication is intended to offer legislators and the public a convenient source of information about how Kentucky compares to other states on a wide variety of public elementary and secondary education indicators. Included are ranking tables, information about the authors and data sources, and discussions of data limitations and other issues intended to enhance readers' use of the report. Future compendiums will be updated and issued annually.

Robert Sherman Director

Legislative Research Commission Frankfort, Kentucky August 12, 2008

# Contents

Summary		xi
Chapter 1:	Introduction to the 2008 Compendium	1
- T	Changes From the 2007 Compendium	1
	Peer States	1
	Trends	2
	Organization of This Compendium	2
	How To Use This Compendium	3
	Rankings in Context	5 A
	Magnitude of Differences	<del>-</del> 1
	Palative Versus Absolute Levels of Quality	4 1
	Zero sum Geme Among States	4
	Drawing Delicy Implications	4
	Different Indicators Address Different Dalies Overtiges	3 5
	Unintended and Unenticipated Concentrations	
	Unintended and Unanticipated Consequences	
	Simpson's Paradox	5
	Steady Gains on Multiple Measures	6
	Data Comparability	6
	Terms Used in This Compendium	7
	Rank	7
	Fiscal and School Years	7
	States and United States Totals	7
	Data Not Available or Inapplicable	7
	Conclusion	7
Chapter 2.	Education Outcomes	9
Chapter 2.	Graduation Rates	9
	Data Quality Issues	ر ۵
	Challenges of Tracking High School Students	10
	Pacammandad Graduatian Formula Using Longitudinal Data	12
	KECONINENDED Classification Pote	.12
	Exemples Independent of State Education Systems	.15
	Formulas independent of State Education Systems	.13
	NCES Averaged Freshman Graduation Rate	.14
	Education Week Cumulative Promotion Index	.10
	State-reported Kates	.19
	Policy Implications of Graduation Rate Calculations	.20
	Dropout Rates	.20
	Data Quality Issues	.20
	Annual Dropout Rate	.21
	Status Dropout Rate	.22
	Achievement Assessments	.23
	National Assessment of Educational Progress	.24
	Reading	.25
	Writing	.29

	Mathematics	31
	Science	34
	NAEP Achievement Gaps	
	Sampling Error	
	Achievement Levels Still Used on Trial Basis	
	Exclusions and Accommodations	
	Comparing NAEP to State Assessments	43
	Students Can Perform Better on State Tests Than on NAEP	
	State Tests Align With Students' Opportunities To Learn	44
	High Stakes on State Tests Can Boost Performance	45
	Performance Standards for NAEP Appear To Be Higher	
	Than for State Tests	46
	Research Suggests NAEP Standards Too High	46
	Many Deem State Standards Too Low	46
	Comparing NAEP to State Tests: Conclusions	47
	College-readiness Exams	47
	The Higher the Participation, the Lower the Scores	47
	Test Design	48
	ACT	48
	SAT	49
	Advanced Placement	51
	Education Week's Quality Counts Achievement Index	53
	Chapter Summary and Conclusions	55
Chanter 3	Characteristics of Students and Their Families and Communities	57
Chapter 5	Enrollment	
	K-12 Enrollment Trends	58
	Preschool Enrollment	
	Racial and Ethnic Composition of Students	
	Incomes	
	Family Income	
	Child Poverty Rates	63
	Caveats and Limitations	
	NCES Comparable Wage Index	
	Student Participation in Selected Programs	
	Programs for Disadvantaged Students and Their Schools	
	Fnolish I anouage Learners	
	Disabilities	
	Annie E. Casev Child Well-being Index and Measures	
	Education Week's Quality Counts Chance-for-Success Index	
	Caveats and Limitations	
	Chapter Summary and Conclusions	73
	- · · · · · · · · · · · · · · · · · · ·	

Office of Education Accountability	
------------------------------------	--

Chapter 4	: Characteristics of Staff, Schools, and Districts	75
-	Student-Teacher Ratio	75
	Teachers as a Percentage of Staff	
	Types of Staff	77
	Education Week's Quality Counts Teaching Profession Index	
	Accountability for Quality	
	Incentives and Teacher Allocation	
	Building and Supporting Teacher Capacity	
	Preschools	
	Title I Schools	
	Technology	
	About Education Week's Technology Counts	
	Access to Technology	
	Use of Technology	
	Capacity To Use Technology	
	Caveats and Limitations	
	Uncertain Benefits To Student Learning	
	Data Comparability	
	Urban, Suburban, and Rural Locations	
	School Crime and Safety	
	Teacher-reported Incidents	
	Student-reported Incidents	
	Chapter Summary and Conclusions	95
Chapter 5	: Finance	97
	Revenues	97
	Revenues Per Pupil	97
	Revenue Sources	
	Expenditures	
	Current Spending Per Pupil	
	Preschool Spending	
	Current Spending Per \$1,000 in Personal Income	
	Spending on Instruction and Other Functions	
	Support Services	
	All Other Functions	
	Instruction	
	Teacher Salaries	
	Funding Equity	111
	Education Week's Equity/Disparity Measures	111
	Wealth-neutrality Score	
	McLoone Index	
	Coefficient of Variation	
	Restricted Range	
	Education Trust's Poverty and Minority Gap Measures	114
	Education Week's Overall School Finance Index	

Chapter 6: Indices Composed of Multiple Topics	121
Quality Counts Overall Grade and Score	121
Kentucky Long-Term Policy Research Center Indices	123
Obstacles to Cost-effective Educational Spending Index	123
Index of NAEP Proficiency Purchasing Power Relative to Obstacles	123
State of the Commonwealth Index	124
Measuring Up	126
Camelot Index	129
Healthy Economy	129
Healthy People	129
Crime-free State	129
Educated Population	129
Healthy Society	130
Prudently Managed Government	130
Smartest State Index	131
Chapter Summary and Conclusions	132
Works Cited	133
Appendix A: Rankings for All States	143
Appendix B: Information About Data Sources	201
Appendix C: State Abbreviations and Names	207

#### List of Tables

Appendix D: National Assessment of Educational Progress Achievement Level Definitions ... 209

1
6
10
15
17
18
19
21
22
23
25
26
27
29
30
31
11122 2222233

2.15	Students At or Above Proficient on NAEP Math Assessment, 2007	32
2.16	Framework for NAEP Science Assessment, 2005	34
2.17	Students At or Above Proficient on NAEP Science Assessment, 2005	35
2.18	Percentages of Students in NAEP Sample Excluded From Assessment, Kentucky and U.S., 1990-2007	41
2.19	Increases and Decreases in Average NAEP Scores From 2003 to 2005	
	With and Without Exclusions.	
2.20	ACT Test Items by Content Area	48
2.21	ACT Participation Rates and Average Scores for High School Graduates Tested, 2007	<u>4</u> 9
2.22	Percentage of High School Graduates Participating in SAT and Average SAT Scores 2007	50
2 23	Advanced Placement Exam 2000 and 2007	
2.23	Education Wook's Quality Counts Achievement Index 2008	52
2.24	Education Week's Quality Counts Achievement Index, 2000	
2.23	Summary, 2008.	54
3.1	Projected Percent Changes in Public P-12 Enrollment, Fall 2000-Fall 2016	58
3.2	Enrollment of 3- and 4-Year-Olds in State-funded Preschool, 2007	60
3.3	Racial and Ethnic Composition of Students, FY 2006	61
3.4	Median Family Income in Nominal Dollars, 1989, 1999, and 2006	62
3.5	Children Living Below the Federal Poverty Line, 1989, 1999, and 2006	63
3.6	NCES Comparable Wage Index, 2005	65
3.7	Students in Title I Schools or Eligible for National School Lunch Program,	
20	FY 2000	00
5.0 2.0	Students Receiving English Language Learner Services, FY 2000	0/
5.9 2.10	Students with individualized Education Plograms, FY 2000	08
5.10 2.11	Overall Child Well being 2007 KIDS COUNT Child Well-being Index, 2007	09
5.11 2.12	Education Weal's Ouglity Counts Change for Success Index Components	70
5.12	Summary 2008	71
3 1 3	Education Week's Quality Counts Chance-for-Success Index	
4 1	Student-Teacher Ratio Total and by Grade Level FY 2006	72
4.2	Teachers as a Percentage of All Staff Fall 1985 1995 and 2005	
43	Full-time Equivalent Staff Members Per 1 000 Students FY 2006	78
4 4	Teachers With Advanced Degrees FY 2004	80
4 5	<i>Quality Counts</i> Teaching Profession Index 2008	81
4.6	Quality Counts Teacher Accountability Measures 2008	82
47	<i>Quality Counts</i> Teacher Incentives and Allocation Measures 2008	83
4 8	<i>Quality Counts</i> Building and Supporting Teacher Capacity 2008	
49	Title I-Eligible Schools and Schoolwide Programs as a Percentage	
•••	of All Schools FY 2006	86
4 10	Education Technology Overall Grade Access Use and Capacity 2008	
4 11	Access to Technology 2006 and 2007	88
4 12	Use of Technology FY 2008	90
4.13	Capacity To Use Technology FY 2008	
4.14	Students in Rural Areas, FY 2006	
	,	-

4.15	Public School Teachers Threatened With Injury by a Student at School,	
	FY 1994 and FY 2004	94
4.16	Public School Students Involved in Violence or Substance Abuse,	
	Average of 2003 and 2005	95
5.1	Revenues Per Pupil, FY 2006	98
5.2	Revenues by Source, FY 2006	99
5.3	Current Spending Per Pupil in P-12 Fall Enrollment, FY 2006	100
5.4	Current Spending Per \$1,000 in Personal Income, FY 2006	103
5.5	Spending on Instruction, Support Services, and Other Functions as Percentages	
	of Current Spending, FY 2006	107
5.6	Spending on Instruction and Instruction-related, Student Support Services,	
	Administration, and Operations Functions as Percentages of	
	Current Spending, FY 2006	108
5.7	Average Public School Teacher Salaries, Unadjusted and Adjusted for	
	Geographic Cost Differences, FY 2006	109
5.8	Education Week's Equity/Disparity Measures, FY 2005	112
5.9	Education Trust Poverty and Minority Funding Equity and Gaps, 2005	115
5.10	Poverty Funding Gap Trends, 1999 to 2005	116
5.11	Minority Funding Gap Trends, 1999 to 2005	117
5.12	Components of Education Week's School Finance Index, 2008	118
5.13	Education Week's Overall School Finance Index, 2008	119
6.1	Education Week's Quality Counts Overall Grade	122
6.2	Components of the Obstacles to Cost-effective Educational	
	Spending Index, 2008	123
6.3	Index of NAEP Proficiency Purchasing Power Relative to Obstacles, 2008	124
6.4	Components of the State of the Commonwealth Index, 2006	125
6.5	Measuring Up Grades and Scores, 2006	126
6.6	Summary of Indicators Used in Measuring Up	127
6.7	Camelot Index, 2008	130
6.8	Smartest State Index, FY 2003-FY 2007	132

# **List of Figures**

2.A	Outcomes for a Hypothetical High School Class	11
2.B	Graduation Rate Formula Preferred by U.S. Department of Education Task Force	
	Using Longitudinal Individual Student Data	12
2.C	Kentucky's Departure Classification Graduation Rate Formula	13
2.D	Averaged Freshman Graduation Rate Formula	14
2.E	NCES Averaged Freshman Graduation Rate Trends, FY 1991-FY 2005	16
2.F	Formula for the Cumulative Promotion Index	17
2.G	NAEP Reading, Percent Proficient, U.S. and Kentucky, 1992 to 2007	28
2.H	NAEP Grade 8 Writing, Percent Proficient, U.S. and Kentucky, 1998, 2002,	
	and 2007	30
2.I	NAEP Math, Percent Proficient, U.S. and Kentucky, 1990 to 2007	33
2.J	NAEP Achievement Gaps by Gender, Most Recent Assessments	36
2.K	NAEP Achievement Gaps by Race, Most Recent Assessments	37

2.L 2 M	NAEP Poverty Gaps Percentages of Students in NAEP Sample Identified and Excluded or Assessed	38
2.111	With or Without Accommodations on the Basis of Disability or English	
	Language Learner, Kentucky and U.S., 2007	42
2.N	Percentages of Students in NAEP Sample Identified and Excluded or Assessed	
	With or Without Accommodations on the Basis of Disability,	
	Kentucky and U.S., 2007	42
3.A	Total P-12 Student Enrollment (in thousands), FY 2008	57
3.B	Enrollment of 3-Year-Olds in State-funded Preschool, 2002 to 2007	59
3.C	Enrollment of 4-Year-Olds in State-funded Preschool, 2002 to 2007	59
4.A	Pre-Kindergarten Quality Standards, FY 2007	85
4.B	Number of Students Per Instructional Computer, Kentucky and U.S., 1999-2006	89
5.A	Current Spending Per Pupil in P-12 Average Daily Attendance, 1960-2005	01
5.B	State Spending Per Child Enrolled, FY 2002-FY 20071	02
5.C	Teacher Salary Trends, U.S. and Kentucky, 1970-2005, Nominal Dollars1	10
5.D	Teacher Salary Trends, U.S. and Kentucky, 1970-2005, Constant 2006 Dollars	
	(Inflation Adjusted)1	10

# Summary

The second annual compendium of public elementary and secondary education data offers legislators and the public a convenient source of information about how Kentucky's education system compares to those in other states. Included are rankings, historical trends, information about data sources, and discussions of data limitations and other issues intended to enhance readers' use of the report.

In the spirit of continuous improvement, several changes were made to this year's compendium to make it more user friendly. While tables in the first compendium spanned two pages in order to rank all states, this year's more compact tables focus on Southern Regional Education Board and bordering states. Rankings for all 50 states and the District of Columbia are provided in Appendix A. In addition, this year's compendium examines more trends and organizes all information by the following topics:

- Chapter 1: Introduction
- Chapter 2: Outcomes (graduation rates and achievement and college-readiness tests)
- Chapter 3: Student, family, and community characteristics
- Chapter 4: School and staff characteristics
- Chapter 5: Finances (revenues, expenditures, and equity)
- Chapter 6: Multitopic indices that combine a wide variety of measures

As in the first compendium, this second edition discusses some data limitations that are helpful to bear in mind when interpreting the information presented. Some are summarized below.

- Rankings are often quoted out of context; it is essential to examine the measures on which rankings are based and the context, such as state differences in poverty, urbanization, and funding.
- A rank, as an isolated number, does not indicate how far apart states are from each other.
- Ranks indicate whether a state is better or worse than Kentucky but not whether this performance is good or bad in absolute terms.
- Rankings treat educational progress as a zero-sum game. A state's movement upward in rankings comes at the expense of other states, even if those states are also improving.
- Improvements in one area can have unintended and unanticipated consequences in other areas. For example, when dropout prevention programs are successful, achievement scores and college-going rates may stagnate or decline.
- This compendium contains multiple indicators of how Kentucky compares to other states. Rather than choosing one measure to gauge Kentucky's success or failure, readers are encouraged to weigh all of the available evidence.

## Highlights

**Graduation Rates.** Policy makers' efforts to monitor high school completion are complicated by data quality issues and competing formulas. Nevertheless, across multiple measures, Kentucky's graduation rate recently rose slightly above the national rate, after hovering slightly below for several years. As in the nation as a whole, Kentucky's graduation rates are higher for females than for males and higher for whites than for African Americans and Hispanics. Compared to the total U.S., Kentucky's gender gaps are larger, but racial/ethnic gaps are smaller.

Achievement. In the most recent National Assessment of Educational Progress (NAEP) exams, Kentucky students performed above the national average in grades 4 and 8 science and grade 4 reading, equal to the national average in grade 4 writing, and below the national average in grades 4 and 8 math and grade 8 writing. Compared to previous years, the latest NAEP results show improvements for the Commonwealth in all above-mentioned areas except for one: grade 8 reading declined, indicating a need for more focus on reading skills in middle schools. As in the nation as a whole, Kentucky's female students scored higher than males in reading and writing, males scored higher in math and science, whites scored higher than African Americans in all subjects, and impoverished students scored lower than others in all subjects. Compared to the nation, Kentucky's gender gaps are larger, but racial and poverty gaps are smaller.

**Students.** Kentucky students face more socioeconomic barriers to achievement than those in most other states. Their parents typically have lower incomes, education, and employment rates. Disabilities, teen parenthood, and teen death rates are above the national average. On the other hand, Kentucky has fewer students with limited English proficiency.

**Teachers.** For many years, in Kentucky and the nation as a whole, the number of teachers employed in schools has increased faster than enrollment, a trend attributed to schools' increasingly complex and diverse offerings. Kentucky's student-teacher ratio is better (lower) than the national average in elementary schools but slightly worse in middle and high schools. In terms of teachers as a percentage of all staff, Kentucky is again ranked last in the nation, due to high percentages of instructional aides and support staff. *Education Week* gives Kentucky high marks for policies relating to the teaching profession. Among all states, Kentucky has the 3<sup>rd</sup> highest percentage of teachers with advanced degrees.

**Schools.** Compared to the nation, a higher percentage of Kentucky students attend rural schools, which have special challenges as well as some advantages. The Commonwealth continues to perform well with regard to technology access, use, and capacity. School climates are generally better than average in Kentucky, with students less likely to report being in fights, being threatened with weapons, having drugs available, or using alcohol.

**Finance.** Even after adjusting for Kentucky's lower cost of living, Kentucky's teacher salaries and per-pupil revenues and expenditures are below the national average. The exception is per-pupil preschool spending which, due to a \$25 million budget increase, spiked above the national average in fiscal year 2007. Notwithstanding Kentucky's below-average education funding, the Commonwealth is a national leader in allocating those funds equitably among districts.

# Chapter 1

# Introduction to the 2008 Compendium

This second annual compendium of P-12 education rankings and related information is meant to provide a convenient reference tool. Changes to this year's compendium include focusing on a subset of peer states, examining more trends, and arranging all data by topic. This second annual compendium published by the Office of Education Accountability (OEA) presents a broad array of elementary and secondary public education data from federal, state, and nonprofit sources. The No Child Left Behind Act (NCLB) has increased the visibility of education, prompting a number of organizations to publish state rankings on education spending, performance, and other indicators. Frequently, legislators are asked about these rankings and Kentucky's education standing in general. The purpose of this compendium is to provide a convenient reference tool.

## **Changes From the 2007 Compendium**

The first annual compendium was published in 2007. This year's compendium provides updated tables and a wider array of topics. In addition, some changes were made in order to make it more user friendly. Changes include focusing analyses on a subset of peer states, examining more trends, and arranging all data by topic.

#### **Peer States**

In the 2007 compendium, each ranking of the 50 states and the District of Columbia spanned two pages. Tables in this edition are more compact, comparing Kentucky to surrounding states and member states of the Southern Regional Education Board, as listed in Table 1.1. Appendix A provides rankings for all 50 states and the District of Columbia.

		Table 1.1			
<b>Peer States</b>	Used for	Comparison	in	<b>Body of Repor</b>	t

Southern	Bordering States Not in Southern Regional		
Alabama (AL)	West Virginia (WV)	Georgia (GA)	<b>Education Board</b>
Mississippi (MS)	Delaware (DE)	South Carolina (SC)	Illinois (IL)
Virginia (VA)	North Carolina (NC)	Louisiana (LA)	Indiana (IN)
Arkansas (AR)	Florida (FL)	Tennessee (TN)	Missouri (MO)
Maryland (MD)	Oklahoma (OK)	Texas (TX)	Ohio (OH)

Source: Staff compilation of information from the Southern Regional Education Board.

The revised format is intended to make the report concise and focused on states facing challenges similar to those in Kentucky. Many southern and bordering states share similar geographic, demographic, cultural, and rural characteristics.

#### Trends

For some topics, charts are used to display trends for Kentucky and the United States. The graphics highlight distinct differences between Kentucky and the nation.

#### **Organization of This Compendium**

Rankings are organized by topic. Most rankings are based on data from the National Center for Education Statistics (NCES), the primary federal entity for collecting, analyzing, and reporting data related to education in the United States. Some rankings are from other sources, which are profiled briefly in Appendix B.

The remainder of Chapter 1 discusses how to use the compendium. It includes important issues to consider when interpreting the rankings.

Chapter 2 presents information on education outcomes, specifically graduation rates, dropout rates, assessments, and college-readiness exams. Extra attention is devoted to summarizing opposing positions taken in the highly controversial debates surrounding graduation rates and proficiency levels.

Chapter 3 focuses on characteristics of students, their families, and their communities. Topics include enrollment growth, racial/ethnic composition, income, and participation in selected programs.

Chapter 4 presents characteristics of schools and staff, including the student-teacher ratio, instructors as a percentage of all staff, Title I eligibility, technology, and school safety.

Chapter 5 reports education revenues and expenditures data, such as revenues by source, spending per pupil, spending by function, teacher salaries, and measures of equity.

Chapter 6 discusses several multitopic indices that combine a wide variety of measures, including noneducation measures.

Most rankings in this compendium use data from the National Center for Education Statistics. They are grouped by topic, with graduation rates and test scores in Chapter 2, student and family characteristics in Chapter 3, school and staff data in Chapter 4, finances in Chapter 5, and multitopic indices in Chapter 6. Appendices provide rankings for all states and other information. Appendices A, B, C, and D contain rankings for all 50 states and the District of Columbia, information about data sources, a key to state abbreviations, and National Assessment of Educational Progress (NAEP) achievement level definitions, respectively.

#### How To Use This Compendium

State rankings give legislators and other policy makers multiple perspectives on how Kentucky's education system compares to those in other states. Such comparisons provide insights into the state's current situation, progress made in the recent past, and potential for progress in the future. They also shed light on issues related to educational performance, such as Kentucky's ability to compete with other states in attracting and retaining skilled workers and business investments.

Rankings may be controversial because they depend on the measures chosen, the statistical methods used, and the ways in which measures are standardized. The same state can appear to perform well on one ranking and poorly on another (Olson. "An 'A"").

Rankings that seem to support or oppose controversial policies often stimulate debates about possible biases on the part of the publishers and sponsors. Bias is not always blatant or deliberate. Those who read and use rankings should bear in mind that the simple act of choosing which indicators to report is a subjective judgment.

All rankings are subject to data and methodology issues that can limit their reliability and validity. When interpreted with their limitations in mind, they can offer valuable insights. This compendium provides caveats with some of the rankings to aid in their interpretation.

Some data, such as results from NAEP and the Census Bureau's annual American Community Survey, are based on randomly selected samples instead of entire populations; therefore, they are subject to sampling error. Compendium tables based on such sample data indicate whether each state is significantly higher than Kentucky, significantly lower, or not statistically different, assuming a 95 percent significance level. A statistically significant difference between another state and Kentucky indicates a 95 percent likelihood that the difference is real and not simply due to random sampling error. In compendium tables based on sample data, states are listed in descending order but are not ranked.

Comparing Kentucky's education system to those in other states provides insights into progress made, potential for further progress, and Kentucky's competitive position.

A state can perform well on one ranking and poorly on another, depending on the measures and methods chosen. Simply choosing what measures to report is a subjective judgment, which should be considered by those who read and use rankings.

All rankings are subject to reliability and validity limitations. Data based on samples are tested statistically to determine whether observed differences are likely to be real or due to sampling error.

# **Rankings in Context**

The underlying measures and context of ranks should always be considered. As an isolated number, a rank does not indicate how far apart states are or whether a state is good or bad in absolute terms. Ranks should be used with caution. It is essential to examine the measures on which the ranks are based and the context affecting the measures. For instance, education outcomes are influenced by state-to-state variations in poverty levels, urbanization, taxes, and funding. The following caveats are other factors to consider when analyzing rankings.

# **Magnitude of Differences**

A rank, as an isolated number, does not indicate how far apart states are from each other. For example, in 2005, Kentucky ranked 42<sup>nd</sup> on gender equity in grade 4 math scores (U.S. Dept. of Ed. Inst. Natl. *NAEP Data*). At first glance, Kentucky seems to perform poorly with respect to gender equity. In reality, the spread of scores from the top-ranked state to the lowest-ranked state is miniscule, and none of the differences is statistically significant. In contrast, states that are only one rank apart can differ considerably. For example, New Mexico was ranked 2<sup>nd</sup> after North Carolina in terms of the increase in preschool enrollment from 2004 to 2005. Although the two states were only one rank apart, North Carolina's growth rate was nearly three times that of New Mexico's (Natl. Inst. *The State* 17). It should be noted that this compendium reserves the terms "significantly" and "significant" for differences that are statistically significant.

# **Relative Versus Absolute Levels of Quality**

A rank indicates whether Kentucky's performance is better or worse than performances of other states, but not whether this performance is "good" or "bad" in absolute terms. If every state were performing well on a measure, then even a low-ranked state may be meeting student needs. Conversely, if all states performed poorly, the top-ranked state may not be meeting student needs.

# Zero-sum Game Among States

In effect, rankings treat educational progress as a zero-sum game among states. A state's movement upward in rankings comes at the expense of another state. Education reform is an ongoing process across the country. It is possible for Kentucky to improve steadily on a measure and yet lose ground in the rankings when other states improve more rapidly.

It is possible for Kentucky to improve steadily on a measure and yet lose ground in the rankings if other states improve at faster rates.

#### **Drawing Policy Implications**

#### **Different Indicators Address Different Policy Questions**

Different indicators are useful for different purposes, and each offers unique insights into Kentucky's challenges and outcomes. For example, education spending per pupil indicates the resources available to help each child, whereas education spending per \$1,000 in personal income reflects taxpayers' efforts relative to their ability to pay for education. Neither measure indicates the cost effectiveness of education systems. High levels of spending are often commended as a sign of extraordinary commitment to education, but they can also indicate wasteful spending, needier students, and/or simply a higher cost of living (Natl. Forum 4).

#### **Unintended and Unanticipated Consequences**

Policy implementation does not occur in a vacuum. Attempts to improve one indicator have impacts on other indicators within the system (Natl. Forum 5). For example, many states are striving for more participation in ACT or Advanced Placement tests. Success at this endeavor is usually associated with lower average scores as the pool of test takers expands beyond the highest achieving students. Similarly, raising graduation rates by preventing at-risk students from dropping out is usually accompanied by declines in test scores and college-going rates. This phenomenon was recently observed in Kentucky (Commonwealth. Dept. of Ed. *Briefing* 1; Innes. *Is Kentucky*).

## Simpson's Paradox

Aggregate rankings may mask important differences between subgroups when the composition of the overall group is shifting over time. For example, between 1981 and 2005, the national average SAT verbal score rose only 4 points for all test takers. However, scores for each ethnic group rose faster than the total (see Table 1.2). Underlying the paradox is the shift in the composition of test takers; whites, the highest-scoring subgroup, made up 85 percent of test takers in 1981, but only 63 percent in 2005 (Bracey 64-65).

Different indicators are useful for different purposes and insights.

Policy initiatives to improve one indicator can cause side effects for other indicators.

Rankings on the overall group may mask important differences between subgroups.

Gains for Eth	Gains for Ethnic Groups—SAT Verbal									
Ethnic Group	1981	2005	Gain	1981	2005					
White	519	529	10	85	63					
Black	412	433	21	9	12					
Asian	474	511	37	3	11					
Mexican	438	453	15	2	5					
Puerto Rican	437	460	23	1	1					
American Indian	471	489	18	0	1					
All Groups	504	508	4							

# Table 1.2Simpson's Paradox: SAT Verbal Scores by Ethnic Group and<br/>Composition of Test Takers, 1981-2005

Source: Bracey 64-65.

#### **Steady Gains on Multiple Measures**

Most rankings are snapshots of performance at a given time. Measures tend to vary from year to year, and sometimes these changes are not statistically significant. Test scores, in particular, can be very volatile. When possible, researchers analyze multiple years of data in order to smooth out random fluctuations (Way).

A large increase in funding can boost per-pupil spending substantially from one year to the next. However, most performance variables, such as graduation rates, do not improve immediately. Successful policies should achieve steady gains in multiple measures over time. Gains should be consistent and reinforcing rather than episodic and inconsistent.

This compendium contains multiple indicators of how Kentucky compares to other states. Rather than choosing one measure to gauge Kentucky's success or failure, readers are encouraged to weigh all of the available indicators.

#### **Data Comparability**

Indicators from different sources may appear to be the same but may differ in the ways variables are defined, collected, analyzed, and reported. Even rankings that appear to use exactly the same source can conflict if they reflect different points in time.

Measures fluctuate over time. It is best to track them over a number of years. Successful policies should achieve steady gains in multiple measures over time.

Rankings on the same measure may seem to contradict if they use slightly different points in time or assumptions.

# Terms Used in This Compendium

#### Rank

Unless otherwise noted, rankings and U.S. totals refer to the 50 states and the District of Columbia. Fiscal year is synonymous with school year; "n.a." indicates data that are not available; and "--" indicates "does not apply." Unless otherwise noted, ranks reported in this compendium are out of 51—the 50 states and the District of Columbia. When two or more states have the same value, they are assigned the same rank and are listed in alphabetical order.

## **Fiscal and School Years**

Since most districts' fiscal years correspond to school years, usually starting July 1 and ending June 30, this compendium refers to fiscal years only. The only exceptions are in Alabama, which follows the federal fiscal year ending September 30; and in Nebraska and Texas where fiscal years end August 31. The federal government usually does not adjust data for states that have fiscal years different from July 1 to June 30 (U.S. Dept. of Ed. Inst. Natl. *An Historical Overview*). This compendium identifies fiscal year by the ending year, for example, the 2005-2006 fiscal year is referred to as FY 2006.

#### **States and United States Totals**

Unless otherwise noted, U.S. totals reported in tables refer to the 50 states and the District of Columbia. If data are not available for all states, the U.S. entry summarizes all available data.

## Data Not Available or Inapplicable

The abbreviation "n.a." indicates data were not available because they were not collected, not reported, or not reliable. In contrast, two dashes (--) indicate that a measure does not apply. For the U.S. as a whole, two dashes appear in place of a state rank.

## Conclusion

Despite their limitations, state education rankings provide a useful summary of Kentucky's educational progress and challenges. This chapter discussed some of the issues to consider when interpreting rankings. The following chapters present measures of student performance, demographic trends, school characteristics, and finances. For each table in the body of the report, Appendix A provides a corresponding table ranking all states.

# Chapter 2

# **Education Outcomes**

This chapter examines graduation rates, dropout rates, and test scores, including an examination of differences between statereported information and information from other sources.

Across multiple measures, Kentucky's graduation rate recently rose slightly above the national rate, after hovering slightly below for years. Compared to the nation, Kentucky's gender gaps are larger, but racial/ethnic gaps are smaller.

Policy makers' efforts to monitor high school completion are complicated by data quality issues and competing formulas. Kentucky's state-reported graduation rates are higher than those from sources using alternative methodologies. This chapter examines graduation rates, dropout rates, and test scores. These indicators frequently stimulate national debate. Graduation rates vary depending upon data quality and the formulas used to calculate them. State-reported graduation rates often disagree with rates produced by other sources. Results of state tests often disagree with national test results. This chapter devotes special attention to examining possible reasons for discrepancies between different measurements.

## **Graduation Rates**

High school graduation is correlated with a wide range of benefits to the student and to society. Compared to dropouts, high school graduates have steadier employment and higher incomes. In addition, they report better health, are more likely to vote, and are less likely to go to prison (U.S. Dept. of Ed. Inst. Natl. *Dropout* 1; U.S. Dept. of Commerce. Census. *Voting*. Table 5).

Across multiple measures, Kentucky's graduation rate recently rose slightly above the national rate, after hovering slightly below for several years. Like other states, Kentucky's graduation rates are higher for females than for males and higher for whites than for African Americans and Hispanics. Compared to the nation, Kentucky's gender gaps are larger, but racial/ethnic gaps are smaller.

#### **Data Quality Issues**

Policy makers' efforts to monitor high school completion are complicated by data quality issues and competing formulas. Recent reviews suggest that the Kentucky Department of Education's formula may overstate graduation rates (Commonwealth. Auditor). As Table 2.1 shows, while the department has reported rates above 80 percent since 2004, sources using alternative methodologies reported rates that range from 5-20 percentage points lower. It is important to note, however, that Kentucky is improving on multiple measures.

	2002		2003		2004		2005	
<b>Calculation Method</b>	KY	U.S.	KY	U.S.	KY	U.S.	KY	U.S.
Kentucky Department of Education calculation	80.8	n.a.	79.2	n.a.	81.3	n.a.	82.9	n.a.
NCES Averaged Freshman Graduation Rate	69.8	72.6	71.7	73.9	73.0	74.3	75.9	74.7
<i>Education Week</i> /Urban Institute Cumulative Promotion Index	72.1	68.7	69.7	69.6	70.0	69.9	71.5	70.6
Manhattan Institute/Greene rate	68.0	71.0	68.9	69.9	n.a.	n.a.	n.a.	n.a.
National Center for Higher Education Management Systems Cohort Survival Rate	62.3	68.2	65.4	69.7	64.8	69.7	71.3	68.8

Table 2.1 Graduation Rates Using Various Formulas, Kentucky and U.S. FY 2002-FY 2005

Note: The Kentucky Department of Education definitions changed slightly in 2005 to comply with National Center of Education Statistics requirements.

Sources: Commonwealth. Dept. of Ed. Briefing 16; U.S. Dept. of Ed. Inst. Natl. Digest 2007 Table 102; Editorial Projects in Education. Education Week's Diplomas; Greene; Natl. Ctr. for Higher Ed. Preparation.

Accurate graduation rates require careful tracking of students. In addition to graduating with a standard diploma in 4 years, students take many other paths throughout their high school years.

Challenges of Tracking High School Students. Kentucky is not alone in reporting rates that conflict with other sources. The reporting of graduation rates is controversial throughout the nation. Accurate graduation rates require careful tracking of students throughout their high school years. The U.S. Department of Education (USDOE) defines graduation as completing a standard diploma within 4 years. In addition, Kentucky and some other states have permission from the USDOE to include in graduation rates students earning standard diplomas in more than 4 years if the students have Individualized Education Programs (IEPs) that stipulate more time (Commonwealth. Dept. of Ed. Approved 17; U.S. Government. Education Could 4). As Figure 2.A illustrates, student progress can be interrupted by repeating a grade; needing more than 4 years to graduate; transferring between schools, districts, or states; dropping out and sometimes returning and then dropping out for a second time; or earning an alternative credential by passing the General Educational Development (GED) tests or by some other means.



Figure 2.A Outcomes for a Hypothetical High School Class

Source: Staff adaptation of example from U.S. Government. Education Could 13.

Most states are only beginning to implement systems for accurate tracking, and few conduct audits of the data on which graduation rates are based. Most states are only beginning to implement systems that track transfers within their own boundaries, and a national system for confirming transfers from one state to another has not been developed. Data inaccuracies can raise or lower a school's graduation rate, yet fewer than half of states audit the data used to calculate rates (U.S. Government. *Education Could* 25-27).

#### **Recommended Formula Using Longitudinal Data**

A federal task force recommends tracking individual students over time and calculating the graduation rate by dividing the number of 4-year standard diplomas by the number of grade 9 students 4 years earlier, adjusted for transfers and deaths. Accurate graduation rates require tracking individuals within each cohort of students for 4 years starting from the day they enter grade 9. Using such longitudinal data, the graduation rate can be calculated by dividing the number of 4-year standard diplomas by the number of grade 9 students 4 years earlier, adjusted for transfers into and out of the cohort and deaths during those 4 years (U.S. Dept. of Ed. Inst. Natl. *National Institute* 12). This formula, which USDOE will require all states to adopt in the near future, is shown in Figure 2.B.

#### Figure 2.B Graduation Rate Formula Preferred by U.S. Department of Education Task Force Using Longitudinal Individual Student Data

Craduation Pata	Number of students receiving standard 4-year diplomas in year G*
Graduation Kate	
for year G	Number of students entering grade 9 for the first time in fall of year G-3
5	+ Number of students transferring into grade 9 during year G-3
	- Number of students excluded from grade 9 in G-3
	+ Number of students transferring into grade 10 during G-2
	- Number of students excluded from grade 10 during year G-2
	+ Number of students transferring into grade 11 during G-1
	- Number of students excluded from grade 11 during year G-1
	- Number of students excluded from grade 12 during year G

Notes: \*The USDOE also allows the definition of graduation to include students who require more than 4 years to earn standard diplomas if their IEPs specify more time. Year G refers to the year of the graduation rate. Year G-1 is the previous year, G-2 is two years previous, and G-3 is three years previous. For example, to calculate the graduation rate for the class of 2008, the denominator starts with the count of first-time 9<sup>th</sup> graders in 2005 and follows that cohort through their high school years from 2005 to 2008. The number of students transferring into the cohort during that time is added and the number excluded (usually because they transferred out to another school) is subtracted.

Source: Staff adaptation from U.S. Dept. of Ed. Inst. Natl. National Institute 12.

The number of states using cohort graduation rates grew from 12 in FY 2005 to 16 in FY 2007. All states have pledged to work toward such a rate. However, Kentucky and 31 other states still use a rate that depends on dropout data. All states have pledged to work toward implementing individuallevel longitudinal data systems that will yield accurate graduation rates for each cohort (Natl. Governors). The number of states that had implemented some type of cohort method was 12 in FY 2005 and 16 by FY 2007 (U.S. Dept. of Ed. Inst. Natl. *National Institute* 18; Editorial. *Implementing*). As of FY 2007, Kentucky and 31 other states still used a "departure classification" rate, which depends on dropout data. The USDOE helped to develop the departure classification rate formula but recently instructed states to transition to a cohort rate by FY 2011 (U.S. Dept. of Ed. *High School* 1).

# **KDE Departure Classification Rate**

Kentucky's graduation rate calculation depends on accurate dropout rates.

Kentucky's departure classification formula is shown in Figure 2.C. The denominator includes annual dropout rates, graduates needing more than 4 years, and other measures. Thus, an accurate graduation rate depends on the accuracy of other reported data.

#### Figure 2.C Kentucky's Departure Classification Graduation Rate Formula

Graduation Rate	Graduates in year G (Number of students receiving standard diplomas within 4 years, or longer if the individualized education program specifies a longer time)
for year G =	Graduates in year G as defined above
+	"Graduates Plus" (standard diplomas earned in more than 4 years without IEP) in year G
	+Certificates of Completion in year G
	+ Grade 12 Dropouts in year G
	+ Grade 11 Dropouts in year G-1
	+ Grade 10 Dropouts in year G-2
	+ Grade 9 Dropouts in year G-3

Note: Certificates of completion include students earning GEDs through in-school programs. They also include those earning GEDs outside the school setting by October 1 of the year after they dropped out. Source: Commonwealth. Dept. of Ed. *Approved State Accountability Plan*.

The method used by Kentucky and 31 other states tends to yield higher graduation rates than the cohort method.

The Kentucky Department of Education should have cohort data by the spring of 2009, but conversion to a new student identification system may cause delays.

The National Center of Education Statistics (NCES) and other organizations have developed formulas that do not depend on accurate dropout rates. However, being less effective at adjusting for student transfers, they can underor overestimate graduation rates. Although the departure classification method is still used in 31 states, the U.S. Government Accountability Office (GAO) estimates that it yields a graduation rate that is 12 percent greater than the cohort method (*Education Could* 16).

The Kentucky Department of Education's implementation of a unique student identifier in FY 2005, which began tracking students entering grade 9 in FY 2006, was expected to provide the first true cohort measures of graduation rates in the spring of 2009. However, the conversion to a new student identification system may cause delays until the spring of 2014.

#### Formulas Independent of State Education Systems

Several organizations have attempted to develop graduation rate formulas that are comparable across states and not dependent on the accuracy of individual schools' record keeping on dropout rates. The most widely accepted among these is the Averaged Freshman Graduation Rate (AFGR) developed and used by NCES. Another well-known measure is the Cumulative Promotion Index, which was developed by the Urban Institute and is used in *Education Week* publications. These measures do not depend on the accuracy of dropout rates, but they are less effective than departure classification rates at adjusting for student mobility. They can underestimate graduation rates in areas or subgroups that are losing population and overestimate rates in areas or subgroups that are gaining population. For example, in 2005, the AFGR for Asians in Kentucky was over 100 percent and the AFGR for Hispanics was 91 percent.

The averaged freshman graduation rate introduced by NCES in 2005 adjusts somewhat for the impact of student transfers. **NCES Averaged Freshman Graduation Rate.** In July of 2005, NCES introduced the AFGR, an estimate of the percentage of an entering freshman class graduating in 4 years. For FY 2005, the AFGR equals the total number of diploma recipients in 2005 divided by the average membership of the grade 8 class in FY 2001, the grade 9 class in FY 2002, and the grade 10 class in FY 2003. Averaging the membership of the three classes adjusts somewhat for the impact of transfers in and out of the cohort over time. The AFGR formula is shown in Figure 2.D.

#### Figure 2.D Averaged Freshman Graduation Rate Formula

AECD for wood C -	Number of students receiving standard 4-year diplomas in year G
AFGK for year G =	Average of Grade 8 membership in year G-4,
	Grade 9 membership in year G-3, and Grade 10 membership in year G-2

Source: Staff compilation based on U.S. Dept. of Ed. Inst. Natl. Dropout 9.

Using NCES graduation rates, Kentucky ranks 29<sup>th</sup> and is slightly above the national average. Kentucky's racial and ethnic gaps are smaller than those for the nation as a whole. As Table 2.2 shows, Kentucky ranks 29<sup>th</sup>, with an AFGR slightly above the national rate. Gaps by race and ethnicity are smaller in Kentucky than in the nation. This reflects not only higher graduation rates for minorities but also lower rates for whites in Kentucky, perhaps due to higher rates of poverty in Kentucky than in most other states. The Hispanic AFGR is likely distorted by the group's small size and relatively rapid migration into the state, which can cause the AFGR to overestimate the graduation rate.

It should be noted that the graduation rate is higher for females than for males, yet NCES does not publish separate male and female graduation rates at the state level.

All R	aces/Eth	nnicities	Whit	te Non-H	ispanic	Blac	k Non-H	ispanic		ic	
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
15	MO	80.6	8	IL	87.5	8	MD	71.1	2	KY	91.0
16	OH	80.2	14	OH	83.6	10	KY	69.9	3	MO	86.4
18	VA	79.6	15	MD	83.5	12	WV	69.5	5	AR	79.6
19	IL	79.4	16	MO	82.9	12	AR	69.5	6	MD	78.6
20	MD	79.3	18	VA	82.3	14	VA	69.0	9	VA	76.7
25	WV	77.3	22	TX	81.5	17	TX	68.6	13	LA	71.9
26	OK	76.9		U.S.	80.4	18	OK	68.2	15	OH	71.2
29	KY	75.9	30	OK	77.9	22	MO	67.5	15	OK	/1.3
30	AR	75.7	31	WV	77.4	26	NC	65.7	20	TN	69.4
	U.S.	74.7	32	DE	77.3	27	DE	64.4	21	MS	69.0
35	TX	74.0	33	AR	76.9	34	OH	61.3	22	NC	66.0
36	IN	73.2	34	KY	76.8	35	TN	59.8	22	IL	00.9
37	DE	73.1	36	NC	76.4	36	MS	59.7	24	TX	66.3
39	NC	72.6	37	IN	76.3	37	IL	58.6		U.S.	64.2
41	TN	68.5	40	LA	72.2		U.S.	58.1	30	IN	63.7
42	AL	65.9	43	TN	71.0	39	AL	57.2	33	FL	62.7
45	FL	64.6	44	AL	70.5	40	LA	54.1	34	DE	62.4
47	LA	63.9	45	FL	70.2	42	GA	52.4	37	AL	59.9
48	MS	63.3	16	GA	66.6	43	IN	52.2	42	GA	51.2
49	GA	61.7	40	MS	00.0	44	FL	51.8	no	SC	no
50	SC	60.1	n.a.	SC	n.a.	n.a.	SC	n.a.	11.ä.	WV	11. <b>ä</b> .

Table 2.2NCES Averaged Freshman Graduation Rate, FY 2005

Notes: The averaged freshman graduation rate estimates the percentage of an entering freshman class that graduates with a standard diploma in 4 years. For FY 2005, it equals the total number of diploma recipients in FY 2005 divided by the average membership of the 8<sup>th</sup>-grade class in FY 2001, the 9<sup>th</sup>-grade class in FY 2002, and the 10<sup>th</sup>-grade class in FY 2003. Rates for Asian students are not reported because Kentucky's AFGR exceeded 100%; this can happen for small, growing subgroups.

Source: U.S. Dept. of Ed. Inst. Natl. "Data Tables."

Kentucky's graduation rate has risen slightly each year since 2002.

Figure 2.E illustrates trends in AFGRs. Rates have not changed substantially over the past 15 years, although Kentucky's rate shows a slight increase each year since 2002.



Figure 2.E

Source: U.S. Dept. of Ed. Inst. Natl. Digest 2006 165 and Digest 2007 165.

The Cumulative Promotion Index (CPI) uses the most up-to-date data for each of high school transition point. It does not reflect the experiences of a real cohort, but it describes current conditions. **Education Week Cumulative Promotion Index.** The Cumulative Promotion Index (CPI), shown in Figure 2.F, resembles a cohort estimate because it combines measures for each of the critical transitions a cohort makes during high school: promotion from grade 9 to grade 10, from grade 10 to grade 11, and from grade 11 to 12, culminating in graduation at the end of grade 12. However, while cohort estimations like the AFGR use measures going back several years to reconstruct the experience of a cohort, the CPI uses a "snapshot" of all high school cohorts at approximately the same point in time. As a composite of several cohorts, the CPI does not accurately capture the true experiences of any one cohort, but it has the advantage of better describing current conditions, since it uses the most up-to-date data for each of the high school transition points.

	Grade 10 enrollment in fall of year G+1	Grade 11 enrollment in fall of year G+1	Grade 12 enrollment in fall of year G+1		Graduates in year G
CPI for year G =	Grade 9 enrollment in fall of year G	Grade 10 enrollment in fall of year G	Grade 11 enrollment in fall of year G	Х	Grade 11 enrollment in fall of year G

Figure 2.F Formula for the Cumulative Promotion Index

Source: Swanson 7.

From 2001 to 2005, Kentucky's CPI showed the 3<sup>rd</sup> fastest improvement, rising above the national rate. However, gender gaps are larger than the nation's; females have graduation rates 9.8 percentage points higher than males. As Table 2.3 shows, while Kentucky's CPI was below the national average in 2001, it was above the national average by 2005. In terms of improvement, Kentucky ranks 3<sup>rd</sup> nationally in positive CPI change between 2001 and 2005. However, the gender gap grew during this time. For the nation as a whole, the graduation rate for boys is 7.5 percentage points lower than the rate for girls. In Kentucky, this gap is 9.8 percentage points.

Table 2.3Cumulative Promotion Index for All Students and by Gender, FY 2001 and FY 2005

			All	Student	S				Gender					
	2001			2005		Change (2005 Minus 2001)			Males 2005			Females 2005		
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
14	MD	75.3	13	IL	76.7	1	TN	7.9	11	MO	73.5	11	MO	79.0
15	IL	74.8	15	MO	76.5	2	FL	7.8	16	OH	72.6	13	MD	78.4
20	MO	73.0	16	OH	75.9	3	KY	6.2	17	IL	72.0	17	OH	78.1
21	VA	72.6	22	IN	72 6	8	OH	5.0	21	OK	70.0	18	IL	77.9
22	IN	72.5	23	MD	/3.0	10	SC	4.9	21	WV	/0.0	22	VA	76.7
27	OH	70.9	26	AR	73.2	13	MS	3.8	26	AR	69.2	23	AR	76.5
28	AR	70.8	27	VA	72.9	14	TX	3.6	28	VA	68.9	24	IN	76.3
29	WV	70.7	28	WV	72.8	16	MO	35	29	MD	68.3	24	WV	70.5
30	OK	70.1	29	KY	71.5	10	NC	5.5		U.S.	67.8	26	KY	76.0
	U.S.	68.0	31	OK	70.8	20	GA	2.6	30	IN	67.7		U.S.	75.3
35	KY	65.3		U.S.	70.6		U.S.	2.6	33	KY	66.2	35	NC	73 3
37	TX	64.9	36	TX	68.5	21	AR	2.4	35	TX	64.9	55	OK	15.5
38	DE	64.3	40	NC	67.0	24	WV	2.1	39	NC	61.3	37	TX	72.7
39	LA	64.1	41	TN	65.4	25	IL	1.9	40	FL	56.1	40	MS	68.6
41	NC	63.5	42	MS	61.8	30	IN	1.2	41	MS	55.4	41	FL	65.5
44	AL	61.6	43	AL	61.3	32	OK	0.7	42	DE	55.0	42	DE	64.0
46	MS	58.0	44	FL	60.8	33	VA	0.3	43	GA	52.8	43	GA	63.8
47	TN	57.5	45	DE	60.1	37	AL	-0.4	46	LA	48.3	44	LA	60.5
48	GA	55.5	46	GA	58.1	41	MD	-1.7		AL			AL	
50	FL	53.0	48	SC	55.6	46	DE	-4.2	n.a.	SC	n.a.	n.a.	SC	n.a.
51	SC	50.8	49	LA	54.7	51	LA	-9.4		TN			TN	

Notes: Ranks for males are out of 44; data are not available for NH, NV, NY, OR, PA, SC, and TN. Ranks for females are out of 43; data are not available for DC, NH, NV, NY, OR, PA, SC, and TN. Source: Editorial Projects in Education. *Education Week's Diplomas Count*.

Racial and ethnic gaps are smaller in Kentucky than in most other states. This may reflect Kentucky's smaller income gaps and more equitable school funding because income is related to educational attainment. Table 2.4 breaks down the CPI for various racial and ethnic groups. In both Kentucky and the nation, Asians are the most likely to graduate, followed in order by whites, Hispanics, and blacks. In Kentucky, the gaps between whites and minorities are smaller; while Kentucky's white students graduate at rates below the national average, all other racial groups graduate at rates above the national average.

Researchers attribute racial graduation gaps, in part, to economic disadvantages among minorities. Kentucky's smaller graduation gaps may stem from smaller gaps in economic opportunity. Black-white differences in income are smaller in Kentucky than in most other states. In addition, Kentucky's Support Education Excellence in Kentucky (SEEK) formula minimizes school funding gaps between high-minority/high-poverty districts and low-minority/low-poverty districts, as will be discussed in Chapter 5.

	Asian			Hispanic		Black/A	frican A	merican		White		
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	
1	MD	91.8	2	KY	66.5	3	AR	64.1	11	IL	82.5	
2	TX	86.9	3	MD	64.9	6	MD	62.0	13	MD	81.8	
3	LA	85.0	4	LA	62.5	7	TX	61.7	18	OH	78.8	
5	KY	84.5	7	WV	59.9	8	VA	61.0	20	VA	78.2	
7	IL	83.4	9	IL	59.8	9	WV	60.7		U.S.	76.2	
8	VA	83.0	10	FL	59.0	11	KY	59.2	21	MO	78.0	
9	FL	82.2	11	TX	57.9	13	MS	58.0	27	TX	75.6	
	U.S.	80.2		U.S.	57.8	10	NC	57.0	28	OK	75.0	
13	OK	79.3	12	MO	57.4	10	OK		29	AR	74.9	
19	NC	75.5	13	VA	57.1	21	MO	55.8	30	IN	74.6	
20	GA	75.4	22	NC	53.8	22	LA	54.6	32	WV	72.6	
22	OH	75.3	25	OK	52.6		U.S.	53.4	33	NC	71.7	
26	WV	70.8	28	IN	50.2	27	IL	51.8	36	KY	70.3	
27	IN	69.8	32	MS	47.5	28	AL	49.9	38	DE	69.0	
30	AL	66.2	34	OH	46.8	29	OH	48.0	40	LA	66.2	
34	MS	63.4	37	DE	41.4	32	FL	46.7	41	FL	66.0	
	AR		39	GA	38.6	36	GA	44.4	42	MS	65.6	
	DE		41	AL	36.5	38	IN	41.6	43	AL	65.2	
n.a.	MO	n.a.		AR		n.a.	DE		45	GA	59.2	
	SC		n.a.	SC	n.a.		SC	n.a.		SC	n.a.	
-	TN			TN			TN		n.a.	TN		

Table 2.4
Cumulative Promotion Index by Race and Ethnicity, FY 2005

Notes: Ranks for Asians are out of 38; data are not available for AR, DE, ID, IA, MO, MT, ND, SD, NH, NV, NY, SC, and TN. Ranks for Hispanics are out of 42; data are not available for AR, ME, MN, VT, NV, NH, NY, SC, and TN. Ranks for blacks/African Americans are out of 40; data are not available for DE, ID, ME, MT, SD, VT, NV, NH, NY, SC, and TN. Ranks for whites are out of 46; data are not available for NV, NH, NY, SC, and TN. Source: Editorial Projects in Education. *Education Week's Diplomas Count*.

#### **State-reported Rates**

State-reported graduation rates are often higher than alternative measures. These discrepancies are smaller in Kentucky than in many other states. The graduation rates that states report for purposes of No Child Left Behind are often higher than the AFGR, CPI, and other alternative measures, as shown in Table 2.5. Some critics argue that states deliberately obscure their true graduation rates in order to avoid embarrassment and sanctions (Dillon; Hall). GAO attributes discrepancies to inadequacies of data systems, coding errors, and the use of different formulas (U.S. Government. *Education Could* 24-27). In response to criticism for its lack of guidance to states, the USDOE recently proposed requiring all states to use the same formula (U.S. Dept. of Ed. U.S. Secretary). It is important to note that gaps between the Kentucky Department of Education's graduation rate and alternative methodologies are smaller than in many peer states.

Table 2.5
State-reported Graduation Rates Compared to AFGR and CPI, FY 2005

State-reported Graduation Rate			NCES Averaged Freshman Graduation Rate (AFGR)			<i>Edu</i> Cumula In	<i>Education Week</i> Cumulative Promotion Index (CPI)			State-AFGR Percentage Point Difference			State-CPI Percentage Point Difference		
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	
1	NC	95.0	15	MO	80.6	13	IL	76.7	1	NC	22.4	2	NC	28.0	
7	IN	89.9	16	OH	80.2	15	MO	76.5	2	MS	21.7	3	DE	23.6	
13	IL	87.4	18	VA	79.6	16	OH	75.9	4	SC	17.0	4	MS	23.2	
19	OH	86.2	19	IL	79.4	22	IN	72 (	5	IN	16.7	5	SC	21.5	
20	MO	85.8	20	MD	79.3	23	MD	/3.0	9	DE	10.6	8	IN	16.3	
21	MS	85.0	25	WV	77.3	26	AR	73.2	12	ΤХ	10.0	10	TX	15.5	
25	MD	84.8	26	OK	76.9	27	VA	72.9	13	TN	9.4	15	TN	12.5	
27	WV	84.3	29	KY	75.9	28	WV	72.8	16	IL	8.0	19	OK	11.6	
28	TX	84.0	30	AR	75.7	29	KY	71.5	17	GA	7.7	20	WV	11.5	
29	DE	83.7	35	TX	74.0	31	OK	70.8	19	WV	7.0	21	GA	11.2	
30	KY	82.8	36	IN	73.2	36	TX	68.5	20	KY	6.9	21	KY	11.5	
31	OK	82.4	37	DE	73.1	40	NC	67.0	25	OH	6.0	24	MD	11.2	
35	AR	81.3	39	NC	72.6	41	TN	65.4	26	AR	5.6	25	IL	10.7	
38	VA	79.5	41	TN	68.5	42	MS	61.8	28	MD	5 5	27	OH	10.3	
40	TN	77.9	42	AL	65.9	43	AL	61.3	28	OK	5.5	31	MO	9.3	
41	SC	77.1	45	FL	64.6	44	FL	60.8	30	MO	5.2	36	FL	8.2	
45	GA	69.4	47	LA	63.9	45	DE	60.1	34	FL	4.4	37	AR	8.1	
46	FL	69.0	48	MS	63.3	46	GA	58.1	45	VA	-0.1	44	VA	6.6	
na	AL	na	49	GA	61.7	48	SC	55.6	na	AL	na	na	AL	na	
11.a.	LA	11.a.	50	SC	60.1	49	LA	54.7	11.a.	LA	11.a.	11.a.	LA	11.a.	

Note: A positive difference indicates that the state reports a rate that is higher than the rate to which it is compared (the AFGR or the CPI).

Sources: Editorial Projects in Education. *Implementing* and *Education Week's Diplomas*; U.S. Dept. of Ed. Inst. Natl. *Public* 13-14.

Instead of the United States Department of Education's definition of "graduation"-earning a standard diploma in 4 yearssome advocate giving credit for late and alternative diplomas.

Kentucky and California have been found to underreport dropout rates. The full magnitude of underreporting in these states and others is unknown because the data are rarely audited.

#### **Policy Implications of Graduation Rate Calculations**

In calculating graduation rates, the USDOE defines graduation strictly as earning a standard diploma within 4 years. Some question this approach, which gives no more credit for late diplomas, GEDs, and certificates of completion than it does for dropouts. Some advocate using the U.S. Census Bureau's definition of high school completion, which counts all standard diplomas, alternative diplomas, and GEDs. Policy makers and think tanks will continue to debate the merits of various measurement tools. However, until data systems using unique student identifiers are fully implemented, all graduation and dropout rates will remain flawed. Moreover, the GAO study suggests that even after these systems are in place, there will be flaws unless school personnel are trained to input data correctly and data are audited periodically (U.S. Government. Education Could 24-27).

#### **Dropout Rates**

A student is considered a dropout if he or she was enrolled in school at some time during the previous school year but was not enrolled at the beginning of the current year; has not graduated from high school or completed a state- or district-approved education program; and does not meet any conditions for exclusion. The conditions for exclusion are the documented transfer to another public school district, private school, or a stateor district-approved education program; temporary absence due to suspension or school-approved illness; or death (U.S. Dept. of Ed. Inst. Natl. Numbers and Rates 8).

## **Data Quality Issues**

Although NCES collects and publishes state-reported dropout rates, it does not audit the accuracy of these rates. A recent review by Kentucky's Auditor of Public Accounts found that the student information system was not tracking all dropouts. The full magnitude of the underreporting is unknown (Commonwealth. Auditor: U.S. Government. *Education Could*). Other states have had similar issues. California's new student-tracking system found a dropout rate of 24 percent, instead of the 13 percent estimated by an older method that had been used for years (Asimov).
### **Annual Dropout Rate**

Kentucky's state-reported dropout rates are below the national average except for students in grade 11. Table 2.6 shows state-reported dropout rates. Kentucky's annual dropout rates are below the national average for all grades except grade 11.

Table 2.6
Annual State-reported High School Dropout Rate by Grade, FY 2005

Total Grades 9-12		9-12	G	Grade 9 G		Gi	rade 10		Grade 11		Grade 12			
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
3	LA	7.5	1	LA	8.1	4	LA	6.6	3	LA	6.7	5	LA	8.3
7	GA	5.6	2	DE	7.2	5	GA	5.9	5	GA	5.8	11	AR	6.3
8	DE	5.3	5	GA	5.6	6	NC	5.6	7	NC	5.4	16	TX	5.0
9	NC	5.2	6	NC	5.2	9	DE	5.2	9	AR	5.3	17	GA	10
12	IL	4.5	8	MD	4.4	12	WV	4.4		DE		1 /	IL	4.9
15	AR	4.3	9	IL	4.3	16	IL	4.3	14	IL	15	-	U.S.	4.9
17	WV	4.1	15	OH	3.7	17	AR	4.0	14	MO	4.5	20	NC	4.7
19	MD	3.9	16	SC	3.6	10	MD	2.0		WV		22	OH	4.6
	U.S.	3.9	17	WV	3.4	10	MO	5.9	22	KY	4.0		FL	
22	MO	3.7	19	FL	3.3	21	KY	3.8	24	OK	3.9	23	MO	4.3
24	TX	3.6	20	OK	3.1		U.S.	3.8		U.S.	3.8		WV	
	FL			U.S.	3.1	23	OK	3.7	27	MD	26	27	KY	4.1
25	KY	2.5	25	TX	2.6	25	SC	26	27	TX	5.0	28	TN	4.0
23	OH	5.5	26	MO	2.5	23	TX	5.0	20	FL	2.5	31	IN	3.9
	OK			AR		27	FL	3.3	29	OH	5.5		DE	
31	SC	3.3	28	KY	2.4	29	AL	3.1	32	SC	3.3	36	MD	3.5
25	AL	20		MS		30	MS	3.0	24	IN	2 2		OK	
55	MS	2.0	21	AL	22	37	OH	2.3	34	TN	3.2	40	VA	3.4
38	TN	2.7	51	VA	2.3		IN		27	AL	2 1	42	AL	3.0
41	IN	25	35	TN	1.8	39	TN	2.2	57	MS	3.1	45	MS	2.7
41	VA	2.3	41	IN	1.3		VA	]	46	VA	2.1	46	SC	2.4

Notes: Ranks for the Total and Grade 9 are out of 46; data are not available for CT, DC, MN, NJ, and OR. Ranks for Grade 10 are out of 48; data are not available for DC, NJ, and OR. Ranks for Grades 11 and 12 are out of 49; data are not available for DC and OR. Ungraded students (those not in a standard grade) who drop out are assigned by the local education agency to the dropout count that most closely matches the grade they would have been in based on age. Ungraded student enrollments are prorated into grades based on graded enrollments to calculate denominators for dropout rates. Individual state total dropout rates are included only if the state reports dropouts and membership for each grade 9-12.

Source: U.S. Dept. of Ed. Inst. Natl. Numbers and Rates 7.

Over the past 3 years, dropout rates have not changed significantly for Kentucky or for the nation.

Table 2.7 presents annual dropout rates for FY 2003-FY 2005, the three most recent years available. During this time, rates have fluctuated but not changed significantly for Kentucky or for the nation as a whole.

	FY 2003			FY 2004		FY 2005			
Rank <sup>a</sup>	State	%	<b>Rank</b> <sup>b</sup>	State	%	Rank <sup>c</sup>	State	%	
3	LA	7.5	1	LA	7.9	3	LA	7.5	
6	GA	5.8	5	DE	6.1	7	GA	5.6	
7	IL	5.7	8	GA	5.4	8	DE	5.3	
8	DE	5.5	10	IL	5.3	9	NC	5.2	
10	NC	5.2	11	NC	5.2	12	IL	4.5	
13	AR	4.6	14	AR	4.7	15	AR	4.3	
17	OK	4.0	17	WV	4.3	17	WV	4.1	
	U.S.	3.9	19	MD	4.1	19	MD	3.9	
22	MS	3.7		U.S.	4.1		U.S.	3.9	
23	WV		20	OK	3.9	22	MO	3.7	
25	MD	2.6	24	TX	3.6	24	TX	3.6	
23	TX	5.0	25	FL	2.4		FL		
28	AL	3.5	23	SC	3.4	25	KY	2.5	
31	FL	3.4		AL		23	OH	5.5	
22	KY	2.2		KY			OK		
52	MO	5.5	29	MO	3.3	31	SC	3.3	
26	SC	2.2		OH		25	AL	20	
50	TN	5.2		TN		35	MS	2.0	
41	OH	2.0	36	MS	2.9	38	TN	2.7	
41	VA	5.0	38	VA	2.8	41	IN	2.5	
45	IN	2.2	42	IN	2.5	41	VA	2.5	

Table 2.7Annual High School Dropout Rate Trends, FY 2003 to FY 2005

Notes: <sup>a</sup>Rank out of 50; data not available for DC. <sup>b</sup>Rank out of 44; data not available for CT, DC, IA, MN, NJ, OR and WI. <sup>c</sup>Rank out of 46; data not available for CT, DC, MN, NJ, and OR. Source: U.S. Dept. of Ed. Inst. Natl. *Numbers and Rates 9*.

The Census Bureau's status dropout rate reflects the cumulative impact of annual dropout rates. The percentage of teenagers who have no type of high school credential and are not enrolled in school is 2.2 points higher in Kentucky than in the nation as a whole. Kentucky ranks 8<sup>th</sup> with respect to this measure.

### **Status Dropout Rate**

The status dropout rate, available from the Census Bureau, provides an overall measure of the cumulative impact that annual dropout rates have on educational attainment. Annual American Community Surveys, conducted by the Census Bureau, provide estimates of the percentage of high school-aged teenagers who are neither enrolled in school nor have attained a high school credential. It is important to note that the Census Bureau uses a much broader definition of high school credential, including not only standard diplomas earned within 4 years but also standard diplomas requiring more than 4 years, certificates of completion, GEDs, and other alternatives. As Table 2.8 shows, Kentucky's status dropout rate is ranked 8<sup>th</sup> in the nation and is 2.2 percentage points above the national average.

Table 2.8
Teenagers Aged 16-19 Not Enrolled in School and Having No
High School Credential, 2006

Rank	State	%
1	Louisiana	11.4%
2	Mississippi	10.4%
5	Georgia	9.1%
8	Alabama, Kentucky	8.8%
11	West Virginia	8.3%
12	Florida	7.9%
13	Indiana, South Carolina	7.6%
15	Oklahoma	7.5%
16	Texas	7.4%
17	North Carolina	7.3%
22	Delaware	6.9%
	United States	6.6%
26	Maryland	6.4%
29	Arkansas, Missouri, Tennessee	6.1%
37	Illinois	5.5%
39	Ohio	5.4%
40	Virginia	5.2%

Source: U.S. Dept. of Commerce. Census. "American Community Survey."

### **Achievement Assessments**

Achievement tests determine the degree to which students have accomplished the broad array of high school learning goals. Tests may be norm-referenced (comparing a student's performance to a distribution of other students' scores) or criterion-referenced (comparing a student's performance to specific standards and goals) or both. They may be nationally standardized or customized to a particular state and its unique core content.

While testing is indispensable to accountability, all tests have limitations, which users of test results should take into account. Even the most rigorously designed and administered tests are estimates, rather than exact measures, of a student's knowledge and skills. Some measurement error is inevitable. For example, a test is a sample of student abilities on one day, but student performance varies from day to day, depending on such factors as motivation, health, and distractions. Similarly, questions on any given test are only a sample of the entire domain of knowledge and

Tests are designed for different purposes. Achievement tests measure student mastery of a broad array of learning goals. College-readiness exams focus on measures that predict success in postsecondary education.

While testing is indispensable to accountability, all tests have limitations. They are estimates, rather than exact measures, of a student's knowledge and skills. The National Assessment of Educational Progress (NAEP) remains the only nationally representative and continuing assessment of American students' performance in reading, mathematics, science, writing, U.S. history, civics, geography, and the arts.

Participation in state-level NAEP tests was voluntary until 2003, when the No Child Left Behind Act (NCLB) began requiring all states to participate every 2 years in grades 4 and 8 math and reading tests.

NAEP scores are not reported for individual students or schools because they are based on a sample of approximately 2,500 students in 100 public schools per grade, per subject in each state.

Statistical testing indicates the likelihood that differences are real rather than due to sampling error. Because these tests take into account each state's sample size and variations in scores, two states with the same average score can have different levels of significance when compared to Kentucky. skills that could be tested. The constraints of a test environment make it impossible to adequately assess certain skills, such as the design and execution of complex projects (Way; Natl. Research).

### National Assessment of Educational Progress

The National Assessment of Educational Progress was created in 1964 to provide a national picture of student achievement. Today, NAEP remains the only nationally representative assessment of student achievement. NAEP assesses reading, mathematics, science, writing, U.S. history, civics, geography, and the arts (U.S. Dept. of Ed. Inst. Natl. *NAEP Overview*). Although it has limitations, which will be discussed below, NAEP is widely respected for its history as a national indicator, the quality of its design, its ability to assess both content and critical thinking, and the rigor of its standards (Barth; Pellegrino "Should"; Standard & Poor's).

NAEP assessments have been conducted at the national level since 1969, but state-level NAEP tests were not implemented until the early 1990s (U.S. Dept. of Ed. Inst. Natl. "About"). Since 2003, the No Child Left Behind Act has mandated biennial participation in NAEP grade 4 and grade 8 reading and math tests, for the stated purposes of comparing states' academic achievement and evaluating states' accountability systems (PL 107–110 Sec. 1501(a)(1)-(3); U.S. Dept. of Ed. Inst. Natl. "More About").

NAEP scores are not reported for individual students or schools because the tests are administered to samples of approximately 2,500 students in 100 public schools per grade per subject in each state. Samples are supplemented as needed to ensure representation of unique areas such as the state's only large city or only large concentration of minorities (U.S. Dept. of Ed. Inst. Natl. *How*).

Because NAEP results are based on samples of students, USDOE provides statistical tests that indicate, with 95 percent certainty, whether differences are likely real or simply due to random sampling error. In this compendium, statistical significance is shown in table columns headed "Sig.," with > indicating states that are significantly better than Kentucky, = indicating states that are not significantly different, and < indicating states that are significantly worse. Since statistical tests use unrounded percentages and take into account each state's sample size and variations in scores, two states with the same average score can have different levels of significance when compared to Kentucky.

NAEP policies are set by the National Assessment Governing Board (NAGB), an independent group of state, local, and federal officials, educators, business representatives, and members of the general public appointed by the U.S. Secretary of Education (U.S. Dept. of Ed. Natl. Assessment. *What*). NAGB oversees the testing frameworks and definitions of achievement levels. The Below Basic, Basic, Proficient, and Advanced achievement levels correspond to specified ranges of scores in each subject and grade. The general policy definitions shown in Table 2.9 are generic versions that provide the basis for more detailed definitions. Detailed definitions for each content area and grade can be found in Appendix D.

 Table 2.9

 General Policy Definitions of NAEP Achievement Levels

Advanced	Superior performance
Proficient	Solid academic performance for the grade tested; demonstrated competency over challenging subject matter, including subject-matter knowledge, application of such knowledge to real world situations, and analytical skills appropriate to the subject matter
Basic	Partial mastery of prerequisite knowledge and skills that are fundamental for proficient work at the grade tested

Notes: The "Below Basic" level is reported but not defined. See Appendix D for definitions tailored to each content area and grade.

Source: U.S. Dept. of Ed. Inst. Natl. "Interpreting."

NAEP assesses four aspects of students' reading abilities in three contexts.

**Reading.** The most recent state-level NAEP reading assessment took place in 2007. NAEP assesses four aspects of students' reading abilities in three contexts. These are described in Table 2.10, arranged in order of increasing difficulty for students.

Assessed of Deading of a Demonstrate of Test House, Dr. Cuede	Grade 4	Grade 8	
Aspects of Reading as a Percentage of Test Items, By Grade	Percent		
Form a general understanding: Consider the text as a whole and provide a global understanding of it	600/	550/	
<b>Develop interpretation</b> : Extend initial impressions to develop a more complete understanding; focus on specific parts of the text and link information across those parts	00%	33%	
<b>Make reader/text connections</b> : Think beyond the text, applying it to real-world situations, and own knowledge and experience	15%	15%	
<b>Examine content and structure</b> : Consider why and how the text was developed; consider content, organization, and form; critically evaluate, compare and contrast, and understand the effect of such features as irony, humor, and organization	25%	30%	
Contexts as a Percentage of Test Time, by Grade			
<b>Read for literary experience</b> : Explore events, characters, themes, settings, plots, actions, and the language of literary works by reading novels, short stories, poems, plays, legends, biographies, myths, and folktales	55%	40%	
<b>Read for information</b> : Gain information to understand the world by reading materials such as magazines, newspapers, textbooks, essays, and speeches	45%	40%	
<b>Read to perform a task</b> : Apply what is learned from reading materials such as bus or train schedules, directions for repairs or games, classroom procedures, tax forms (grade 12), maps, and so on (not assessed in grade 4)	0%	20%	

Table 2.10Framework for NAEP Reading Assessment, 2007

Source: Staff compilation based on U.S. Dept. of Ed. Inst. Natl. "Reading"; U.S. Dept. of Ed. Natl. Assessment. *Reading* vi, 6.

Reading proficiency of Kentucky's students in grades 4 and 8 is on par with the nation and compares favorably to peer states.

As Table 2.11 shows, the reading proficiency of Kentucky's grade 4 and grade 8 students is on par with the nation.<sup>1</sup> With about one-third scoring at levels considered proficient, Kentucky's grade 4 students performed well relative to those in peer states. None of the peer states had significantly higher grade 4 proficiency levels than Kentucky, while 10 had significantly lower levels.

In Kentucky, as in most other states, fewer grade 8 students than grade 4 students were deemed proficient. Kentucky's grade 8 reading proficiency level was significantly better than that in four peer states but significantly worse than the level in three other peer states.

<sup>&</sup>lt;sup>1</sup> Appendix D contains a description of the knowledge and skills that correspond to proficiency and other achievement levels for reading at each grade.

	Grade 4			Grade 8	
State	%	Sig.	State	%	Sig.
VA	38		OH	36	
OH	26		VA	34	>
MD	30		MD	33	
FL			IN		
DE	34		DE	31	
DE		_	MO		
KY	22		IL	30	
IN			U.S.	29	
IL	32		FI		
MO	52		1 L	28	
U.S.	32		NC		=
TX	30		KY		
NC			TX		
AL	29		OK		
AR			GA	26	
GA	29		TN		
WV	28	_	AR	25	
TN	27		SC	25	
ОК	27		WV	23	
SC	26		AL	21	
LA	20		LA	19	<
MS	19		MS	17	

Table 2.11Students At or Above Proficient on NAEP Reading Assessment, 2007

Note: > indicates states with proficiency levels significantly better than Kentucky's, = indicates states not significantly different, and < indicates states scoring significantly worse, with 95 percent certainty. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.

Reading proficiency rates are lower and improving at a slower rate for grade 8 students than for grade 4. Some attribute this to a lack of focus on reading in middle and high schools. Some point to a "middle school slump." Figure 2.G shows reading score trends for Kentucky and the U.S. Grade 4 scores have improved significantly since the first statelevel test in 1992, with Kentucky improving faster than the nation's. Grade 8 proficiency rates improved until 2003 but then fell lower than the first year tested. Critics attribute the lack of sustained improvement in grade 8 reading to educators' deeply rooted belief that students should "learn to read" in primary grades but then "read to learn" in later grades. Middle and high schools are urged to teach reading skills (Kuersten; Robb). Some believe a "middle school slump" occurs in most subjects, although the causes and very existence of this slump are debated (Herszenhorn; Leischer).



Figure 2.G NAEP Reading, Percent Proficient, U.S. and Kentucky, 1992 to 2007

Notes: ☆ indicates Kentucky's percent proficient in the indicated year was significantly different from Kentucky's percent proficient in 2007. ★ indicates the U.S. percent proficient in the indicated year was significantly different from 2007. Statistical tests assumed a 95 percent level of significance, used unrounded percentages, and took into account the sample size and variance. No state-level grade 8 reading test was administered before 1998.

Source: U.S. Dept. of Ed. Inst. Natl. NAEP Data and NAEP State.

NAEP writing assessments attempt to address as many facets of writing as possible within time constraints. **Writing.** The most recent state-level NAEP writing assessment took place in 2007, but it was only for grade 8. Grades 4 and 8 were tested at the state level in 2002.<sup>2</sup> In an effort to address as many facets of writing as possible within the time constraints of the NAEP writing test, NAGB defines six objectives, shown in Table 2.12.

## Table 2.12Framework for NAEP Writing Assessment, 2007

#### **Testing Objectives**

- Writing for a variety of purposes (narrative, informative, and persuasive)
- Writing for a variety of tasks and for many different audiences
- Writing with a variety of stimulus materials and within various time constraints
- Generating a draft, revising, and editing ideas and forms of expression
- Making effective choices in the organization of writing, including details to illustrate and elaborate ideas, and using appropriate conventions of written English
- Valuing writing as a communicative activity

	Gra	ade				
Writing Purposes as a Percentage of Test Time, by Grade						
<b>Narrative</b> : Emphasizes the writer's experiences, perceptions, and imagination; includes fictional or nonfictional stories, poems, plays, or personal essays; offers an opportunity to analyze and understand the emotions and actions of oneself and others	40%	33%				
<b>Informative</b> : Emphasizes the subject matter that is being explained; shares knowledge and conveys messages, instructions, and ideas; may be based on personal experience or secondary information and may involve recall, analysis, or evaluation	35%	33%				
<b>Persuasive</b> : Emphasizes the readers, in particular, influencing them and prompting them to take some action; requires awareness of arguments that most affect the audience being addressed and critical thinking skills, such as analysis, inference, synthesis, and evaluation	25%	33%				

Source: Staff compilation based on U.S. Dept. of Ed. Natl. Assessment. Writing 7, 11-14, 54.

Kentucky's grade 8 students performed significantly below the nation in 1998, 2002, and 2007. Kentucky's grade 4 students, tested only in 2002, performed on par with the nation. Table 2.13 presents the percentage of students found to be proficient or advanced in writing. Kentucky's grade 4 students performed on par with the nation in 2002. However, Kentucky's grade 8 students performed significantly below the nation in both 2002 and 2007. As figure 2.I shows, grade 8 writing scores have increased since the first NAEP writing assessment in 1998. However, they improved less between 2002 and 2007 than between 1998 and 2002. Trends are not available for grade 4 writing because it was assessed at the state level only once in 2002.

<sup>&</sup>lt;sup>2</sup>The 2002 and 2007 NAEP writing tests were based on the same framework.

	Grade 4		Grade 8						
	2002			2002		2007			
State	%	Sig.	State	%	Sig.	State	%	Sig.	
DE	35	/	OH	38		IL	37		
FL	33	/	DE	25		FL	36		
NC	32		MD	33		DE	34		
MD	30		NC	34		OH	32		
TX	20		FL	22		VA	31		
VA	29		VA	52		U.S.	31		
OH	28		TX	31		IN	20		
KY	27	=	U.S.	30		TN	30		
U.S.	27		MO	27		GA	20		
IN	26		OK	27		NC	29		
GA	23		IN	26		AR	27		
TN	23		GA	25	=	MO	21	_	
MO	22		KY	23		KY		_	
AR	19		TN	24		OK	26		
WV	17		WV	21		TX			
SC	17		AL	20		AL	24		
OK	16	<	SC	20		SC	23		
AL	15		AR	19	<	WV	22		
LA	14		LA	18		LA	17	/	
MS	13		MS	13		MS	15		
IL	n.a.	n.a.	IL	n.a.	n.a.	MD	n.a.	n.a.	

Table 2.13Students At or Above Proficient on NAEP Writing Assessment, 2002 and 2007

Notes: > indicates states with proficiency levels significantly better than Kentucky's, = indicates states not significantly different, and < indicates states scoring significantly worse, with 95 percent certainty. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.



Figure 2.H NAEP Grade 8 Writing, Percent Proficient, U.S. and Kentucky 1998, 2002, and 2007

Notes: **★** indicates the U.S. percent proficient in 1998 was significantly different from 2007. Changes in Kentucky's percent proficient were not statistically significant. Statistical tests assumed a 95 percent level of significance, used unrounded percentages, and took into account the sample size and variance. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.

NAEP tests five math content areas at three levels of complexity. **Mathematics.** The most recent NAEP math assessment took place in 2007. Grades 4 and 8 had sufficiently large sample sizes for reporting at the state level. As shown in Table 2.14, NAEP tests five mathematics content areas, at three levels of complexity (U.S. Dept. of Ed. Natl. Assessment. *Mathematics* 9, 35).

## Table 2.14Framework for NAEP Mathematics Assessment, 2007

Content Areas as a Percentage of Test Items, by Grade	4	8			
	Per	cent			
Number Properties and Operations	40	20			
Measurement	20	15			
Geometry	15	20			
Data Analysis and Probability	10	15			
Algebra	15	30			
Levels of Complexity as a Percentage of Test Items (same for all grades)					
Low-complexity item may ask a student to recall a property.	2	5			
Moderate-complexity item may ask a student to make a connection between two properties.	5	0			
High-complexity item may ask a student to analyze the assumptions made in a mathematical model.	2	5			

Source: Staff compilation based on U.S. Dept. of Ed. Inst. Natl. "Mathematics"; U.S. Dept. of Ed. Natl. Assessment. *Mathematics* 9, 35-42.

In 2007, Kentucky's grade 4 and grade 8 students performed significantly below the national average in math. In both Kentucky and the nation, proficiency is lower for grade 8 than for grade 4. Kentucky's grade 4 and grade 8 students performed significantly below the national average on the 2007 NAEP math test, as evident in Table 2.15. For both Kentucky and the nation, proficiency levels are consistently lower for grade 8 students than for grade 4 students.

	Grade 4		Grade 8				
State	%	Sig.	State	%	Sig.		
IN	16		MD	27			
OH	40		VA	57			
VA	42		IN				
NC	41		OH	35			
DE			TX				
FL	40		NC	34			
MD	40	>	SC	32			
TX			U.S.	31			
U.S.	39		DE				
MO	38		IL	51			
AR	37		MO	30			
IL	26		FL	27			
SC	50		KY	21	=		
OK	22		GA	25			
WV	55		AR	24			
GA	32	=	TN	23			
KY	31		OK	21			
TN	29		LA	10	_		
AL	26		WV	19			
LA	24	<	AL	18			
MS	21		MS	14			

Table 2.15Students At or Above Proficient on NAEP Math Assessment, 2007

Notes: > indicates states with achievement levels significantly better than Kentucky's, = indicates states that are not significantly different, and < indicates states that are significantly worse than Kentucky, based on statistical testing with a 95 percent significance level. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.

Kentucky has made steady progress in math but remains below national proficiency rates.

Figure 2.J presents trends in math proficiency levels from 1992 to 2007 for grade 4 and 1990 to 2007 for grade 8. In Kentucky, both grades made steady progress every year except for a 1-year drop for grade 8 students in 2005. The national proficiency rate for grade 4 rose even faster than Kentucky's, widening the gap between Kentucky and the nation. For grade 8 students, the gap narrowed slightly between Kentucky and the nation.



Figure 2.I NAEP Math, Percent Proficient, U.S. and Kentucky, 1990 to 2007

Notes: State-level NAEP math assessments started in 1992 for grade 4 and 1990 for grade 8.  $\Rightarrow$  indicates Kentucky's percent proficient in the indicated year was significantly different from Kentucky's percent proficient in 2007.

 $\star$  indicates the U.S. percent proficient in the indicated year was significantly different from 2007. Statistical tests assumed a 95 percent level of significance, used unrounded percentages, and took into account the sample size and variance.

Source: U.S. Dept. of Ed. Inst. Natl. NAEP Data.

Science. After a review of curriculum guides, frameworks, and other course outlines from around the nation, NAGB noted that the traditional approach to teaching science tends to emphasize rote memorization of facts without connection or organization. While acknowledging the value of factual knowledge, NAGB argues that science is easier to remember and more useful when information is organized into broad conceptual understandings and when knowledge and skills are applied to practical situations (U.S. Dept. of Ed. Natl. Assessment. Science 11-12). The NAEP science assessment reflects this position.

Table 2.16 summarizes the content of the NAEP science assessment, in which each exercise measures one of the elements of knowing and doing within one of the fields of science. In addition, every grade level is given a few interdisciplinary exercises that integrate earth, physical, and life sciences, along with content pertaining to technology (U.S. Dept. of Ed. Natl. Assessment. Science 13-14). Half of the students in each school receive one of three hands-on tasks and related questions. These students conduct actual experiments using materials provided to them and record their observations and conclusions in their test booklets by responding to multiple-choice and open-response questions (U.S. Dept. of Ed. Inst. Natl. "Science").

Table 2.	16
Framework for NAEP Scie	nce Assessment, 2005

Percentage of Test Items Devoted to Each Element, by Grade	Gr	ade	
	4	8	
Fleids of Science	Percentage		
Earth Science (such as geology, meteorology)	33.3%	30%	
Physical Science (physics)	33.3%	30%	
Life Science (biology, ecology)	33.3%	40%	
Knowing and Doing Science (exercises often combine more than one)			
Conceptual Understanding: Demonstrate knowledge and skills relating to facts learned in class and in nature; scientific concepts, principles, laws, and theories; procedures for conducting scientific inquiries; application of knowledge in practical tasks; and interactions between science, technology, and society	45%	45%	
Scientific Investigation: Demonstrate abilities to acquire new information, plan investigations, use scientific tools, and communicate results to a variety of audiences	45%	30%	
Practical Reasoning: Demonstrate abilities to think abstractly, consider hypothetical situations, consider several factors simultaneously, take an objective view, and realize the importance of practical reasoning and experience	10%	25%	
Nature of Science: Themes of Systems, Models, and Patterns of Change	A few ex	ercises at	
(interdisciplinary exercises integrating the three sciences, along with technology topics)	each grade level		

Source: U.S. Dept. of Ed. Inst. Natl. "Science"; U.S. Dept. of Ed. Natl. Assessment. Science 11-27.

Each NAEP science exercise measures one of the elements of knowing and doing within one of the fields of science. In addition, a few interdisciplinary exercises integrate the sciences, along with technology topics.

In 2000, Kentucky's science scores were on par with the nation. Five years later, Kentucky performed significantly above the national average, with science scores unsurpassed by any other state NAEP assessed science at the state level in 2000 and 2005. Proficiency rates are shown in Table 2.17. In 2000, Kentucky's grade 4 and grade 8 students performed on par with the nation. Five years later, Kentucky's grade 4 and grade 8 students performed significantly above the national average. When statistical significance is considered, no other state surpassed Kentucky's performance on grade 4 science in 2005.

		20	00					20	05		
	Grade 4	-		Grade 8			Grade 4			Grade 8	
State	%	Sig.									
MO	34	>	OH	39		VT	38		SD	41	
IN	22		IN	22	>	KY	26		VT	41	/
VA	32		MO	33		MO	50	=	MO	22	
IL	21		IL	20		SD	35		UT	33	
OH	51		VA	29		UT	33		OR	32	
KY	28	=	U.S.	29		DE			KY	31	=
OK	26		KY	28		IL	27		DE	20	
U.S.	26		MD	27 -	_	IN	21		IN	29	
MD			NC	25		MD			IL	27	
TN	24		OK	23		U.S.	27		U.S.	27	
WV			TN	24		FL	FL 26		MD	26	
AR			WV	24		OR			GA	25	
GA	22		AL	22	<	GA			OK	23	
NC	23		GA	25		NC	25		AR		
TX		<	TX	23	=	OK	23		TX	23	/
AL	22		AR	22		TX			WV		
SC	20		SC	20		AR	24		NC	22	
LA	18		LA	18	<	WV	24		FL	21	
MS	13	1	MS	15	1	AL	21		AL	10	
DE	l l		DE			LA	20		LA	19	
FL	n.a.	n.a.	FL	n.a.	n.a.	MS	12		MS	14	

## Table 2.17Students At or Above Proficient on NAEP Science Assessment, 2005

Notes: The NAEP science test remains voluntary. The number of states choosing to participate was 39 in the 2000 grade 4 test, 38 in the 2000 grade 8 test, and 44 in the 2005 grades 4 and 8 tests. Statistical significance can differ for states with the same average score because statistical tests use unrounded percentages and take into account each state's sample size and variation in scores. > indicates states scoring significantly better than Kentucky, = indicates states that are not significantly different, and < indicates states scoring significantly worse than Kentucky, based on statistical testing with a 95 percent significance level.

Source: U.S. Dept. of Ed. Inst. Natl. NAEP Data.

**NAEP Achievement Gaps.** This section discusses subgroup differences that are statistically significant at the 95 percent level for the latest NAEP assessment in each content area and grade. The subgroups compared were males versus females, whites versus African Americans, and free or reduced price-lunch-eligible versus noneligible.

Females score higher than males on reading and writing, while males score higher on math and science. In grade 4 writing and math, Kentucky's gender gaps are wider than the nation's. As Figure 2.J shows, females score higher than males in reading and writing, while males have small but statistically significant leads in math and science. The only gender gap that is not statistically significant is in grade 8 math in Kentucky. In grade 4 writing and math, gender gaps are significantly wider in Kentucky than the U.S.. In both Kentucky and the nation, gender gaps have not changed significantly since state-level testing began in 1990.

Figure 2.J NAEP Achievement Gaps by Gender, Most Recent Assessments



Notes: All gender differences are statistically significant except grade 8 math in Kentucky. Scores are on a 0-to-500 scale.

Source: U.S. Dept. of Ed. Inst. Natl. NAEP Data.

Whites score higher than African Americans in all content areas. However, gaps have been narrowing steadily. Kentucky's racial gaps are smaller than the nation's. Figure 2.K also shows that, for the nation as a whole, whites score significantly higher than African Americans in all content areas. However, gaps have been narrowing steadily in grade 4 reading and were narrower in the most recent assessments of grades 4 and 8 writing, grade 8 math, and grade 4 science. Racial gaps in Kentucky are smaller than national gaps and have not changed significantly over time.



Figure 2.K NAEP Achievement Gaps by Race, Most Recent Assessments

Notes: All race differences are statistically significant. Scores are on a 0-to-500 scale. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.

Students eligible for free or reduced-price lunch score lower than other students in all content areas. Kentucky's gaps are smaller than the nation's. As Figure 2.L shows, students eligible for free or reduced-price lunch—a proxy for poverty—score lower than ineligible students in all content areas. Kentucky's gaps are smaller than the nation's. Over time, poverty gaps for science have narrowed for Kentucky and for the nation as a whole. In the other content areas, gaps have fluctuated but have not changed significantly.



Figure 2.L NAEP Poverty Gaps

Notes: F/RL=Free or reduced-price lunch. All differences between eligible and ineligible students are statistically significant. Scores are on a 0-to-500 scale. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.

Due to sampling error, differences in NAEP scores must be tested for statistical significance. The definitions for proficiency and other achievement levels have not yet been officially approved. Exclusions and accommodations raise questions about comparability over time and across states. Although NAEP is widely respected, it shares the limitations of all assessments. In addition to those already discussed, other limitations include sampling error, achievement levels still being used on a trial basis, and uncertainty about the impact of exclusions and accommodations.

**Sampling Error.** NAEP assessments are administered at about 100 randomly selected public schools in each state for each subject at each grade for the sample. Consequently, test results are subject to sampling error, and differences should be tested for statistical significance.

Achievement Levels Still Used on Trial Basis. The basic, proficient, and advanced achievement levels have never been formally approved (U.S. Dept. of Ed. Inst. Natl. "The Status"). By law, they are still being used on a trial basis, awaiting a determination that they are "reasonable, valid, and informative to the public" (PL 107-110).

For the first two decades of NAEP's existence, results were reported only as numeric scores. Concerned that these scores were difficult to interpret, Congress voted in 1988 to require the development of "appropriate achievement goals" (PL 100-297). Controversy has surrounded the creation of achievement levels from the outset. Many approaches to defining and setting achievement standards are available to policy makers, but consensus as to what constitutes the best approach has not been achieved (Vinovskis 41, 84; U.S. Government. *Educational*). Congressionally mandated evaluations of NAEP by such organizations as the National Academy of Education, the GAO, and the National Academy of Sciences have found a number of technical flaws that they believe have not been fully addressed (Shepard; U.S. Government. *Educational*; Pellegrino, Jones, and Mitchell).

One concern is that the cut scores (minimum scores) that define proficiency are set unreasonably high, a charge that resonates with many testing critics (Pellegrino "Should"; Rothstein; U.S. Government. *Educational*). Two studies suggest that countries that rank at the top on international assessments would show relatively low proficiency levels on NAEP (Phillips 9; Rothstein 32).

**Exclusions and Accommodations**. NAEP, like many assessments, permits students with disabilities or limited English proficiency to use certain accommodations, such as Braille or bilingual test forms, in order to fairly and accurately demonstrate their abilities. If it appears that a student cannot meaningfully participate even with permitted accommodations, the student is excluded from the assessment altogether. Decisions regarding accommodations and exclusions are made by personnel working at the individual schools participating in NAEP. In an effort to guide these decisions, NAEP stipulates that a questionnaire be filled out for each student by the person most knowledgeable about that student. However, studies have found that the person filling out the questionnaire is often the school's test coordinator, who may not have detailed knowledge about students affected by decisions (U.S. Dept. of Ed. Inst. Natl. *NAEP Inclusion Policy*; Stancavage 22).

NAEP exclusions can be affected by state policies. For example, if a state test permits a certain accommodation that is not allowed by NAEP, students who use that accommodation on state assessments can be excluded from NAEP. For Kentucky, the most notable difference is that Kentucky permits reading test items to be signed or read aloud, while NAEP does not<sup>3</sup> (U.S. Dept. of Ed. Inst. Natl. *NAEP Inclusion Policy* and *Students with Disabilities* 7; Commonwealth. Dept. of Ed. *Inclusion*).

Inconsistent practices with respect to exclusions and accommodations among states and over time have raised concerns about comparability. In recent years, policy makers have become increasingly concerned about variations in NAEP exclusion and

<sup>&</sup>lt;sup>3</sup>The justification for allowing readers for reading tests, stated in documentation incorporated by reference to 703 KAR 5:070, is that the intent of the KCCT reading test is to assess comprehension.

accommodation rates from state to state and from year to year within a state. Several studies, many commissioned by NAGB, have investigated these concerns, with no clear conclusions. A new federal ad hoc committee convened recently to investigate these issues further (Cavanagh).

Concerns have been raised that Kentucky excludes and accommodates more students than the national average and that these practices inflate NAEP scores to such an extent that they cannot be trusted (Innes. "'CATS-SCAN'"). To date, there is conflicting evidence with respect to these claims, as will be discussed below. This issue warrants continued monitoring due to its potential for skewing results.

As Table 2.18 shows, Kentucky's exclusion rates are not consistently above the national average. In fact, before 2002, Kentucky's exclusion rates were below national rates. Since 2002, exclusion rates for reading have been between 1 and 3 percentage points higher than the nation, while math exclusion rates have fluctuated between 1 percentage point below the nation and 3 points above. An analysis focusing on students with disabilities shows a similar pattern.

For years, the percentages of students excluded from NAEP due to disabilities or language barriers in Kentucky's have mirrored the nation as a whole. However, exclusion rates have risen slightly in recent years, especially for reading. Exclusions based on disabilities alone show similar patterns.

Chapter 2

					•									
		1990	1992	1994	9661	1996*	1998	*8661	2000	2000*	2002*	2003*	2005*	2007*
Destine	KY		4	4			9	7			9	9	8	8
Grade 4	U.S.		6	6			10	7		6	7	6	7	6
Glade 4	KY – U.S.		-2	-2			-1	0			2	3	1	
Destine	KY						5	3			7	7	7	8
Grade 8	U.S.		7	7			6	4			6	5	5	5
Glade 8	KY – U.S.						-1	-1			1	2	2	3
	KY		3		6				8	3		3	3	3
Math Grade 4	U.S.		7		6	4			7	4		4	3	3
Glade 4	KY – U.S.		-4		0				1	-1		-1	0	0
Math Grada 8	KY	5	5		5				9	4		4	3	7
	U.S.		6		5				7	4		4	4	4
Grade o	KY – U.S.		-1		0				2	0		0	-1	3

 
 Table 2.18

 Percentages of Students in NAEP Sample Excluded From Assessment, Kentucky and U.S., 1990-2007

Note: \*Accommodations Permitted. "KY - U.S."=Kentucky exclusion rate minus U.S. rate; this number is positive when Kentucky excludes a relatively higher percentage of students and negative when Kentucky excludes a smaller percentage of students.

Source: U.S. Dept. of Ed. Inst. Natl. State Exclusion.

In 2007, Kentucky's exclusion rates tended to be slightly above national rates, while accommodation rates were slightly below. Kentucky's rates would mirror the nation if some students currently being excluded were tested with accommodations. NAEP allows accommodations in order to include as many students as possible. Figure 2.M shows accommodations and exclusions for students with disabilities and English Language Learners in the U.S and Kentucky. Because Kentucky has relatively few English Language Learners, Figure 2.N shows exclusions and accommodations on the basis of disability only. As both figures show, Kentucky has below-average accommodation rates and above-average exclusion rates. This suggests that Kentucky could mirror the nation if some of the students who are being excluded were, instead, offered appropriate accommodations.



Notes: Percentages may not add to 100 percent due to rounding. Source: U.S. Dept. of Ed. Inst. Natl. *Mathematics Report Card* and *Reading Report Card*.

### Figure 2.N





Notes: Percentages may not add to 100 percent due to rounding. Source: U.S. Dept. of Ed. Inst. Natl. *Mathematics Report Card* and *Reading Report Card*.

Extensive studies found the impact of accommodations on scores to be within acceptable limits, but the impact of exclusions is still being studied. After 6 years of study, the USDOE found the impact of accommodations on scores to be within acceptable limits. It is for this reason that NAEP now permits accommodations for the entire sample (U.S. Dept. of Ed. Inst. Natl. *A Closer*).

Exclusions are still being studied closely because of their potential impact and their variation from state to state and from year to year. Since students with disabilities or limited English proficiency tend to score lower on NAEP, it is logical to assume that excluding more of these students will lead to higher test scores. However, to date, numerous studies exploring this topic have not found consistent or dramatic evidence to warrant this conclusion (U.S. Dept. of Ed. Inst. Natl. *Investigating* and *A Closer*). As Table 2.19 shows, special analyses suggest that exclusions had small and inconsistent effects on changes in Kentucky's scores between 2003 and 2005. Exclusions may have made reading scores look better but made math scores look worse.

# Table 2.19Increases and Decreases in Average NAEP Scores From2003 to 2005 With and Without Exclusions

	Official Results (With Exclusions)	Estimated Results For Full Population (Without Exclusions)
Reading, Grade 4	0.9	0.7♥
Reading, Grade 8	-2.3	-2.8♥
Math, Grade 4	2.8	2.9
Math, Grade 8	-0.3	0.2

Notes: Scores are based on 0-to-500 scales. Official results are the number of points on the scale that were gained or lost (if negative number) between the 2003 and 2005 assessments. Estimated results are the number of points on the scale that would have been gained or lost if exclusions were not permitted; these are based on a statistical model.  $\Psi$ =Scores would have improved less if there had been no exclusions.  $\Lambda$ =Scores would have improved more if there had been no exclusions.

Source: U.S. Dept. of Ed. Inst. Natl. Investigating.

**Comparing NAEP to State Assessments.** Proficiency rates on NAEP assessments are usually lower than rates on tests that states design, administer, score, and report for No Child Left Behind purposes. The discrepancies vary greatly by state and appear to be growing over time (Cary; Peterson; Fuller and Wright; Fuller et al. *Is the*; Olson. "Gaps").

Given the complexities of public policy and the education testing systems, many factors may be responsible for these discrepancies.

NAEP consistently reports fewer proficient students than do the assessments that states design, administer, and report for NCLB purposes. Several factors could be operating simultaneously, and the mix of factors could be different in each state. Some possible explanations for the discrepancies are discussed below.

For the purposes of this discussion, it is important to note the distinction between "content standards" and "performance standards." A state's content standards—detailed documents describing what students should know and apply—are meant to guide the design of curriculum, instruction, and assessment within a particular state. In contrast, performance standards pertain specifically to assessment and are typically in the form of descriptions and cut scores (minimum scores) for each achievement level, such as NAEP's basic, proficient, and advanced or Kentucky's novice, apprentice, proficient, and distinguished. Performance standards are not necessarily tied to a state's curriculum.

**Students Can Perform Better on State Tests Than on NAEP**. Differences in proficiency on NAEP and states' own tests reflect, in part, true differences in student performance. Given differences in test content and higher stakes for test takers and administrators, students can perform better on state tests than on NAEP, as discussed below.

**State Tests Align With Students' Opportunities To Learn**. This issue is central for many who oppose comparing NAEP and state test results. The No Child Left Behind Act requires state assessments to be aligned with the subject areas, depth of knowledge, and skills specified in the state's content standards, which also guide the curriculum and teaching in that state (PL 101-110 Sec. 1111(b)(3)(C)(ii)). Thus, a student should perform better on a state test that is tailored to the student's everyday learning experiences than on a "one size fits all" standardized test.

A 2003 study identified several differences between the Kentucky Core Content Test (KCCT) and NAEP, including the standards each test is designed to measure, the content and format of test items, the language used within the items, test administration guidelines, and psychometric processing of the raw test scores (Koger). Yet despite these differences, another study found correlations of student performance on KCCT and NAEP to be moderate to strong, ranging from .55 for elementary level reading

Higher proficiency levels on state tests compared to NAEP are partly due to better alignment of state tests with what students have an opportunity to learn in each state. However, it is unknown how much these differences in alignment contribute to differences in scores. to .74 for high school math (Hoffman 6).<sup>4</sup> The study authors concluded that

Student ability tends to be highly correlated irrespective of the measurement instrument used or the content measured. Students who score well on one reading test tend to score well on all reading tests, and in fact tend to score well on tests of mathematics and other subjects, as well (Hoffman 1).

Studies have found a number of differences between NAEP and other state assessments. However, none of these findings provides an unambiguous indication as to whether NAEP is aligned enough with state assessments to permit accurate comparisons and rankings. A recent analysis of test scores suggested that threefourths of state tests have "at least a moderate degree of alignment" with NAEP, with correlations ranging from .62 to .86 (U.S. Dept. of Ed. Inst. Natl. *Mapping* 59). However, USDOE recommended that this analysis be supplemented by examining alignment of test frameworks between NAEP and state tests. One research team found that state content standards do not agree with each other, much less with NAEP, yet the team is not pessimistic:

> On the question of whether state content standards represent a de facto national curriculum, the answer is 'no', but that is not to say that each state's content standards have content messages completely non-overlapping with each other state's content standards. Some might say that it's a question of seeing the glass two thirds empty or one third full (Porter, Polikoff, and Smithson 29).

Most testing experts stress the crucial importance of alignment but rarely venture to say how much alignment is enough. Answering that question is as difficult as deciding how much statistical confidence is enough. There is no absolute criterion (Porter. *Curriculum* 14).

**High Stakes on State Tests Can Boost Performance**. NAEP has been called a "no stakes" test (Standard & Poor's). In contrast, the sanctions and rewards associated with some states' tests can motivate districts, schools, teachers, and students to try harder and to target more resources where needed (Stecher 90-91, 96-97). In addition, the incentives surrounding high stakes testing can promote practices that are undesirable but that boost scores, such as reassigning the best teachers to the assessed grades, narrowing

Higher proficiency rates on state tests compared to NAEP may be due, in part, to the higher stakes of state tests. Whereas NAEP has been called a "no stakes" test, the sanctions and rewards that accompany state tests might motivate districts, schools, teachers, and students to try harder.

<sup>&</sup>lt;sup>4</sup> The correlation coefficient, a statistical measure of the strength of a relationship, can range between a perfect negative correlation of -1 and a perfect positive correlation of 1.

instruction to subjects and topics that are tested, making classroom instruction more like a test, and spending an excessive amount of time familiarizing students with the format of test questions and how to record answers (Stecher 91-97).

**Performance Standards for NAEP Appear To Be Higher Than for State Tests**. Although differences between tests make comparisons difficult, many believe that performance standards for NAEP are higher than for state assessments. If this is true, it raises the question: Are state assessment standards too low or are NAEP standards too high?

**Research Suggests NAEP Standards Too High**. As mentioned earlier, some say NAEP's cut scores (minimum scores) for defining proficiency may be set too high. NAEP seems to have higher standards than two internationally accepted tests (Phillips 9; Rothstein 32).

Many Deem State Standards Too Low. Others attribute the discrepancies to lower state standards, sometimes even intentional inflation of performance in order to avoid embarrassment and sanctions (Cary; Innes "CATS-SCAN"; Olson. "Gaps"; Peterson; Ravitch).

A recent analysis attempted to quantify differences in standards by "mapping" state proficiency levels onto the NAEP scale (U.S. Dept. of Ed. Inst. Natl. *Mapping*). It is important to note that critics have pointed out several limitations to the analysis, especially the inability to account for nonalignment between NAEP and state tests (Ho *(Over)-Interpreting* and *Apples*). However, the USDOE insists that the analysis is a useful, albeit rough, indicator of state performance standards.

The analysis of grade 4 reading suggests that states' proficiency standards are all below NAEP's proficiency standard; in fact, most are even below NAEP's standard for basic. Kentucky is one of the states whose proficiency standard falls below basic; yet Kentucky's standard is the 10<sup>th</sup> highest among the 32 states examined.

States' grade 8 math standards for proficiency appear to be somewhat closer to NAEP's. Three states—Missouri, South Carolina, and Massachusetts—have a proficiency standard above NAEP's. Most others, including Kentucky, have proficiency standards above NAEP basic but below NAEP's proficiency cut

Many believe that NAEP has higher performance standards than do state tests. There is some evidence that NAEP standards may be too high, but there is more evidence that state standards are too low. score. Kentucky's proficiency standard ranks 8<sup>th</sup> among the 36 states examined.

**Comparing NAEP to State Tests: Conclusions**. Although both NAEP and states such as Kentucky use the term "proficient," the term refers to different definitions and measures. When NAEP and state tests appear to conflict, there are compelling reasons to favor NAEP, including its ample funding for hiring top experts; its nationwide scrutiny, which encourages continual improvement; and its lack of incentive to keep standards low (intentionally or unintentionally) in order to avoid embarrassment and sanctions. On the other hand, as discussed earlier, expert evaluators have strongly criticized the process whereby NAEP performance standards were set. There is no universally accepted definition for proficiency. Given the measurement challenges discussed above, comparisons between NAEP and state test results should be made with caution and backed up with multiple sources of evidence.

Some policy makers have called for replacing the 50 different sets of state standards with one national set of standards based on NAEP (Hoxby; Olson. "Standards"). However, these efforts have stalled thus far due to strong resistance to what is perceived as federal intrusion into state education policy. After voting to establish a National Education Standards and Improvement Council, Congress reversed its decision and abolished the council before ever naming any of its members (Olson. "An 'A"").

### **College-readiness Exams**

College-readiness exams were originally designed for use by college admissions officers, in combination with high school grades and other information, to gauge a student's chances for success in college (College Board. "SAT Program"). An increasing number of states are incorporating these tests into their accountability systems, as an indicator of how well the K-12 education system is preparing students for postsecondary education. These tests may be norm referenced, criterion referenced, or both.

### The Higher the Participation, the Lower the Scores

Caution should be used when comparing college entrance exam scores across states because the pool of test takers varies widely. Participation rates can be as low as 3 percent or as high as 100 percent, and the higher the participation rate, the lower the average

Caution should be used when comparing college entrance exam scores because higher participation is associated with lower average scores. Participation rates vary from 3 percent to 100 percent. score. This is particularly true for the SAT, with about threefourths of the differences among scores being accounted for by state participation rates.<sup>5</sup>

### **Test Design**

Another important caveat to consider is that the ACT and SAT were designed primarily to predict the success of college-bound high school seniors. The tests are meant to be the most sensitive to differences among top students rather than among struggling students. In contrast, achievement tests are designed to measure a school's success at teaching a comprehensive range of content to students at all ability levels. The tests provide different insights into content delivery and mastery.

### ACT

The ACT test consists of multiplechoice tests that cover English, mathematics, reading, and science. The test also includes a relatively new optional writing test. The ACT consists of multiple-choice tests that cover English, mathematics, reading, and science.<sup>6</sup> The test also includes an optional writing exam that entails planning and writing a short essay. The maximum score for each test is 36. The ACT composite score is the average of the scores on the four multiple-choice tests. Table 2.20 shows the distribution of items by content area.

Table 2.20ACT Test Items by Content Area

ACT College Readiness Test	Number of
<b>Content Areas</b>	Multiple-choice Items
English	75
Math	60
Reading	40
Science	40
Total	215

Source: ACT. ACT Newsroom.

<sup>&</sup>lt;sup>5</sup> The correlation between ACT participation rates and average ACT composite scores is -0.45. The correlation between SAT participation rates and average scores is even stronger: -.9 for reading and -.87 for math. The correlation coefficient, a measure of the strength of a relationship, can range between a perfect negative correlation of -1 and a perfect positive correlation of 1. The square of these correlation coefficients serves as an estimate of the amount of variation in average scores that can be explained by different participation rates. Therefore, participation rates explain about one-fifth of state differences in average ACT scores and about three-fourths of differences in SAT scores. <sup>6</sup>Appendix B contains a brief profile of ACT, Inc., the organization that creates and scores the ACT test.

Average ACT scores are not comparable across all states because participation rates vary. Comparability is better among states that have similar participation rates. It is evident from Table 2.21, which presents ACT participation rates and test scores, that participation rates vary widely across states. Because different levels of participation indicate different pools of test-takers, scores are not comparable across all states. Comparability is better among states with similar participation rates; for example, Kentucky's scores are relatively comparable to those of Louisiana and Arkansas because the percentages of public and private school students taking the ACT in those states were similar to the percentage in Kentucky. However, it should be noted that even when two states have identical participation rates, there is no guarantee that the characteristics of test takers are the same.

 Table 2.21

 ACT Participation Rates and Average Scores for High School Graduates Tested, 2007

Par	ticipat	tion	Co	ompos	ite	ŀ	Englis	h		Math		R	leadin	g	Science		
Rank	State	%	Rank	State	Score	Rank	State	Score	Rank	State	Score	Rank	State	Score	Rank	State	Score
1	IL	100	15	IN	22.0	16	IN	21.5	14	IN	22.0	10	IN	22.5	16	IN	21.7
2	MS	06	23	DE	21.7	10	MO	21.3	21	DE	21.6	24	MO	22.1	19	OH	21.6
3	TN	90	25	MO	21.6	23	DE	21.2	26	NC	21.4	28	OH	22.0	22	MO	21.5
6	AL	81	23	OH	21.0	27	OH	21.0	28	OH	21.3	30	DE	21.9	25	DE	21.4
7	LA	79	32	VA	21.4	21	VA	21.0	31	VA	21.2	34	VA	21.7	33	VA	21.1
9	KY	77		U.S.	21.2	20	TN	20.8	35	MO	21.0		U.S.	21.5		U.S.	21.0
13	AR	75	35	NC	21.0	29	WV	20.8		U.S.	21.0	35	NC	21.4	35	NC	20.7
14	MO	74		KY			U.S.	20.7	36	ΤХ	20.8	36	OK	21.3	36	KY	20.6
15	OK	71	36	OK	20.7	26	AR	20.5	37	IL	20.4	27	KY	21.2	27	OK	20.5
20	OH	68		TN		50	OK	20.3	38	GA	20.3	57	WV	21.2	57	WV	20.5
21	WV	66	39	WV	20.6		AL		40	FL	20.0	39	TN	21.1		IL	
26	FL	54		AR		38	KY	20.3	40	KY	20.0	40	AR	20.9	39	TN	20.4
27	SC	43	40	IL	20.5		LA		42	AR	10.0	43	AL	20.7		ΤХ	
	U.S.	42		ΤХ		41	IL	20.2	42	TN	19.9	4.4	GA	20.6	43	AR	20.2
28	GA	34	11	AL	20.2	41	NC	20.2	44	OK	10.9	44	ΤХ	20.0	45	AL	20.1
30	ΤХ	30	44	GA	20.5	44	GA	19.9	44	SC	19.0	16	FL	20.5	45	GA	20.1
34	IN	21	47	LA	20.1	47	ΤX	19.5		AL		40	IL	20.3	47	LA	19.9
37	VA	18	48	FL	19.9	48	FL	19.1	47	LA	19.5	48	LA	20.2	18	FL	10.5
40	NC	16	49	SC	19.6	40	MS	10.0		WV	]	49	SC	19.8	40	SC	19.5
50	DE	9	50	MS	18.9	49	SC	19.0	51	MS	18.0	51	MS	19.1	50	MS	18.7

Note: Because participation rates are negatively correlated with average scores, use caution when comparing across states.

Source: ACT. 2007 Average.

### SAT

The SAT measures critical reading, mathematical reasoning, and writing skills that students have developed over time and that they need to be successful in college. The College Board releases As is true of average ACT scores, average SAT scores are not comparable across all states because participation rates vary widely. Comparability is better among states with similar participation rates. annual reports on the average SAT scores and participation rates by state.<sup>7</sup> That information is presented in Table 2.22.

Like the earlier table of ACT scores, this table lists SAT scores in order of participation rate. (It bears repeating that because participation varies widely across states, scores are not comparable across all states.) Comparability is better among states with similar participation rates; for example, Alabama would be the closest match to Kentucky.

Kentucky's average scores are high, as one would expect given our low participation rate. The percentage of Kentucky high school graduates who took the SAT was only 10 percent, in contrast to the 77 percent who took the ACT.

 Table 2.22

 Percentage of High School Graduates Participating in SAT and Average SAT Scores, 2007

Parti	cipation	in SAT	Crit	ical Rea	ding		Math		Writing			
Rank	State	%	Rank	State	Score	Rank	State	Score	Rank	State	Score	
9	VA	73%	2	IL	504	2	IL	611	1	IL	588	
10	DE	72%	5	MO	594	7	MO	594	2	MO	587	
11	NC	71%	10	AR	570	11	OK	571	7	TN	568	
12	MD	70%	10	OK	5/8	13	TN	569	9	AR	565	
13	GA	69%	12	TN	574	14	LA	567	10	LA	563	
16	FL	65%	13	LA	569	15	AR	566	13	MS	560	
17	IN	620/	14	MS	568	16	KY	565	14	OK	559	
1 /	SC	0270	16	KY	567	18	AL	556	15	AL	554	
22	TX	52%	18	AL	563	20	MS	549	16	KY	553	
	U.S.	48%	24	OH	536	23	OH	542	22	OH	522	
28	OH	27%	30	WV	516		U.S.	515	30	WV	505	
30	WV	20%	33	VA	511	34	VA	511	33	VA	498	
32	TN	13%		U.S.	502	36	NC	509	35	MD	496	
34	KY	10%	35	MD	500		IN			U.S.	494	
35	AL	9%		DE		37	TX	507	39	DE	486	
38	IL	8%	38	FL	497		WV		40	GA	183	
41	LA	7%		IN		43	MD	502	40	IN	405	
12	MO	60/	42	NC	495		DE		42	NC	187	
42	OK	070	44	GA	494	46	FL	496	42	TX	402	
47	AR	5%	46	TX	492		SC		47	FL	479	
48	MS	4%	48	SC	488	49	GA	495	48	SC	475	

Note: Average scores are negatively correlated with participation rates. The College Board strongly discourages the comparison or ranking of states on the basis of SAT scores alone. The denominators for participation rates are Western Interstate Commission for Higher Education estimates of high school graduates.

Source: College Board. *College-Bound Seniors 2007* Table 3. Copyright (c) 2007-2008 The College Board, www.collegeboard.com. Reproduced with permission.

<sup>&</sup>lt;sup>7</sup> Appendix B contains a brief profile of the College Board, the organization that creates and scores the SAT and Advanced Placement tests.

Advanced Placement courses and exams provide high school students with early access to college-level learning in 37 subject areas.

In 2007, 15.2 percent of the nation's high school students earned a passing score on an AP exam. This compares to 9.7 percent in Kentucky. **Advanced Placement** 

Advanced Placement (AP) courses and exams provide high school students with early access to college-level learning in 37 subject areas. Most colleges and universities in the United States and many other countries use AP exam results in the admissions process to gauge student's ability and also award college credit or placement into higher-level college courses (College Board. *Advanced*).

The composite score for each AP exam reflects the grade that a student could be expected to earn in a college course. The score is reported on a scale of 1 to 5, corresponding to the letter grades F, D, C, B, and A, respectively. Statistical reports often focus on scores of 3 or higher because these correspond to passing grades eligible for college credit.

As Table 2.23 shows, in 2007, 15.2 percent of the nation's high school students took an AP exam and scored 3 or higher (a score that demonstrates college-level competency). In Kentucky, the percentage was 9.7 percent, placing Kentucky 33<sup>rd</sup> among all states.

	Hi	gh School (	Class of	2000		High School Class of 2007						
Took	At Lea	st One AP	De Colleg	emonstr ge-level	<sup>.</sup> ated Mastery	Took	At Lea	st One AP	D Colle	emonst ge-level	monstrated e-level Mastery	
Exan	Exam in High School			Scored	3-5)	Exan	n in Hiş	gh School	(	(Scored 3-5)		
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	
2	VA	25.0	3	VA	15.9	2	FL	38.0	2	MD	22.4	
4	FL	22.7	6	MD	14.1	4	MD	35.3	3	VA	21.5	
6	MD	20.2	8	FL	13.5	5	VA	34.4	4	FL	20.3	
7	NC	19.7	12	NC	11.3	6	AR	32.2	11	NC	18.5	
12	SC	17.7		U.S.	10.2	7	NC	31.9	15	GA	15.3	
14	GA	17.2	16	SC	10.0	13	GA	28.6		U.S.	15.2	
15	TX	16.6	17	IL	0.0	16	DE	27.4	17	IL	14.9	
	U.S.	15.9	1/	ΤХ	9.9	17	ΤХ	27.3	10	DE	14.5	
22	IL	13.4	19	GA	9.7		U.S.	24.9	19	TX	14.5	
24	DE	13.3	25	DE	7.6	22	SC	22.7	22	SC	13.3	
27	IN	11.9	28	OH	7.1	23	IL	22.0	28	OH	11.0	
29	OH	11.3	34	TN	6.2	27	OK	19.8	31	TN	10.0	
33	KY	10.6	36	IN	6.0	29	KY	19.6	22	IN	0.7	
36	TN	10.4	39	KY	5.5	32	IN	19.0	33	KY	9.7	
40	OK	9.5	40	OK	5.4	33	TN	18.3	36	AR	9.6	
41	WV	8.4	42	WV	4.6	34	OH	18.0	37	OK	9.3	
42	AR	8.1	45	AR	4.3	42	WV	15.2	46	WV	7.0	
43	AL	7.2	46	AL	3.9	46	MS	11.5	47	MO	6.7	
48	MS	5.6	48	MO	3.7	47	AL	11.4	48	AL	6.4	
49	MO	5.5	50	MS	2.3	48	MO	10.6	50	MS	3.7	
51	LA	3.2	51	LA	1.9	51	LA	5.7	51	LA	2.7	

Table 2.23Advanced Placement Exam, 2000 and 2007

Sources: College Board. Advanced Placement Report to the Nation 2005 48. Copyright (c) 2005 The College Board, www.collegeboard.com. Reproduced with permission. College Board. The 4<sup>th</sup> Annual AP Report to the Nation 52. Copyright (c) 2008 The College Board, www.collegeboard.com. Reproduced with permission.

Several states, including Kentucky, are considering policies to increase the number of students taking AP courses and exams. If these policies are implemented and successful at increasing participation, policy makers should expect a decline in average scores. This phenomenon has been observed for the ACT and the SAT, so it will likely happen with the AP. The Achievement Index is composed of 15 measures. Kentucky ranks 33<sup>rd</sup>, which is below average but not in the bottom tier of states.

### Education Week's Quality Counts Achievement Index

*Education Week's Quality Counts*, published by Editorial Projects in Education, reports an Achievement Index, shown in Table 2.24. This index is a composite of several types of scores, summarized in Table 2.25.

Among all states, Kentucky ranks 33<sup>rd</sup> on the Achievement Index. Kentucky's total score of 66.5 is below a majority of the peer states. Both Kentucky and the U.S. were awarded the same letter grade, a D+.

Rank	State	Total Score	Grade
1	Maryland	82.5	В
6	Virginia	76.2	С
7	Florida	75.2	С
13	Texas	72.6	С
14	Ohio	72.4	C-
16	Delaware	72.2	C-
25	Indiana	70.0	C-
	United States	69.4	D+
27	Illinois	69.1	D+
28	Georgia	68.1	D+
29	Tennessee	67.0	D+
31	North Carolina	66.7	D+
33	Kentucky	66.5	D+
34	Arkansas	66.3	D
36	Oklahoma	66.0	D
38	Missouri	64.7	D
40	South Carolina	64.5	D
46	Louisiana	60.3	D-
48	Alabama	59.1	F
49	West Virginia	58.1	F
51	Mississippi	55.9	F

Table 2.24Education Week's Quality Counts Achievement Index, 2008

Note: Components of this index are summarized in Table 2.20. Source: Editorial Projects in Education. *Education Week's Quality Counts* 2008.

Table 2.25 summarizes the components of the Achievement Index. Kentucky's greatest weaknesses are relatively lower math proficiency levels and smaller gains in grade 8 reading.

Table 2.25
Education Week's Quality Counts Achievement Index Components, Summary, 2008

Indicator	U.S. %	KY %	KY Rank
Achievement Levels, NAEP			
Grade 4 Math Proficiency 2007 <sup>a</sup>	38.6	30.8	42
Grade 8 Math Proficiency 2007 <sup>a</sup>	31.0	27.3	37
Grade 4 Reading Proficiency 2007 <sup>a</sup>	31.7	33.5	27
Grade 8 Reading Proficiency 2007 <sup>a</sup>	29.2	27.7	33
Achievement Gains, NAEP			
Grade 4 Math Scale Score Percent Change 2003-2007 <sup>a</sup>	+5.1	+6.4	16
Grade 8 Math Scale Score Percent Change 2003-2007 <sup>a</sup>	+4.1	+4.4	17
Grade 4 Reading Scale Score Percent Change 2003-2007 <sup>a</sup>	+3.2	+3.4	20
Grade 8 Reading Scale Score Percent Change 2003-2007 <sup>a</sup>	-0.3	-4.2	49
Achieving Excellence, NAEP			
Grade 8 Math Percent Advanced <sup>a</sup>	6.6	5.0	36
Grade 8 Math Percent Advanced - Percent Change 2003-2007 <sup>a</sup>	+1.6	+1.3	30
High School Graduation, Public Schools			
Graduation Rates, School Year 2004 <sup>a</sup>	69.9	70.0	33
Graduation Rate Change School Years 2000-2004 <sup>a</sup>	+3.1	+6.3	5
Advanced Placement (Per 100 Grade 11 and 12 Students)			
Advanced Placement Scores of 3+ 2006 <sup>b</sup>	16.9	10.8	29
Advanced Placement Scores of 3+ Change 2000-2006 <sup>b</sup>	+6.0	+5.6	20
NAEP Achievement Poverty Gap based on National School Lunch Program			
Grade 4 Reading Poverty Gap 2007 <sup>a</sup>	26.8	22.0	34
Grade 8 Math Poverty Gap 2007 <sup>a</sup>	26.0	21.1	36
Grade 4 Reading Poverty Gap Change 2003-2007 <sup>a</sup>	-1.1	+2.2	13
Grade 8 Math Poverty Gap Change 2003-2007 <sup>a</sup>	-2.4	-1.4	28
Achievement Index Total Score	69.4	66.5	33

Notes: Origins of data that appear in *Quality Counts*: <sup>a</sup>U.S. Dept. of Ed.; <sup>b</sup>College Board. Source: Editorial Projects in Education. *Education Week's Quality Counts* 2008.

### **Chapter Summary and Conclusions**

Policy makers' efforts to monitor high school completion are complicated by data quality issues and competing formulas. Nevertheless, across multiple measures, Kentucky's graduation rate recently rose slightly above the national rate, after hovering slightly below for several years. As in the nation as a whole, Kentucky's graduation rates are higher for females than for males and higher for whites than for African Americans and Hispanics. Compared to the total U.S., Kentucky's gender gaps are larger but racial/ethnic gaps are smaller.

In the most recent NAEP exams, Kentucky students performed above the national average in grades 4 and 8 science and grade 4 reading, equal to the national average in grade 4 writing, and below the national average in grades 4 and 8 math and grade 8 writing. Compared to previous years, the latest NAEP exams show improvements for the Commonwealth in all these areas except for one: grade 8 reading declined, indicating a need for more focus on reading skills in middle schools. As in the nation as a whole, Kentucky's female students scored higher than males in reading and writing, males scored higher in math and science, whites scored higher than African Americans in all subjects, and impoverished students scored lower than others in all subjects. Compared to the nation, Kentucky's gender gaps are larger but racial and poverty gaps are smaller.

This chapter presented a variety of outcome measures, highlighting some strengths and weaknesses to keep in mind when interpreting the data. Different measures may not always agree but most lead to the same conclusion: that Kentucky has made significant strides in recent years but must intensify efforts to improve student learning, especially in middle and high school.
### Chapter 3

### Characteristics of Students and Their Families and Communities

The impact of demographics and other external factors on academic performance is well known. Although data are not available to rank states on such personal factors as student motivation, data are available for family and community characteristics that influence achievement. This chapter focuses on those characteristics, as well as on enrollment and participation in selected programs.

#### Enrollment

The four largest states account for one-third of the nation's enrollment. Kentucky has the 26<sup>th</sup> largest enrollment in the nation. The four largest states—California, Texas, Florida, and New York—account for one-third of the nation's elementary and secondary student enrollment. Kentucky has the 26<sup>th</sup> largest enrollment in the nation. Figure 3.A presents enrollment for Kentucky and the peer states.



Figure 3.A Total P-12 Student Enrollment (in thousands), FY 2008

Source: U.S. Dept. of Ed. Inst. Natl. Projections, 2007 45.

Legislative Research Commission Office of Education Accountability

Birth rates have the most impact on enrollment trends. Birth rate fluctuations, such as the drop during the Great Depression and the upswing following World War II, have impacts on enrollment for generations.

#### K-12 Enrollment Trends

Public school enrollment trends reflect a variety of factors, especially birth rates and migration. Fluctuations in birth rates affect enrollment for generations. For example, after falling during the Great Depression, U.S. birth rates rose dramatically during the two decades of prosperity that followed World War II and then dropped again. This post-war baby boom, which strained the capacity of the education system, has had ongoing impact as baby boomers' children and grandchildren have reached school age (Bloom).

Kentucky's enrollment growth rate of approximately one-half percent per year since 2000 is expected to continue through 2016. Between 2000 and 2008, Kentucky's enrollment growth averaged about a half percent per year, as shown in Table 3.1. This rate was the 19<sup>th</sup> fastest in the nation. The same growth rate is projected for 2008-2016, but Kentucky's rank is projected to drop to 32<sup>nd</sup> as other states experience faster growth.

Table 3.1Projected Percent Changes in Public P-12 Enrollment, Fall 2000-Fall 2016

2	2000-200	)8	2008-2016					
Rank	State	%	Rank	State	%			
4	TX	17.1	2	TX	18.6			
5	GA	16.3	6	FL	16.1			
6	FL	14.7	7	GA	15.3			
7	NC	13.2	9	NC	11.9			
10	VA	8.9	12	VA	9.1			
11	DE	8.7	13	TN	8.9			
13	TN	6.8	14	DE	8.8			
14	AR	6.2	18	AR	7.3			
15	SC	6.1		U.S.	7.0			
	U.S.	5.5	21	OK	6.4			
16	IN	5.5	22	MD	6.1			
19	KY	3.9	26	MO	5.1			
20	IL	3.8	29	SC	4.7			
21	OK	2.7	31	LA	4.1			
25	MD	1.4	32	KY	3.9			
26	MO	0.2	38	MS	1.8			
28	MS	0.8	39	IN	1.5			
34	OH	-0.7	40	WV	0.7			
35	AL	-0.9	40 IL		0.7			
38	WV	-1.4	42	AL	0.5			
41	LA	-2.3	44	OH	0.3			

Source: U.S. Dept. of Ed. Inst. Natl. *Projections*, 2007 44-45.

In recent years, Kentucky has been, and continues to be, a national leader in preschool access. In FY 2007, the Commonwealth had the 5th highest enrollment of 3-year-olds and the 11th highest enrollment of 4-year-olds.

Nationwide attention is increasingly focusing on preschool as an opportunity to help at-risk students early in their educational development. Yet Kentucky is one of only 38 states with preschool programs and one of only 5 requiring preschool in all districts. Kentucky became a national leader in comprehensive preschool reform following the Kentucky Education Reform Act of 1990 (Natl. Institute. The State 64). As Figure 3.C and Table 3.2 show, the Commonwealth has been a national leader in preschool access in recent years and continued to be in FY 2007, with the 5<sup>th</sup> highest enrollment of 3-year-olds and the 11<sup>th</sup> highest enrollment of 4year-olds.

Figure 3.B Enrollment of 3-Year-Olds in State-funded Preschool, 2002 to 2007

**Preschool Enrollment** 



Source: Natl. Institute. The State 4, 64.

Figure 3.C Enrollment of 4-Year-Olds in State-funded Preschool, 2002 to 2007



Year and Jurisdiction

Chapter 3

Source: Natl. Institute. The State 4, 64.

	3-Year-Olds			4-Year-Olds	
Rank	State	%	Rank	State	%
1	IL	18.5	1	OK	68.4
4	AR	10.8	2	FL	56.7
5	KY	10.7	3	GA	53.3
7	WV	5.1	4	WV	45.8
10	TX	4.4	5	TX	45.2
	U.S.	3.2	7	SC	37.8
13	MO	2.2	10	MD	34.0
18	OH	1.3	11	KY	29.3
20	MD	1.2	12	IL	26.7
21	TN	1.0	14	LA	24.4
24	SC	0.6		U.S.	21.8
	OK		15	AR	21.4
	FL		18	TN	15.6
	GA		21	NC	14.8
27	LA	0.0	23	VA	12.5
<i>∠ I</i>	NC	0.0	27	DE	7.6
	VA		32	MO	4.3
	DE		35	OH	3.4
	AL		38	AL	1.8
<b>n</b> 0	IN	No program	<b>n</b> 0	IN	No program
II.a.	MS	No program	II.a.	MS	No program

Table 3.2Enrollment of 3- and 4-Year-Olds in State-funded Preschool, 2007

Note: Rankings are out of 38 states that have state-funded preschool programs. Source: Natl. Inst. *The State* 15.

#### **Racial and Ethnic Composition of Students**

Kentucky's minority student population is smaller and is growing more slowly than the nation's. As Table 3.3 shows, Kentucky's elementary and secondary students are less racially diverse than the rest of the nation. The minority population has grown more slowly than in the nation as a whole. Unlike some states, Kentucky has not experienced rapid Hispanic population growth over the past decade.

Amer Ala	<sup>.</sup> ican Ind ska Nati	lian/ ive	Asia Is	an/Pacifi slander	ic	Hispanic		Nor	Black Non-Hispanic			White Non-Hispanic					
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%			
2	OK	18.9	0	MD	5.2	3	TX	45.3	2	MS	51.2	3	WV	93.6			
18	NC	1.4	9	VA	3.2	7	FL	23.9	3	LA	44.4	7	KY	86.3			
	U.S.	1.2		U.S.	4.6		U.S.	19.8	4	SC	40.3	13	IN	80.3			
21	AL	0.8	13	IL	3.8	9	IL	19.0	5	GA	39.2	14	OH	76.6			
21	LA	0.8	17	TX	3.1	21	DE	9.2	6	MD	38.1	18	MO	69.5			
24	AR	0.7	20	DE	20	23	OK	8.9	7	AL	36.0	26	TN	69.5			
20	MD	0.4	20	GA	2.8	24	GA	8.7	8	DE	32.5	27	AR	68.2			
30	MO	0.4	26	FL	2.2	25	NC	8.4	9	NC	31.5	30	VA	59.8			
	DE		27	NC	2.1	26	VA	7.7	10	VA	27.0	31	OK	59.6			
	FL		31	OK	1.7	27	MD	7.6	11	TN	25.1	32	AL	59.4			
24	IN	0.2	32	MO	1.6	28	AR	6.8	12	FL	23.9		U.S.	57.1			
54	SC	0.5		AR		32	IN	5.7	13	AR	23.0	34	NC	56.6			
	TX		35	35	35	35	35	OH 1.4	37	SC	4.0	14	IL	20.6	36	IL	56.4
	VA			TN		38	TN	3.8	17	MO	18.2	37	DE	55.1			
	IL		40	LA	1 2	39	MO	3.2		U.S.	17.2	38	SC	54.0			
12	KY	0.2	40	SC	1.3	40	AL	2.8	19	OH	17.1	40	LA	51.5			
42	MS	0.2	43	IN	1.2	42	OH	2.4	21	TX	14.7	41	FL	49.6			
	TN		46	AL	1.0	44	KY	2.1	23	IN	12.5	42	GA	49.2			
	GA		48	KY	0.9	44	LA	2.1	25	OK	10.9	43	MD	48.6			
47	ОН	0.1	50	MS	0.8	48	MS	1.4	26	KY	10.6	45	MS	46.5			
	WV		51	WV	0.6	51	WV	0.7	38	WV	5.0	47	TX	36.5			

# Table 3.3Racial and Ethnic Composition of Students, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Common.

#### Incomes

#### **Family Income**

Kentucky's median family income is among the lowest in the nation.

As Table 3.4 shows, Kentucky's median family income is among the lowest in the nation. In 2005, incomes were significantly lower in only three peer states.

	1989			1999			200	6	
Rank	State	\$	Rank	State	\$	Rank	State	\$	Sig.
4	MD	45,034	3	MD	61,876	3	MD	77,839	
9	DE	40,252	10	IL	55,545	8	VA	66,886	
12	IL	38,664	11	DE	55,257	14	IL	63,121	
13	VA	38,213	12	VA	54,169	15	DE	62,623	
	U.S.	35,225	21	IN	50,261		U.S.	58,526	
23	OH	34,351		U.S.	50,046	27	OH	56,148	
24	IN	34,082	22	OH	50,037	28	GA	56,112	
25	GA	33,529	24	GA	49,280	30	IN	55,781	>
31	FL	32,212	31	NC	46,335	34	FL	54,445	
33	MO	31,838	33	MO	46,044	36	MO	53,026	
36	TX	31,553	34	TX	45,861	38	TX	52,355	
37	NC	31,548	36	FL	45,625	39	NC	52,336	
38	SC	30,797	38	SC	44,227	42	SC	50,334	
39	TN	29,546	40	TN	43,517	43	TN	49,804	
42	AL	28,688	43	AL	41,657	44	AL	49,207	
43	OK	28,554	44	KY	40,939	45	KY	48,726	_
47	KY	27,028	45	OK	40,709	46	LA	48,261	_
48	LA	26,313	47	LA	39,774	48	OK	47,955	
49	WV	25,602	49	AR	38,663	49	AR	45,093	
50	AR	25,395	50	MS	37,406	50	WV	44,012	<
51	MS	24,448	51	WV	36,484	51	MS	42,805	

## Table 3.4Median Family Income in Nominal Dollars, 1989, 1999, and 2006

Notes: Incomes for 1989 and 1999 are from decennial censuses, and are therefore not subject to sampling error. Incomes for 2006 are from the 2006 American Community Survey, which is subject to sampling error. Each difference between Kentucky and another state was tested for statistical significance with a 95 percent confidence level; > indicates states with significantly higher incomes than Kentucky, = indicates states not significantly different, and < indicates states with significantly lower incomes than Kentucky. Statistical tests used unrounded percentages and took into account each state's sample size and variance; therefore, states with the same percentages can have different levels of significance.

Sources: U.S. Dept. of Commerce. Census. "American Community Survey" and "Decennial Census."

### **Child Poverty Rates**

More than one in five Kentucky children live in poverty.

More than one in five Kentucky children live in families whose incomes are below the federal poverty line. As Table 3.5 shows, in 2005, Kentucky had the 10<sup>th</sup> highest child poverty rate in the nation. This compares to a rank of 6<sup>th</sup> in 1989 and 7<sup>th</sup> in 1999, based on the last two decennial censuses.

	1989			1999			20	)06	
Rank	State	%	Rank	State	%	Rank	State	%	Sig.
1	MS	25.2	2	MS	19.9	2	MS	29.5	
2	LA	23.6	3	LA	19.6	3	LA	27.8	
4	WV	19.7	5	WV	17.9	5	WV	25.2	
5	AR	19.1	6	AL	16.1	6	AR	24.2	
6	KY	19.0	7	AR	15.0	0	OK	24.5	
7	AL	18.3	/	KY	13.8	8	TX	23.9	
8	TX	18.1	9	TX	15.4	9	AL	23.0	_
10	OK	16.7	10	OK	14.7	10	KY	22.8	_
13	TN	15.7	14	SC	14.1	11	TN	22.7	
15	SC	15.4	16	TN	13.5	12	SC	22.1	
16	GA	14.7	18	GA	13.0	12	GA	20.2	
18	MO	13.3	19	FL	12.5	15	NC	20.2	
	U.S.	13.1		U.S.	12.4	17	OH	18.7	
21	NC	13.0	20	NC	12.3	18	MO	18.6	
23	FL	12.7	24	MO	11.7		U.S.	18.3	
24	OH	12.5	29	IL	10.7	21	IN	17.9	<
27	IL	11.9	31	OH	10.6	23	FL	17.5	
37	IN	10.7	37	VA	9.6	25	IL	17.1	
39	VA	10.2	38	IN	9.5	29	DE	15.8	
46	DE	8.7	44	DE	9.2	43	VA	12.2	
47	MD	8.3	47	MD	8.5	50	MD	9.7	

# Table 3.5Children Living Below the Federal Poverty Line, 1989, 1999, and 2006

Notes: Poverty rates for 1989 and 1999 are from decennial censuses. Poverty rates for 2006 are from the American Community Survey, which is subject to sampling error. Each difference between Kentucky and another state in 2006 was tested for statistical significance with a 95 percent confidence level; > indicates states with significantly higher poverty rates than Kentucky, = indicates states not significantly different, and < indicates states with significantly lower poverty rates than Kentucky. Statistical tests used unrounded percentages and took into account each state's sample size and variance; therefore, states with the same percentages can have different levels of significance. Source: U.S. Dept. of Commerce. Census. "American Community Survey" and "Decennial Census."

**Caveats and Limitations**. Although poverty rates are widely used, they do not take into account geographic differences in the cost of living nor do they include noncash benefits such as food stamps, subsidized housing, Medicaid, and subsidized school lunches (U.S. Dept. of Commerce. Census. "Characteristics" 7). Because one national set of income thresholds is used for the entire country, poverty will be overstated in areas like Kentucky where the cost of

living is lower. In addition, poverty statistics provide little information about the distribution of income. For example, two states could have the same percentage below the poverty level, but one could have income extremes far above and below the poverty level, while the other could have incomes concentrated just above and below the poverty level.

#### **NCES Comparable Wage Index**

NCES developed the Comparable Wage Index (CWI) to allow researchers to adjust comparisons of financial data for geographic differences in costs. The CWI compares the average income earned by a college-educated worker in a given labor market to the national average in 1999.<sup>8</sup> This measure, which has several advantages over cost-of-living indices, is appropriate because salaries make up the majority of education costs. In order to avoid the potential for education systems to manipulate the CWI by raising salaries, NCES excludes wages of education professionals when calculating the CWI (U.S. Dept. of Ed. Inst. Natl. "NCES Comparable"). As Table 3.6 shows, Kentucky's CWI is 34<sup>th</sup> in the nation.

NCES developed the Comparable Wage Index to allow researchers to adjust comparisons of financial data for geographic differences in costs.

<sup>&</sup>lt;sup>8</sup> A CWI of 1 was assigned to the national average salary of college-educated workers in 1999; a CWI greater than 1 indicates an average salary above the 1999 national average, and a CWI less than 1 indicates an average salary below the 1999 national average. The 1.265 CWI for the U.S. in 2005 indicates that the national average income for college-educated workers increased by 26.5 percent between 1999 and 2005.

			Range of CWI Across All Labor Markets in State								
		Average CWI for All	Rank by		Lowest	Highest	Range, Highest				
Rank	State	Labor Markets in State	Range	State	Market	Market	Minus Lowest				
7	VA	1.369		U.S.	0.833	1.669	0.835				
8	MD	1.368	2	WV	0.980	1.554	0.574				
10	IL	1.306	4	TX	0.855	1.389	0.534				
11	DE	1.286	5	MD	1.063	1.554	0.491				
	U.S.	1.265	6	VA	1.066	1.554	0.488				
14	TX	1.251	9	FL	0.833	1.265	0.432				
15	GA	1.242	11	IL	0.964	1.387	0.424				
19	OH	1.211	13	TN	0.894	1.290	0.395				
24	NC	1.194	14	AR	0.910	1.290	0.380				
26	FL	1.171	15	KY	0.913	1.293	0.379				
27	TN	1.163	16	NC	0.942	1.308	0.365				
31	MO	1.144	17	MS	0.930	1.290	0.360				
32	SC	1.140	18	MO	0.349	0.897	1.246				
33	IN	1.123	23	GA	0.991	1.305	0.314				
34	KY	1.117	24	IN	0.979	1.293	0.313				
36	AL	1.108	25	DE	1.031	1.338	0.308				
37	LA	1.097	29	OH	0.992	1.293	0.300				
38	WV	1.071	30	LA	0.913	1.203	0.290				
42	OK	1.064	35	SC	1.079	1.308	0.229				
45	MS	1.051	36	OK	0.905	1.128	0.223				
46	AR	1.041	40	AL	0.970	1.174	0.205				

Table 3.6NCES Comparable Wage Index, 2005

Source: U.S. Dept. of Ed. Inst. Natl. "NCES Comparable."

#### **Student Participation in Selected Programs**

#### Programs for Disadvantaged Students and Their Schools

Title I funds can be used for targeted assistance to specific students or for schoolwide programs. Kentucky has the 15<sup>th</sup> highest percentage of students enrolled in Title I-eligible schools and the 5<sup>th</sup> highest percentage of children in schoolwide programs.

Because income data for specific students is difficult to obtain, researchers and policy makers often use proxies, such as participation in programs that help low-income students and the schools they attend. Title I, the first section of the Elementary and Secondary Education Act, refers to federally funded programs aimed at America's most disadvantaged students. Title I, Part A funds can be used for targeted assistance to specific students. However, schools with at least 40 percent of students living below the poverty level are encouraged to combine Title I, Part A funds with other federal, state, and local funds to operate a comprehensive schoolwide program that upgrades the entire educational program in the school (U.S. Dept. of Ed. Office of Elementary). Through the National School Lunch Program, children from families with incomes at or below 130 percent of the poverty level are eligible for free meals, and those with incomes between 130 percent and 185 percent of the poverty level are

Above-average percentages of Kentucky students attend Title I schools and are eligible for subsidized lunches. eligible for reduced-price meals (U.S. Dept. of Ed. Inst. Natl. *Numbers and Types of Public Elementary and Secondary Schools* 4-5).

Table 3.7 presents the percentages in 2006 of students who are enrolled in Title I schools as well as those eligible for free or reduced-price lunch. All of these rankings reflect Kentucky's high rates of child poverty. With more than 60 percent of students enrolled in Title I-eligible schools and more than 50 percent in schools with schoolwide programs, Kentucky ranks 15<sup>th</sup> and 5<sup>th</sup>, respectively, in the nation. With 52.4 percent of students eligible for free or reduced-price lunches, Kentucky is well above the national average and ranks 7<sup>th</sup>.

Table 3.7Students in Title I Schools or Eligible for National School Lunch Program, FY 2006

All Title I Schools			Title I Schools with Schoolwide Programs			Eligible for Free Lunch		Eligible for Reduced-price Lunch		or ·ice	Eligi or Re	ble for duced- Lunch	Free price	
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
2	IN	96.1	2	MS	61.6	1	MS	62.3		TN	n.a.	1	MS	69.5
6	AR	67.1	3	TX	60.4	2	LA	53.9	2	OK	10.4	2	LA	61.2
8	OK	65.8	5	KY	52.8	5	OK	44.1	6	WV	9.8	4	OK	54.5
9	MS	65.6	6	LA	51.0	6	AR	44.0	11	AR	8.8	6	AR	52.9
10	TX	64.7	8	OK	46.1	7	KY	43.8	12	FL	8.7	7	KY	52.4
11	OH	63.8	9	AR	45.8	8	SC	43.7	12	KY	86	8	AL	51.7
15	KY	60.6	10	AL	45.1	9	AL	43.0	15	AL	8.0	9	SC	51.5
19	LA	57.7	11	TN	38.6	10	GA	41.5	17	GA	8.2	10	GA	49.8
21	AL	55.5	13	GA	37.5	11	TX	40.0	17	ΤХ	8.2	11	WV	49.1
23	IL	54.4	14	SC	34.4	12	WV	39.3	22	IN	7.0	13	TX	48.2
	U.S.	52.3	15	FL	34.1	14	FL	37.1	22	SC	1.9	14	TN	47.1
29	GA	45.9		U.S.	31.3	17	NC	35.2	25	MO	71	15	FL	45.8
30	TN	45.3	18	DE	29.8		U.S.	32.4	23	NC	/.4	19	NC	42.6
32	DE	44.6	19	NC	29.7	10	IL	31.5		U.S.	7.4		U.S.	41.6
33	MO	42.7	20	VA	27.4	19	MO	51.5	30	LA	73	22	MO	39.1
40	NC	37.3	23	WV	26.2	21	DE	30.7	50	MS	1.5	25	IL	37.2
41	SC	36.9	24	IL	24.8	27	IN	28.2	33	MD	7.2	27	DE	36.1
42	WV	36.7	26	OH	21.6	30	OH	26.4	35	VA	6.8	21	IN	30.1
43	FL	34.9	38	MD	16.8	36	MD	24.4	40	OH	6.2	35	OH	32.5
46	VA	27.4	40	MO	15.8	38	VA	24.3	42	IL	5.7	39	MD	31.6
48	MD	20.1	48	IN	6.8	n.a.	TN	n.a.	45	DE	5.4	43	VA	31.1

Notes: Ranks for all Title I schools and Title I schools with schoolwide programs are out of 50; data were not available for NJ. Ranks for eligibility for free lunch and eligibility for reduced-price lunch are out of 48; data not available for NV, TN, or WI.

Source: U.S. Dept. of Ed. Inst. Natl. Common.

Kentucky has relatively few students who lack English proficiency, ranking 44<sup>th</sup> out of 46 states reporting.

#### **English Language Learners**

An English language learner (ELL) is a student who comes from an environment in which a language other than English has had a significant impact on his or her level of English language proficiency (U.S. Dept. of Ed. Inst. Natl. *Overview* 38). Table 3.8 ranks peer states with respect to the percentage of students receiving ELL services. The percentages are high in southwestern and western states. Although some Kentucky districts have concentrations of ELL students, the overall percentage for the state is low, giving Kentucky a rank of 44<sup>th</sup> out of the 46 states that reported data on this variable.

Table 3.8 Students Receiving English Language Learner Services FY 2006

Rank	State	%
4	Texas	15.7
	United States	8.6
11	Florida	8.3
12	Oklahoma	7.5
19	Virginia	6.0
20	Indiana	5.5
21	Georgia	5.4
24	North Carolina	5.2
26	Delaware	4.9
29	Arkansas	4.4
32	Maryland	3.7
37	Alabama	2.2
38	South Carolina	2.1
39	Missouri	2.0
41	Louisiana	1.8
43	Ohio	1.6
44	Kentucky	1.5
45	West Virginia	0.7
46	Mississippi	0.6
no	Illinois	no
11.a.	Tennessee	11.a.

Notes: Ranks are out of 46; data are not available for IL, ND, NH, PA, and TN.

Source: U.S. Dept. of Ed. Inst. Natl. Common.

The percentage of students receiving ELL services has been growing over time. This growth has been slower for Kentucky than for the nation as a whole.

#### Disabilities

A frequently used proxy for the percentage of students with disabilities is the percentage of students with Individualized Education Programs (IEPs). An IEP is a written instructional plan for a student with a disability (U.S. Dept. of Ed. Inst. Natl. *Overview* 39). This is not a very precise proxy for the disability rate because IEPs are provided for a wide range of disability types and severity levels, including speech difficulties, attention deficit/hyperactivity disorder, and severe cognitive disabilities.

As Table 3.9 shows, Kentucky is above the national average with respect to the percentage of students with IEPs, and ranks 12<sup>th</sup> overall.

Rank	State	%
6	WV	17.6
8	IN	17.1
10	AL	16.8
12	KY	16.0
13	SC	15.6
15	IL	15.3
16	OK	15.2
18	FL	14.9
22	DE	14.7
25	OH	14.5
26	VA	14.4
31	MS	13.7
	U.S.	13.6
32	NC	13.6
35	TN	13.3
39	LA	13.0
40	MD	12.8
41	GA	12.4
42	AR	12.3
46	TX	11.3
51	MO	0.0

Table 3.9Students With Individualized Education Programs, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Common.

The percentage of students with IEPs has been growing over time. Growth has been faster in Kentucky than in the nation as a whole.

Kentucky is above the national average with respect to the percentage of students with Individualized Education Programs due to disabilities.

#### Annie E. Casey Child Well-being Index and Measures

Since 1990, the Annie E. Casey Foundation has released an annual *KIDS COUNT Data Book* on the well-being of American children. Although the data and rankings from this publication do not report directly on education, they are included in this compendium because child well-being is related to educational success. Even though the education system has little or no control over these factors, they have an important impact on a child's ability to benefit from education.

Kentucky has a low ranking on several child well-being measures, indicating more obstacles to student achievement in Kentucky than in most other states. *KIDS COUNTS* ranks states with respect to a child well-being index. The 10 key indicators that make up this index are summarized in Table 3.10. Kentucky has a low ranking on several components, reflecting a number of obstacles to achievement.

 Table 3.10

 Summary of Components of 2007 KIDS COUNT Child Well-being Index, 2007

Key Indicator	KY	U.S.	KY Rank
Percentage of live births with low birth weight (under 5.5 pounds), 2004 <sup>a</sup>	8.8	8.1	38
Infant mortality rate (deaths per 1,000 live births), 2004 <sup>a</sup>	6.8	6.8	27
Child death rate (deaths per 100,000 children ages 1-14), 2004 <sup>b</sup>	24	20	32
Teen death rate (deaths per 100,000 teens ages 15-19), 2004 <sup>b</sup>	95	66	44
Teen birth rate (births to 15-19-year-olds per 1,000 females), 2004 <sup>b</sup>	49	41	37
Percentage of teens ages 16-19 who are high school dropouts, 2005 <sup>a</sup>	9	7	36
Percentage of teens ages 16-19 not attending school and not working, 2005 <sup>a</sup>	11	8	45
Percentage of children with no parent working full time, year-round, 2005 <sup>a</sup>	38	34	44
Percentage of children in poverty, 2005 <sup>a</sup>	22	19	41
Percentage of children in single-parent families, 2005 <sup>a</sup>	31	32	26

Notes: Poverty rates do not take into account noncash benefits or geographic cost differences. Origins of data that appear in *KIDS COUNT*: <sup>a</sup> U.S. Centers for Disease Control and Prevention, Natl. Ctr. for Health Statistics; <sup>b</sup> Death rates from U.S. Centers for Disease Control and Prevention, and population data from U.S. Census Bureau. Source: Annie E. Casey 28, 33.

As shown in Table 3.11, Kentucky's Overall Child Well-being Index rank's 40<sup>th</sup> in the nation. This illustrates the overall impact of challenges facing Kentucky.

Overall Rank	State
14	Virginia
24	Maryland
26	Illinois
28	Ohio
31	Indiana
32	Florida
34	Missouri
35	Delaware
37	Texas
39	North Carolina
40	Kentucky
41	Georgia
42	Oklahoma
43	Tennessee
44	West Virginia
45	Arkansas
46	South Carolina
48	Alabama
49	Louisiana
50	Mississippi

# Table 3.11 Overall Child Well-being, 2007 KIDS COUNT Data Book

Source: Annie E. Casey.

#### Education Week's Quality Counts Chance-for-Success Index

Kentucky students are fortunate to have few language barriers, but other measures of their chances for success are low. *Education Week*'s Chance-for-Success Index is based on 13 indicators of students' socioeconomic status, participation in education, and employment opportunities; these are summarized in Table 3.12. The index includes more education measures and fewer health measures than the *KIDS COUNT* study. Kentucky has strong linguistic integration but struggles with other factors, especially parental employment, kindergarten enrollment, and adult educational attainment.

Table 3.12Education Week's Quality Counts Chance-for-Success Index Components, Summary, 2008

			Percenta Chang		
	KY	U.S.	previou	ıs year	KY
Indicator	%	%	KY	U.S.	Rank
Early Foundations	24.8	25.0			36
Family Income 2005 (at or above 200% of poverty level) <sup>a</sup>	56.4	60.1	+2.7	+0.3	38
Parent Education 2006 (at least one parent has a 2- or 4-year					
postsecondary degree) <sup>a</sup>	39.0	43.3	+1.8	+0.8	37
Parental Employment 2006 (at least one parent employed full time,					
year-round) <sup>a</sup>	68.6	71.8	+2.1	+1.2	44
Linguistic Integration 2006 (parents speak fluent English) <sup>a</sup>	97.5	84.3	+0.5	0.0	6
The Schooling Years	32.9	34.5			33
Preschool Enrollment 2006 (3- and 4-year-olds enrolled) <sup>a</sup>	40.9	46.1	-1.3	+1.3	34
Kindergarten Enrollment 2006 (5- and 6-year-olds enrolled) <sup>a</sup>	73.0	75.7	-2.2	+0.4	44
Elementary Reading Achievement 2007 (grade 4 public school					
students who scored at or above proficient)	33.5	31.7	+2.7	+1.9	27
Middle School Math Achievement 2007 (grade 8 public school					
students who scored at or above proficient) <sup>b</sup>	27.3	31.0	+4.8	+2.5	37
High School Graduation FY 2004 (public high school students					
graduating on time with standard diploma) <sup>c</sup>	70.0	69.9	+0.3	+0.3	33
Postsecondary Participation 2006 (adults ages 18-24 who have a					
postsecondary credential or are enrolled in college/university) <sup>a</sup>	47.4	51.8	+3.9	+4.0	35
Adult Outcomes (ages 25-64)	17.0	18.8			43
Adult Educational Attainment 2006 (associate degree or higher) <sup>a</sup>	29.3	37.2	+0.6	-0.2	46
Annual Income 2005 (personal income at/above national median) <sup>a</sup>	44.1	50.1	+0.4	+0.1	37
Steady Employment 2006 (working full time year-round) <sup>a</sup>	68.4	68.2	+1.0	+1.0	26
	74.7	78.4			
Chance-for-Success Index Total Score	(C)	(C+)			38

Notes: Income differences do not take into account geographic cost differences. Origins of *Quality Counts* 2008 data: <sup>a</sup>Editorial Projects in Education analysis of data from the U.S. Census Bureau's 2006 American Community Survey; <sup>b</sup>Natl. Assessment of Educational Progress, Natl. Ctr. for Education Statistics, U.S. Dept. of Ed., 2005; <sup>c</sup>Calculated using Editorial Projects in Education Research Center's Cumulative Promotion Index formula and data from the U.S. Dept. of Ed.'s *Common Core of Data*.

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

Table 3.13 presents state rankings on the Chance-for-Success-Index. Kentucky's rank is 38<sup>th</sup>, earning a C letter grade, compared to the national average of C+.

Rank	State	Score	Grade
6	Maryland	88.2	B+
8	Virginia	85.9	В
13	Delaware	82.6	В
19	Illinois	81.3	B-
24	Ohio	79.6	B-
	U.S.	78.4	C+
28	Missouri	77.5	C+
29	Indiana	77.4	C+
31	North Carolina	77.1	C+
32	Florida	76.6	C+
36	Georgia	75.2	С
38	Kentucky	74.7	С
39	South Carolina	74.3	С
41	Texas	73.3	С
42	Alabama	72.1	C-
43	Tennessee, Oklahoma	71.9	C-
45	Arkansas	71.7	C-
47	West Virginia	70.8	C-
50	Louisiana	67.9	D+
51	Mississippi	67.7	D+

	Τ	Table 3.1	13
Education	Week's Quality	<b>Counts</b>	Chance-for-Success-Index

Notes: Components of this index are summarized in Table 3.14. Source: Editorial Projects in Education. *Education Week's Quality Counts* 2008.

#### **Caveats and Limitations**

More than half of the components of the Chance-for-Success-Index reported in Table 3.14 are outside the control of educators. However, in its 2007 edition of *Quality Counts, Education Week* justified including them because of their impact on a child's ability to benefit from educational opportunities:

A child who comes to school malnourished, from a poor household, having a mother with less than a high school education, or a parent whose primary language is not English is much more likely than a classmate without those factors to have academic and behavioral problems later on (Editorial. *Education Week's Quality Counts* 2007 20).

However, Education Trust charges that the index in *Quality Counts* encourages the defeatist notion that "demographics are destiny"

*Education Week* justifies examining noneducation indicators because of their impact on a child's ability to and benefit from educational opportunities. However, Education Trust criticizes this focus, asserting that personal circumstances do not preordain a child to failure or success. and diminishes the "critical role of educators and public schools in preparing young people to become contributing citizens despite the obstacles they face outside of school" (Education Trust. *Education*).

#### **Chapter Summary and Conclusions**

As this chapter has shown, Kentucky's students face many obstacles to educational success before they even set foot in a classroom. Their parents typically have lower incomes, education, and employment rates. Disabilities, teen parenthood, and teen death rates are above the national average. On the other hand, Kentucky has fewer students with limited English proficiency. Despite these challenges, some of Kentucky's high-poverty schools have demonstrated that it is possible to achieve high performance by focusing on the delivery of challenging content and creating a strong learning environment.

### **Chapter 4**

### Characteristics of Staff, Schools, and Districts

This chapter presents characteristics of schools and staff, including the student-teacher ratio, instructors as a percentage of all staff, salaries, Title I eligibility of schools, safety, and technology.

#### **Student-Teacher Ratio**

The student-teacher ratio is the number of students enrolled in the fall of the school year divided by the full-time equivalent count of teachers. A low student-teacher ratio is considered an indicator of quality, giving students more opportunities for personal attention. As shown in Table 4.1, Kentucky's 16.0 ratio overall is close to the national average of 15.7; Kentucky is ranked 36<sup>th</sup>. Kentucky's ratio is better (lower) than the national average in elementary schools, but worse (higher) than the national average in middle and high schools.

Table 4.1
Student-Teacher Ratio, Total and by Grade Level, FY 2006

	Total		Ele	mentar	у	Mid	dle Sch	ool	High School		
Rank	State	Ratio	Rank*	State	Ratio	Rank*	State	Ratio	Rank*	State	Ratio
6	VA	12.6	6	VA	13.1	9	VA	13.2	6	AR	12.9
8	AL	12.8	13	MO	13.7	17	AR	14.1	17	VA	14.5
14	MO	13.7	18	GA	14.3	20	WV	14.3	18	OK	14.6
19	WV	14.1	20	AL	14.4	21	MO	14.7	19	ΤX	15.0
20	AR	14.4	22	SC	14.6	23	TX	14.8	22	LA	15.6
22	SC	14.6		WV	14.0	24	NC	14.0	22	MO	15.7
24	GA	147	24	NC	14.7	24	SC	14.9	23	WV	13.7
24	LA	14./	25	LA	14.9	27	GA	15.0	25	SC	15.8
26	NC	14.8	26	KY	15 1	28	LA	15.2	26	NC	15.9
28	ΤX	15.0	20	MD	13.1	29	MD	15.4	29	IL	16.4
30	DE	15.1	28	TN	15.3	20	MS	15.5	20	AL	16.5
21	MD	15.2	20	DE	15 /	50	OK	15.5	50	DE	10.5
51	OK	13.2	29	TX	13.4	22	IL	15.9	32	GA	16.7
33	OH	15.6	31	FL	15.5	52	OH	13.0	33	MS	16.8
	U.S.	15.7	32	AR	15.6		U.S.	16.0	-	U.S.	17.0
34	MS	15.7		U.S.	15.9	35	KY	16.3		KY	
35	IL	15.8	35	OK	16.2	38	DE	16.7	35	MD	17.2
36	KY	16.0	39	IL	17.0	39	TN	17.0		OH	
50	TN	10.0	40	MS	17.1	41	AL	17.4	41	IN	18.8
40	FL	16.8	42	OH	17.5	42	IN	17.6	43	TN	19.1
43	IN	17.1	44	IN	17.6	43	FL	17.7	44	FL	19.8

Note: \*Ranks by grade level are out of 50; data not available for AZ.

Sources: U.S. Dept. of Ed. Inst. Natl. Common and Numbers and Types of Public Elem. and Sec. Schools 8-9.

Kentucky has 16 enrolled students per full-time equivalent teacher, compared to a national student-teacher ratio of 15.7. This places Kentucky 36<sup>th</sup> among all states. The steady decline in the ratio of students to teachers for the past several decades is attributed to the need for more teachers to provide schools' increasing complex and diverse offerings. Legislative Research Commission Office of Education Accountability

Student-teacher ratios decline when the number of teachers employed grows at a faster rate than enrollment. The studentteacher ratio has declined steadily in Kentucky and the nation since 1987, the first year for which comparable data are available for all states. This trend is attributed to increasingly complex and diverse school offerings, such as special education and enrichment programs, requiring increasing numbers of specialized teachers (U.S. Dept. of Ed. Inst. Natl. *120 Years* 28-29).

#### **Teachers as a Percentage of Staff**

Kentucky has the nation's lowest percentage of teachers as a percentage of all staff. Nationally, the number of nonteaching staff has been increasing slightly faster than the number of teachers. As Table 4.2 shows, teachers as a percentage of staff declined from 53 percent in 1985 to 51.2 percent in 2005. Kentucky fell from a rank of 38<sup>th</sup> among 48 states in 1985 to a rank of 51<sup>st</sup> in 1995. In 2005, the most recent year data are available, Kentucky still ranked last, with teachers making up 43.3 percent of staff, compared to 51.2 percent of staff for the U.S.

	1985			1995		2005		
Rank	State	%	Rank	State	%	Rank	State	%
14	SC	56.7	9	OH	55.2	1	SC	71.5
15	IL	56.6	10	WV	54.5	7	AL	55.7
18	DE	55.4	10	DE	54.5	13	NC	52.5
19	WV	55.0	13	MD	54.4	14	VA	52.2
20	VA	54.3	14	VA	512	14	WV	52.5
21	OH	54.0	14	IL	54.5	16	TN	52.2
23	MD	53.9	18	TN	54.0	17	MO	52.1
25	OK	53.5	19	AR	53.8	20	DE	51.7
26	AR	53.4	22	SC	53.3	-	U.S.	51.2
27	MO	53.3	27	AL	52.9	23	OK	51.1
	U.S.	53.0	31	NC	52.2	24	MD	51.0
31	GA	52.8		U.S.	52.0	27	FL	50.6
32	AL	52.4	33	TX	52.0	21	IL	50.0
32	NC	32.4	39	LA	50.5	30	GA	49.6
36	FL	51.6	43	FL	48.3	31	OH	49.4
38	KY	51.1	45	GA	48.2	34	TX	48.7
50	TN	51.1	16	IN	48.0	37	LA	48.2
41	IN	50.8	40	MO	40.0	43	AR	46.7
43	TX	50.7	48	MS	47.6	44	MS	46.5
46	LA	48.2	49	OK	47.0	48	IN	45.5
47	MS	47.0	51	KY	46.3	51	KY	43.3

Table 4.2Teachers as a Percentage of All StaffFall 1985, 1995, and 2005

Notes: For 1985, ranks are out of 48 states; CT, MT, and NV are not shown because they underreported support staff. The U.S. value includes estimates for these three states.

Source: Staff compilation of data from U.S. Dept. of Ed. Inst. Natl. Digest.

During the 1980s, the percentage of staff who were teachers in Kentucky was close to the national average, at a little over 50 percent. However, since 1990, Kentucky's percentage has fallen more quickly than the U.S. percentage.

#### **Types of Staff**

Table 4.3 presents full time equivalent staff in each major category per 1,000 students in membership. Student support and other support staff include library support, student support services, and all other nonadministrative support staff. Administrative support includes district- and school-level administrative support staff.

Kentucky's higher numbers of support staff and instructional aides may be due to higher preschool enrollments and student disability rates. While the number of teachers per 1,000 students in Kentucky is below the national average, resulting in a rank of 37<sup>th</sup>, Kentucky is above the national average for several other types of staff. The Commonwealth ranks 5<sup>th</sup> for student support and other support

staff, 5<sup>th</sup> for administrative support staff, 8<sup>th</sup> for the number of instructional aides, and 14<sup>th</sup> for librarians.

Staffing reflects many factors, especially policies written into statutes and regulations. For example, Kentucky's high rate of instructional aides could be related to the following four factors:

- Preschools use more instructional aides than do K-12 classes, and the percentage of children enrolled in preschool is higher in Kentucky than in most other states.
- Until recently, Kentucky allowed instructional aides to teach in preschools. The statutes and regulations now require certification, but those already in the position were allowed to stay. In contrast, some other states have only certified teachers in their preschools.
- KRS 157.360 requires one instructional aide for every 24 fulltime equivalent kindergarten students.
- Kentucky has an above-average proportion of students who receive special education services, which are more labor-intensive than regular education services; instructional aides provide many extra services.

Table 4.3	
Staffing Rates	
Full-time Equivalent Staff Members Per 1,000 Students, FY 2	006

	All Staf	f		Teacher	s	Ins	structio Aides	nal	Instruc Su	tional (	Coord./ ors	Guidance Counselors		
Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate
10	VA	151.4	7	VA	79.2	8	KY	20.7	4	LA	2.7	3	LA	4.5
11	AR	149.0	8	AL	77.9	9	NC	20.3	5	DE	2.4	8	AR	3.0
14	KY	144.1	14	MO	73.1	11	IN	19.2	10	IN	16	10	MO	2.9
18	LA	141.5	19	WV	71.0	17	MS	17.5	12	MD	1.0	12	MD	2.7
19	MO	140.3	20	AR	69.6	20	IL	16.6	15	MS	1.5	14	NC	2.6
21	AL	139.9	22	SC	68.7	24	GA	16.0	16	AR	1.4		OK	
22	ΤX	137.1	24	LA	68.2	25	LA	15.8	10	VA	1.4	16	SC	2.5
23	GA	137.0	25	GA	67.9	26	AR	15.6		AL			WV	
24	MS	136.7	26	NC	67.5	27	VA	15.5	20	KY	1.3	19	AL	2.4
25	WV	135.8	29	TX	66.8	30	TN	14.3		WV		27	DE	2.2
30	OH	129.9	30	DE	66.1		U.S.	14.2	20	MO	1.0	27	ΤX	2.5
31	MD	129.3	21	MD	(5.0	32	DE	14.1	28	SC	1.0	20	GA	2.2
32	OK	129.0	51	OK	03.9	33	TX	13.5		U.S.	1.0	50	VA	2.2
22	IN	120 6	33	OH	64.1	36	MO	13.4	35	OK	0.8		U.S.	2.1
33	NC	128.0		U.S.	63.9	39	OK	12.4	38	NC	0.7		FL	
35	DE	127.9	34	MS	63.5	40	MD	12.2	41	IL	0.6		KY	
36	IL	125.3	35	IL	63.4	43	WV	11.7	42	TN	0.5	34	MS	2.1
	U.S.	124.7	36	TN	62.5	44	FL	10.9	44	TX	0.4		OH	
39	TN	119.7	37	KY	62.4	48	OH	9.7		FL			TN	
42	FL	117.5	40	FL	59.4	50	AL	9.1	46	GA	0.3	43	IN	1.7
47	SC	96.1	43	IN	58.5	51	SC	5.2		OH		46	IL	1.5

Continued on next page.

Librarians		ans	Studen Su	t Suppor Ipport St	rt/Other taff	Sch	School Admin.		School District Admin.			Admin. Support Staff		
Ran k	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate
3	AR	2.1	2	AR	45.5	2	TX	7.0	2	OH	4.3	2	OH	16.3
5	MS	2.0	5	KY	39.3	6	SC	4.8	7	DE	2.7	5	KY	11.9
8	AL	1.9	8	ΤX	37.2	0	AL	4.0	12	MS	2.0	6	FL	11.5
10	LA	18	9	GA	36.9	9	GA	4.0	17	IL	18	7	AL	11.2
10	MO	1.0	12	MS	36.2	15	LA	3.0	17	TX	1.0	8	VA	11.1
12	NC	17	13	LA	35.9	15	MD	5.9	21	WV	1.6	14	OK	10.4
12	VA	1.7	14	VA	35.7	18	TN	37	22	MO	1.5	15	MO	9.6
	KY		15	WV	35.1	10	WV	5.7	24	AR	14		U.S.	8.7
14	OK	16	16	IN	35.0	21	MS	3.6	24	GA	1.7	20	IL	84
17	SC	1.0	18	MD	34.4	- 24	AR	35		U.S.	1.3	20	SC	0.7
	TN		20	MO	33.7	24	NC	5.5	28	KY	13	23	MS	8.3
	GA		25	AL	31.6		U.S.	3.4	20	VA	1.5	25	LA	8.2
21	MD	1.4	26	OK	30.9		MO		30	NC	1.2	30	IN	76
	WV		28	DE	29.8	26	OK	3.4	32	MD	1.1	50	WV	7.0
	U.S.	1.1	29	OH	29.7		VA		37	IN	1.0	33	NC	7.4
32	DE	11		U.S.	29.0	29	KY	3.3	51	OK	1.0	36	ΤX	7.1
52	ΤX	1.1	35	IL	28.9	31	DE	3.2	43	FL	0.7	37	GA	6.9
36	FL	1.0	36	FL	28.8	33	IL	3.1	45	LA	0.5	38	AR	6.8
50	IL	1.0	37	TN	28.3	36	IN	2.9	47	SC	0.4	41	DE	63
41	IN	0.9	42	NC	23.7	40	FL	2.7	50	AL	03		TN	0.5
44	OH	0.8	51	SC	3.4	44	OH	2.6	50	TN	0.5	43	MD	6.1

#### Table 4.3 continued

Source: U.S. Dept. of Ed. Inst. Natl. Public 5-6 and 15-16.

27.2

Due to Kentucky's recertification requirements, Kentucky has the 3<sup>rd</sup> highest rate of teachers holding advanced degrees. Table 4.4 shows the percentage of teachers who have a master's, doctorate, or other degree beyond the bachelor's. Kentucky has the 3<sup>rd</sup> highest rate of advanced degree-holding teachers. Unlike many other states, Kentucky requires that teachers complete an advanced degree within their first 10 years of teaching.

Rank	State	%
	United States	48.1
3	Kentucky	70.6
4	Indiana	61.9
5	West Virginia	61.1
6	Alabama	60.6
9	Maryland	56.3
15	Illinois	53.5
16	Delaware	53.4
17	Georgia	50.7
17	Ohio	52.7
19	Tennessee	52.3
22	South Carolina	51.0
23	Missouri	50.8
36	Virginia	39.4
37	Arkansas	38.4
39	Florida	36.6
40	Mississippi	35.5
43	Louisiana	33.9
45	Oklahoma	33.4
47	North Carolina	31.7

Table 4.4Teachers With Advanced Degrees, FY 2004

Source: U.S. Dept. of Ed. Inst. Natl. Digest 2007, Table 64.

#### Education Week's Quality Counts Teaching Profession Index

Texas

The 2008 edition of *Education Week's Quality Counts* included an index of teaching profession measures in three categories: accountability for quality, incentives and teacher allocation, and building and supporting capacity. As Table 4.5 shows, Kentucky received an overall rank of 9<sup>th</sup> and a grade of B-. It is interesting to note that most of the top-ranked states are members of Southern Regional Education Board.

Kentucky's grade reflects the implementation of 28 of the 50 policies recommended in *Quality Counts*. These are discussed briefly in the following pages.

Education Week's Quality Counts gave Kentucky a rank of 9<sup>th</sup> and a grade of B- on its overall teaching profession index. 49

Rank	State	Score	Grade
1	SC	91.9	A-
2	AR	88.9	B+
3	NC	84.8	
4	FL	83.2	Б
5	GA LA	83.0	В
7	VA	82.0	
8	AL	81.1	
9	KY	80.9	В-
10	OK	80.1	
13	WV	78.9	
14	OH	78.1	C+
19	DE	76.8	
21	TN	75.1	С
22	MO	75.0	С
23	TX	73.1	С
	U.S.	73.0	C
26	MD	72.0	C
28	IN	71.9	
35	IL	68.1	D+
44	MS	65.1	D

Table 4.5Quality Counts Teaching Profession Index, 2008

Source: Editorial Projects in Education. *Education Week's Quality Counts* 2008.

#### Accountability for Quality

Quality Counts gives Kentucky credit for having implemented many of the recommended teacher accountability measures. Table 4.6 summarizes the 16 teacher-related accountability policies that were included in the *Quality Counts* index. Kentucky has implemented nine of these policies, including limitations on out-of-field teaching, a policy present in only three other states. Of the seven policies Kentucky does not have in place, only one is in a majority of other states: written testing of prospective teachers' basic skills.

# Table 4.6Quality Counts Teacher Accountability Measures, 2008

	Implement	ted
Policies Relating to Accountability for Quality	# of States	KY
Initial Licensure Requirements for All Prospective Teachers (FY 2008)		
State requires substantial formal coursework in subject area(s) taught	27	✓
Prospective teachers must pass written tests		
Basic skills	39	
Subject-specific knowledge	42	✓
Subject-specific pedagogy	6	
State requires clinical experiences during teacher training		
Student teaching <sup>a</sup>	39	✓
Other clinical experiences <sup>b</sup>	13	
Discouraging Out-of-Field Teaching for All Schools (FY 2008)		
Parental notification of out-of-field teachers	5	
State has ban or cap on the number of out-of-field teachers	4	✓
Evaluation of Teacher Performance (FY 2008)		
State requires all teachers' performance to be formally evaluated	43	✓
Teacher evaluation is tied to student achievement	12	
Teacher evaluation occurs on an annual basis	12	
State requires evaluators to receive formal training	26	✓
Accountability for Effectiveness of Teacher Education Programs (FY 2008)		
State publishes pass rates/rankings of teacher-preparation institutions	30	✓
Programs accountable for graduates' performance in classroom setting	18	✓
Data Systems to Monitor Quality (2007)		
State assigns a unique identification number to each teacher	46	$\checkmark$
Teacher and student records can be matched by course/subject and state assessment	12	

Notes: <sup>a</sup>Kentucky requires 12 weeks, which is the same as the median for the 39 states with student teacher training.

<sup>b</sup>The median for the 13 states requiring other clinical experience is 100 hours.

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

#### **Incentives and Teacher Allocation**

Kentucky has implemented 8 of the 17 recommended policies with respect to incentives and teacher allocation. Table 4.7 summarizes the 17 policies relating to teacher incentives and allocation of talent. Kentucky has implemented eight of these policies. Of the nine policies not in place, two are in a majority of other states: teacher license reciprocity agreements with other states and monitoring of highly qualified teachers by school poverty level.

## Table 4.7 Quality Counts Teacher Incentives and Allocation Measures, 2008

	Implemen	nted
Policies Relating to Incentives and Teacher Allocation	# of States	KY
Reducing Entry and Transfer Barriers (FY 2008)		
State finances/regulates an alternative-route teacher-preparation program to recruit candidates with at least a bachelor's degree	47	~
State has teacher-license reciprocity or portability agreement with other state(s)	38	
State policy allows portability of teacher pension across state lines	20	
Teacher Salaries		
Pay Parity (2006): average teacher salary is 100% of average in comparable occupations <sup>a</sup>	10	
State requires all districts to report average teacher salaries at the school level (FY 2008)	12	
Incentives for Teacher Leadership and Performance (FY 2008)		
State has pay-for-performance program or pilot rewarding teachers for raising student achievement	7	
State formally recognizes differentiated roles for teacher leaders	20	✓
State provides incentives or rewards to teachers taking on leadership roles	17	✓
State provides incentives for teachers to earn national board certification	38	✓
Monitoring the Distribution of Teaching Talent by School Poverty Level (FY 2008)		
Fully licensed teachers	42	$\checkmark$
Highly qualified teachers	31	
First-year teachers	35	$\checkmark$
National-board-certified teachers	25	✓
Managing the Allocation of Talent (FY 2008)		
State provides incentives to teachers who work in targeted assignments		
Targeted schools	20	
Targeted teaching assignment areas	16	✓
State provides incentives for national-board-certified teachers to work in targeted schools	10	
State provides incentives to principals who work in targeted schools	10	

Note: <sup>a</sup>With Kentucky teachers paid 94.3% of what comparable workers earn, Kentucky ranks 18<sup>th</sup>. Source: Editorial Projects in Education. *Education Week's Quality Counts* 2008.

### **Building and Supporting Teacher Capacity**

Kentucky has 11 of the 17 policies recommended for building and supporting teacher capacity.

Table 4.8 summarizes the 17 policies related to building and supporting teacher capacity. Kentucky has implemented 11 of these policies. Of the six policies not implemented, two are present in a majority of other states: penalties for incidents of school violence and state-financed program to reduce school violence.

# Table 4.8Quality Counts Building and Supporting Teacher Capacity, 2008

	Implemented		
Policies Relating to Building and Supporting Teacher Capacity	# of States	KY	
Supports for Beginning Teachers (FY 2008)			
All new teachers are required to participate in a state-funded induction program	22	✓	
Mentoring programs for all beginning teachers			
All new teachers are required to participate in a state-funded mentoring program	25	$\checkmark$	
State has standards for selecting, training, and/or matching mentors	20	✓	
State has a reduced-workload policy for first-year teachers	2		
Professional Development (FY 2008)			
State has formal professional-development standards	41	$\checkmark$	
State finances professional development for all districts	24	$\checkmark$	
State requires districts/schools to set aside time for professional development	16	$\checkmark$	
State requires districts to align professional development with local priorities and goals	30	$\checkmark$	
School Leadership (2006)	-		
State has standards for licensure of school administrators	48	$\checkmark$	
State requirements for initial administrator licensure			
Supervised internship	28	$\checkmark$	
Participation in induction or mentoring program	14	$\checkmark$	
School Working Conditions			
State has a class-size-reduction program or regulations to limit class size (FY 2008)	18		
Median student-teacher ratio in primary-level schools is 15.0 or lower (2005) <sup>a</sup>	26	$\checkmark$	
State tracks condition of all school facilities (FY 2008)	22		
State collects and publicly reports school-level information on school climate and working conditions (FY 2008)	3		
State policy imposes penalties for incidents of school violence (FY 2008)	36		
State finances program to reduce school violence (FY 2008)	28		

Note: <sup>a</sup>The median student-teacher ratio in Kentucky's primary-level schools is 15.0. Source: Editorial Projects in Education. *Education Week*'s *Quality Counts* 2008.

#### Preschools

Kentucky meets 8 out of the 10 preschool quality standards recommended by the National Institute for Early Education Research. The National Institute for Early Education Research reports that Kentucky meets 8 out of 10 recommended quality standards, the same as last year. As Figure 4.A shows, the two standards Kentucky does not meet are requiring teachers to have a Child Development Associate credential and having the Kentucky Department of Education visit preschool sites.



#### Figure 4.A Pre-Kindergarten Quality Standards, FY 2007

Notes: \*Monitoring refers to department of education personnel making periodic site visits to preschools. \*\*CDA means Child Development Associate credential.

Source: Staff compilation based on data from Natl. Institute. The State 19.

#### **Title I Schools**

The high percentage of students enrolled in Title I schools and schoolwide programs was discussed in Chapter 3. This table shows Kentucky's high percentage of schools that are eligible for Title I. Chapter 3 discussed the high percentages of students enrolled in Title I schools and schoolwide programs. Table 4.9 ranks states by the percentages of schools eligible for Title I and schoolwide programs. Kentucky ranks 17<sup>th</sup> with over 60 percent of schools eligible for Title I and ranks 5<sup>th</sup> with over half of schools operating schoolwide programs.

#### Table 4.9

### Title I-Eligible Schools and Schoolwide Programs as a Percentage of All Schools, FY 2006

T As a l	itle I-Eligible Scho Percentage of All S	ols chools	Title I Schoolwide Programs As a Percentage of All Schools					
Rank	State	%	Rank	State	%			
2	IN	90.1	2	MS	59.9			
5	AR	72.6	3	TX	58.4			
3	OK	/3.0	5	KY	55.0			
10	10 OH		5	OK	55.9			
16	MS	63.5	7	LA	55.5			
17	KY	62.0	8	AR	53.3			
1 /	LA	02.9	10	AL	46.4			
19	TX	62.7	11	TN	45.1			
	U.S.	55.3	12	SC	42.4			
25	AL	54.8	13	GA	39.9			
28	TN	53.4	14	NC	39.2			
29	IL	52.5	15	VA	37.1			
30	MO	49.3	16	FL	36.4			
32	GA	48.4	18	WV	34.8			
34	NC	47.8		U.S.	31.4			
37	WV	45.7	19	DE	30.6			
38	DE	45.0	25	OH	24.6			
39	SC	44.9	27	MD	22.9			
43	FL	37.6	32	IL	21.4			
45	VA	37.1	40	MO	18.0			
47	MD	27.0	48	IN	8.7			

Note: Ranks are out of 50; data were not available for NJ. Title I-eligible schools are encouraged to create schoolwide programs if 40 percent or more of their students are eligible for free or reduced-price lunch. Otherwise, assistance is targeted to individual students who are eligible for free or reduced-price lunch. Source: Staff compilation of data from U.S. Dept. of Ed. Inst. Natl. *Numbers and Types of Public Elem. and* 

Secondary Schools 4-5.

#### Technology

#### About Education Week's Technology Counts

Kentucky is considered a national leader in school technology, ranking 4<sup>th</sup> overall.

*Technology Counts* ranks states on 14 indicators covering three major areas of state technology policy and practice: access, use, and capacity. The overall rankings for these areas are presented in Table 4.10. Subsequent tables show state scores on the indicators that are used to calculate scores. As Table 4.10 shows, Kentucky is a national leader in school technology, ranking 4<sup>th</sup> overall, 13<sup>th</sup> for access to technology, 5<sup>th</sup> for use of technology, and 3<sup>rd</sup> for capacity to use technology.

Overall		Access				Us	e	Capacity			
Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)
1	WV	A (95.3)	3	WV	A (96.3)	1	GA	$\Lambda$ (100.0)	1	GA	A (100 0)
3	GA	A- (91.2)	7	VA	A- (90.0)	1	NC	A (100.0)	1	WV	A (100.0)
4	KY	B+ (88.5)	12	IN	B (82 5)		AR		3	KY	A (93.2)
5	VA	B+ (88.7)	15	KY	D (82.3)		FL			AR	
6	FL	B (85.4)	16	FL	D (90.0)		KY	A- (89.8)	4	MD	B- (79.5)
8	LA	B- (82.0)	10	SC	D- (80.0)		LA			OH	
10	NC	B- (81.5)	19	NC	C+ (78.8)	5	MD			OK	
11	OK	B- (80.9)	23	OH	C (75.0)		MO			SC	
13	AR	B- (80.1)	26	GA	C(72.5)		OK			FL	
15	SC	B- (79.7)	20 OK C (73.3)	VA			IL				
17	MO	C+ (77.0)	20	IL	C(72.5)	W	WV		16	LA	B (86.3)
19	MD	C+ (77.9)	29	ΤХ	C(12.3)		AL			ΤХ	
20	IN	C+ (78.2)	32	AR	C- (71.0)		IL			VA	
24	IL	C + (70.4)	22	LA	$C_{1}(70.0)$		IN			AL	
24	ΤХ	C+(79.4)	33	TN	C-(70.0)	17	MS	B- (79.5)	25	DE	C (72.7)
34	OH	C (74.6)	35	MO	D+ (68.5)		SC			IN	
36	TN	C (74.1)	39	AL	D (65.5)		TN			MO	
39	AL	C (72.6)	41	MD	D (64.5)		ΤХ			MS	
44	MS	C- (70.4)	46	DE	D- (60.5)	26	DE	$D \pm (60.2)$		TN	
47	DE	D+ (67.5)	47	MS	F (59.0)	36	OH	D⊤ (09.3)	38	NC	D (65.8)

Table 4.10Education Technology Overall Grade, Access, Use, and Capacity, 2008

Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

#### Access to Technology

Kentucky is above the national average for all technology access measures, ranking 13<sup>th</sup> overall.

The access index is made up of the percentage of students who say they have access to a computer in the classroom, the number of students per instructional computer, and the number of students per high-speed Internet-connected computer. State rankings on these measures are reported in Table 4.11. Kentucky is above the national average on all access measures, with a rank of 3<sup>rd</sup> for access to computers in grade 4 classrooms, 7<sup>th</sup> for access to computers in grade 8 classrooms, 20<sup>th</sup> for the student-computer ratio, and 19<sup>th</sup> for the number of students per high-speed Internetconnected computer.

Table 4.11Access to Technology, 2006 and 2007

Access to Technology Grade and Score		Percentage of 4 <sup>th</sup> Grade Students With Access to Computers 2007		Percentage of 8 <sup>th</sup> Grade Students With Access to Computers 2007			Students Per Instructional Computer 2006			Students Per High-speed Internet-connected Computer 2006					
Grade		Grade							<b>D</b> 1	<b>G</b> ( )	D (1				
Rank	State	(Score)		State	<b>%</b>	Rank	State	<b>%</b>	Rank	State	Ratio	Rank	State	Ratio	
3	WV	A (96.3)	1	WV	100	2	WV	97	2	DE	5.2	2	MS	5.0	
7	VA	A- (90.0)	3	KY	98	3	VA	95	4	MS	5.0	4	DE	4.9	
13	IN	B (82.5)	_	NC		7	KY	92	6	AL	4.8	5	AL	4.8	
	KY	2 (02.0)		AL			GA		10	MD	4.5	9	MD	4.4	
16	FL	$B_{-}(80.0)$	7	FL	97	11	IN	90	15	LA	11	11	LA	4.3	
10	SC	D- (80.0)		SC			SC		15	TN	4.1	14	TN	4.1	
20	NC	C+ (78.8)		IN		17	IL	88	19	IL	4.0	18	IL	3.9	
23	OH	C (75.0)		LA	96	10	LA	07	20 K	KY	2.0		AR		
•	GA			OK		19	19	NC	87	8/ 20	NC	3.9	19	KY	3.8
26	OK	C (73.5)	12	TN		90	MD		6 23	AR		1	NC	1	
	IL			ТХ		24	OH	86		GA	3.8	23	GA	37	
29	TX	C (72.5)		VA	<u> </u>	28	TN	83		SC		25	SC	3.6	
32	AR	$C_{-}(71.0)$		AR		31	TX	82		MO		28	OH	3.5	
52	ΙΔ	0 (71.0)	22	II	95	51	DE	02		OH		20	OH	5.5	
33	TN	C- (70.0)		MD		34	DL EI	81	29	OK	3.5	30	OK	31	
25	MO	D + (69.5)	31		94	94 20	TL MO	70			-	50		5.4	
20		$D^{+}(08.3)$				38	MO	79				24		2.2	
39	AL	D (05.5)		GA		41	MD	/8	35	FL DI	3.3	34	IIN	3.3	
41	MD	D (64.5)	37	MO	93	43	AL	76		IN		38	FL	3.2	
46	DE	D- (60.5)		MS			OK		38	WV	3.2	44	VA	3.0	
47	MS	F (59.0)	45	DE	91	46	AR	75	41	VA	3.1		WV	5.0	

Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

Figure 4.B shows trends in the number of students per instructional computer for Kentucky and the U.S. A lower number of students indicates better access. Kentucky provided better access than the

nation in 1999 and 2000 but lost this advantage in 2001 and has stayed on par with the nation since then.





Source: Staff compilation of data from Editorial Projects in Education. EdCounts Database.

Kentucky has implemented all of the recommended policies regarding use of technology except the assessment of students' technology skills.

#### Use of Technology

Use of technology is based on four policies: state includes technology in academic standards for students, state tests students on technology, state has established a virtual school, and state offers computer-based assessments. In Table 4.12, checkmarks indicate the states in which the policies are in place. The only policy that Kentucky lacks is assessment of students' technology skills, a policy in place in only five states.

Use of Technology Grade and Score			State Standards For Students	State Tests	State Has	State Offers	
Rank	State	Grade (Score)	Total Implemented	Include Technology	Students On Technology	Established a Virtual School	Computer-based Assessments
1	GA	А	4	$\checkmark$	✓	✓	✓
1	NC	(100.0)	4	$\checkmark$	✓	✓	✓
	AR			$\checkmark$		✓	✓
	FL			$\checkmark$		✓	✓
	KY			$\checkmark$		$\checkmark$	✓
	LA		3	$\checkmark$		✓	✓
5	MD	A-		$\checkmark$		✓	$\checkmark$
	MO	(09.0)		$\checkmark$		✓	✓
	OK			$\checkmark$		$\checkmark$	$\checkmark$
	VA			$\checkmark$		✓	$\checkmark$
	WV			$\checkmark$		✓	$\checkmark$
	AL			$\checkmark$		✓	
	IL		2	$\checkmark$		✓	
	IN			$\checkmark$			$\checkmark$
17	MS	B- (79.5)				✓	$\checkmark$
	SC	(7).5)		$\checkmark$		✓	
	TN			$\checkmark$			$\checkmark$
	TX			$\checkmark$			$\checkmark$
26	DE	D+	1	$\checkmark$			
- 36	OH	(69.3)	1	$\checkmark$			
Number of States/I		s/DC	48	5	25	27	

Table 4.12Use of Technology, FY 2008

Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

Kentucky ranks 3<sup>rd</sup> with respect to the capacity to use technology, lacking only a policy that would require administrators to demonstrate technology skills or take technology-related training.

#### **Capacity To Use Technology**

This indicator is based on six policy indicators, shown in Table 4.13, designed to measure the extent to which states include technology in their personnel requirements. Improving students' ability to benefit from using computers is closely tied to improving professional development for teachers in the area of technology use (U.S. Dept. of Ed. Inst. Natl. *Monitoring* 28). Kentucky ranks 3<sup>rd</sup>, lacking only a policy that would require administrators to demonstrate technology skills or take technology-related training or professional development.

Capacity To Use Technology Grade and Score		Total # of	State Standards otal # of Include Technology		Requirem Initial Include T Coursewor	ents for an License echnology k or a Test	State Requires Technology-related PD or Training or Testing for Recertification		
Rank	State	Grade (Score)	Policies Implemented	Teachers	Admin.	Teachers	Admin.	Teachers	Admin.
	GA	A		√	√	✓	✓	✓	√
1	WV	(100.0)	6	✓	✓	✓	~	~	✓
3	KY	A (93.2)	5	~	~	✓	✓	✓	
	FL			~	~	✓	~		
	IL	D		✓	✓	✓	✓		
4	LA	В (863)	4	✓	~	✓			$\checkmark$
	TX	(00.2)		✓	$\checkmark$	✓	$\checkmark$		
	VA			$\checkmark$	$\checkmark$	✓	✓		
	AR			✓				✓	$\checkmark$
	MD	D		✓	$\checkmark$	✓			
13	OH	B- (79.5)	3	~	$\checkmark$	✓			
	OK	(,,,,,,,)		✓	~	✓			
	SC			~	~			~	
	AL			✓	~				
	DE			~	~				
25	IN	С	2	~	~				
23	MO	(72.7)	2	~	~				
	MS			✓	✓				
	TN			✓	~				
38	NC	D (65.8)	1	<b>√</b>					
<b>Total States With Indicated Policies</b>			51	49	43	39	37	36	

Table 4.13Capacity To Use Technology, FY 2008

Note: PD = professional development.

Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

The data source for studentcomputer ratios undercounts computers. This may be partially offset by the fact that many computers included in the count may be old and of limited use. It is also important to remember that overall national and state averages obscure the uneven distribution of computers.

#### **Caveats and Limitations**

A recent NCES report found that the data sources for studentcomputer ratios tend to undercount computers. However, this undercount is partially offset by the fact that many computers that are counted should not have been included because they are old and have limited utility.

Overall student-to-computer ratios for the nation and for states obscure the fact that computers are not evenly distributed; some districts have many computers, while others have few. This is not evident in an overall average student-to-computer ratio. Finally, some states that score high on technology indicators are not ranked high in educational performance. The link between technology and student outcomes remains unknown (U.S. Dept. of Ed. Inst. Natl. *Monitoring* 28).

**Uncertain Benefits To Student Learning**. There is widespread agreement that classrooms need computers so that students can gain computer skills to succeed in today's workplace. However, there is less consensus regarding the nature and degree of other benefits that may be gained, such as critical thinking skills and knowledge of other content areas (U.S. Dept. of Ed. Inst. Natl. *Monitoring* 28).

**Data Comparability**. Table 4.12 includes a measure of whether the state offered computer-based assessments in FY 2008 and indicates that Kentucky has this policy in place. However, the availability of computer-based assessments is neither statewide nor to the majority of students. Kentucky has been conducting pilot programs to test computer-based assessment. The state also offers it for some special needs students. The comparability of computerbased assessment across states is unknown because *Technology Counts* does not discuss specific implementation differences.

#### Urban, Suburban, and Rural Locations

As Table 4.14 shows, Kentucky has the 8<sup>th</sup> highest percentage of students in rural districts. Rural schools have special challenges as well as some advantages. Costs per student are higher due to underenrolled schools and long bus routes to carry geographically dispersed students. Compared to other locales, rural schools offer fewer Advanced Placement and International Baccalaureate courses. However, achievement levels are typically higher, and school climates are better in rural schools than in urban schools (U.S. Dept. of Ed. Inst. Natl. *Status of Education*).

There is widespread agreement that computers are needed for gaining computer skills, but there is less consensus about the nature and degree of other learning benefits.

Kentucky has the 8<sup>th</sup> most rural student population. Rural schools have unique advantages and disadvantages over other schools.
Chapter 4

Office of Education Accountability

Rank	State	%
3	MS	51.3
4	WV	48.6
5	NC	47.0
8	KY	40.1
9	AR	38.4
13	AL	31.5
14	ОК	31.1
15	VA	29.6
16	SC	29.2
17	LA	28.1
20	IN	27.2
22	TN	26.7
23	OH	25.3
25	MO	24.2
27	GA	23.0
31	DE	17.7
	U.S.	17.4
33	TX	12.4
38	IL	11.2
40	MD	9.1
43	FL	7.4

Table 4.14Students in Rural Areas, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Numbers and Types of Public Elem. and Secondary Agencies 12.

#### **School Crime and Safety**

NCES collaborates with the U.S. Bureau of Justice Statistics to produce a report on crime and safety that brings together the perspectives of students, teachers, principals, and the public. The sources include several federally funded collections such as the National Crime Victimization Survey, Youth Risk Behavior Survey, School Survey on Crime and Safety, and School and Staffing Survey. In these surveys, students and teachers report incidents occurring at school and on the way to and from school (U.S. Dept. of Ed. Inst. Natl. *Indicators*).

#### **Teacher-reported Incidents**

Table 4.15 shows the percentages of teachers reporting that students threatened them with bodily injury. This is a more common occurrence in cities than in rural and suburban areas; consequently, percentages are high in only a handful of states that have large cities. In 2004, 7.9 percent of teachers in Kentucky

Incidents of students threatening teachers with injury are less common in rural and suburban areas than in cities. In 2004, 7.9 percent of Kentucky teachers reported threats, compared to 7.5 percent for the U.S. reported threats, compared to 7.5 percent for the U.S. Although Kentucky had the 15<sup>th</sup> highest rate in 2004, this rate differed by only a few percentage points from rates in most other states. The incidence of threats has been decreasing dramatically; between 1994 and 2004, the percentage of teachers reporting threats dropped 43.6 percent in Kentucky, compared to a drop of 41.4 percent for the U.S.

Table 4.15
Public School Teachers Threatened With Injury by a Student at School,
FY 1994 and FY 2004

	FY 1994			FY 2004		Percent Change						
Rank	State	%	Rank	State	%	Rank	State	%				
2	FL	20.1	2	MD	13.5	5	AR	-65.3				
3	MD	19.9	3	FL	11.2	7	ОН	-59.3				
4	DE	18.7	5	LA	9.9	8	MS	-58.9				
5	NC	17.1	10	NC	8.7	9	DE	-58.8				
6	LA	17.0	11	SC	8.6	10	WY	-57.8				
8	SC	15.3	12	MO	8.3	11	VA	-56.3				
9	OH	15.2	14	IL	8.0	12	AL	-54.3				
10	VA	14.9	15	KY	7.9	13	GA	-54.2				
11	GA	14.0	17	DE	77	16	NC	-49.2				
11	KY	14.0	1 /	TX	1.1	19	IN	-47.8				
12	AR	12.9		U.S.	7.5	20	TN	-47.2				
15	IN	15.8	21	IN	7.2	24	OK	-44.7				
17	MS	13.4	21	WV	1.2	27	FL	-44.4				
19	AL	13.3	26	TN	6.6	28	KY	-43.6				
	U.S.	12.8	27	VA	6.5	29	SC	-43.6				
25	TX	12.7	28	GA	6.4	32	LA	-41.7				
26	MO	12.6	30	OH	6.2		U.S.	-41.4				
27	TN	12.5	21	AL	6.1	34	TX	-39.4				
31	WV	11.4	51	OK	0.1	39	MO	-33.9				
34	OK	11.0	37	MS	5.5	40	MD	-32.0				
36	IL	10.8	45	AR	4.8	43	IL	-26.2				

Note: Staff calculated the percent change using unrounded percentages for FY 1994 and FY 2004. Source: Staff compilation of data from U.S. Dept. of Ed. Inst. Natl. *Indicators* 79.

#### **Student-reported Incidents**

Student involvement in violence and substance abuse in Kentucky is below the national average. Table 4.16 shows several types of undesirable incidents reported by students, including fighting, being threatened with weapons, having access to drugs, and using alcohol. Kentucky's rates are lower than the national rates for all measures.

				On		Anywhere								
V Dhy	Vere in	a iaht	Were	Threater	ned or	Н	ad Drug	gs	Use	d Alaak	vol	Uaz	d Alaal	hol
Pily	Sical FI		Damle			Dank		e 0/	Dank	u Alcol	0/	Dank	State	101 07
Kank	State	<b>%</b> 0	Rank	State	<i>%</i> 0	Kank	State	<b>%</b> 0	Rank	State	<b>%</b> 0	Rank	State	<i>%</i> 0
3	MD	14.9	2	MD	11.7	1	GA	32.0	8	SC	6.0	1	IX	47.3
4	TX	14.5	4	SC	10.1	9	OH	31.0	10	TX	5.7	13	MO	45.0
5	AR	13.9	6	AR	9.6	11	ΤX	30.7	11	WV	5.3	15	DE	44.2
6	AL	13.7	7	ΤX	9.3	13	NC	29.6	12	AR	5.2	15	OK	44.2
	U.S.	13.2	11	AL	8.9	14	AR	29.2	12	DE	5.2		U.S.	44.1
9	SC	12.7		U.S.	8.6	15	SC	29.1	16	MS	4.9	21	IN	12.2
11	FL	12.4	16	MO	0.2	16	MD	28.9	19	FL	4.8	21	SC	43.2
17	OK	11.7	10	WV	0.5	17	IN	28.6		U.S.	4.7	23	AR	43.1
18	GA	11.6	18	GA	8.2	20	DE	27.0	25	NC	4.5	24	WV	43.0
19	TN	11.5	19	FL	8.1		U.S.	27.0	27	AL	4.3	28	OH	42.3
20	KY	11.4	22	OH	8.0	22	AL	26.1	28	KY	4.1	30	MS	41.8
24	WV	11.2	24	TN	7.9	24	WV	25.6	32	GA	4.0	31	TN	41.4
26	IN	11 1	25	IN	7.8	25	TN	25.5	34	TN	3.9	32	KY	41.3
20	NC	11.1	29	NC	7.5	27	KY	25.1	27	IN	2.6	33	FL	41.2
29	OH	10.8	35	DE	6.9	29	FL	24.4	57	OH	5.0	35	NC	40.9
30	DE	10.6	38	OK	6.7	34	MS	22.3	39	OK	3.5	36	AL	20.8
33	MS	10.3	20	KY	6.6	40	OK	20.3	40	MD	3.2	50	MD	39.0
37	MO	10.0	39	MS	0.0	41	MO	19.9	41	MO	2.9	38	GA	38.8
	IL			IL			IL			IL			IL	
n.a.	LA	n.a.	n.a.	LA	n.a.	n.a.	LA	n.a.	n.a.	LA	n.a.	n.a.	LA	n.a.
	VA			VA			VA			VA			VA	

Table 4.16Public School Students Involved in Violence or Substance AbuseAverage of 2003 and 2005

Note: Because some states reported only in 2003 and some reported only in 2005, staff averaged the 2003 and 2005 data; states that failed to report in both 2003 and 2005 are indicated by n.a. Ranks are out of 42 for use of alcohol on school property and out of 43 for all other measures.

Source: Staff compilation using data from U.S. Dept. of Ed. Inst. Natl. Indicators 76, 91, 101, and 105.

#### **Chapter Summary and Conclusions**

This chapter presented data on school characteristics, studentteacher ratios, and participation in such programs as subsidized lunches, English language learner services, and IEPs. Compared to the nation, a higher percentage of Kentucky students attend rural schools, which have special challenges as well as some advantages. The Commonwealth continues to perform well with regard to technology access, use, and capacity. School climates are generally better than average in Kentucky, with students less likely to report being in fights, being threatened with weapons, having drugs available, or using alcohol. For many years, in Kentucky and the nation as a whole, the number of teachers employed in schools has increased faster than enrollment, a trend attributed to schools' increasingly complex and diverse offerings. Kentucky's student-teacher ratio is better (lower) than the national average in elementary schools but slightly worse (higher) in middle and high schools. In terms of teachers as a percentage of all staff, Kentucky is again ranked last in the nation because of high percentages of instructional aides and support staff. Additional research is needed to better understand the relatively high percentage of nonteacher staff employed in Kentucky schools.

Kentucky receives high marks for policies relating to the teaching profession. Among all states, Kentucky has the 3<sup>rd</sup> highest percentage of teachers with advanced degrees.

## **Chapter 5**

## Finance

While policy makers acknowledge the critical role of education in the global economy, education competes with other issues for resources. This chapter presents data on revenues, expenses, funding equity, and teacher salaries. FY 2006 is the most recent year for which data are available. Given this lag time in collecting and reporting data, these tables do not reflect the recent downturn in the economy.

#### Revenues

#### **Revenues Per Pupil**

Table 5.1 reports combined federal, state, and local revenues per pupil before and after adjusting for geographic differences in the cost of living. Kentucky ranks 43<sup>rd</sup> with respect to unadjusted revenues per pupil. When revenues are adjusted, Kentucky rises to 38<sup>th</sup>.

Kentucky ranks 43<sup>rd</sup> in terms of revenues per pupil. When revenues are adjusted for geographic differences in the cost of living, Kentucky rises to 38<sup>th</sup>.

	1		1	1	[
Rank	State	Unadjusted \$	Rank	State	Cost-adjusted \$
10	DE	13,143	12	DE	12,924
12	MD	12,430	17	IN	12,418
16	OH	11,606	19	OH	12,120
18	IN	11,028	20	LA	12,060
	U.S.	10,771	21	WV	11,850
21	VA	10,672	26	MD	11,496
23	IL	10,506	29	AR	10,886
24	LA	10,456		U.S.	10,771
26	GA	10,113	31	SC	10,700
27	WV	10,032	32	MO	10,597
32	SC	9,643	34	MS	10,408
33	MO	9,585	35	FL	10,307
34	FL	9,542	36	GA	10,301
39	TX	9,210	37	IL	10,179
40	AR	8,960	38	KY	9,995
43	KY	8,828	39	VA	9,860
44	MS	8,644	40	AL	9,771
45	AL	8,560	42	OK	9,597
46	NC	8,434	44	TX	9,311
47	OK	8,069	48	NC	8,937
49	TN	7,512	50	TN	8,174

Table 5.1Revenues Per Pupil, FY 2006

Source: U.S. Dept. of Commerce. Census. Public 2006 11.

## **Revenue Sources**

Compared to other states, Kentucky draws a smaller share of revenues from local sources than from state and federal sources. Kentucky ranks 40<sup>th</sup> in terms of the percentage of revenues from local sources but 16<sup>th</sup> for the percentage from federal sources and 13<sup>th</sup> for the percentage from state sources. Table 5.2 shows the percentages of revenues that come from federal, state, and local sources. Compared to other states, Kentucky draws a smaller share of revenues from local sources than from state and federal sources. Kentucky ranks 40<sup>th</sup> in terms of the percentage of revenues from local sources, compared to 16<sup>th</sup> with respect to federal sources and 13<sup>th</sup> with respect to state sources.

	Federal			State		Local					
Rank	State	%	Rank	State	%	Rank	State	%			
1	MS	20.1	3	AR	73.4	2	IL	59.1			
2	LA	18.8	6	DE	64.5	6	TX	55.5			
8	OK	12.8	9	WV	58.8	8	MD	54.6			
9	WV	12.3	10	NC	58.5	10	VA	53.7			
12	TX	11.6	13	KY	57.3	14	OH	50.4			
14	AL	11.5	17	AL	55.6	15	FL	50.1			
15	AR	11.4	21	OK	50.2	20	MO	47.9			
16	KY	11.3	22	MS	49.4	21	GA	46.8			
17	TN	11.1	23	IN	47.5	22	IN	45.9			
20	NC	10.1		U.S.	46.6	24	TN	45.7			
23	SC	9.9	27	SC	44.8	26	SC	45.3			
24	FL	9.7	28	GA	44.2		U.S.	44.4			
26	GA	9.0	31	MO	43.6	30	LA	39.6			
	U.S.	9.0	32	TN	43.2	32	OK	37.0			
28	IL	8.5	35	OH	42.3	37	AL	32.9			
29	MO	8.4	36	LA	41.5	39	NC	31.4			
37	DE	7.3	39	FL	40.2	40	KY	31.3			
39	OH	7.2	41	VA	39.6	42	MS	30.5			
43	VA	6.7	42	MD	39.2	44	WV	28.9			
44	IN	6.5	47	TX	32.9	45	DE	28.2			
45	MD	6.2	49	IL	32.3	48	AR	15.2			

Table 5.2Revenues by Source, FY 2006

Source: U.S. Dept. of Commerce. Census. Public, 2006 5.

Historical trends reveal that the state share seen today is a recent phenomenon. In the late 19<sup>th</sup> century and early 20<sup>th</sup> century, U.S. education funds came from states; not federal funds. Since the 1930s, states have increased their contributions rapidly, and the federal government started providing some funds. It was not until the late 1970s that state and local funds made up roughly equal shares of U.S. education revenues.

The Kentucky Education Reform Act changed Kentucky's funding landscape abruptly in 1990; state funds were increased to almost 70 percent of Kentucky's revenues, while local and federal funds constituted 20 percent and 10 percent, respectively. However, the state share of funds has declined gradually and steadily, so that the 2006 shares were 57 percent from state sources, 31 percent from local sources, and 11 percent from federal sources.

#### **Expenditures**

#### **Current Spending Per Pupil**

Current spending includes expenditures for day-to-day operations, payments made by the state government on behalf of districts, and employer contributions made by the few school systems that administer their own retirement funds. It excludes capital outlay and interest on debt. (U.S. Dept. of Commerce. Census. *Public* 2006 vi).

Table 5.3 displays current spending per pupil before and after adjusting for geographic cost differences. Kentucky's current expenditures per pupil in FY 2006 were far below the national average. When these expenditures are adjusted for geographic differences in costs, Kentucky moves up only one rank, from 40<sup>th</sup> to 39<sup>th</sup>. However, it should be noted that, unlike other states, Kentucky does not report spending from school fees and activity funds, which are substantial. It is not known what Kentucky's rank would be if spending from these funds were reported.

Table 5.3	
Current Spending Per Pupil in P-12 Fall Enrollment, FY 20	06

Rank	State	Unadjusted \$	Rank	State	Cost-adjusted \$
8	DE	11,633	10	DE	11,439
12	MD	10,670	12	WV	11,047
17	OH	9,598	21	OH	10,023
19	VA	9,447		IN	9,901
20	WV	9,352	25	MD	9,869
21	IL	9,149	27	LA	9,691
	U.S.	9,138	28	AR	9,632
23	IN	8,793		U.S.	9,138
27	GA	8,565	32	SC	8,977
30	LA	8,402	33	MO	8,964
33	MO	8,107	34	IL	8,864
34	SC	8,091	25	VA	° 77°
37	AR	7,927	33	AL	0,720
39	FL	7,759	37	GA	8,724
40	KY	7,662	38	MS	8,695
42	AL	7,646	39	KY	8,675
43	TX	7,561	41	FL	8,381
44	NC	7,388	42	OK	8,280
46	MS	7,221	44	NC	7,828
47	OK	6,961	46	TX	7,643
48	TN	6,883	48	TN	7,489

Sources: Unadjusted numbers are from U.S. Dept. of Commerce. Census. *Public* 8; Staff calculated adjusted numbers using the Comparable Wage Index from U.S. Dept. of Ed. Inst. Natl. "NCES Comparable Wage."

Current spending pertains to operations; it excludes capital outlay and interest on debt. Kentucky's current spending per pupil is below the national average, even after adjusting for geographic cost differences. As Figure 5.A shows, current spending per pupil has been increasing rapidly over the past 45 years. This is true in both Kentucky and the nation as a whole.

Figure 5.A Current Spending Per Pupil in P-12 Average Daily Attendance In Constant 2007 Dollars, 1960-2005



Notes: There are discrepancies in average daily attendance reporting practices from state to state. Source: U.S. Dept. of Ed. Inst. Natl. *Digest*, 2007 258-259.

In recent years, preschool spending declined in Kentucky as well as in the nation as whole. Kentucky's per-child spending was below the national average, even after adjusting for Kentucky's lower costs. However, a budget increase in 2007 pushed Kentucky's per-child spending above the national average.

#### **Preschool Spending**

Eligibility for state-funded preschool is extended to Kentucky's 4-year-olds from low-income families and all 3- and 4-year olds who have disabilities.<sup>9</sup> Children not meeting the criteria for state funds may participate if districts use their own funds or parents pay tuition (Natl. Institute. *The State* 64). Figure 5.B shows inflation-adjusted state spending per child enrolled in preschool from FY 2002 to FY 2007 for Kentucky compared to all state programs in the nation. Before 2007, spending had been declining in Kentucky as well as in the nation, and Kentucky's per-child spending was below the national average, even after adjusting for Kentucky's lower costs. In FY 2007, a budget increase of \$25 million pushed Kentucky's per-child spending above the national average.

<sup>&</sup>lt;sup>9</sup> Kentucky recently raised the income criterion for low-income families from 130 percent to 150 percent of the federal poverty level, making more children eligible.



Source: Natl. Institute The State 4, 64.

#### **Current Spending Per \$1,000 in Personal Income**

Adjusting spending for each state's personal income takes into account what taxpayers are able to afford. Relative to personal income, Kentucky's spending is 16<sup>th</sup> in terms of teacher salaries and 21<sup>st</sup> with respect to teacher benefits. Spending on administration is close to the national average.

Table 5.4 presents current spending per 1,000 in personal income of Kentucky residents. This is another way to adjust for geographic cost differences because it takes into account what taxpayers are able to afford. Relative to the personal income, Kentucky is  $16^{th}$  in terms of spending on teacher salaries and  $21^{st}$  with respect to spending on benefits. Spending on administration is close to the national average.

|          |             | ration                                      | 8   | 3.00   | 2.93   | 2.80  
   
  | 2.75   
   
   | 2.70          | 2.68  | 2.65          | 2.64     | 2.63       | 2.60  | 2.52  
   | 2.49  | cv c  
   
  | 4.46   
   
  | 2.40   
   | 2.38  | 2.33  | 2.32  | 2.24          | 2.11  | 1.97  
   
   | 1.90   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|----------|-------------|---|---|--|--
--
--
--
--
--
--|---------------|-------|---------------|----------|------------|---|---|---
--
--
--
---
--
--|-------|-------|-------|---------------|-------
--
---	--
--
--|---
---
---|--
--|--|---|---|---|---|
|          |             | dminist                                     | State   | GA   | WV   | ME  
   
  | MD   
   
   | $\mathbf{SC}$ | AR    | IN            | AL       | MS         | НО  | LA  
   | NC  | DE  
   
  | ΤX   
   
  | KY   
   | SU    | МО    | VA    | OK            | П     | LΝ  
   
   | FL   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          |             | School Ad                                   | Rank  | 4  | 6  | 8   
   
  | 6  
   
   | 11            | 12    | 15            | 16       | 17         | 19  | 22  
   | 24  | γL  
   
  | 707  
   
  | 28   
   | -     | 35    | 37    | 39            | 41    | 43  
   
   | 45   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          |             | ion   | \$  | 1.45   | 1.41   | 1.39  
   
  | 1.38   
   
   | 1.29          | 1 72  | 67.1          | 1 16     | 1.10       | 1.09  | 1.06  
   | 0.86  | 0.82  
   
  | 0.75   
   
  | 0.67   
   | 0.64  | 0.63  | 0.60  | 0.57          | 0.47  | 0.42  
   
   | 0.34   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | General     | unistrat                                    | State   | AR   | IL   | WV  
   
  | MS   
   
   | НО            | ОМ    | OK            | AL       | LA         | ME  | КҮ  
   | IN  | NS  
   
  | TN   
   
  | ТΧ   
   | NC    | GA    | VA    | $\mathbf{SC}$ | DE    | MD  
   
   | FL   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          |             | Adm   | Rank  | 5  | 9  | 7   
   
  | 8  
   
   | 12            | 16    | 10            | , 1<br>1 | <b>4</b> 1 | 24  | 25  
   | 30  | 1   
   
  | 32   
   
  | 33   
   | 34    | 35    | 37    | 39            | 45    | 48  
   
   | 50   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          |             | ly  | S   | 10.34  | 8.90   | 7.95  
   
  | 7.18   
   
   | 6.94          | 6.19  | 5.99          | 5.92     | 5.90       | 5.81  | 5.68  
   | 5.63  | 5.35  
   
  | 5.33   
   
  | 5.21   
   | 4.62  | 4.35  | 4.33  | 4.30          | 4.08  | 3.56  
   
   | 3.22   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | (<br>i      | efits On                                    | State   | WV   | IN   | ME  
   
  | DE   
   
   | LA            | GA    | KY            | HO       | AL         | MD  | SU  
   | MS  | AR  
   
  | SC   
   
  | VA   
   | Π     | OK    | ОМ    | NT            | NC    | FL  
   
   | XT   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | f           | Ben   | Rank  | 1  | 4  | 8   
   
  | 10   
   
   | 14            | 19    | 21            | 22       | 23         | 25  | -   
   | 28  | 31  
   
  | 32   
   
  | 34   
   | 38    | 40    | 41    | 42            | 45    | 48  
   
   | 50   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
| u        |             | ly  | S   | 21.94  | 21.51  | 20.74   
   
  | 20.15  
   
   | 20.01         | 19.87 | 19.52         | 18.99    | 18.65      | 18.43   | 17.93   
   | 17.89   | 17.87   
   
  | 17.65  
   
  | 17.57  
   | 17.49 | 16.65 | 16.55 | 16.33         | 15.91 | 15.85   
   
   | 12.87  |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
| structio |             | aries On                                    | State   | ME   | GA   | AR  
   
  | ΤX   
   
   | WV            | SM    | $\mathbf{SC}$ | LA       | КҮ         | НО  | NS  
   | МО  | IN  
   
  | NC   
   
  | Π  
   | VA    | DE    | MD    | AL            | TN    | OK  
   
   | FL   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
| In       |             | Sala  | Rank  | 4  | 5  | 7   
   
  | 8  
   
   | 10            | 11    | 13            | 15       | 16         | 19  | :   
   | 25  | 26  
   
  | 31   
   
  | 33   
   | 34    | 36    | 37    | 38            | 41    | 42  
   
   | 49   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | •           | tion  | S   | 33.87  | 32.73  | 30.28   
   
  | 30.23  
   
   | 28.36         | 28.31 | 27.64         | 27.51    | 27.42      | 26.70   | 26.43   
   | 26.20   | 26.05   
   
  | 25.00  
   
  | 24.64  
   | 24.45 | 24.40 | 23.81 | 23.76         | 23.01 | 22.89   
   
   | 19.80  |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          |             | Instruc                                     | State   | ME   | WV   | AR  
   
  | GA   
   
   | LA            | SM    | IN            | HO       | SC         | TX  | SU  
   | DE  | KY  
   
  | AL   
   
  | МО   
   | VA    | IL    | MD    | NC            | TN    | OK  
   
   | FL   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | [           | Total                                       | Rank  | 4  | 5  | 7   
   
  | 8  
   
   | 13            | 14    | 16            | 18       | 19         | 24  | -   
   | 27  | 28  
   
  | 33   
   
  | 35   
   | 36    | 37    | 41    | 42            | 43    | 44  
   
   | 50   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | nt          |   | S   | 54.56  | 52.03  | 50.10   
   
  | 48.77  
   
   | 48.21         | 48.18 | 47.72         | 47.07    | 46.63      | 44.26   | 44.02   
   | 43.34   | 42.83   
   
  | 42.75  
   
  | 41.47  
   | 41.45 | 40.81 | 39.93 | 39.48         | 38.42 | 35.55   
   
   | 33.51  |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | al Curre    | pending                                     | State   | WV   | ME   | AR  
   
  | LA   
   
   | GA            | MS    | НО            | SC       | IN         | ΤX  | КҮ  
   | SU  | DE  
   
  | AL   
   
  | OK   
   | П     | МО    | VA    | MD            | NC    | $\mathbf{TN}$   
   
   | FL   |   |  
   |   
  |   |  
  |  
  |  |  |  |   |   
   |   |   |
|          | Tot         |   | Rank  | 4  | 9  | 8   
   
  | 11   
   
   | 13            | 14    | 15            | 17       | 18         | 21  | 22  
   | -   | 26  
   
  | 27   
   
  | 33   
   | 34    | 36    | 38    | 40            | 43    | 48  
   
   | 50   |   |  
   |   
  |   |  
  |  
  |  |  |  |   | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | |
   |   |   |
|          | Instruction | Instruction     General       Total Current | Total CurrentInstructionEnergiesSpendingTotal InstructionSalaries OnlyBenefits OnlyAdministrationSchool AdministrationSchool Administration | Total Current     Instruction       Total Current     General       Spending     Total Instruction     Salaries Only     Benefits Only     Administration     School Administration       Rank     State     \$     \$     \$     Rank     State     \$     < | Total CurrentImage: Image: Im | Total CurrentInstructionSpendingColspan="6">Colspan="6">Colspan="6"SpendingSpendingSchool AdministrationSpendingSpendingSchool AdministrationSpending <th colspan="6" s<="" th=""><th>Total CurrentImatter constructionSchool AdministrationTotal CurrentGeneralSpendingSpendingSchool AdministrationSpendingState SRankState SRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankState&lt;</th><th>InstructionInstructionTotal CurrentImage: Image: Imalited Image: Image: Image: Image: Image: Image: Imag</th><th>Instruction         Instruction         I</th><th>Internation         Internation         Internation         Internation         Internation           Total Current         Internation         Salaries Only         Benefits Only         Administration         School Administration           Spending         Total Instruction         Salaries Only         Benefits Only         Administration         School Administration           Rank         State         \$\$         \$\$         \$\$         \$\$         State         \$\$         State         \$\$         Rank         State         \$\$         Rank         State         \$\$         State         \$\$         State         \$\$         State         \$\$         \$\$         \$\$         \$\$<!--</th--><th>InstructionInstructionInstructionInstructionInstructionSelaries OnlySelaries OnlySelon AministrationSelaries OnlySelaries Only<th <="" colspan="6" th=""><th>Instruction         Instruction         Select of II Instruction         School Administration           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S<th>Instruction         Instruction           Rank         State         State</th><th>Intruction         Intruction         State Stat</th><th>Instruction         Instruction         Instruction         Shole Administration           Total Current         Total Current         Salaries Only         Salaries Only         Administration         School Administration           Potelling         Total Instruction         Salaries Only         Administration         School Administration           Rank         State         S         R         R         State         S         R         R         S</th><th>Instruction         Instruction         State SDIP         Administration         State SDIP           Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         S         S         S         S<th>Instruction         Instruction           Total Current         Target on the state on the state</th><th>Introduction         Introduction         <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th></th></th></th></th></th></th></th> | <th>Total CurrentImatter constructionSchool AdministrationTotal CurrentGeneralSpendingSpendingSchool AdministrationSpendingState SRankState SRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankState&lt;</th> <th>InstructionInstructionTotal CurrentImage: Image: Imalited Image: Image: Image: Image: Image: Image: Imag</th> <th>Instruction         Instruction         I</th> <th>Internation         Internation         Internation         Internation         Internation           Total Current         Internation         Salaries Only         Benefits Only         Administration         School Administration           Spending         Total Instruction         Salaries Only         Benefits Only         Administration         School Administration           Rank         State         \$\$         \$\$         \$\$         \$\$         State         \$\$         State         \$\$         Rank         State         \$\$         Rank         State         \$\$         State         \$\$         State         \$\$         State         \$\$         \$\$         \$\$         \$\$<!--</th--><th>InstructionInstructionInstructionInstructionInstructionSelaries OnlySelaries OnlySelon AministrationSelaries OnlySelaries Only<th <="" colspan="6" th=""><th>Instruction         Instruction         Select of II Instruction         School Administration           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S<th>Instruction         Instruction           Rank         State         State</th><th>Intruction         Intruction         State Stat</th><th>Instruction         Instruction         Instruction         Shole Administration           Total Current         Total Current         Salaries Only         Salaries Only         Administration         School Administration           Potelling         Total Instruction         Salaries Only         Administration         School Administration           Rank         State         S         R         R         State         S         R         R         S</th><th>Instruction         Instruction         State SDIP         Administration         State SDIP           Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         S         S         S         S<th>Instruction         Instruction           Total Current         Target on the state on the state</th><th>Introduction         Introduction         <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th></th></th></th></th></th></th> |               |       |               |          |            | Total CurrentImatter constructionSchool AdministrationTotal CurrentGeneralSpendingSpendingSchool AdministrationSpendingState SRankState SRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankStateSRankState< | InstructionInstructionTotal CurrentImage: Image: Imalited Image: Image: Image: Image: Image: Image: Imag | Instruction         I | Internation         Internation         Internation         Internation         Internation           Total Current         Internation         Salaries Only         Benefits Only         Administration         School Administration           Spending         Total Instruction         Salaries Only         Benefits Only         Administration         School Administration           Rank         State         \$\$         \$\$         \$\$         \$\$         State         \$\$         State         \$\$         Rank         State         \$\$         Rank         State         \$\$         State         \$\$         State         \$\$         State         \$\$         \$\$         \$\$         \$\$ </th <th>InstructionInstructionInstructionInstructionInstructionSelaries OnlySelaries OnlySelon AministrationSelaries OnlySelaries Only<th <="" colspan="6" th=""><th>Instruction         Instruction         Select of II Instruction         School Administration           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S<th>Instruction         Instruction           Rank         State         State</th><th>Intruction         Intruction         State Stat</th><th>Instruction         Instruction         Instruction         Shole Administration           Total Current         Total Current         Salaries Only         Salaries Only         Administration         School Administration           Potelling         Total Instruction         Salaries Only         Administration         School Administration           Rank         State         S         R         R         State         S         R         R         S</th><th>Instruction         Instruction         State SDIP         Administration         State SDIP           Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         S         S         S         S<th>Instruction         Instruction           Total Current         Target on the state on the state</th><th>Introduction         Introduction         <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th></th></th></th></th></th> | InstructionInstructionInstructionInstructionInstructionSelaries OnlySelaries OnlySelon AministrationSelaries OnlySelaries Only <th <="" colspan="6" th=""><th>Instruction         Instruction         Select of II Instruction         School Administration           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S<th>Instruction         Instruction           Rank         State         State</th><th>Intruction         Intruction         State Stat</th><th>Instruction         Instruction         Instruction         Shole Administration           Total Current         Total Current         Salaries Only         Salaries Only         Administration         School Administration           Potelling         Total Instruction         Salaries Only         Administration         School Administration           Rank         State         S         R         R         State         S         R         R         S</th><th>Instruction         Instruction         State SDIP         Administration         State SDIP           Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         S         S         S         S<th>Instruction         Instruction           Total Current         Target on the state on the state</th><th>Introduction         Introduction         <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th></th></th></th></th> | <th>Instruction         Instruction         Select of II Instruction         School Administration           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S<th>Instruction         Instruction           Rank         State         State</th><th>Intruction         Intruction         State Stat</th><th>Instruction         Instruction         Instruction         Shole Administration           Total Current         Total Current         Salaries Only         Salaries Only         Administration         School Administration           Potelling         Total Instruction         Salaries Only         Administration         School Administration           Rank         State         S         R         R         State         S         R         R         S</th><th>Instruction         Instruction         State SDIP         Administration         State SDIP           Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         S         S         S         S<th>Instruction         Instruction           Total Current         Target on the state on the state</th><th>Introduction         Introduction         <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th></th></th></th> |       |       |       |               |       | Instruction         Select of II Instruction         School Administration           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S <th>Instruction         Instruction           Rank         State         State</th> <th>Intruction         Intruction         State Stat</th> <th>Instruction         Instruction         Instruction         Shole Administration           Total Current         Total Current         Salaries Only         Salaries Only         Administration         School Administration           Potelling         Total Instruction         Salaries Only         Administration         School Administration           Rank         State         S         R         R         State         S         R         R         S</th> <th>Instruction         Instruction         State SDIP         Administration         State SDIP           Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         S         S         S         S<th>Instruction         Instruction           Total Current         Target on the state on the state</th><th>Introduction         Introduction         <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th></th></th> | Instruction           Rank         State         State | Intruction         State Stat | Instruction         Instruction         Instruction         Shole Administration           Total Current         Total Current         Salaries Only         Salaries Only         Administration         School Administration           Potelling         Total Instruction         Salaries Only         Administration         School Administration           Rank         State         S         R         R         State         S         R         R         S | Instruction         State SDIP         Administration         State SDIP           Rank         State         S           Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         Rank         State         S         S         S         S         S <th>Instruction         Instruction           Total Current         Target on the state on the state</th> <th>Introduction         Introduction         <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th></th> | Instruction         Instruction           Total Current         Target on the state | Introduction         Introduction <th colspa<="" th=""><th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th><th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th><th>Introduct of the state of the stat</th><th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th><th>Instruction           Total Turner           Instruction           Specifie         Specifie</th><th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th><th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th></th> | <th>Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select</th> <th>Instruction         Instruction         Instruction         Instruction         Select of the state of the</th> <th>Introduct of the state of the stat</th> <th>Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim</th> <th>Instruction           Total Turner           Instruction           Specifie         Specifie</th> <th>Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing</th> <th>Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe</th> | Instruction         Instruction         Instruction         Selection         Selection         Selection           Total Lutrent         Salarise Only         Selection         Selection         Selection           Selection         Salarise Only         Selection         Select | Instruction         Instruction         Instruction         Instruction         Select of the state of the | Introduct of the state of the stat | Intruction         Intruction           Intraction         Intruction           Specific Time         Addition         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Time         Specific Time           Specific Tim | Instruction           Total Turner           Instruction           Specifie         Specifie | Instruction         Instruction           Instruction         Instruction           Instruction         Instruction           Spacing         Amount struction         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Amount struction           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacing         Spacing           Spacing         Spacing         Spacin         Spacing | Instruction         Instruction         Instruction         Instruction           Instruction         Instruction         Instruction         Instruction         Instruction           Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific S         Specific Specific Specific Specific S         Specific Spe |

Table 5.4Current Spending Per \$1,000 in Personal Income, FY 2006

Source: U.S. Dept. of Commerce. Census. Public2006.

Spending on Instruction and Other Functions

The next few pages describe how NCES and the Census Bureau break out current expenditures by function.

**Support Services.** This category includes a wide variety of activities that occur outside the classroom. The services include payments from all funds for salaries, employee benefits (paid by the school or the state), supplies, materials, and contractual services associated with the following activities:

- General Administration includes the board of education and executive administration (office of the superintendent) services.
- Instructional Staff Support includes supervision of instruction service improvements; curriculum development; instructional staff training; and media, library, audiovisual, television, and computer-assisted instruction services.
- Operation and Maintenance of Plant includes building services (heating, electricity, air conditioning, property insurance), care and upkeep of grounds and equipment, nonstudent transportation vehicle operation and maintenance, and security.
- Pupil Support Services include attendance record keeping, social work, student accounting, counseling, student appraisal, record maintenance, and placement services. This category also includes medical, dental, nursing, psychological, and speech services.
- Pupil Transportation Services include transportation of public school students including vehicle operation, rider monitoring, and vehicle servicing and maintenance.
- School Administration includes office of principal services.
- Other Support Services include business and central support and other support services. Business support services include payments for fiscal services, purchasing, warehousing, supply distribution, printing, publishing, and duplicating. Central support services include planning, research, development, and evaluation services. They also include information services, staff services (recruitment, staff accounting, noninstructional in-service training, and staff health), and data processing services.
- Nonspecified Support Services include expenditures that pertain to more than one of the above categories. In some cases, reporting units could not provide distinct expenditure amounts for each support services category. This expenditure is included in "nonspecified" instead of "other support services" (U.S. Dept. of Commerce. Census. *Public* A-5).

Support services include a variety of activities that occur outside the classroom, such as administration, training, library services, counseling, operations, and transportation. The spending category called all other functions or noninstruction includes food services, enterprise operations, community services, and adult education.

Instruction expenditures include teacher compensation, purchased services, tuition, and supply costs for the interaction between teachers and students.

Salaries and benefits make up 90 percent of instructional expenditures. All Other Functions. This category, which is also sometimes called noninstruction, includes all expenditures not related to instruction or support services, such as food services, enterprise operations, community services, and adult education (U.S. Dept. of Commerce. Census. *Public* 2006 A-3).

**Instruction**. Instruction expenditures include teacher salaries and benefits, purchased services, tuition payments, and supply costs incurred for year-round activities dealing directly with the interaction between teachers and students (U.S. Dept. of Ed. Inst. Natl. *Natl. Public* 2006 50-51).

Instructional activities may occur in a classroom, in another location such as a home or hospital, in other learning situations such as cocurricular activities, or through an approved medium such as television or correspondence between teachers and students. Teachers' coaching and supervising of cocurricular and extracurricular activities is considered instructional (U.S. Dept. of Ed. Inst. Natl. *Natl. Public* 2006 51, 54).

Salaries and benefits make up 90 percent of instructional expenditures for the U.S. as a whole and 94 percent in Kentucky (U.S. Dept. of Commerce. Census. *Public* 6). Instruction personnel include not only full-time classroom teachers but also part-time, substitute, and home- or hospital-based teachers; teachers on sabbatical leave; classroom assistants; clerks; and graders.

The distinction between instruction and noninstruction is sometimes blurred, with the same activities classified in different ways depending on who performs them and whether they generate self-sustaining revenues. For example, nonteachers are excluded even when engaged in duties that teachers also could perform, such as librarians who teach students about conducting research or guidance counselors who help students with job-readiness skills.

Controversy has arisen over the NCES definition of instruction as a result of a national movement dubbed the "65 Percent Solution." This plan calls for 65 percent of expenditures to be devoted to instruction. Proponents of the 65 Percent Solution were accused of "putting sports before education" because the NCES definition of instruction includes coaches and extracurricular activities but not librarians, library expenses, guidance counselors, or professional development (Elliott). In response, Texas changed its definition to include librarians (State of Texas). In July 2006, NCES introduced a new "instruction and instructionrelated" category that includes librarians. NCES said that this and the other new categories "provide a clearer picture of how education dollars are spent" (U.S. Dept. of Ed. Inst. Natl. *Current Expenditures*).

Under the new instruction and instruction-related definition, 30 states spent 65 percent or more on instruction in FY 2004, compared to just two states under the old definition in FY 2003. Although subsequent NCES publications have used the new categories, the Census Bureau continues to break out spending using the old definition (U.S. Dept. of Ed. Inst. Natl. *Overview* and *Revenues*; U.S. Dept. of Commerce. Census. *Public* 2006). This compendium breaks out current spending by the old categories in Table 5.5 and the new categories in Table 5.6.

Under the older definition still used by the Census Bureau, Kentucky spent 58.5 percent on instruction in FY 2006, which is below the national average and gives Kentucky a rank of 37<sup>th</sup>. Table 5.5 shows the distribution of current spending by the older categories of instruction, support services, and all other functions. By these definitions, only New York spent 65 percent or more on instruction in FY 2006. Kentucky spent 58.5 percent on instruction, which was below the national average of 60.2 percent and which gives Kentucky a rank of 37<sup>th</sup>.

	Instruction		Su	pport Servio	es	All Other					
Rank	State	%	Rank	State	%	Rank	State	%			
3	TN	64.0	5	OH	38.4	1	OK	10.7			
8	GA	62.5	8	IL	37.4	4	AL	8.4			
11	NC	61.5	11	IN	36.5	7	WV	7.0			
14	VA	60.9	13	OK	36.4		FL				
16	DE	60.6	14	SC	36.0	8	MS	6.8			
	U.S.	60.2	15	LA	35.9		KY				
20	MD	60.2	19	FL	35.4	11	SC	6.4			
22	AR	60.0	23	MD	35.0	10	LA	6.2			
23	TX	59.8	26	KY	347	12	MO	0.2			
28	MO	50.2	20	MS	54.7	14	NC	6.0			
20	WV	59.2	28	DE	34.6	14	TN	0.0			
33	IN	58.9	20	MO	54.0	18	TX	5.8			
25	MS	58.5		U.S.	34.6	20	AR	5.6			
55	KY	56.5		VA			U.S.	5.2			
38	IL	58.4	31	TX	34.4	23	GA	5.2			
39	LA	57.9		AR		28	MD	4.9			
40	FL	57.8	35	AL	34.2	21	OH	18			
42	SC	57.6	36	WV	33.9	51	DE	4.0			
43	AL	57.4	42	NC	32.5	34	IN	17			
46	OH	56.8	43	GA	32.3	34	VA	4./			
50	OK	52.9	47	TN	30.1	39	IL	4.2			

Table 5.5Spending on Instruction, Support Services, and Other FunctionsAs Percentages of Current Spending, FY 2006

Source: U.S. Dept. of Commerce. Census. Public 2006.

Under the newer definition used by NCES, Kentucky spent 65.1 percent on instruction and instruction-related expenditures in FY 2006, which was close to the national average and ranked Kentucky 22<sup>nd</sup>. Table 5.6 shows the distribution of current spending by the newer categories of instruction and instruction-related (including librarians), student support services, administration, and operations. The number of states spending 65 percent or more on instruction is 24 by this definition, in contrast to just one by the older definition shown in Table 5.5. Kentucky spent 65.1 percent on instruction and instruction-related activities, which was close to the national average and ranked Kentucky 22<sup>nd</sup>.

Ins Instru	struction uction-re	elated	Stu	dent Su Service	pport es	Ad	lministra	ation	Operations			
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	
2	TN	69.9	8	SC	6.9	4	DE	13.2	1	WV	23.1	
6	GA	68.0	10	OK	6.5	5	OH	13.1	2	LA	22.4	
8	VA	67.7	12	IL	6.3	11	IL	11.9	4	OK	21.6	
12	AR	(( (	15	OH	6.0	13	IN	11.7	5	MS	21.2	
15	MD	00.0	24	NC	5.4	19	OK	11.2	6	KY	20.9	
	U.S.	65.9		U.S.	5.2	20	NC	11.1	8	IN	20.7	
17	FL	(5.0	26	AL	5.1	25	AL	10.9	9	AL	20.6	
1 /	NC	03.9	27	TX	4.9		U.S.	10.8	10	DE	20.4	
20	MO	65.2		DE		26	AR	10.8	14	MO	19.6	
22	KY	65.1	29	GA	4.8	28	TX	10.6	11	FL	20.1	
23	TX	65.0		VA			GA		15	TX	19.5	
25	SC	64.9	22	FL	17	30	MD	10.5	21	MD	10.7	
31	MS	63.9	32	MO	4./		MO		21	VA	18.7	
34	WV	63.8	34	AR	4.6	34	MS	10.4	27	IL	10.2	
35	OH	63.7	37	MS	4.5	36	LA	10.2	21	SC	16.2	
37	IL	63.5	38	IN	4.4	37	KY	10.0		U.S.	18.1	
38	AL	63.4	40	MD	4.2	38	SC	9.9	31	AR	18.1	
39	LA	63.3	40	KY	4.1	43	WV	9.5	26	NC	17.6	
41	IN	63.2	42	LA	4.1	45	FL	9.3	30	TN	1/.0	
46	DE	61.7	48	WV	3.6	47	TN	9.1	39	OH	17.2	
50	OK	60.7	49	TN	3.4	48	VA	8.9	41	GA	16.7	

Table 5.6Spending on Instruction and Instruction-related, Student Support Services,Administration, and Operations Functions as Percentages of Current Spending, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Revenues 13.

In FY 2005, the average teacher salary in Kentucky was \$42,592, and Kentucky ranked 33<sup>rd</sup> in the nation. Adjusting salaries for state cost differences brings Kentucky up to a rank of 31<sup>st</sup>. Both the adjusted and unadjusted salaries are below the national average.

## **Teacher Salaries**

Comparing salaries across states can be misleading because states with higher costs usually offer higher salaries. In Table 5.7, states are ranked by unadjusted teacher salaries and then by salaries adjusted for cost differences, using the NCES Comparable Wage Index discussed in Chapter 3.<sup>10</sup> When salaries are adjusted, some states' rankings change more than others. Going from unadjusted to adjusted salaries, Virginia's rank drops from 28<sup>th</sup> to 51<sup>st</sup>. However, Kentucky's rank changes only slightly, from 33<sup>rd</sup> to 31<sup>st</sup>. Both the adjusted and unadjusted salaries for Kentucky are below the national average. However, as discussed earlier, Kentucky ranked 16<sup>th</sup> in terms of spending on salaries per \$1,000 in personal income. Whereas the CWI adjustment takes into account only the

<sup>&</sup>lt;sup>10</sup> Cost-adjusted salaries were calculated by dividing each state's average teacher salary by its CWI and then multiplying by the national CWI.

wages of college-educated professionals, personal income is from the full range of workers.

Table 5.7
Average Public School Teacher Salaries, Unadjusted and Adjusted
for Geographic Cost Differences, FY 2006

	Unadjusted		Adjusted for Geographic Cost Differences			
Rank	State	Dollars	Rank	State	Dollars	
4	Illinois	58,686	1	Illinois	56,861	
10	Maryland	54,333	12	Delaware	53,361	
11	Delaware	54,264	13	Indiana	53,211	
14	Ohio	50,314	14	Ohio	52,540	
	United States	49,026	16	Arkansas	51,966	
18	Georgia	48,300	21	Maryland	50,253	
19	Indiana	47,255	23	Georgia	49,198	
27	North Carolina	43,922		United States	49,026	
28	Virginia	43,823	27	Mississippi	48,857	
29	Florida	43,302	31	Kentucky	48,222	
31	South Carolina	43,011	34	South Carolina	47,723	
32	Arkansas	42,768	37	Florida	46,774	
33	Kentucky	42,592	38	North Carolina	46,538	
34	Tennessee	42,537	40	Tennessee	46,284	
35	Texas	41,744	41	Louisiana	46,172	
41	Mississippi	40,576	42	Oklahoma	46,118	
42	Missouri	40,462	43	Alabama	46,056	
44	Alabama	40,347	45	West Virginia	45,223	
45	Louisiana	40,029	46	Missouri	44,738	
48	Oklahoma	38,772	50	Texas	42,198	
49	West Virginia	38,284	51	Virginia	40,488	

Note: Staff calculated adjusted salaries using NCES Comparable Wage Index.

Source: Natl. Ed. Assoc. *Rankings & Estimates 2006-2007*, Rankings, Table C-11. Data used with permission of the National Education Association © 2007. All rights reserved.

Figures 5.C and 5.D show year-to-year changes in teacher salaries for Kentucky and the nation as a whole, from 1970 to 2005. In nominal dollars, salary increases appear large, especially for the period between 1970 and 2000. However, adjusting for inflation reveals modest increases from 1970 to 2000, followed by increases that just keep up with inflation.

Chapter 5



Figure 5.C Teacher Salary Trends, U.S. and Kentucky, 1970-2005 Nominal Dollars

Source: U.S. Dept. of Ed. Inst. Natl. Digest 110.



Source: U.S. Dept. of Ed. Inst. Natl. Digest 110 and "NCES Comparable."

## **Funding Equity**

Funding equity between poorer and wealthier districts has occupied policy makers' attention for four decades, as court cases have challenged the constitutionality of education systems that rely on local property taxes. Since the passage of the Kentucky Education Reform Act of 1990, Kentucky's SEEK formula has provided state funds to reduce funding disparities among districts.

There is no consensus as to which equity indicator is best. Some are easier to understand than others. Few distinguish undesirable inequity (wealthier districts spending more) from that which is intentional and policy driven (poorer districts spending more, sometimes called "progressive inequity").

## Education Week's Equity/Disparity Measures

Table 5.8 presents equity measures published by *Education Week's Quality Counts*. Kentucky is better than the national average on three of the four measures. On two of the four measures, Kentucky is among the top three states. These favorable ratings could be seen as indicators of the success of Kentucky's SEEK funding mechanism. Brief descriptions of the equity measures follow the table.

Kentucky's ranks highly on several funding equity measures, which may indicate the success of Kentucky's Support Education Excellence in Kentucky funding mechanism.

Wealth-neutrality Score			McLoone Index			Coeffic	ient of V	ariation	Restricted Range			
Relationship Between District Funding and Local Property Wealth (Lower Value is Better)			Actual Spending as Percentage of Amount Needed To Bring All Districts to Median Level (Higher Value is Better)			Amoun Spe District Gre	nt of Disp nding Ac s (Lower eater Equ	arity in ross Value = iity)	Difference in Per-pupil Spending Levels at the 95 <sup>th</sup> and 5 <sup>th</sup> Percentiles (Lower Value is Better)			
Rank	State	Score	Rank	State	Index	Rank	State	CV	Rank	State	\$	
12	OK	0.016	5	NC	95.6	1	WV	0.058	1	WV	1,626	
13	IN	0.017	7	LA	95.3	2	FL	0.074	2	KY	1,920	
14	AR	0.033	9	MD	95.2	3	KY	0.091	3	AL	1,980	
16	LA	0.045	12	FL	94.4	4	DE	0.002	4	TN	2,092	
19	KY	0.070	13	WV	94.3	4	AL	0.092	5	AR	2,355	
20	WV	0.075	14	GA	94.1	6	LA	0.094	6	MS	2,396	
21	MO	0.077	16	AL	94.0	7	TN	0.095	8	OH	2,644	
	U.S.	0.085	10	AR	02.0	9	NC	0.098	9	FL	2,837	
28	TN	0.093	10	TX	95.9	11	GA	0.099	11	SC	3,060	
29	OH	0.094	20	SC	93.7	12	AR	0.103	12	NC	3,090	
21	GA	0 127	24	TN	93.3	13	MD	0.105	13	LA	3,335	
51	MS	0.127	26	VA	92.8	14	SC	0.112	15	DE	3,521	
36	FL	0.148		U.S.	92.6	14	MS	0.112	16	GA	3,530	
38	ΤХ	0.156	29	MS	92.5	23	IL	0.135	18	MO	3,659	
39	DE	0.159		DE		26	VA	0.137	19	MD	3,696	
43	IL	0.170	33	OH	92.1	29	TX	0.143	20	IN	3,785	
44	AL	0.171		KY		30	OH	0.144	23	OK	4,062	
45	NC	0.201	39	OK	90.9	31	IN	0.145	26	VA	4,163	
46	SC	0.212	40	IL	90.7		U.S.	0.147		U.S.	4,725	
48	MD	0.283	42	MO	89.7	34	MO	0.162	31	IL	4,743	
49	VA	0.288	44	IN	89.5	38	OK	0.180	32	TX	4,756	

Table 5.8Education Week's Equity/Disparity Measures, FY 2005

Note: Rankings are out of 49 because DC and HI have only one district. Source: Editorial Projects in Education. *Education Week's Quality Counts*. **Wealth-neutrality Score.** This equity measure examines the relationship between districts' property wealth and revenues per pupil. If these two measures have a high positive correlation, funding is inequitable. A negative value indicates that more funding is targeted at poorer districts. Of the four equity measures used by *Education Week*, this appears to be the only one that rewards, instead of penalizing, states for this progressive inequity. As Table 5.8 shows, this index ranks Kentucky 19<sup>th</sup> with respect to equity (Editorial. *Education Week's Quality*).

**McLoone Index**. This equity measure focuses on districts that spend less than the state median, calculating the amount that would be required were they to spend at the median (Costrell). By focusing on only those districts spending below the median, it is less likely than other meaures to be distorted if some districts spend considerably more than what might be considered necessary to provide adequate educational services (Hussar 17). The maximum value for this measure is 100 percent, indicating the same level of spending for all districts. Kentucky's index of 92.1 percent means Kentucky would need to spend 7.9 percent more to bring all districts up to the median. In Table 5.8, Kentucky ranks 33<sup>rd</sup> on this index.

**Coefficient of Variation.** This is a measure of the amount of variation in spending among districts. Kentucky ranks 3<sup>rd</sup> with respect to this measure. This measure has the advantage of being simple to calculate and understand—it is the standard deviation divided by the mean—and was chosen by Congress as an equity factor for Title I legislation (Hussar 13). On the negative side, it has the potential to penalize states if they spend more in poorer districts. However, *Education Week* did not find Kentucky to be spending more for poorer districts (Editorial. *Education Week's Quality*). As will be discussed later in this chapter, Education Trust did find that Kentucky spends more for poorer districts, after making certain adjustments to improve comparability.

**Restricted Range**. This measures the difference in per-pupil spending between the 95<sup>th</sup> percentile and the 5<sup>th</sup> percentile. The smaller this range, the better (Editorial. *Education Week's Quality*). Kentucky ranks  $2^{nd}$  on this measure.

Education Trust uses per-student revenues to calculate funding gaps between highest- and lowest-poverty districts and highest- and lowest-minority districts. Revenues are adjusted for geographic variations in costs and the number of special education students enrolled.

Kentucky is reported to be relatively good at ensuring equity for students in poverty and for minority students. Instead of a funding gap, Kentucky's highestpoverty districts and highestminority districts receive more money per student than wealthier and less diverse districts.

## **Education Trust's Poverty and Minority Gap Measures**

Education Trust examines differences in state and local revenues per pupil between districts with relatively high numbers of impoverished or minority students and districts with relatively few impoverished or minority students. To classify districts by poverty, Education Trust ranked them by the percentage of students living below the poverty level and then divided them into quartiles. The top 25 percent were the highest-poverty districts and the bottom 25 percent were the lowest-poverty districts. Highest- and lowestminority districts were identified in a similar way.

The poverty gap is equal to revenues received by the highestpoverty districts minus revenues received by the lowest-poverty districts. Gaps are adjusted for geographic cost differences and the number of special education students enrolled because higher costs are associated with these students. For some tables, Education Trust also made adjustments that assumed 40 percent greater costs for low-income students. This is more than the 20 percent adjustment that *Education Week* uses.

As Education Trust interprets its analysis, positive numbers indicate greater funding of districts that need it most, while negative numbers indicate less funding for these neediest districts. The minority gap is calculated in a similar way; positive numbers indicate greater funding of districts with the highest percentages of minorities while negative numbers indicate less funding in highminority districts.

As Table 5.9 shows, Kentucky ranks 7<sup>th</sup> and 12<sup>th</sup>, respectively, for funding to highest-poverty and highest-minority districts relative to the lowest-poverty and lowest-minority districts. The positive numbers indicate that more funds go to Kentucky's needier districts.<sup>11</sup> In contrast, Illinois had large funding gaps, with the highest-poverty districts receiving an average of \$2,827 less per student than the lowest-poverty districts and the highest-minority districts.

<sup>&</sup>lt;sup>11</sup> *Education Week's* equity measures did not indicate that Kentucky spends more for poorer districts than for wealthier districts. This discrepancy may be due to the fact that *Education Week* uses only a 20 percent adjustment for low-income students, while the Education Trust uses a 40 percent adjustment.

Gap, Low-pove	erty Minus High-p	overty Districts	Gap, Low-minority Minus High-minority Districts				
Rank*	State	Gap (\$)	Rank*	State	Gap (\$)		
7	KY	462	4	OH	1,032		
9	AR	230	6	AR	547		
11	TN	155	7	MO	535		
12	OH	73	8	IN	428		
13	SC	-19	10	WV	238		
14	MD	-23	11	LA	229		
16	OK	-72	12	KY	152		
19	FL	-199	13	GA	134		
22	IN	-238	14	SC	81		
23	MS	-267	15	TN	-81		
26	WV	-383	18	FL	-133		
27	GA	-436	20	MS	-166		
30	VA	-526	22	VA	-250		
31	LA	-560	23	OK	-294		
35	AL	-692	25	AL	-500		
38	TX	-796	31	NC	-738		
40	MO	-803	32	MD	-803		
42	NC	-825	35	DE	-933		
44	DE	-1,126		U.S.	-1,275		
	U.S.	-1,532	42	TX	-1,385		
48	IL	-2,827	46	IL	-2,021		

Table 5.9Education Trust Poverty and Minority Funding Equity and Gaps, 2005

Notes: \*Rank out of 49; DC and HI each has only one district. A negative dollar amount indicates that fewer dollars were provided to high-poverty or high-minority districts. A positive amount indicates more dollars for high-poverty or high-minority districts. Education Trust made a 40 percent adjustment for low-income students. Source: Education Trust. *The Funding Gap* 9.

In both 1999 and 2005, Kentucky was a national leader in providing extra funding for high-poverty districts.

As Table 5.10 shows, Kentucky was a national leader in providing extra funding for high-poverty districts in both 1999 and 2005. Between 1999 and 2005, Kentucky's extra funding for the highest-poverty districts increased by \$77 per pupil, the 18<sup>th</sup> largest increase.

1999 Ga Hig	ap, Low-pover h-poverty Dis	ty Minus tricts	2005 ( Hi	Gap, Low-povert igh-poverty Dist	ty Minus ricts	1999-2005 Poverty Gap, Change in Dollars			
Rank*	State	Gap (\$)	Rank*	State	Gap (\$)	Rank*	State	Change (\$)	
4	KY	801	6	KY	878	3	MD	1,376	
6	TN	729	7	OH	833	4	OH	910	
12	MO	480	11	AR	541	6	AR	523	
16	FL	350	13	TN	454	11	IN	197	
17	NC	337	14	MD	395	12	LA	180	
18	OK	312	15	IN	322	15	SC	136	
19	TX	280	16	SC	302	17	DE	98	
20	GA	245	17	OK	271	18	KY	77	
22	VA	234	21	MS	151	25	AL	-19	
24	MS	192	24	GA	82	28	WV	-40	
26	SC	166	26	FL	-18	29	MS	-41	
27	IN	126	27	WV	-19	30	OK	-42	
29	WV	22	28	MO	-104		U.S.	-90	
30	AR	18	29	VA	-122	33	GA	-162	
36	ОН	-77	32	ТХ	-165	34	TN	-275	
40	AL	-309	35	LA	-241	37	VA	-356	
41	LA	-421	38	AL	-328	38	FL	-368	
	U.S.	-848	42	NC	-603	40	TX	-445	
45	MD	-981		U.S.	-938	43	MO	-584	
46	DE	-1,052	45	DE	-954	46	IL	-668	
48	IL	-1,568	48	IL	-2,235	48	NC	-939	

Table 5.10Poverty Funding Gap Trends, 1999 to 2005

Notes: \*Rank out of 49; DC and HI each has only one district. A negative dollar amount indicates that fewer dollars were provided to high-poverty districts. A positive amount indicates more dollars for high-poverty districts. Education Trust made no adjustment to these numbers for low-income students. Source: Education Trust. *The Funding Gap* 6.

Between 1999 and 2005, Kentucky eliminated a minority funding gap. High-minority districts now receive more funding than low-minority districts. As Table 5.11 shows, Kentucky eliminated a minority funding gap between 1999 and 2005. In 1999, Kentucky's highest-minority districts had \$162 per student less than the lowest-minority districts. By 2005, there was a "progressive gap," with the highestminority districts receiving \$44 more per pupil.

1999 Gaj High	p, Low-minori -minority Dis	ty Minus tricts	2005 Gap High-	, Low-mino minority Di	rity Minus stricts	1999-2005 Minority Gap, Change in Dollars			
Rank*	State	Gap (\$)	Rank*	State	Gap (\$)	Rank*	State	Change (\$)	
3	MO	1,446	4	OH	1,520	3	OH	800	
5	GA	957	6	MO	788	4	LA	578	
7	IN	766	7	IN	721	7	SC	445	
8	OH	720	8	AR	707	9	AR	400	
9	VA	521	9	GA	496	12	KY	206	
12	AR	308	12	SC	297	15	WV	109	
13	MS	215	13	LA	293		U.S.	73	
14	WV	91	14	WV	200	17	AL	67	
18	FL	31	15	MS	163	19	TN	51	
19	OK	17	20	KY	44	23	IN	-45	
22	TN	-41	21	TN	10	24	FL	-46	
25	NC	-75	23	FL	-15	25	MS	-52	
28	SC	-148	24	OK	-56	28	OK	-73	
29	KY	-162	25	VA	-57	29	MD	-157	
30	LA	-285	29	AL	-280	33	TX	-327	
31	AL	-347	31	MD	-578	36	GA	-461	
32	DE	-385	33	NC	-663	37	DE	-507	
34	MD	-421		U.S.	-877	40	VA	-579	
39	TX	-586	39	DE	-892	41	NC	-587	
42	IL	-864	40	TX	-912	42	MO	-658	
	U.S.	-950	45	IL	-1,623	45	IL	-758	

Table 5.11Minority Funding Gap Trends, 1999 to 2005

Notes: \*Rank out of 49; DC and HI each has only one district. A negative dollar amount indicates that fewer dollars were provided to high-minority districts. A positive amount indicates more dollars for high-minority districts. Education Trust made no adjustment to these numbers for low-income students. Source: Education Trust. *The Funding Gap* 6.

## **Education Week's Overall School Finance Index**

*Education Week*'s 2008 issue of *Quality Counts* included an overall school finance index, based on the four equity measures discussed earlier in this chapter and four spending measures, shown in Table 5.12. One weakness of this index is that some of the equity measures fail to distinguish "progressive" inequity and therefore penalize states like Kentucky that provide poorer districts with more funding than wealthier districts.

# Table 5.12Components of *Education Week's* School Finance Index, 2008

			KY
Indicator	U.S.	KY	Rank
Equity			
Wealth-neutrality Score, 2005 (lower value is better; negative value indicates higher funding for poorer districts)*	0.085	0.070	19
McLoone Index, 2005 (actual spending as a percentage of amount needed to bring all students to median level)*	92.6	92.1	33
Coefficient of Variation, 2005 (amount of disparity in spending across districts, with lower value indicating greater equity)*	0.147	0.091	3
Restricted Range, 2005 (difference in per-pupil spending levels at the 95 <sup>th</sup> and 5 <sup>th</sup> percentiles)*	\$4,725	\$1,920	2
Spending			
Per-pupil Expenditures (PPE), adjusted for regional cost differences, 2005	\$8,973	\$7,978	37
Percentage of Students in Districts in Which PPE is At or Above U.S. Average, 2005*	46.1%	9.0%	41
Spending Index, 2005 (per-pupil spending levels weighted by the degree to which districts meet or approach the national average for expenditures, cost and student-need adjusted)*	92.7	91.0	35
Percentage of Total Taxable Resources Spent on Education, 2005	3.6%	3.4%	32
School Finance Index Total Score	77.6	74.0	29

Note: \*Adjusted for regional cost differences and special-needs students.

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

Education Week's 2008 issue of Quality Counts gave Kentucky a ranking of 29<sup>th</sup> and a grade of C on the overall school finance index. Table 5.13, which presents state rankings based on the overall school finance index, shows that Kentucky ranks 29<sup>th</sup> and has a grade of C. Even if the equity measures in this index did not penalize states for extra funding for high-poverty schools, Kentucky's rank and grade would not improve much because per-pupil funding is relatively low even after adjusting for geographic cost differences.

Rank	State	Total Score	<b>Overall Grade</b>
1	WV	92.5	A
10	MD	84.9	
11	IN	84.3	В
12	DE	83.5	
13	OH	82.0	D
16	AR	81.1	D-
21	GA	78.7	
24	VA	77.6	$C^{\perp}$
	U.S.	77.6	
25	LA	77.2	
26	SC	76.1	
29	KY	74.0	
31	AL	73.8	С
32	IL	73.2	
33	MO	72.9	
34	MS	72.3	
38	FL	70.6	
39	NC	70.3	C-
40	TX	69.9	
41	TN	69.7	
42	OK	69.5	D+

# Table 5.13Education Week's Overall School Finance Index, 2008

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

Kentucky spends less on education than other states, even after adjusting for cost differences. However, the below-average funding is allocated more equitably than in most other states.

## **Chapter Summary and Conclusions**

Even after adjusting for Kentucky's lower cost of living, Kentucky's teacher salaries and per-pupil revenues and expenditures are below the national average. The exception is perpupil preschool spending which, due to a \$25 million budget increase, spiked above the national average in FY 2007. Notwithstanding Kentucky's below-average education funding, the Commonwealth is a national leader in allocating those funds equitably among districts.

## Chapter 6

## **Indices Composed of Multiple Topics**

Indices composed of multiple topics are popular overall measures, but due to different measures and methodologies they lead to widely different conclusions about Kentucky's position in the nation. This chapter discusses indices that combine several measures, including some topics discussed earlier in this compendium as well as noneducation topics. Many organizations grade and rank states based on broad indices that combine data across a spectrum of topics including education, economic conditions, crime, health, and other quality-of-life measures. Instead of focusing on education in a vacuum, these indices provide a multidisciplinary frame of reference.

However, multi-topic indices should be used with caution. They vary widely, depending on the measures chosen and the methodology used for combining them. The same state can be graded an A by one organization but an F by another (Olson. "An 'A'"). In addition, organizations that produce indices on a broad array of topics may not have the specialized expertise to choose the best indicators for each specific topic, such as education. Some measures may be redundant, giving certain factors too much weight, or important measures may be overlooked.

## **Quality Counts Overall Grade and Score**

The 2008 edition of *Education Week*'s *Quality Counts* assigned overall grades to states by averaging the scores for six categories of measures, with each category given equal weight in the overall grade. The categories were

- Chance-for-Success-Index, made up of 13 measures
- K-12 Achievement Index, calculated from 18 measures
- Standards, assessment, and accountability, based on 25 policies
- Transitions and alignment, based on 14 policies relating to early childhood education, college readiness, and the economy and the workforce
- Teaching profession, reflecting 50 policies related to accountability, incentives and allocation, and building and supporting capacity
- School finance, derived from 8 measures (Editorial. *Education Week*'s *Quality*).

*Education Week's Quality Counts* 2008 gave Kentucky, overall, a C grade and a score that is just above the national score. As Table 6.1 shows, Kentucky received a C grade overall and a total score that is just above the national score. Many peer states received higher grades than Kentucky, although differences among states are small. Kentucky's B+ grade for standards, assessment, and accountability reflects the implementation of 18 of the 25 policies recommended by *Education Week*. Kentucky's implementation of half of the 14 recommended transitions and alignment policies earned a grade of C. Having put in place 28 of the 50 recommended policies regarding the teaching profession, Kentucky received a grade of B-. The Chance-for-Success-Index, K-12 Achievement Index, and School Finance Index are discussed in detail in this compendium, in chapters 3, 2, and 5, respectively.

 Table 6.1

 Education Week's Quality Counts Overall Grade

		Over-	<b>T</b> ( )	Chance	K-12	Standards,	Transitions		
Rank	State	all Grade	Total Score	for Success	Achieve- ment	Assessment, and Accountability	and Alignment	Teaching Profession	School Finance
3	MD	В	83.5	B+	В	В	B+	С-	В
5	VA	B-	82.3	В	С	А	C+	B-	C+
6	WV	B-	80.8	C-	F	А	B+	C+	Α
7	OH	B-	80.6	B-	C-	А	C+	C+	B-
8	AR	B-	80.5	C-	D	B+	В	B+	B-
9	SC	B-	80.4	С	D	А	B-	A-	С
11	GA	B-	80.2	С	D+	A-	В	В	C+
12	IN	B-	79.7	C+	C-	А	C+	C-	В
14	FL	C+	79.2	C+	С	A-	C+	В	C-
16	TN	C+	78.0	C-	D+	A-	Α	С	C-
10	DE	C+	/8.0	В	С-	B+	D	C+	В
21	LA	С	76.5	D+	D-	А	С	В	C+
22	KY	С	76.4	C	D+	B+	С	B-	C
23	TX	С	75.9	С	С	B+	C+	С	C-
	U.S.	С	75.9	C+	D+	В	C	С	C+
24	NC	С	75.8	C+	D+	B+	D+	В	C-
26	AL	С	75.7	C-	F	A-	C+	В-	С
28	OK	С	75.5	C-	D	A-	С	В-	D+
35	IL	С	73.4	B-	D+	C+	C-	D+	С
40	MO	C-	72.1	C+	D	С	D+	С	С
48	MS	D+	68.9	D+	F	В	D+	D	C-

Notes: The total score is the average of the scores across the six individual categories; each category receives equal weight in the overall grade. Because DC and HI are single-district jurisdictions, there is no measure of financial equity among districts; therefore, neither DC nor HI has a grade for school finance. Source: Editorial Projects in Education. *Education Week*'s *Quality Counts* 2008.

#### Kentucky Long-Term Policy Research Center Indices

The Kentucky Long-Term Policy Research Center (KLTPRC) tracks Kentucky's progress relative to other states using several indices, some of which focus on education. The center's indices of obstacles to cost-effective spending, NAEP proficiency purchasing power, and state of the Commonwealth are discussed below.

## **Obstacles to Cost-effective Educational Spending Index**

KLTPRC created this index using eight measures, summarized in Table 6.2, that are correlated with academic achievement. As mentioned in Chapter 3, most Kentucky students are proficient in English, but they struggle with the other seven obstacles. In particular, Kentucky has the highest student obesity rate in the nation and the 9<sup>th</sup> highest rates of lunch subsidies and rural school settings. Consequently, Kentucky is ranked 4<sup>th</sup> with respect to the Obstacles to Cost-Effective Educational Spending Index (Commonwealth. Legislative. Kentucky Long-Term. *Reducing*).

According to the Kentucky Long-Term Policy Research Center, Kentucky children face the 4<sup>th</sup> highest level of obstacles.

Table 6.2
<b>Components of the Obstacles to Cost-effective Educational Spending Index, 2008</b>

Student Characteristic	KY	U.S.	KY Rank
No parent with postsecondary degree, 2006 <sup>a</sup>	61.0%	56.7%	13
Eligible for free or reduced-price lunch, FY 2006	49.5%	41.4%	9
Limited English proficiency, FY 2006	1.6%	8.6%	45
Attends rural school, FY 2005	42.3%	21.4%	9
Obese or overweight, 2003	38.0%	31.0%	1
Fair or poor health (as reported by parent), 2003	3.1%	3.2%	17
Disability, FY 2005	15.8%	13.8%	11
Missed 11 or more school days, 2003	6.1%	5.2%	17
Index Value	0.71686	0.50000	4

Note: <sup>a</sup>This table presents the reciprocal of the published parent education measure in order to make all measures consistent (all negatively correlated with achievement).

Source: Staff compilation of data from Commonwealth. Legislative. Kentucky Long-Term. Reducing.

# Index of NAEP Proficiency Purchasing Power Relative to Obstacles

After adjusting for obstacles, the Kentucky's return on investment in education was deemed to be 18 percent above what would be expected. KLTPRC assesses states' returns on investments in education by dividing the average NAEP proficiency rate by per-pupil spending. On this measure, as Table 6.3 shows, Kentucky is just below the national average and is ranked 25<sup>th</sup>. However, when this measure is examined relative to levels of obstacles in each state, Kentucky rises to a rank of 8<sup>th</sup>. Using a statistical analysis, KLTPRC estimates that Kentucky's return on investment in education is 18

percent above what would be expected, given the obstacles the Commonwealth faces.

Table 6.3Index of NAEP Proficiency Purchasing Power Relative to Obstacles, 2008

Per-p Adjus Livii	upil Sp ted for 1g Diffe FY 200	ending, Cost-of- erences )5	NAEP in Gr Readi 200	Profic ades 4 ing & 1 5 & 20	ficiency NAE 4 & 8 Purch 2 Math (% p 2007 \$1,00		<b>EP Proficiency</b> chasing Power proficient per 000 spending)		Obstacles to Cost-Effective Educational Spending Index			NAEP Proficiency Purchasing Power Relative to Obstacles		
Rank	State	\$	Rank	State	%	Rank	State	%	Rank	State	Value	Rank	State	%
5	DE	10,661	11	OH	37.4	9	NC	4.33	1	WV	0.81392	4	TX	125.3
10	WV	10,073	14	VA	37.0	12	TX	4.19	3	MS	0.75334	5	NC	121.5
14	MD	9,829	22	MD	34.4	14	FL	4.10	4	KY	0.71686	8	KY	118.2
15	IN	9,542	24	IN	34.1	16	VA	4.04	5	AR	0.70718	10	FL	116.1
16	OH	9,441	25	DE	33.2	18	OH	3.96	6	AL	0.68972	11	OK	112.3
20	VA	9,169		U.S.	32.9	21	MO	3.80	7	OK	0.68328	12	SC	110.2
26	AR	8,790	28	NC	32.6		U.S.	3.78	8	LA	0.67382	17	TN	107.3
	U.S.	8,701	29	ΤХ	32.2	23	IL	3.62	9	TN	0.66423	19	IN	105.7
28	GA	8,658	31	MO	31.5	25	KY	3.59	10	SC	0.66394	21	AR	105.1
29	IL	8,621	32	IL	31.2	26	IN	3.57	12	ΤX	0.60727		U.S.	103.6
30	LA	8,582	33	FL	30.9	29	OK	3.52	14	IN	0.59493	23	OH	103.2
34	SC	8,339	35	SC	29.3	30	SC	3.51	17	FL	0.54216	24	MO	103.1
35	MO	8,276	37	KY	28.7	31	MD	3.50	19	NC	0.53064	25	IL	100.8
36	KY	7,978	38	AR	28.3	34	TN	3.42	20	IL	0.52025	26	VA	99.7
38	AL	7,924	39	GA	26.8	40	AR	3.22	21	GA	0.50806	36	AL	88.3
39	ΤX	7,687	40	OK	25.8	41	DE	3.12		U.S.	0.50000	38	WV	86.3
40	FL	7,539	41	TN	25.7	42	GA	3.09	24	MO	0.48487	40	GA	85.3
41	NC	7,525	43	WV	24.0	44	AL	2.75	27	DE	0.46491	42	DE	83.2
42	MS	7,513	47	AL	21.8	47	WV	2.39	29	OH	0.43336	45	MD	80.1
43	TN	7,506	48	LA	20.4	48	LA	2.38	39	VA	0.3575	46	MS	79.6
45	OK	7,331	50	MS	17.6	49	MS	2.34	49	MD	0.24172	48	LA	75.4

Notes: KLTPRC assigned a value of 0.50000 to the average Obstacles to Cost-Effective Educational Spending Index of all states. Staff used spending and proficiency data from NCES to calculate purchasing power for the U.S. Source: Staff compilation using data from Commonwealth. Legislative. Kentucky Long-Term. *Reducing*; U.S. Dept. of Ed. Inst. Natl. *NAEP Data* and *Common*.

State of the Commonwealth Index

Kentucky is among the 20 best in terms of crime, homeownership, air and water quality, achievement gaps, and government efficiency. However, Kentucky is in the bottom 20 for such factors as smoking, obesity, college degrees, and women legislators. The 32 "quality of life" measures that make up the State of the Commonwealth Index are summarized in Table 6.4. Kentucky is among the 20 best states in terms of crime, homeownership, achievement gaps, air and water quality, and government efficiency. However, the Commonwealth is among the 20 worst for multiple factors, with higher smoking and obesity rates, lower employment of persons with disabilities, and fewer women legislators.

Subindex or Indicator Name	Rank	Among 5	0 States	Rank Ar	nong 17 P	eer States <sup>a</sup>
			1990-2003			1990-2003
	2003	1990	Change in	2003	1990	Change in
Communities (209/)	Rank	Rank	Rank	Rank	Rank 10	
Communities (20%)	39	39	0	2	10	-1
1. Crime index	11	4	-/		2	0
2. Employment of Persons with Disabilities	4/	40	-/	14	9	-5
3. Homeownership Kate	8	31	+23	6	13	+/
4. Health Insurance Rate	25	26	+1	/	6	- l
5. Teen Birth Kate	50	30	0	9	8	-l
6. Smoking Rate	50	49	-l	1/	16	-1
7. Obesity	46	36	-10	14	8	-6
8. Charitable Contributions	3/	32	-5	14	13	-1
Education (20%)	41	44	+3	9	12	+3
9. High School Attainment Rate	35	4/	+12	9	14	+5
10. College Attainment Rate	44	45	+1	14	12	-2
11. ACT Average Composite Score	40	43	+3	10	11	+1
12. NAEP Grade 8 Math Results	35	43	+8	9	12	+3
13. Educational Achievement Gap	17	4	-13	3	1	-2
14. Arts Occupations	45	45	0	14	15	+1
Economy (20%)	43	47	+4	12	14	+2
15. Per Capita Income	40	44	+4	11	13	+2
16. Poverty Rate	41	44	+3	11	12	+1
17. Per Capita Gross State Product	40	41	+1	11	13	+2
18. Entrepreneurial Depth	28	25	-3	8	8	0
19. U.S. Patents	37	40	+3	12	15	+3
20. Transportation Index	35	44	+9	10	14	+4
21. Home Computer Access	38	45	+7	10	12	+2
22. Internet Access	34	46	+12	8	13	+5
23. Home Broadband Access*	44	11	-33	14	1	-13
Environment (20%)	37	46	+9	8	10	+2
24. Toxic Air Emissions	42	38	-4	9	8	-1
25. Toxic Surface Water Discharges	32	32	0	6	8	+2
26. Toxic Releases to Land	41	15	-26	11	1	-10
27. Air Quality	15	18	+3	3	7	+4
28. Water Quality	15	46	+31	5	16	+11
29. Motor Fuel Use	39	39	0	13	13	0
Government (20%)	41	34	-7	14	10	-4
30. Efficiency (persons served per 100 state						
and local government employees)	15	6	-9	5	3	-2
31. Women in State Legislatures	48	47	-1	15	14	-1
32. Voter Participation Rate	35	42	+7	8	11	+3
Index	43	45	+2	11	11	0

 Table 6.4

 Components of the State of the Commonwealth Index, 2006

Note: <sup>a</sup>Peer states chosen by KLTPRC: AL, AR, FL, GA, IL, IN, LA, MI, MS, MO, NC, OH, SC, TN, VA, and WV. Source: Commonwealth. Legislative. Kentucky Long-Term. "The State" Table 1 and Appendix A.

## **Measuring Up**

Measuring Up 2006 ranks states in the five categories listed below.

- Preparation for education and training beyond high school
- Participation in education and training beyond high school
- Affordability of higher education
- Completion of certificates or degrees in a timely manner
- Benefits to the state from a highly educated population

Kentucky ranks lowest on preparation for and completion of postsecondary education, as well as the benefits of education. As Table 6.5 shows, Kentucky ranks below the national average on all measures. The lowest ranks pertain to preparation for and completion of postsecondary education and training, as well as the ensuing benefits to the state of a highly educated population.

Preparation		Participation		Affordability		Completion			Benefits					
Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)
5	MD	A- (91)	5	IL	A (96)	8	IL	F (59)	7	GA	A (95)	2	MD	A (99)
7	VA	A- (90)	7	MD	A (95)		IN	F (57)	10	FL	A (94)	3	VA	A (97)
10	NC	B+ (87)	) 20	DE	B (86)	10	NC	F (57)	14	DE	A- (90)	6	MO	A (95)
15	IL	B (85)	21 MO VA	D (94)	10	TX	F (57)	15	MO	B+ (89)	9	IL	A (93)	
21	OH	B- (81)		VA	B (84)		VA	F (57)		IL	B+ (88)	21 O	OH	B+ (87)
25	TX	B- (80)		U.S.	B (83)	15	OK	F (55)		IN		24	FL	B (84)
	U.S.	C+(79)	26	KY	B- (81)	17	AR	E (54)	16	NC			U.S.	B (84)
•	GA	CL (77)	20	NC	D (00)	1/	DE	F (54)		SC		25	AL	B (83)
28	SC	C+(//)	28	OH	B- (80)	22	MD	F (53)		VA			NC	
30	DE		30 5) 33 (1) 36	OK	C+(79)		U.S.	F (52)	25	MD	B (86)	20	DE	D (02)
	FL	C (75)		IN	C+ (78)	26	GA	F (51)	26	OH	B (85)	28	GA	<b>Б-</b> (82)
	IN			TX			KY			U.S.	B (85)	30	OK	B- (81)
	MO			AL	0.00	20	LA	F (50)	F (70)	30	MS	B (84)	32	TX
38	KY	C-(71)		AR	C (76)	29	MS		32	TN	B (83)	26	KY CL	
40	TN	C (70)	39	FL	C (75)	34	FL	F (49)	38	AL	B- (81)	36	TN	C+(78)
	WV	C-(/0)		LA	C- (70)	36	MO	F (47) F (46)	41	KY	C+ (78) C+ (77)	40	MS	C (76)
44	AR	D+ (69)	43	TN	C- (70)		TN			WV		41	AR	C (75)
45	OK	D+(67)		WV	C- (70)	41	WV		43	TX			SC	
47	MS	D- (62)	47 49	SC	D+ (69)	42	AL	F (42)	45	AR	C (76)	45	IN	C (74)
48	AL	D- (61)		GA	D+(67)	42	SC	г (43)	43	OK		49	LA	
50	LA	F (56)	50	MS	D (66)	45	OH	F (42)	47	LA	C- (72)		WV	D+ (68)

Table 6.5Measuring Up Grades and Scores, 2006

Note: Staff calculated U.S. average scores and grades. Source: Natl. Ctr. for Public Policy. "Compare." Table 6.6 summarizes the *Measuring Up* indicators that were weighted and summed to create a numeric score for each state. Letter grades reflect each state's standing relative to the average score of the top five states.

		KY	U.S.	Top 5	KY				
Indicators	Weight	Score	Avg.	Avg.	Rank*				
Preparation									
High School Completion	20%								
18- to 24-year-olds with high school credential (diploma or GED), 2002-									
2004 <sup>a</sup>	20%	87%	87%	94%	25				
K-12 Course Taking	35%								
Students in grades 9–12 taking at least one upper-level math course, FY	0.750/	520/	520/	C 10/	13				
2004°	8.75%	53%	53%	64%	(of 35)				
Students in grades $9-12$ taking at least one upper-level science course,	12 1250/	2004	210/	400/	18				
<u>F12004</u>	13.12370	29/0	31/0	4070	(01 33)				
Grade 8 students taking algebra FY 2004 <sup>b</sup>	8 75%	12%	22%	35%	(of 31)				
Grade 12 students taking at least one upper-level math course FY 2004 <sup>b</sup>	4 375%	na	n a	66%	na				
K-12 Student Achievement	35%	11.4.	11.u .	0070	11.u.				
Students at or above proficient on NAEP Grade 8 math 2005 <sup>c</sup>	3.5%	23%	28%	38%	38				
Students at or above proficient on NAEP Grade 8 reading 2005 <sup>c</sup>	3.5%	31%	29%	38%	24				
	5.570	5170	2970	5070	23				
Students at or above proficient on NAEP Grade 8 science, 2005 <sup>c</sup>	3.5%	31%	27%	41%	(of 46)				
					26				
Students at or above proficient on NAEP Grade 8 writing, 2002 <sup>c</sup>	3.5%	25%	30%	41%	(of 43)				
Low-income students at/above proficient on NAEP Grade 8 math, 2005 <sup>c</sup>	3.5%	14%	13%	22%	24				
Scores in top 20% nationally on SAT/ACT exam per 1,000 high school									
graduates, 2005 <sup>d,e</sup>	8.75%	156	184	237	34				
Scores of 3-5 on Advanced Placement test per 1,000 high school juniors									
and seniors, 2005 <sup>d,c</sup>	8.75%	96	147	217	32				
Teacher Quality	10%								
Students in grades 7-12 whose teachers majored in subject, FY 2000 <sup>e</sup>	10%	62%	70%	81%	40				
Overall Score and Grade for Preparation	100%	71 C-	79 C+	99 A	38				
Participation	-	1							
Young Adults	60%								
Chance for college by age 19 (percentage of 9 <sup>th</sup> graders completing high	100/								
school in 4 years & immediately going to college), 2002	40%	38.5%	38.0%	53.8%	25				
18- to 24-year-olds enrolled in college, 2002-2004 <sup>a</sup>	20%	32%	35%	41%	35				
Working-age Adults	40%								
25- to 49-year-olds enrolled part time, any postsecondary education, 2003 <sup>a</sup>	40%	3.6%	3.9%	5.1%	27				
Overall Score and Grade for Participation	100%	81 B-	83 B	98 A	27				
Affordability									
Family Ability To Pay consists of the 3 indicators below, weighted by		(For indicators below, low percent							
number of students enrolled in each sector:	50%	11	ndicates af	fordability	/)				
Percentage of income needed to pay for expenses (minus financial aid) at $200(\frac{a}{c},\frac{b}{c},\frac{b}{c})$	enroll-	260/	240/	150/	22				
Community coneges, FY 2000	ment	20%	24%0	13%	33				
references of income for expenses (infinus infancial and) at public 4-year colleges/universities EV $2006^{a,c,g,h,j}$	ment	30%	310/-	16%	25				
Percentage of income needed to pay for expenses (minus financial aid) at	enroll_	5070	J1/0	10/0	23				
private 4-vear colleges/universities FY 2006 <sup>a,c,g,h,j</sup>	ment	61%	72%	32%	26				
private : jear conegeo, anteronaco, i i 2000	1110110	01/0	/ 2/0	52/0	20				

Table 6.6Summary of Indicators Used in Measuring Up

Continued on next page.

Table 6.6	continued
-----------	-----------

Indicators	Waight	KY	U.S.	Top 5	KY Donk*
	weight	Score	Avg.	Avg.	Капк"
Affordability (continue	<b>d)</b>				
Strategies for Affordability	40%		1	1	1
State investment in need-based financial aid as compared to the federal $EV 200(c^{h})$	200/	420/	400/	200/	15
At lowest priced colleges, the share of income that the peerest families need	2070	4270	4070	8970	15
A lowest-priced coneges, the share of income that the poolest families need to pay for tuition FY $2006^{a,c,g}$	20%	24%	16%	7%	38
Reliance on Loans	10%	2170	1070	//0	50
Average loan amount undergraduates borrow each year FY 2005 <sup>c</sup>	10%	\$3 210	\$3.619	\$2 619	8
Overall Score and Grade for Affordability	100%	51 F	52 F	67 D	27
Completion	10070	011	021	07 D	
Development	200/				
First war community college students returning their second year fall	20 /0				
2004 <sup>ce.g</sup>	10%	51%	53%	62%	27
Freshmen at 4-year colleges/universities returning for sophomore year,					
Fall 2004 <sup>c,e,g</sup>	10%	70%	77%	82%	40
Completion	80%				
First-time, full-time students completing a bachelor's degree within 6 years					
of college entrance, FY 2004 <sup>c</sup>	30%	38%	55%	64%	47
Certificates, degrees, and diplomas awarded at all colleges and universities	500/	17	17	20	22
per 100 undergraduates, FY 2004 <sup>c,s,m</sup>	50%	17	17	20	23
Overall Score and Grade for Completion	100%	78 C+	85 B	102 A	41
Benefits Of Education	1				
Educational Achievement	37.5%				
Population ages 25 to 65 with a bachelor's degree or higher, 2002–2004 <sup>c,k,m</sup>	37.5%	23%	30%	37%	43
Economic Benefits	31.25%			1	
Increase in total personal income as a result of percentage of population with a bachelor's degree 2002, 2004, and $2005^{a,k,m}$	10 750/	09/	100/	120/	10
a bacheloi s degree, 2005, 2004, and 2005	18.7370	9%	1070	1270	19
some college but no bachelor's, 2003, 2004, and 2005 <sup>a,k,m</sup>	12.5%	2%	2%	3%	8
Civic Benefits	31.25%				
Residents voting in 2002 and 2004 national elections <sup>a</sup>	10.5%	54%	51%	64%	19
Of those who itemize on federal taxes, percent with charitable gifts, 2003 <sup>n</sup>	10.375%	86%	87%	91%	30
Increase in volunteering rate as a result of college education, 2003–2005 <sup>a</sup>	10.375%	16%	18%	22%	43
Overall Score and Grade for Benefits	100%	78 C+	84 B	98 A	36

Notes: Origins of data that appear in *Measuring Up 2006*: <sup>a</sup>Census Bureau; <sup>b</sup>Rolf K. Blank and Doreen Langesen. *State Indicators of Science and Mathematics Education* 2005 and unpublished data from the authors and from the Council of Chief State School Officers; <sup>c</sup>U.S. Dept. of Ed.; <sup>d</sup>College Board; <sup>e</sup>ACT, Inc.; <sup>f</sup>Thomas Mortenson. "Chance for College by Age 19 by State in 2002." Postsecondary Education Opportunity Web site; <sup>g</sup>Natl. Ctr. for Higher Ed. Management Systems' special analysis of IPEDS Peer Analysis System data from Natl. Ctr. for Ed. Statistics; <sup>h</sup>Natl. Ctr. for Higher Ed. Management Systems' Annual Survey of State Grant Aid Programs; <sup>j</sup>Natl. Assoc. of State Student Grant and Aid Programs; <sup>k</sup> Pinkerton Computer Consultants; <sup>m</sup>Research Triangle Institute; <sup>n</sup>U.S. Dept. of the Treasury. Internal Revenue Service.

Source: Natl. Ctr. for Public Policy and Higher Ed. "Compare" and Technical Guide.
#### **Camelot Index**

A founding editor of Federal Funds Information for States (FFIS) assembled the Camelot Index from an array of indicators, to counter a tendency for rankings to focus on just one area of states' performance. FFIS provides little information regarding its methodology and criteria for selecting indicators. A brief technical note mentions that the index is based on annual state rankings that were published by Morgan Quitno, which was acquired by Congressional Quarterly in 2007. After this change in ownership resulted in the discontinuation of some tables, FFIS went directly to the sources of those tables (Federal. "The 2008" 19). The overall rank is derived by averaging states' ranks on the six quality of life indices discussed below.

### **Healthy Economy**

The Healthy Economy index is based on poverty rates, employment growth, population growth, income growth, per-capita federal tax liability (a proxy for income), robustness of the state tax base, and average retail salaries.

### **Healthy People**

Just three measures— infant mortality rates, age-adjusted death rates, and uninsured rates—make up the Healthy People index.

### **Crime-free State**

The Crime-free State index is based on the ratio of violent and property crimes to the number of residents. As such, it is inflated in states with heavy tourism. The influx of tourists increases the population at risk of crime, but the number of residents in the denominator does not reflect this larger population.

#### **Educated Population**

The Educated Population index is derived from Armed Forces Qualification test rankings, pupil-teacher ratios, high school graduation rates, instate college costs, and ACT and SAT scores. As discussed in Chapter 2, states' average ACT and SAT scores are highly dependent on participation rates. The Armed Forces test is voluntary and likely to have similar or other limitations. The reasons for not including NAEP scores and adult educational attainment are unclear.

#### **Healthy Society**

The Healthy Society index reflects homeownership, voter turnout, births to unwed mothers, single-parent families, and welfare recipients.

#### **Prudently Managed Government**

The Prudently Managed Government index is based on the tax burden (state and local taxes as a percentage of income), solvency (state assets minus debt and unfunded pension liabilities), structural surplus/deficit (anticipated revenue growth compared to the amount needed to maintain current services), and bond ratings.

Table 6.7 shows Kentucky and peer state rankings overall and for each of the six indices. Based on these, Kentucky's greatest strengths are low crime and prudently managed government, with ranks of 8<sup>th</sup> and 13<sup>th</sup>, respectively. The greatest challenges are the economy, which is ranked 45<sup>th</sup>, and education, ranked 37<sup>th</sup>.

State	<b>Overall Rank</b>	Economy	Health	Crime	Education	Society	Government
VA	14	18	29	11	26	13	2
MD	22	14	26	37	34	20	7
MO	24	28	31	39	13	28	16
KY	27	45	33	8	37	28	13
DE	29	12	39	36	45	33	2
IL	31	24	28	27	38	20	44
IN	35	42	38	25	31	36	25
FL	36	20	34	46	49	35	15
OK	37	39	48	32	21	33	28
WV	38	50	43	11	32	25	43
NC	39	28	42	41	46	38	10
GA	40	31	47	35	47	42	5
OH	40	46	31	27	35	32	36
ΤХ	42	25	40	42	44	40	19
AR	43	49	44	43	17	38	23
AL	45	44	46	34	39	30	29
TN	46	43	41	49	24	49	20
SC	48	38	45	50	50	37	25
MS	49	48	49	23	41	44	45
LA	50	34	49	46	36	47	50

### Table 6.7Camelot Index, 2008

Source: Federal. "The 2008" 18.

According to the Camelot Index, Kentucky's greatest strengths are low crime and prudently managed government. The greatest challenges are the economy and education.

#### **Smartest State Index**

Morgan Quitno, acquired by Congressional Quarterly in 2007, has published state rankings and indices since 1990. The annual Smartest State Award is derived from the following 21 factors:

- K-12 revenue per \$1,000 personal income
- Percentage of current expenditures used for instruction
- District administrators as a percentage of staff
- Teacher salaries compared to all workers' salaries
- Percentage of school-age population in public schools
- Average daily attendance as a percentage of fall enrollment
- NAEP grade 4 reading proficiency
- NAEP grade 8 reading proficiency
- NAEP grade 4 writing proficiency
- NAEP grade 8 writing proficiency
- NAEP grade 4 math proficiency
- NAEP grade 8 math proficiency
- Average class size in public elementary schools
- Average class size in public secondary schools
- Median pupil-teacher ratio in public primary schools
- Median pupil-teacher ratio in public middle schools
- Median pupil-teacher ratio in public high schools
- Special education pupil-teacher ratio
- High school dropout rate
- Averaged freshman graduation rate
- Percentage of population graduated from high school (Congressional. *Smartest*).

Computing the Smartest State index for each state begins with comparing each indicator to the national average; a positive or negative score is assigned based on the extent to which the indicator is better or worse than the national average. The scores for all 21 indicators are then summed.

Kentucky is ranked 31<sup>st</sup> on the Smartest State Index.

Table 6.8 shows rankings for Kentucky and peer states. In FY 2007, Kentucky ranked 31<sup>st</sup>. This is a slight improvement over recent years but still below FY 2003.

		Rankings										
State*	FY 2003	FY 2004	FY 2005	FY 2006	FY 2007	Score						
VA	37	17	12	7	6	10.07						
MD	30	18	18	19	18	2.27						
MO	31	28	26	21	22	0.94						
NC	24	21	25	22	23	0.84						
IN	9	13	17	26	24	0.06						
TX	16	34	33	24	25	-0.11						
SC	36	41	32	29	26	-1.19						
DE	43	19	27	25	28	-2.47						
FL	47	40	39	36	29	-2.90						
TN	39	42	41	41	30	-3.01						
KY	28	37	37	35	31	-3.24						
AR	38	38	36	37	32	-3.44						
OH	41	22	20	31	34	-4.00						
IL	33	27	24	32	35	-4.32						
OK	32	39	40	39	36	-5.81						
WV	18	29	33	34	37	-5.82						
GA	40	36	38	40	41	-6.92						
LA	49	47	46	45	44	-10.95						
AL	41	46	44	43	45	-11.00						
MS	48	48	47	49	48	-14.78						

Table 6.8Smartest State Index, FY 2003-FY 2007

Note: \*States are listed in order of their 2007 score. Source: Congressional. *Smartest.* 

#### **Chapter Summary and Conclusions**

Indices based on multiple topics provide a multidisciplinary frame of reference. However, they have many disadvantages. They vary widely, depending on the measures and methodology used. In addition, organizations that produce indices on a broad array of topics may not have the specialized expertise to choose the best indicators for each specific topic, such as education.

The composite indices reviewed in this chapter represent a fraction of the state indices and rankings that are published each year. Kentucky's rank varies widely depending on the measures and methodologies used by each organization.

### Works Cited

ACT, Inc. 2007 Average ACT Scores by State. Iowa City: ACT, Inc., 2008. <a href="http://www.act.org/news/data/07/states-text.html">http://www.act.org/news/data/07/states-text.html</a> (accessed Jan. 18, 2008).

---. ACT Newsroom: Facts About the ACT. <a href="http://www.act.org/news/aapfacts.html">http://www.act.org/news/aapfacts.html</a> (accessed Feb. 23, 2007).

Annie E. Casey Foundation. 2007 KIDS COUNT Data Book. Baltimore: AECF, 2007. <a href="http://www.kidscount.org/datacenter/databook.jsp">http://www.kidscount.org/datacenter/databook.jsp</a> (accessed Jan. 18, 2008).

Asimov, Nanette. "California high school dropout rate far higher than expected." *San Francisco Chronicle* July 17, 2008. <a href="http://www.sfgate.com">http://www.sfgate.com</a>> (accessed Aug. 7, 2008).

Barth, Patte. *Score wars: Comparing the National. Assessment of Educational Progress with state assessments.* Alexandria, VA: Center for Public Education, National School Boards Association, March 29, 2006. <a href="http://www.centerforpubliceducation.org/site/c.kjJXJ5MPIwE/b.1577019/k.A07C/Score\_wars\_Comparing\_the\_National\_Assessment\_of\_Educational\_Progress\_with\_state\_assessments.htm">http://www.centerforpubliceducation.org/site/c.kjJXJ5MPIwE/b.1577019/k.A07C/Score\_wars\_Comparing\_the\_National\_Assessment\_of\_Educational\_Progress\_with\_state\_assessments.htm</a>> (accessed May 22, 2007).

Bloom, David E., and David Canning. "Booms, Busts, and Echoes." *Finance & Development* 43.3 (Sept. 2006). <a href="http://www.imf.org/external/pubs/ft/fandd/2006/09/bloom.htm">http://www.imf.org/external/pubs/ft/fandd/2006/09/bloom.htm</a> (accessed June 6, 2006).

Bracey, Gerald W. *Reading Educational Research: How to Avoid Getting Statistically Snookered*. Portsmouth, New Hampshire: Heinemann, Feb. 9, 2006.

Cary, Kevin. *The Evidence Suggests Otherwise: How States Inflate Their Educational Progress Under NCLB*. Washington, DC: Education Sector, May 2006. <a href="http://www.educationsector.org/analysis/analysis\_show.htm?doc\_id=373044">http://www.educationsector.org/analysis/analysis\_show.htm?doc\_id=373044</a>> (accessed May 11, 2007).

Cavanagh, Sean. "Testing Officials Again Tackle Accommodations and Exclusions for Special Student Populations." *Education Week* Vol. 27, Issue 43 (July 11, 2008): 1, 16.

College Board. *Advanced Placement Report to the Nation 2005*. New York: College Board, 2005. <a href="http://www.collegeboard.com/prod\_downloads/about/news\_info/ap/2005/ap-report-nation.pdf">http://www.collegeboard.com/prod\_downloads/about/news\_info/ap/2005/ap-report-nation.pdf</a>> (accessed March 18, 2008).

---. *College Board History*. New York: College Board, 2008. <http://www.collegeboard.com/about/association/history.html> (accessed March 19, 2008).

---. *College-Bound Seniors 2007*. New York: College Board, 2007. <a href="http://professionals.collegeboard.com/data-reports-research/sat/cb-seniors-2007/tables">http://professionals.collegeboard.com/data-reports-research/sat/cb-seniors-2007/tables</a> (accessed March 18, 2008).

---. "SAT Program." *Higher Ed. Recruitment and Admission.* <http://www.collegeboard.com/highered/ra/sat/sat.html> (accessed March 2, 2007).

---. *The 4<sup>th</sup> Annual AP Report to the Nation*. New York: College Board, 2008. <a href="http://professionals.collegeboard.com/profdownload/ap-report-to-the-nation-2008.pdf">http://professionals.collegeboard.com/profdownload/ap-report-to-the-nation-2008.pdf</a>> (accessed March 18, 2008).

Commonwealth of Kentucky. Auditor of Public Accounts. *Kentucky's Dropout Rate: October 2006- Performance Audit.* <a href="http://www.auditor.ky.gov/Public/Audit\_Reports/Archive/2006Dropoutreport.pdf">http://www.auditor.ky.gov/Public/Audit\_Reports/Archive/2006Dropoutreport.pdf</a> (accessed Jan. 31, 2008).

---. Department of Education. *Approved State Accountability Plan*. <http://www.ed.gov/admins/lead/account/stateplans03/index.html> (accessed Feb. 4, 2008). ---. Briefing Package, Nonacademic Data: Dropout, Retention, Transition to Adult Life, Attendance and Graduation Rates, 1993 to 2007, State Totals. Frankfort, KY: Comm. of KY Dept. of Ed., May 28, 2008. <a href="http://education.ky.gov/NR/rdonlyres/32B50296-B273-4943-BA0E-E4EEC97C80CF/0/">http://education.ky.gov/NR/rdonlyres/32B50296-B273-4943-BA0E-E4EEC97C80CF/0/</a> Nonacademic\_Brief\_2007rev.pdf> (accessed Aug. 7, 2008).

---. *Inclusion of Special Populations in the State-Required Assessment and Accountability Programs: 703 KAR* 5:070. Frankfort: Kentucky Board of Education. Feb. 2004. <a href="http://education.ky.gov/kde/administrative+resources/testing+and+reporting+/cats/policies+and+regulations/inclusion+of+special+populations+document.htm">http://education.ky.gov/kde/administrative+resources/testing+and+regulations/inclusion+of+special+populations+document.htm</a>> (accessed June 27, 2008).

---. Legislative Research Commission. Kentucky Long-Term Policy Research Center. *Reducing Obstacles Will Yield Even Higher Academic Returns to Educational Investments. Policy Notes* No. 26. Author Michael Childress. Frankfort: KLTPRC, April 2008. <a href="http://www.kltprc.net/policynotes/pn0026\_education\_funding.pdf">http://www.kltprc.net/policynotes/pn0026\_education\_funding.pdf</a>> (accessed May 12, 2008).

---. ---. "The State of the Commonwealth Index: Kentucky Makes Steady Quality-of-Life Progress." *Foresight* No. 45. Author Amy L. Watts. Frankfort: KLTPRC, 2006. <a href="http://www.kltprc.net/foresight/Chpt\_86.htm">http://www.kltprc.net/foresight/Chpt\_86.htm</a> (accessed July 17, 2008).

Congressional Quarterly. CQ Press Acquires Morgan Quitno Press. Washington, DC: CQ, June 21, 2007.

---. "Mission." About CQ. <a href="http://www.cq.com/corp/show.do?page=about\_mission">http://www.cq.com/corp/show.do?page=about\_mission</a>> (accessed Nov. 27, 2006).

---. *Smartest State 2006-2007*. Washington, DC: CQ, 2007 < http://www.statestats.com/edrank.htm> (accessed July 21, 2008).

Costrell, Robert. "Equity v. Equity: Why Education Week and the Education Trust don't agree." *Education Next* Vol. 5 No. 3(Summer 2005). <a href="http://www.hoover.org/publications/ednext/3219926.html">http://www.hoover.org/publications/ednext/3219926.html</a> (accessed Aug. 29, 2008).

Dillon, Sam. "States' Data Obscure How Few Finish High School." *The New York Times*. March 20, 2008. <*www.nytimes.com/2008/03/20/education/20graduation.html>* (accessed April 7, 2008).

Editorial Projects in Education. About EPE. < http://www2.edweek.org/info/about/> (accessed April 5, 2007).

---. *Education Week's Diplomas Count*. Bethesda, MD: EPE, Inc. For years 2006-2008. <http://www.edweek.org/rc> (accessed June 5, 2008).

---. *Education Week's Quality Counts*. Bethesda, MD: EPE, Inc. For years 2007 and 2008. <<u>http://www.edweek.org/rc> (accessed March 18, 2008)</u>.

---. *Education Week's Technology Counts 2008*. Bethesda, MD: EPE, Inc. March 27, 2008. <a href="http://www.edweek.org/ew/toc/2008/03/27/index.html">http://www.edweek.org/ew/toc/2008/03/27/index.html</a> (accessed Aug. 29, 2008).

---. *EdCounts Database*. Bethesda, MD: EPE, Inc., 2008. <a href="http://www.edweek.org/rc/2007/06/07/edcounts.html">http://www.edweek.org/rc/2007/06/07/edcounts.html</a> (accessed Aug. 29, 2008).

---. *Implementing Graduation Accountability Under NCLB*. Bethesda, MD: EPE, Inc. June 2007. <http://www.edweek.org> (accessed July 23, 2008).

Education Trust. *Education Trust Response to Education Week's "Quality Counts 2007" Demographics Aren't Destiny: What Schools Do Matters*. March 16, 2007. <a href="http://www2.edtrust.org/EdTrust/Press+Room/Quality+Counts+Response.htm">http://www2.edtrust.org/EdTrust/Press+Room/Quality+Counts+Response.htm</a> (accessed April 5, 2007).

---. *The Funding Gap 2007*. Washington, DC: Education Trust. Jan. 2008. <a href="http://www2.edtrust.org/">http://www2.edtrust.org/</a> NR/rdonlyres/5AF8F288-949D-4677-82CF-5A867A8E9153/0/FundingGap2007.pdf">http://www2.edtrust.org/</a> NR/rdonlyres/5AF8F288-949D-4677-82CF-5A867A8E9153/0/FundingGap2007.pdf</a> (accessed Jan. 18, 2007).

---. *What is The Education Trust?* <<u>http://www2.edtrust.org/edtrust/about+the+ed+trust></u> (accessed April 3, 2007).

Elliott, Janet. "65% plan to include librarians. Some criticized Perry's classroom spending proposal as putting sports before education." *Houston Chronicle*. April 7, 2006, section B, page 4. <a href="http://www.firstclasseducation.org/tx.asp">http://www.firstclasseducation.org/tx.asp</a> (accessed June 5, 2007).

Federal Funds Information for States. *FFIS Services*. Washington, DC: FFIS. <a href="http://www.ffis.org/47401.html">http://www.ffis.org/47401.html</a> (accessed July 21, 2008).

---. "The 2008 Camelot Index." State Policy Reports Vol. 26, Issue 7(April 2008): 1-19.

Fuller, Bruce, and Joseph Wright. *Diminishing Returns? Gauging the Achievement Effects of Centralized School Accountability*. Presentation. American Educational Research Association, Chicago, April 11, 2007. Berkeley, California: Policy Analysis for California Education, April 10, 2007. <a href="http://pace.berkeley.edu/reports/Fuller\_AERA\_Lecture\_04-10.pdf">http://pace.berkeley.edu/reports/Fuller\_AERA\_Lecture\_04-10.pdf</a>> (accessed May 22, 2007).

Fuller, Bruce, Kathryn Gesicki, Erin Kang, and Joseph Wright. *Is the No Child Left Behind Act Working? The Reliability of States' Tracking of Achievement*. Working Paper 0 6–1 Berkeley, California: Policy Analysis for California Education, 2006. <a href="http://eric.ed.gov/ERICDocs/data/ericdocs2/content\_storage\_01/000000b/80/33/75/06.pdf">http://eric.ed.gov/ERICDocs/data/ericdocs2/content\_storage\_01/000000b/80/33/75/06.pdf</a>> (accessed May 22, 2007).

Garasky, Steven. "The effects of family structure on educational attainment: do the effects vary by the age of the child?" American Journal of Economics and Sociology. Jan. 1, 1995.

Greene, Jay P., and Marcus A. Winters. *Leaving Boys Behind: Public High School Graduation Rates*. Civic Report No. 48. New York: Manhattan Institute, April 2006. <a href="http://www.manhattan-institute.org/html/cr\_48.htm">http://www.manhattan-institute.org/html/cr\_48.htm</a> (accessed May 8, 2008).

Hall, Daria. *Getting Honest About Grad Rates: How States Play the Numbers and Students Lose*. Washington, DC: Education Trust, June 2005.

Herszenhorn, David M. "New York English Scores Drop Sharply in 6th Grade." *The New York Times*. Sept. 22, 2006. <<a href="https://www.nytimes.com/2006/09/22/nyregion/22scores.html">www.nytimes.com/2006/09/22/nyregion/22scores.html</a> (accessed June 7, 2008).

Ho, Andrew, and Edward Haertel. (Over)-Interpreting Mappings of State Performance Standards onto the NAEP Scale. Washington, DC: Council of Chief State School Officers, 2007. <a href="http://www.ccsso.org/content/PDFs/Ho%20Haertel%20CCSSO%20Brief1%20Final.pdf">http://www.ccsso.org/content/PDFs/Ho%20Haertel%20CCSSO%20Brief1%20Final.pdf</a>> (accessed June 2, 2008).

---. Apples to Apples? The Underlying Assumptions of State-NAEP Comparisons. Washington, DC: Council of Chief State School Officers, 2007. <a href="http://www.ccsso.org/content/PDFs/">http://www.ccsso.org/content/PDFs/</a> Ho%20Haertel%20CCSSO%20Brief1%20Final.pdf> (accessed June 2, 2008).

Hoffman, Gene R., and Arthur A. Thacker. *Examining the Relationships Between KCCT and NAEP Through the Use of School-Level Scale Scores*. (HumRRO Report No. FR-04-25). Alexandria, VA: Human Resources Research Organization, April 2004.

Hoxby, Caroline. "Inadequate Yearly Progress: Unlocking the secrets of NCLB." Education Next 5.3 (Summer 2005):47-51. <a href="http://media.hoover.org/documents/ednext20053\_46.pdf">http://media.hoover.org/documents/ednext20053\_46.pdf</a>> (accessed May 22, 2007).

Hussar, William, and William Sonnenberg. *Trends in Disparities in School District Level Expenditures per Pupil* (NCES 2000020). Statistical Analysis Report. Washington, DC: NCES, U.S. Dept. of Ed.: Jan. 31, 2000. <a href="http://nces.ed.gov/pubs2000/2000020.pdf">http://nces.ed.gov/pubs2000/2000020.pdf</a>> (accessed April 8, 2008).

Innes, Richard G. "CATS-SCAN' reveals sick education system." *Perspective* 2008-06. Bowling Green: Bluegrass Institute for Public Policy Solutions, 2008. <a href="http://www.bipps.org/pubs/2008/catsscan.pdf">http://www.bipps.org/pubs/2008/catsscan.pdf</a>> (accessed May 22, 2008)

---. *Is Kentucky Socially Promoting Public School Graduates?* Bowling Green: Bluegrass Institute for Public Policy Solutions, 2008. <www.bipps.org/blog/> (accessed May 30, 2008).

Kuersten, Joan. Are Middle and High School Students Reading to Learn or Learning to Read? Chicago: National Parent Teacher Association. <a href="http://www.pta.org/archive\_article\_details\_1118072738531.html">http://www.pta.org/archive\_article\_details\_1118072738531.html</a> (accessed June 7, 2008).

Koger, L.E., A.A. Thacker, M.E. Koger, and R. Deatz. *Comparisons between KCCT and NAEP: Assessment frameworks, item format, item content, test administration, scoring, and reporting* (HumRRO Report No. DFR-03-88). Alexandria, VA: Human Resources Research Organization, 2003.

Leischer, Jennifer. *Middle School Slump*. Washington, DC: Thomas B. Fordham Institute, Sept. 14, 2005. <a href="http://www.edexcellence.net/detail/news.cfm?news">http://www.edexcellence.net/detail/news.cfm?news</a> id=407&id=> (accessed Aug. 7, 2008).

National Center for Higher Education Management Systems. *About Us–NCHEMS*. Boulder, CO: NCHEMS, 2008. <a href="http://www.nchems.org/about/index.php#more">http://www.nchems.org/about/index.php#more</a> (accessed May 7, 2008).

---. *Preparation: Public High School Graduation Rates*. Boulder, CO: NCHEMS, 2007. <www.higheredinfo.org/dbrowser/index.php?submeasure=36&year=2005&level=nation&mode=data&state=0> (accessed May 7, 2008).

National Center for Public Policy and Higher Education. *About Measuring Up: About the National Center*. <a href="http://measuringup.highereducation.org/about/about/aboutnationalcenter.cfm">http://measuringup.highereducation.org/about/aboutnationalcenter.cfm</a> (accessed Nov. 27, 2006).

---. "Compare States: Index Scores (2006)." *Measuring Up 2006: The National Report Card on Higher Education*. San Jose, CA: NCPPHE, 2006. <a href="http://measuringup.highereducation.org/compare/indexpage.cfm?myyear=2006">http://measuringup.highereducation.org/compare/indexpage.cfm?myyear=2006</a> (accessed Nov. 27, 2006).

---. *Measuring Up 2006: The National Report Card on Higher Education*. San Jose, CA: NCPPHE, 2006. <a href="http://measuringup.highereducation.org">http://measuringup.highereducation.org</a> (accessed Nov. 27, 2006).

---. *Technical Guide*. San Jose, CA: NCPPHE, 2006. For years 2000, 2002, 2004, and 2006. <a href="http://measuringup.highereducation.org">http://measuringup.highereducation.org</a> ,(accessed April 4, 2007).

National Education Association. *About NEA*. Washington, DC: NEA. <a href="http://www.nea.org/aboutnea/index.html">http://www.nea.org/aboutnea/index.html</a> (accessed April 2, 2007).

---. Rankings & Estimates 2006-2007: Rankings of the States 2006 and Estimates of School Statistics 2007. Washington, DC: NEA, Dec. 2007. <a href="http://www.nea.org/edstats/images/07rankings.pdf">http://www.nea.org/edstats/images/07rankings.pdf</a> (accessed Jan. 17, 2008).

National Forum on Education Statistics. *Forum Guide to Education Indicators* (NFES 2005-802). Washington, DC: NCES, 2005. <a href="http://nces.ed.gov/pubs2005/2005802.pdf">http://nces.ed.gov/pubs2005/2005802.pdf</a>> (accessed June 6, 2008).

National Governors Association. *Graduation Counts: Compact and Task Force Report Guidance on State Implementation and Reporting*. Washington, DC: NGA, Feb. 2, 2006. <a href="http://www.nga.org/Files/pdf/0602GRADGUIDANCE.pdf">http://www.nga.org/Files/pdf/0602GRADGUIDANCE.pdf</a>> (accessed April 24, 2008).

National Institute for Early Education Research. *About NIEER*. New Brunswick, NJ: NIEER/Rutgers University. <a href="http://nieer.org/about/">http://nieer.org/about/</a> (accessed March 26, 2007).

---. *The State of Preschool: State Preschool Yearbook.* New Brunswick, NJ: NIEER/Rutgers University. For years 2003-2007. <a href="http://nieer.org/yearbook/">http://nieer.org/yearbook/</a>> (accessed June 6, 2008).

National Research Council. *Lessons Learned About Testing: Ten Years of Work at the National Research Council.* Washington, DC: National Academies, 2003. <a href="http://www7.nationalacademies.org/dbasse/">http://www7.nationalacademies.org/dbasse/</a> BOTA\_brochure\_Lessons\_Learned\_PDF.pdf> (accessed June 7, 2008).

Olson, Lynn. "An 'A' or a 'D': State Rankings Differ Widely." *Education Week* 17(April 15, 1998): 1, 18. </br><www.edweek.org/ew/articles/1998/04/15/31stand.h17.html> (accessed June 7, 2008).

---. "Gaps in Proficiency Levels on State Tests And NAEP Found to Grow." *Education Week* 26.33(April 18, 2007): 12. <a href="http://www.edweek.org/ew/articles/2007/04/18/33aeradata.h26.html">http://www.edweek.org/ew/articles/2007/04/18/33aeradata.h26.html</a> (accessed May 4, 2007).

---. "Standards Get Boost on the Hill: Bills before Congress aim to raise the bar in states." *Education Week* 26.19 Jan. 17, 2007: 1, 25. <a href="http://www.edweek.org/ew/articles/2007/01/17/19standards.h26.html">http://www.edweek.org/ew/articles/2007/01/17/19standards.h26.html</a> (accessed May 4, 2007).

Pellegrino, James W. "Should NAEP performance standards be used for setting standards for state assessments?" *Phi Delta Kappan* 88.7 March 2007.

Pellegrino, James W., L.R. Jones, and K.J. Mitchell, eds. *Grading the Nation's Report Card: Evaluating NAEP and Transforming the Assessment of Educational Progress*. National Research Council Committee on the Evaluation of National Assessments of Educational Progress. Washington, DC: National Academy Press, 1998.

Peterson, Paul E., and Frederick M. Hess. "Keeping an Eye on State Standards." *Education Next* 6.3 Summer 2006: 28-29. <a href="http://www.hoover.org/publications/ednext/3211601.html">http://www.hoover.org/publications/ednext/3211601.html</a> (accessed May 11, 2007).

Phillips, Gary W. *Expressing International Educational Achievement in Terms of U.S. Performance Standards: Linking NAEP Achievement Levels to TIMSS.* Washington, DC: American Institutes for Research, 2007. <a href="http://www.air.org/news/documents/naep-timss.pdf">http://www.air.org/news/documents/naep-timss.pdf</a> (accessed May 27, 2008).

Porter, Andrew M. *Curriculum Assessment*. Naperville, IL: North Central Regional Educational Laboratory, Jan. 2, 2004. <a href="http://www.secsupport.org/pdf/curricasses.pdf">http://www.secsupport.org/pdf/curricasses.pdf</a>> (accessed May 27, 2008).

Porter, Andrew M., M. Polikoff, and J. Smithson. *Is There a de Facto National Curriculum?: Evidence from State Standards*. Prepared for the National Research Council Workshop on Assessing the Role of K-12 Academic Standards in States. Washington, DC: National Academies, May 2008. <a href="http://www7.nationalacademies.org/cfe/">http://www7.nationalacademies.org/cfe/</a> Porter\_Smithson%20State%20Standards%20Paper\_Tables.pdf> (accessed May 27, 2008).

Ravitch, Diane. "Every State Left Behind." *The New York Times*, Nov. 7, 2005, Ed. Desk, Sec. A, Page 23. <a href="http://www.nytimes.com/2005/11/07/opinion/07ravitch.html?ex=1289019600&en=8d680876065c7c42&ei=5090&partner=rssuserland&emc=rss">http://www.nytimes.com/2005/11/07/opinion/07ravitch.html?ex=1289019600&en=8d680876065c7c42&ei=5090&partner=rssuserland&emc=rss">http://www.nytimes.com/2005/11/07/opinion/07ravitch.html?ex=1289019600&en=8d680876065c7c42&ei=5090&partner=rssuserland&emc=rss</a> (accessed Dec. 1, 2006).

Robb, Laura. "*The Myth of Learn to Read/Read to Learn*." *Instructor* May/June 2002. <<u>http://content.scholastic.com/browse/article.jsp?id=4260></u> (accessed June 7, 2008).

Rohe, William M., Shannon Van Zandt, and George McCarthy. *The Social Benefits and Costs of Homeownership: A Critical Assessment of the Research*. Working Paper LIHO-01.12. Cambridge, MA: Harvard Univ. Joint Center for Housing Studies, Oct. 2001. <a href="http://www.jchs.harvard.edu/publications/homeownership/liho01-12.pdf">http://www.jchs.harvard.edu/publications/homeownership/liho01-12.pdf</a>> (accessed May 2, 2008).

Rothstein, Richard, Rebecca Jacobsen, and Tamara Wilder. "Proficiency for All' Is an Oxymoron." *Education Week* 26.13 Nov. 29, 2006: 32, 44. <a href="http://www.edweek.org/ew/articles/2006/11/29/13rothstein.h26.html">http://www.edweek.org/ew/articles/2006/11/29/13rothstein.h26.html</a> (accessed Dec. 1, 2006).

Shepard, Lorrie. Setting Performance Standards for Student Achievement. A Report of the National Academy of Education Panel on the Evaluation of the NAEP Trial State Assessment: An Evaluation of the 1992 Achievement Levels. Stanford, CA: National Academy of Education, Stanford Univ., School of Education, 1993.

Southern Regional Education Board. About SREB. Atlanta: SREB, 2008. <a href="http://sreb.org/main/SREB/index.asp">http://sreb.org/main/SREB/index.asp</a> (accessed July 11, 2008).

Stancavage, Fran, Freya Makris, and Megan Rice. SD/LEP Inclusions/Exclusions in NAEP: An Investigation of Factors Affecting SD/LEP Inclusions/Exclusions in NAEP. Final Report. Washington, DC: American Institutes for Research. Jan. 2007. <a href="http://www.air.org/publications/documents/NAEP">http://www.air.org/publications/documents/NAEP</a> inclusion.pdf> (accessed June 27, 2008).

Standard & Poor's. The National Assessment of Educational Progress and State Assessments: What Do Differing Student Proficiency Rates Tell Us? New York: SchoolMatters.com, Fall 2005. <a href="http://www.schoolmatters.com/pdf/naep">http://www.schoolmatters.pdf</a> (accessed May 11, 2007).

State of Texas. Texas Education Agency. Librarians included in proposed 65 percent rule. Austin, TX: Texas Education Agency, April 6, 2006. <a href="http://www.tea.state.tx.us/press/65percentpressrelease.pdf">http://www.tea.state.tx.us/press/65percentpressrelease.pdf</a> (accessed June 4, 2007).

Stecher, Brian M. "Consequences of Large-Scale, High-Stakes Testing On School and Classroom Practice." In Hamilton, Making Sense of Test-Based Accountability in Education. Laura S. Hamilton, Brian M. Stecher, and Stephen P. Klein, eds. Santa Monica, CA: RAND Corp., 2002. 79-100. <a href="http://www.rand.org/pubs/monograph">http://www.rand.org/pubs/monograph</a> reports/MR1554/MR1554.ch4.pdf> (accessed May 22, 2007).

Swanson, Christopher B. Who Graduates? Who Doesn't? A Statistical Portrait of Public High School Graduation, Class of 2001. Washington, DC: Urban Institute, 2003. <a href="http://www.urban.org/UploadedPDF/410934">http://www.urban.org/UploadedPDF/410934</a> WhoGraduates.pdf> (accessed April 16, 2008).

United States. Department of Commerce. Census Bureau. "American Community Survey." American FactFinder. Washington, DC: Census Bureau. <a href="http://factfinder.census.gov/servlet/">http://factfinder.census.gov/servlet/</a> DatasetMainPageServlet? program=ACS& submenuId=& lang=en& ts=> (accessed March 28, 2008).

---. ---. "Characteristics of the Population Below the Poverty Level: 1980." Current Population Reports: Consumer Income. Series P-60, No. 133. <a href="http://www.census.gov/hhes/www/poverty/prevcps/p60-">http://www.census.gov/hhes/www/poverty/prevcps/p60-</a> 133.pdf#page=9> (accessed April 20, 2007).

---. ---. "Decennial Census." American FactFinder. Washington, DC: Census. For census years 1990 and 2000. <http://factfinder.census.gov/servlet/DatasetMainPageServlet? program=DEC& submenuId=& lang=en& ts=> (accessed March 28, 2008).

---. ---. Economic Census. < http://www.census.gov/econ/census02/> (accessed Aug. 1, 2008).

---. ---. Public Education Finances. Washington, DC: Census. For years 2003 to 2006. <http://www.census.gov/govs/www/school.html> (accessed July 10, 2008).

---. ---. Voting and Registration in the Election of November 2004. <a href="http://www.census.gov/population/www/socdemo/voting/cps2004.html">http://www.census.gov/population/www/socdemo/voting/cps2004.html</a> (accessed April 16, 2008).

---. Department of Education. High School Graduation Rate: Non-Regulatory Guidance. Washington, DC: USDOE, Dec. 22, 2008. < http://www.ed.gov/policy/elsec/guid/hsgrguidance.pdf> (accessed Dec. 22, 2008).

----. U.S. Secretary of Education Margaret Spellings Announces Proposed Regulations to Strengthen No Child Left Behind. Washington, DC: USDOE, April 22, 2008.

<a href="http://www.ed.gov/news/pressreleases/2008/04/0422208.html">http://www.ed.gov/news/pressreleases/2008/04/0422208.html</a> (accessed April 1, 2008).

---. ---. Institute of Education Sciences. National Center for Education Statistics. *120 Years of American Education: A Statistical Portrait*. NCES 93-442. Washington, DC: NCES, Jan. 1993. <a href="http://nces.ed.gov/pubs93/93442.pdf">http://nces.ed.gov/pubs93/93442.pdf</a> (accessed July 3, 2008).

---. ---. A Closer Look at Exclusion and Accommodations as Related to Assessment Results. Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/about/effect\_exclusion.asp">http://nces.ed.gov/nationsreportcard/about/effect\_exclusion.asp</a> (accessed May 22, 2007).

---. ---. "About State NAEP." *National Assessment of Educational Progress: The Nation's Report Card.* Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/about/state.asp">http://nces.ed.gov/nationsreportcard/about/state.asp</a> (accessed May 4, 2007).

---. ---. An Historical Overview of Revenues and Expenditures in Public Elementary and Secondary Education, by State. Fiscal Years 1990-2002. Statistical Analysis Report. NCES 2007-317. Authors Elise St. John, Jason Hill, and Frank Johnson. Washington, DC: NCES, Jan. 2007. <a href="http://nces.ed.gov/pubs2007/2007317.pdf">http://nces.ed.gov/pubs2007/2007317.pdf</a> (accessed July 7, 2008).

---. ---. Common Core of Data. Washington, DC: NCES. <a href="http://nces.ed.gov/ccd/">http://nces.ed.gov/ccd/</a> (accessed July 8, 2008).

---. ---. Current Expenditures for Public Elementary and Secondary Education: School Year 2003-04. NCES 2006-352. Washington, DC: NCES, July 2006. <a href="http://nces.ed.gov/pubs2006/2006352.pdf">http://nces.ed.gov/pubs2006/2006352.pdf</a>> (accessed Feb. 7, 2007).

---. ---. "Data Tables: Averaged Freshman Graduation Rates by Race and Ethnicity-School Year 2004-05." *Common Core of Data.* Washington, DC: NCES, 2008. <a href="http://nces.ed.gov/ccd/data\_tables.asp">http://nces.ed.gov/ccd/data\_tables.asp</a> (accessed Feb. 29, 2008).

---. ---. *Digest of Education Statistics*. Washington, DC: NCES. For years 1991 to 2007. <http://nces.ed.gov/programs/digest/> (accessed July 3, 2008).

---. ---. Dropout Rates in the United States: 2005. Compendium Report. NCES 2007-059. Washington, DC: NCES, June 2007.

---. ---. How the Samples of Schools and Students Are Selected for the Main Assessments (State and National). Washington, DC: NCES. April 17, 2008. <a href="http://nces.ed.gov//nationsreportcard/about/nathow.asp">http://nces.ed.gov//nationsreportcard/about/nathow.asp</a> (accessed July 9, 2008).

---. ---. Indicators of School Crime and Safety, 2007. NCES 2008-021/NCJ 219553. Dec. 2007. <http://nces.ed.gov/pubs2008/2008021.pdf> (accessed July 9, 2008).

---. ---. "Interpreting NAEP Results." *National Assessment of Educational Progress: The Nation's Report Card.* Washington, DC: NCES. < http://nces.ed.gov/nationsreportcard/about/interpretresults.asp > (accessed May 7, 2007).

---. ---. Investigating the Potential Effects of Exclusion Rates on Assessment Results. Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/about/2005\_effect\_exclusion.asp">http://nces.ed.gov/nationsreportcard/about/2005\_effect\_exclusion.asp</a> (accessed May 23, 2007).

---. ---. *Mapping 2005 State Proficiency Standards Onto the NAEP Scales*. (NCES 2007-482). U.S. Department of Education. Washington, DC: NCES, 2007. <a href="http://nces.ed.gov/nationsreportcard/pdf/studies/2007482.pdf">http://nces.ed.gov/nationsreportcard/pdf/studies/2007482.pdf</a>> (accessed June 30, 2008).

---. ---. "Mathematics." *National Assessment of Educational Progress (NAEP): The Nation's Report Card.* Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/mathematics/whatmeasure.asp">http://nces.ed.gov/nationsreportcard/mathematics/whatmeasure.asp</a> (accessed May 7, 2007).

---. ---. *Mathematics Report Card*. Washington, DC: NCES. <a href="http://nationsreportcard.gov/math\_2007">http://nationsreportcard.gov/math\_2007</a> (accessed May 22, 2008).

---. ---. *Monitoring School Quality: An Indicators Report*. Washington, DC: NCES, Dec. 2000. <http://nces.ed.gov/pubs2001/2001030.pdf> (accessed April 30, 2007).

--. --. "More About the NAEP Long-Term Trend Assessment." *National Assessment of Educational Progress (NAEP): The Nation's Report Card.* Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/ltt/moreabout.asp">http://nces.ed.gov/nationsreportcard/ltt/moreabout.asp</a> (accessed May 4, 2007).

---. ---. *NAEP Data Explorer, Main NAEP Version*. Washington, DC: NCES. <http://nces.ed.gov/nationsreportcard/nde/> (accessed May 4, 2007).

---. ---. *NAEP Inclusion Policy*. Washington, DC: NCES. May 13, 2008. <http://nces.ed.gov/nationsreportcard/about/inclusion.asp> (accessed June 3, 2008).

---. ---. *NAEP Overview*. Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/about/">http://nces.ed.gov/nationsreportcard/about/</a> (accessed May 1, 2008).

---. ---. *NAEP State Profiles*. Washington, DC: NCES. <<u>http://nces.ed.gov/nationsreportcard/states/</u>> (accessed Aug. 7, 2008).

---. ---. National Institute of Statistical Sciences/Education Statistics Services Institute Task Force on Graduation, Completion and Dropout Indicators (Final Report). Washington, DC: 2005. <a href="http://nces.ed.gov/pubs2005/2005105.pdf">http://nces.ed.gov/pubs2005/2005105.pdf</a>> (accessed April 25, 2008).

---. ---. *National Public Education Financial Survey Instruction Booklet*. Feb. 2005. <a href="http://www2.census.gov/govs/npefs/manual2004.pdf">http://www2.census.gov/govs/npefs/manual2004.pdf</a>> (accessed Nov. 27, 2006).

---. ---. "NCES Comparable Wage Index Data Files." *Education Finance Statistics Center-Cost Adjustments*. Washington, DC: NCES. <a href="http://nces.ed.gov/edfin/prodsurv/data.asp">http://nces.ed.gov/edfin/prodsurv/data.asp</a> (accessed Feb. 5, 2008).

---. ---. Numbers and Rates of Public High School Dropouts: School Year 2004-05, First Look. NCES 2008-305. Washington, DC: NCES, Dec. 2007. <a href="http://nces.ed.gov/pubs2008/2008305.pdf">http://nces.ed.gov/pubs2008/2008305.pdf</a>> (accessed Feb. 29, 2008).

---. ---. Numbers and Types of Public Elementary and Secondary Agencies from the Common Core of Data: School Year 2005-06. NCES 2007-353. Author Lee Hoffman. <a href="http://nces.ed.gov/pubs2007/2007353.pdf">http://nces.ed.gov/pubs2007/2007353.pdf</a> (accessed Feb. 5, 2008).

---. ---. Numbers and Types of Public Elementary and Secondary Schools from the Common Core of Data: School Year 2005–06. First Look. NCES 2007-354rev. Author Lee Hoffman. <http://nces.ed.gov/pubs2007/2007354rev.pdf> (accessed Feb. 5, 2008).

---. ---. Overview of Public Elementary and Secondary Students, Staff, Schools, School Districts, Revenues, and Expenditures: School Year 2004-05 and Fiscal Year 2004. Washington, DC: NCES, Nov. 2006. <a href="http://nces.ed.gov/pubs2007/2007309.pdf">http://nces.ed.gov/pubs2007/2007309.pdf</a> (accessed Feb. 5, 2008).

---. ---. Projections of Education Statistics. Washington, DC: NCES. For years 1991 to 2007. <http://nces.ed.gov/pubsearch/index.asp?PubSectionID=1&HasSearched=1&pubspagenum=1&sort=3&order=0&L 1=&L2=&searchstring=projections+of+education+statistics&searchtype=AND&searchcat2=&searchcat=title&page size=15&searchmonth=1&searchyear=1991&datetype=%3E%3D&pubtype=&surveyname=&surveyid=&centerna me=NCES&center=NCES> (accessed July 1, 2008).

---. ---. Public Elementary and Secondary School Student Enrollment, High School Completions, and Staff From the Common Core of Data: School Year 2005–06 (NCES 2007-352). Authors Jennifer Sable and Anthony Garofano. Washington, DC: NCES, U.S. DOE, 2007. <a href="http://nces.ed.gov/pubsearch/pubsinfo.asp?pubid=2007352">http://nces.ed.gov/pubsearch/pubsinfo.asp?pubid=2007352</a>> (accessed Jan. 31, 2008). ---. ---. "Reading." *National Assessment of Educational Progress (NAEP): The Nation's Report Card.* Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/reading/whatmeasure.asp">http://nces.ed.gov/nationsreportcard/reading/whatmeasure.asp</a> (accessed May13, 2008).

---. ---. *Reading Report Card*. Washington, DC: NCES. <a href="http://nationsreportcard.gov/reading\_2007">http://nationsreportcard.gov/reading\_2007</a> (accessed May 22, 2008).

---. ---. Revenues and Expenditures for Public Elementary and Secondary Education: School Year 2005-06 (Fiscal Year 2006), First Look. NCES 2008328. Washington, DC: NCES, April 15, 2008. <a href="http://nces.ed.gov/pubs2008/2008328.pdf">http://nces.ed.gov/pubs2008/2008328.pdf</a>> (accessed June 13, 2008).

--. ---. "Science." *National Assessment of Educational Progress (NAEP): The Nation's Report Card.* Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/science/whatmeasure.asp">http://nces.ed.gov/nationsreportcard/science/whatmeasure.asp</a> (accessed May 7, 2007).

---. ---. State Exclusion Rate Tables. <a href="http://nationsreportcard.gov/reading\_math\_2005/s0094.asp">http://nationsreportcard.gov/reading\_math\_2005/s0094.asp</a> (accessed July 8, 2008).

---. ---. Status of Education in Rural America. NCES 2007-040. Washington, DC: NCES, July 2007. <a href="http://nces.ed.gov/pubs2007/2007040.pdf">http://nces.ed.gov/pubs2007/2007040.pdf</a>> (accessed July 8, 2008).

---. ---. Students with Disabilities Background Questionnaire: 2007. <http://nces.ed.gov/nationsreportcard/pdf/bgq/sch-sdlep/BQ07-NAEP-SD.pdf> (accessed June 27, 2008).

---. ---. "The Status of Achievement Levels." *National Assessment of Educational Progress (NAEP): The Nation's Report Card.* Washington, DC: NCES. <a href="http://nces.ed.gov/nationsreportcard/achievdev.asp">http://nces.ed.gov/nationsreportcard/achievdev.asp</a> (accessed July 1, 2008).

---. National Assessment Governing Board. *Mathematics Framework for the 2007 National Assessment of Educational Progress*. Washington, DC: NAGB, Sept. 2006. <a href="http://nagb.org/publications/frameworks.htm">http://nagb.org/publications/frameworks.htm</a> (accessed May 16, 2008).

---. *Reading Framework for the 2007 National Assessment of Educational Progress*. Washington, DC: NAGB, Sept. 2006. <a href="http://nagb.org/publications/frameworks.htm">http://nagb.org/publications/frameworks.htm</a>> (accessed May 13, 2008).

---. Science Framework for the 2005 National Assessment of Educational Progress. Washington, DC: NAGB, Sept. 2004. <a href="http://nagb.org/publications/frameworks.htm">http://nagb.org/publications/frameworks.htm</a>> (accessed May 19, 2008).

---. *Writing Framework and Specifications for the 2007 National Assessment of Educational Progress.* Washington, DC: NAGB, 2006. < http://nagb.org/publications/frameworks.htm> (accessed May 15, 2008).

---. ---. What is NAGB? < http://www.nagb.org> (accessed May 16, 2008).

---. Office of Elementary and Secondary Education. *Notice Authorizing Schoolwide Programs To Consolidate Federal Education Funds and Exempting Them From Complying With Statutory or Regulatory Provisions of Those Programs*. <a href="http://www.ed.gov/legislation/FedRegister/other/2004-3/070204a.html">http://www.ed.gov/legislation/FedRegister/other/2004-3/070204a.html</a> (accessed May 15, 2007).

---. Government Accountability Office. *Education Could Do More to Help States Better Define Graduation Rates and Improve Knowledge about Intervention Strategies* (GAO-05-879). Washington, DC: GAO, September 2005. <a href="http://www.gao.gov/new.items/d05879.pdf">http://www.gao.gov/new.items/d05879.pdf</a> (accessed Jan. 31, 2008).

---. *Educational Achievement Standards: NAGB's Approach Yields Misleading Interpretations.* Report to Congressional Requesters. GAO/PEMD-93-12. June 1993. <a href="http://eric.ed.gov/ERICDocs/data/ericdocs2sql/">http://eric.ed.gov/ERICDocs/data/ericdocs2sql/</a> content storage 01/0000019b/80/13/f1/45.pdf> (accessed May 22, 2007).

Vinovskis, Maris A. Overseeing the Nation's Report Card: The Creation and Evolution of the National Assessment Governing Board (NAGB). Washington, DC: U.S. Dept. of Education, 1998.

Way, Walter D. *Precision and Volatility in School Accountability Systems*. Educational Testing Service Research Report RR-06-26. Princeton, New Jersey: ETS, Sept. 2006.

### Appendix A Rankings For All States

All R	aces/Etl	nnicities	White	e Non-H	ispanic	Black Non-Hispanic		spanic		Hispani	c
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
1	NE	87.8	1	WI	91.8	1	ND	96.2	1	ME	97.3
2	WI	86.7	2	NE	91.0	2	VT	92.1	2	KY	91.0
3	IA	86.6	3	ND	89.3	3	ME	91.7	3	MO	86.4
4	VT	86.5	4	MN	89.1	4	AZ	88.0	4	MT	84.4
5	ND	86.3	5	AZ	88.8	5	SD	79.3	5	AR	79.6
6	MN	85.9	6	NJ	88.7	6	RI	76.4	6	MD	78.6
7	NJ	85.1	7	IA	87.7	7	NJ	72.5	7	NJ	78.1
8	AZ	84.7	0	IL	975	8	MD	71.1	8	AZ	76.9
9	UT	84.4	0	UT	87.5	9	DC	70.4	9	VA	76.7
10	PA	82.5	10	SD	86.4	10	KY	69.9	10	NE	76.3
11	SD	82.3	11	PA	86.3	11	WY	69.6	11	ND	75.7
12	MT	81.5	12	CT	86.2	12	WV	69.5	12	IA	73.2
13	ID	81.0	13	MT	84.0	12	AR	07.5	13	LA	71.9
14	CT	80.9	14	OH	83.6	14	VA	69.0	14	HI	71.5
15	MO	80.6	15	MD	83.5	14	IA	07.0	15	OH	71.3
16	OH	80.2	16	MO	82.9	16	CO	68.9	15	OK	71.5
17	NH	80.1	17	MA	82.7	17	TX	68.6	17	RI	71.0
18	VA	79.6	18	VA	82.3	18	OK	68.2	18	WI	70.9
19	IL	79.4	19	CO	82.1		MA		19	SD	70.5
20	MD	79.3	20	DC	81.7	19	UT	68.0	20	TN	69.4
21	KS	79.2	21	KS	81.6		СТ		21	MS	69.0
22	MA	78.7	22	TX	81.5	22	MO	67.5	22	NC	66.9
23	ME	78.6	23	CA	80.6		HI			IL	
24	RI	78.4		U.S.	80.4	24	KS	66.6	24	TX	66.3
25	W V OV	77.3	24	KI VT	80.2	25	PA	66.2	25	PA	66.2
26	OK	/6.9	25	VI NV	79.1	26	NC CA	65./	26	CA	65.6
27	0	76.7	27	NY	70.0	27	CA	64.4	27	OK	65.2
20	W Y VV	75.0	27	MI	79.0	20	DE	64.2	28	MN	65.1
29		75.9	28	ME	78.0	29	INIVI W/A	62.2	29	W I U S	64.0
30	AK HI	75.1	30	OK	77.0	30	WA NE	63.0	30	U.S. IN	63.7
32	W A	75.0	30	WV	77.4	32	MN	62.1	31	WA	63.1
52	US	74.7	32	DF	77.3	33	MT	61.6	32	CT	62.9
33	CA	74.6	33	AR	76.9	34	OH	61.3	33	FL	62.7
34	OR	74.2	34	KY	76.8	35	TN	59.8	34	DE	62.4
35	TX	74.0	35	WA	76.6	36	MS	59.7	35	NM	60.7
36	IN	73.2	36	NC	76.4	37	IL	58.6	36	KS	60.1
37	DE	73.1	37	IN	76.3	38	OR	58.4	37	AL	59.9
38	MI	73.0	38	OR	74.9		U.S.	58.1	37	CO	59.6
39	NC	72.6	39	NM	73.4	39	AL	57.2	39	UT	59.5
40	DC	68.8	40	LA	72.2	40	LA	54.1	39	MA	56.5
41	TN	68.5	41	HI	71.9	41	WI	52.6	41	MI	55.8
42	AL	65.9	42	AK	71.3	42	GA	52.4	42	GA	51.2
43	NM	65.4	43	TN	71.0	43	IN	52.2	43	DC	49.9
44	NY	65.3	44	AL	70.5	44	FL	51.8	44	NY	43.6
45	FL	64.6	45	FL	70.2	45	AK	51.5	45	NV	41.8
46	AK	64.1	46	GA	66.6	46	MI	50.5	46	AK	27.6
47	LA	63.9	40	MS	00.0	47	NY	45.8		ID	
48	MS	63.3	48	NV	62.4	48	NV	42.7		NH	
49	GA	61.7		ID			ID		n.a.	SC	n.a.
50	SC	60.1	n.a.	NH	n.a.	n.a.	NH	n.a.		VT	
51	NV	55.8		SC			SC			WV	

### Table A.1 (corresponds to Table 2.2)NCES Averaged Freshman Graduation Rate, FY 2005

Notes: The averaged freshman graduation rate estimates the percentage of an entering freshman class that graduates with a standard diploma in 4 years. For FY 2005, it equals the total number of diploma recipients in FY 2005 divided by the average membership of the 8th-grade class in FY 2001, the 9th-grade class in FY 2002, and the 10th-grade class

in FY 2003. Rates for Asian students are not reported because Kentucky's AFGR exceeded 100%; this can happen for small, growing subgroups.

Source: U.S. Dept. of Ed. Inst. Natl. "Data Tables."

All Students							Gender							
						(	hange							
	2001			2005		(2005)	Minus 2	001)	м	ales 200	)5	Fer	nales 2(	005
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
1	NJ	83.4	1	NJ	83.3	1	TN	7.9	1	NJ	81.1	1	UT	85.2
2	ND	79.9	2	IA	82.8	2	FL	7.8	2	IA	80.5	2	NJ	84.0
2	ID	70.0	3	WI	80.5	2	KY	()	3	PA	78.1	3	IA	83.4
3	SD	/9.8	4	PA	80.4	3	NY	6.2	4	MN	77.3	4	PA	82.6
5	MN	78.9	5	VT	80.2	5	AZ	6.0	5	WI	77.0	5	WI	82.4
6	IA	78.7	6	NE	79.6	3	WA	0.0	6	ND	76.3	6	MN	81.2
7	UT	78.5	7	ND	79.2	7	ME	5.2	7	UT	76.1	7	СТ	80.8
8	WI	78.1	8	UT	78.6	8	CO	5.0	8	NE	75.0	'	NE	00.0
9	VT	78.0	9	CT	78.1	8	OH	5.0	9	CT	74.9	9	ND	80.2
10	NE	77.7		MN		10	PA	4.9	10	ID	74.4	10	NY	79.2
11	MT	77.4	11	ME	77.2	10	SC	4.1	11	MO	73.5	11	MO	79.0
12		76.8	12	NH	77.1	12	IA	4.1	10	MI	72.2	12	ME	/8./
13	PA	/5.5	13	IL ID	/6./	13	MS	3.8	12	N Y VT	13.2	13	MD	78.4
14	MD	73.3	14	ID MO	76.5	14	MA	3.6	15	V I ME	72 8	15	MD	78.2
15	IL KS	74.0	15	OH	75.0				15	OH	72.6	15	ID	10.2
10	NH	74.5	10	MT	75.7	16	MO	35	10	Ш	72.0	16	OH OH	78.1
17	RI	73.5	17	SD	75.6	10	NC	5.5	17	SD IL	72.0	18	Ш	77.9
10	OR	73.3	10	MA	74.7	19	NH	3.1	19	KS	70.8	10	Δ7	77.5
20	MO	73.0	20	KS	74.3	20	GA	2.6	20	CO	70.3	20	SD SD	77.2
20	VA	72.6	20	CO	74.5		US	2.6	20	OK	70.2	20	OR	76.8
21	IN	72.5	21	WY	74.2	21	AR	2.0	21	WV	70.0	21	VA	76.7
23	WY	72.4		IN		22	WI	2.3		AZ		23	AR	76.5
24	ME	72.1	23	MD	73.6	23	VT	2.2	23	OR	69.8		IN	70.0
	MA		25	AZ	73.3	24	WV	2.1	25	WY	69.4	24	WV	76.3
25	MI	71.0	26	AR	73.2	25	IL	1.0	26	AR	69.2	26	KY	76.0
27	OH	70.9	27	VA	72.9	25	NE	1.9	27	MA	69.0	27	KS	75.8
28	AR	70.8	28	WV	72.8	27	WY	1.8	28	VA	68.9	28	MA	75.5
29	WV	70.7	29	KY	71.5	28	HI	1.4	29	MD	68.3	29	WY	75.3
30	OK	70.1	30	RI	71.1	29	CT	1.3		U.S.	67.8		U.S.	75.3
31	CA	69.2	31	OK	70.8	30	IN	1.2	30	IN	67.7	30	WA	74.6
32	CO	69.1		U.S.	70.6	31	CA	0.9	31	MI	67.3	31	RI	74.3
	U.S.	68.0	32	MI	70.5	32	OK	0.7	32	RI	67.1	32	MI	74.2
33	AZ	67.3	33	OR	70.4	33	VA	0.3	33	KY	66.2	33	CA	74.1
34	HI	66.0	34	CA	70.1	34	UT	0.1	34	CA	65.7	34	VT	74.0
35	KY	65.3	35	WA	68.8	35	KS	0.0	35	TX	64.9	35	NC	73.3
36	DC	65.2	36	TX	68.5	36	NJ	-0.1	36	WA	64.8		OK	
37	TX	64.9	37	NY	68.0	37	AL	-0.4	37	HI	64.6	37	TX	72.7
38	DE	64.3	38	AK	67.6	38	MI	-0.5	38	AK	61.8	38	AK	/1./
39	AK	64.1	39	HI	67.0	39	ND MOT	-0.7	<u> </u>	NC EL	61.3 5(1	<u> </u>	HI	/0.4
41	LA	(2.5	40	NC TN	67.0	40	MN	-0.8	40	FL	56.1	40	MS	68.6
41	INC WA	62.0	41	1 N MS	61.9	41	MD	-1.7	41	MS DE	55.0	41	FL DE	64.0
42	W A NV	61.9	42		61.2	12	DI	2.4	42	GA DE	52.0	42	GA	63.0
43		61.6	43	FI	60.8	43	OP NI	-2.4	43	DC	51.0	43	UA I A	60.5
45	NM	61.0	45	DF	60.0	45	ID	-2.9	45	NM	49.1	45	NM	59.8
46	MS	58.0	46	GA GA	58.1	-5	DF	-5.2	46	I.A	483		AI	57.0
47	TN	57.5	47	DC	57.6	46	SD	-4.2	10	AL	10.5	1	DC	
48	GA	55.5	48	SC	55.6	48	NM	-73		NV			NV	
49	NV	54.7	49	LA	54.7	49	DC	-7.5	n.a.	NH	n.a.	n.a.	NH	n.a.
50	FL	53.0	50	NM	54.1	50	NV	-9.3		SC			SC	
51	SC	50.8	51	NV	45.4	51	LA	-9.4		TN			TN	

# Table A.2 (corresponds to Table 2.3)Cumulative Promotion Index for All Students and by Gender, FY 2004

Source: Editorial Projects in Education. Education Week's Diplomas Count.

	Asian			Hispanic			Black			White	
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
1	MD	91.8	1	NJ	67.0	1	UT	65.5	1	UT	88.6
2	TX	86.9	2	KY	66.5	2	HI	65.0	2	DC	86.3
3	LA	85.0	3	MD	64.9	3	AR	64.1	3	СТ	85.5
4	NJ	84.7	4	LA	62.5	4	CT	63.5	4	NJ	84.6
5	KY	84.5	5	UT	61.7	5	NJ	62.5	5	NE	84.5
6	CA	83.7	6	CA	60.0	6	MD	62.0	6	ND	83.9
7	IL	83.4	7	WV	59.9	7	TX	61.7	7	PA	83.6
8	VA	83.0	8	HI	59.8	8	VA	61.0	8	IA	82.8
9	FL	82.2	0	IL	57.0	9	WV	60.7	9	MN	82.6
10	CO	80.4	10	FL	59.0	10	AK	59.3	,	WI	02.0
	U.S.	80.2	11	TX	57.9	11	KY	59.2	11	IL	82.5
11	UT	79.4		U.S.	57.8	12	DC	58.5	12	SD	82.3
12	СТ	79.3	12	MO	57.4	13	MS	58.0	13	MD	81.8
	OK		13	VA	57.1	14	CO	57.9	14	CO	81.4
14	PA	78.3	14	OR	56.0	15	AZ	57.8	15	VT	80.9
15	VI	77.7	16	WY	<i></i>	16	CA	57.3	16	MA	79.4
10	NE	75.9	10	AK	55.7	1 /	KI NC	57.2	10	MI	70.0
10	MA	/5.8	1/		55.5	18	NU	57.0	18	UH	/8.8
18	OR NC	75.5	18	AZ DI	54.8	20	OK MA	56.6	20	KS VA	78.5
20	GA	75.5	19		54.4	20	MO	55.8	20	MO	78.0
20	KS	73.4	20	NE	53.8	21	I A	54.6	21	ID	/ 8.0
21	OH	75.3	20	NC	55.0	22	NM	54.1	22	WV	77.7
23	WI	74.8		CT		23	IA	54.0	24	CA	76.7
23	WA	72.9	23	WI	53.6		US	53.4		US	76.2
25	NM	72.3	25	OK	52.6	25	KS	53.0	25	ME	76.1
26	WV	70.8	26	ID	52.3	26	PA	52.6	26	MI	75.9
27	IN	69.8	27	MT	51.5	27	IL	51.8	27	TX	75.6
28	MI	69.4	28	IN	50.2	28	AL	49.9	28	OK	75.0
29	AZ	68.1	29	WA	50.1	29	OH	48.0	29	AR	74.9
30	AL	66.2	30	KS	49.0	20	NE	47.4	30	IN	74.6
31	AK	65.7	31	IA	47.6	30	ND	47.4	31	RI	74.3
32	HI	65.3	32	MS	47.5	32	FL	46.7	32	WV	72.6
33	MN	64.6	33	PA	47.3	33	WA	45.7	33	NC	71.7
34	MS	63.4	34	OH	46.8	34	WY	45.1	34	OR	71.1
35	WY	62.0	35	SD	44.2	35	WI	44.8	35	AK	70.5
36	DC	61.3	36	MA	44.0	36	GA	44.4	36	KY	70.3
37	ME	57.2	37	DE	41.4	37	MN	42.7	37	WA	70.0
38	RI	53.7	38	DC	39.7	38	IN	41.6	38	DE	69.0
	AR		39	GA	38.6	39	MI	35.2	39	NM	68.0
	DE		40	MI	37.0	40	DE	32.1	40	LA	66.2
	IA		41	AL	30.5		DE		41	FL MC	00.0
	ID MO		42	ND AD	33.2		ID ME		42	MS	65.0
	MT			ME			ME		43	AL UI	60.0
<b>n</b> 0	ND	<b>n</b> 0		MN					44	GA	50.9
11.a.	NH	11. <b>a</b> .		NH		na	NV	na	45	Δ7	59.2
	NV		na	NV	na	11.a.	NV	11.a.	0	NH	50.5
	NY		11. <b>u</b> .	NY	11. <b>u</b> .		SC			NV	
	SC			SC			SD		n.a	NY	na
	SD			TN			TN			SC	
	TN			VT			VT			TN	

### Table A.3 (corresponds to Table 2.4)Cumulative Promotion Index by Race and Ethnicity, FY 2004

Source: Editorial Projects in Education. Education Week's Diplomas Count.

						E	ducatio	n							
						We	eek/Urb	an							
			NCES	S Avera	ged	I	Institute	1							
			Fr	eshman		Ci	ımulativ	ve	Sta	te-AFG	R	St	ate-CPI	1	
State	-Renort	ed	Grad	ustion I	Rate	Pron	notion I	ndev	Perce	ntage P	oint	Perce	ntage P	oint	
Cradi	ation D	oto	Grau (	AFCD)	vare	1101		IUCA	Dif	foronco	*	Difference*			
Daul			<u>[</u> ]	State	0/	Derl		0/	Durk	State	0/	Darda	State	0/	
	State	<b>%</b> 0		State	<b>%</b> 0		State	<b>%</b> 0		State	<b>%</b> 0		State	<b>%</b> 0	
1	NU	95.0	1	NE	8/.8	1	NJ I A	83.3	1	NC MS	22.4	1	NM	30.9	
2	CT	91.3	2	I A	86.6	2	WI	80.5	2	NM	19.6	2	DE	28.0	
4	IA	90.7	4	VT	86.5	4		80.5	4	SC	17.0	4	MS	23.0	
5	KS	90.2	5	ND	86.3	5	VT	80.2	5	IN	16.7	5	SC	21.5	
6	MN	90.1	6	MN	85.9	6	NE	79.6	6	MI	14.7	6	NV	19.5	
7	IN	89.9	7	NJ	85.1	7	ND	79.2	7	NY	11.7	7	MI	17.2	
8	SD	89.1	8	AZ	84.7	8	UT	78.6	8	KS	11.0	8	IN	16.3	
9	WI	88.8	9	UT	84.4	0	СТ	78.1	9	DE	10.6	9	KS	15.9	
10	NE	88.0	10	PA	82.5	2	MN	/0.1	10	CA	10.4	10	TX	15.5	
11	MI	87.7	11	SD	82.3	11	ME	77.2	11	CT	10.3	11	CA	14.9	
12	PA	87.6	12	MT	81.5	12	NH	77.1	12	TX	10.0	12	RI	13.9	
13	IL	87.4	13	ID	81.0	13	IL	76.7	13	TN	9.4	13	SD	13.5	
14	ME	87.2	14	CT	80.9	14	ID MO	76.6	14	NV	9.1	14	CT	13.1	
16	VI ND	967	15	MU	80.6	15	MU	75.0	15	ME II	8.6	15	IN DC	12.5	
10	ID ID	80.7	10	NH	80.2	10	МТ	75.9	10	IL GA	8.0	10	DC Ш	12.3	
17	NH	86.6	17	VA	79.6	17	SD	75.6	17	OR	7.7	17	MN	12.2	
19	OH	86.2	19	IL.	79.4	19	MA	74.7	19	WV	7.0	19	OK	11.6	
20	MO	85.8	20	MD	79.3	20	KS	74.3	20	KY	6.9	20	WV	11.5	
	NM		21	KS	79.2		CO	74.0	21	SD	6.8		GA		
21	MS	05.0	22	MA	78.7	21	WY	74.2	22	RI	6.6	21	KY	11.3	
21	CA	85.0	23	ME	78.6	22	IN	726	23	NH	6.5		OR	1	
	RI		24	RI	78.4	23	MD	/3.0	24	NJ	6.2	24	MD	11.2	
25	MD	84.8	25	WV	77.3	25	AZ	73.3	25	OH	6.0	25	IL	10.7	
25	MT	04.0	26	OK	76.9	26	AR	73.2	26	AR	5.6	26	WA	10.5	
27	WV	84.3	27	CO	76.7	27	VA	72.9	20	ID	5.0	27	OH	10.3	
28	TX	84.0		WY	75.0	28	WV	72.8	28	MD	5.5	28	ID	10.0	
29	DE	83.7	29	KY	75.9	29	KY	71.5	20	OK	5.2	20	ME	0.5	
30		82.8	30	AK	/5./	30	KI OV	/1.1	30	MO	5.2	30	NH	9.5	
31	UT	82.4	31	WA	75.0	31	US	70.8	31	r A WV	3.1	31	MT	9.3	
33	OR	81.7	32	US	73.0	32	0.3. MI	70.0	33	HI	4.8	33	NY	9.1	
34	WY	81.5	33	CA	74.6	33	OR	70.4	34	FL	4.4	34	NE	8.4	
35	AR	81.3	34	OR	74.2	34	CA	70.1	35	WA	4.3	35	WI	8.3	
36	CO	80.1	35	TX	74.0	35	WA	68.8	36	MN	4.2	36	FL	8.2	
37	HI	79.6	36	IN	73.2	36	TX	68.5	37	IA	4.1	37	AR	8.1	
38	VA	79.5	37	DE	73.1	37	NY	68.0	38	CO	3.4	38	NJ	8.0	
39	WA	79.3	38	MI	73.0	38	AK	67.6	39	MT	3.3	39	IA	7.9	
40	TN	77.9	39	NC	72.6	39	HI	67.4	40	WI	2.1	40	ND	7.5	
41	SC	77.1	40	DC	68.8	40	NC	67.0	41	DC	1.1	41	WY	7.3	
42	NY	77.0	41	TN	68.5	41	TN	65.4	42	VT	0.7	42	PA	7.2	
45	AZ	/5.0	42	AL	65.9	42	MS	61.8	43	ND	0.4	45	VI VA	/.0	
44	DC GA	69.9	43	NM NV	65.2	43	AL	60.9	44	NE VA	0.2	44	VA	0.0 5.0	
45	FI	60.0	44	FI	64.6	-++ //5	DE	60.1	45	VA UT	-0.1	45	UT	3.7	
40	NV	64.9	45	AK	64.0	46	GA GA	58.1	40	AK	-2.3	40	A7	17	
48	AK	61.4	47	LA	63.9	47	DC	57.6	48	AZ	-9.7	48	AK	-6.2	
	U.S.		48	MS	63.3	48	SC	55.6		U.S.			U.S.		
	AL		49	GĂ	61.7	49	LA	54.7		AL			AL		
n.a.	LA	n.a.	50	SC	60.1	50	NM	54.1	n.a.	LA	n.a.	n.a.	LA	n.a.	
	MA		51	NV	55.8	51	NV	45.4		MA			MA	1	

### Table A.4 (corresponds to Table 2.5)State-reported Graduation Rates Compared to AFGR and CPI, FY 2005

Sources: Editorial Projects in Education. *Implementing Graduation Accountability* and *Education Week's Diplomas*; U.S. Dept. of Ed. Inst. Natl. *Public* 13-14.

	Total			Grade 9		G	rade 10		Grade 11			Grade 12		2
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%
1	AK	8.2	1	LA	8.1	1	NY	10.4	1	AK	8.6	1	AK	10.3
2	CO	7.8	2	DE	7.2	2	AK	7.6	2	CO	8.2	2	CO	10.0
3	LA	7.5	3	AK	6.7	3	CO	7.2	3	LA	6.7	3	AZ	9.5
4	AZ	6.2	4	CO	6.1	4	LA	6.6	4	AZ	6.4	4	UT	9.1
5	NV	5.8	5	GA	5.6	5	GA	5.9	5	GA	5.8	5	LA	8.3
6	NY	5.7	6	NC	5.2	6	NC	5.6	6	WY	5.6	6	NV	8.0
7	GA	5.6	7	NV	5.0	7	AZ	5 5	7	HI	5.4	7	HI	67
8	DE	5.3	8	MD	4.4	/	NV	5.5	/	NC	5.4	/	NH	0.7
9	NC	5.2	9	IL	4.3	9	DE	5.2	9	AR	5.3	9	WY	6.6
10	WY	4.8	10	NM	4.2	10	HI	4.7	10	NV	5.2	10	MN	6.4
11	HI	4.7	11	AZ	41	11	NM	4.5	11	WA	4.9	11	AR	6.3
12	IL	4 5		RI	1.1		RI		12	RI	4.8	12	CA	6.1
12	WA	1.5	13	MI	3.9	12	SD	44	13	NY	4.7	13	SD	5.8
14	SD	4.4	14	WA	3.8		WV			DE		14	WA	5.6
15	AR	4.3	15	OH	3.7		WY		14	IL	4.5	15	WI	5.4
16	NM	4.2	16	SC	3.6	16	IL	4.3		MO		16	TX	5.0
17	RI	4.1	17	SD	3.4	17	AR	4.0	1.0	WV			U.S.	4.9
17	WV	4.1		WV			MD	• •	18	NM	4.4	17	GA	4.9
	U.S.	3.9	19	FL	3.3	18	MO	3.9	19	NH	4.3	10		1.0
19	MD	3.9		U.S.	3.1		WA	2.0	20	MA	4.1	19	NY	4.8
21	MI	2.0	20	NY	3.1		U.S.	3.8		SD		20	MA	4.7
21	MA	3.8	22	OK MA	2.0	21		3.8	22		4.0	22	NU	1.6
22	MO	3.7	22	MA	3.0		MI		24	MI	2.0	22	OH	4.6
24	UI	2.6	23	HI	2.8	23	MA	3.7	24	UK	3.9	22	FL MO	4.2
24		3.0	24		2.7		OK SC			U.S. MT	3.8	23	MU	4.5
	ГL VV		23	1A MO	2.0	25		3.6	23	ID	3.8	26	ME	4.2
25	NU	2.5	26	MT	2.5				20	MD	3.1	20	IVIE VV	4.2
23	OH	5.5				27	TL MT	3.3	27	TY	3.6	27		4.1
	OK		28	KV	24	29		3.1		FI		28	MT	4.0
30	MT	3.4	20	MS	2.7	30	MS	3.0	29	OH	3.5	20	TN	ч.0
31	SC	33		AI		31	PA	2.9	31	PA	3.4		IN	
32	CA	3.1	31	VA	2.3	51	ID	2.)	51	ME	5.4	31	PA	3.9
33	ID	3.0	33	CA	2.1	32	VT	2.8	32	SC	3.3		MI	
34	PA	2.9	34	ID	1.9		NE			IN		33	NE	3.7
-	AL		35	TN	1.8	34	NH	2.6	34	TN	3.2	35	IA	3.6
35	ME	2.8	36	NE	1.7	36	ME	2.4	-	VT			DE	
	MS		37	PA	1.6	25	ND			AL			MD	
20	NE	2.7	38	WI	1.5	37	OH	2.3	37	MS	3.1	36	NM	3.5
38	TN	2.7	20	ME	1.4		CA			UT			OK	
40	VT	2.6	39	VT	1.4	20	IN	2.2	40	NE	3.0	40	VA	3.4
41	IN	2.5	41	IN	1.3	39	TN	2.2	41	MN	2.9	41	VT	3.2
41	VA	2.5	42	IA	1.2		VA		42	CA	2.7	42	AL	2.0
43	WI	2.4	42	KS	1.2	12	KS	2.0	43	KS	2.6	42	RI	5.0
44	IA	2.2	44	ND	1.1	43	UT	2.0	44	IA	2.3	44	KS	2.8
45	KS	2.1	45	UT	1.0	15	СТ	1.8	45	СТ	2.2	45	MS	2.7
46	ND	1.9	46	NH	0.9	45	MN	1.0	16	NJ	21	46	SC	2.4
	CT			CT		47	IA	1.7	40	VA	2.1	47	ND	2.1
	DC			DC		48	WI	1.1	48	ND	2.0	48	NJ	2.0
n.a.	MN	n.a.	n.a.	MN	n.a.		DC		49	WI	1.6	49	CT	1.9
	NJ			NJ		n.a.	NJ	n.a.	na	DC	na	na	DC	na
	OR			OR			OR		11.a.	OR	11.a.	11.a.	OR	11.a.

### Table A.5 (corresponds to Table 2.6)Annual High School Dropout Rate by Grade, FY 2005

Notes: Districts assign ungraded dropouts to the grade that most closely matches their ages. Ungraded student enrollments are prorated into grades based on graded enrollments to calculate denominators for dropout rates. A state's total dropout rate is included only if the state reports dropouts and membership for each grade. Source: U.S. Dept. of Ed. Inst. Natl. *Numbers and Rates* 7.

	FY 2003			FY 2004		FY 2005		
Rank	State	%	Rank	State	%	Rank	State	%
1	AZ	8.5	1	LA	7.9	1	AK	8.2
2	AK	7.6	2	AK	7.0	2	СО	7.8
3	LA	7.5	3	AZ	6.7	3	LA	7.5
4	WA	6.2	4	WA	6.5	4	AZ	6.2
5	NV	6.1	5	DE	6.1	5	NV	5.8
6	GA	5.8	6	NV	6.0	6	NY	5.7
7	IL	5.7	7	NY	5.6	7	GA	5.6
8	DE	5.5	8	CO	5.4	8	DE	5.3
8	NY	5.5	8	GA	5.4	9	NC	5.2
10	NC	5.2	10	IL	5.3	10	WY	4.8
11	HI	47	11	NM	5.2	11	HI	4.7
11	NM	7.7	11	NC	5.2	12	IL	4.5
13	AR	4.6	13	HI	4.8	12	WA	ч.5
14	MI	4.5	14	AR	4.7	14	SD	4.4
14	WY	ч.5	15	MI	4.6	15	AR	4.3
16	OR	4.4	15	WY	1.0	16	NM	4.2
17	OK	4.0	17	WV	4.3	17	RI	4.1
1,	RI	1.0	18	SD	4.2	17	WV	
	U.S.	3.9		U.S.	4.1		U.S.	3.9
19	ID	3.9	19	MD	4.1	19	MD	3.9
17	UT	5.9	20	OK	3.9	15	MI	5.9
21	MN	3.8	21	NH	3.8	21	MA	3.8
	NH			UT		22	MO	3.7
23	MS	3.7	23	MA	3.7		UT	
-	WV		24	TX	3.6	24	TX	3.6
	MD			FL			FL	
25	MT	3.6	25	MI	3.4	25	KY	
	TX			RI		25	NH	3.5
20	AL	2.5		SC			OH	
28	0	3.5		AL		20	OK MT	2.4
21		2.4		CA		30	MI	3.4
31	FL	3.4	29	KY MO	3.3	31	SC	3.3
				MU		32		3.1
32	MA	3.3		UH		24	ID DA	3.0
	NIO SD		25	ID	2.1	54	FA AI	2.9
			33	ID MS	5.1	35	ME	28
			36	DA	2.9	35	ME	2.0
36	SC	3.2		NE			NE	
	TN	1	38	VT	28	38	TN	2.7
40	NE	3.1	50	VA	2.0	40	VT	2.6
40	OH	5.1	41	ME	27	-10	IN	2.0
41	VA	3.0	42	INIL	2.7	41	VA	2.5
43	ME	2.8	43	KS	2.2	43	WI	2.4
44	KS	2.0	44	ND	2.0	44	IA	2.2
	IN	,		CT	2.0	45	KS	2.1
45	ND	2.2		DC		46	ND	1.9
47	СТ	2.1		IA			СТ	
48	WI	2.0	n.a.	MN	n.a.		DC	
49	IA	1.9		NJ		n.a.	MN	n.a.
50	NJ	1.8		OR			NJ	
n.a.	DC	n.a.		WI			OR	

Table A.6 (corresponds to Table 2.7)Annual High School Dropout Rate Trends, FY 2003 to FY 2005

Source: U.S. Dept. of Ed. Inst. Natl. Numbers and Rates 9.

#### Table A.7 (corresponds to Table 2.8) Percentage of Teenagers Age 16-19 Not Enrolled in School and Having No High School Credential, 2006

Rank	State	%
1	Louisiana	11.4%
2	Mississippi	10.4%
3	New Mexico	10.3%
4	Nevada	10.0%
5	Georgia. Arizona	9.1%
7	Colorado	9.0%
8	Alabama, Kentucky	8.8%
10	Montana	8.7%
11	West Virginia	8.3%
12	Florida	7.9%
13	South Carolina, Indiana	7.6%
15	Oklahoma	7.5%
16	Texas	7.4%
17	District of Columbia, North Carolina	7.3%
19	Idaho	7.2%
21	Alaska	7.0%
22	Oregon, Delaware, Wyoming	6.9%
25	Rhode Island, South Dakota	6.7%
	United States	6.6%
26	California, Maryland	6.4%
28	Washington	6.3%
29	Arkansas, Tennessee, Missouri	6.1%
32	Michigan, Hawaii	6.0%
34	Utah	5.9%
35	Pennsylvania	5.7%
36	New York	5.6%
37	Illinois, Nebraska	5.5%
39	Ohio	5.4%
40	Virginia	5.2%
41	Wisconsin	4.9%
42	New Jersey	4.7%
43	Iowa	4.5%
44	Kansas	4.4%
45	Maine	4.2%
46	Minnesota, Massachusetts, New Hampshire	4.1%
49	Vermont	4.0%
50	Connecticut	3.9%
51	North Dakota	3.5%

Source: U.S. Dept. of Commerce. Census. "American Community Survey."

	Grade 4		Grade 8					
State	%	Sig.	State	%	Sig.			
MA	49		MA	43	U U			
NJ	43		VT	42				
СТ			MT	20				
NH	41	>	NJ	39				
VT			СТ					
PA	40		ME					
MT	39		MN	37				
VA	38		NH					
MN	37		SD					
СО			IA					
IA			OH	36				
KS			PA		~			
MD			СО		/			
ME	26		KS	35				
NY	50		NE					
OH			OR					
WA			VA	34				
WI			WA					
WY			MD					
ID			WI	33				
ND	35	=	WY					
NE			ID					
DE			ND	32				
FL	34		NY					
SD	54		DE					
UT			IN	31				
IN	33		MO					
KY	55		IL	30				
IL			UT	20				
MI	32		U.S.	29				
MO			FL					
U.S.	32		KY					
RI	31		MI	28				
	30		NC	-	=			
AL	_		1X					
AK	- 29		AK	27				
AR	_		RI					
NC			GA	24				
GA	20			26				
OR	28							
W V OV			AR	25				
	27		SC AZ	24				
		<	AZ	24				
HI SC	26		W V NV	23				
<u> </u>				22				
AL NM	- 24		AL	21				
NV	24		HI	20	<			
CA	23		LA	19	ì			
LA	20		MS	1)				
MS	19		NM	17				
DC	14		DC	12				
20		1	50					

### Table A.8 (corresponds to Table 2.11)Percentage of Students At or Above Proficient on NAEP Reading Assessment, 2007

Note: > indicates states with achievement levels significantly better than Kentucky's, = indicates states that are not significantly different, and < indicates states that are significantly worse than Kentucky, based on statistical testing with a 95% significance level. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.

150

	Grade 4		Grade 8							
	2002			2002			2007			
State	%	Sig.	State	%	Sig.	State	%	Sig.		
СТ	49	- 8	СТ	45		NJ	56	8		
MA	44		MA	42		CT	53			
NY	37	>	VT	41		MA	46			
DE	35		OH	38		VT	40			
FL	33		ME	36		NH	39			
NC			DE	35		CO	38			
VT	32		MD	50		ME	50			
ME			NC	34		IL	37			
KI WA	20		WA	22	>	F	26			
WA MD	50		- OK FI	33		PA	30			
MD VA			FL KS			W I W/A	35	_		
MN			NE	32		WA	35	-		
TX	29		PA	52		DE	34			
PA		TX		KS						
OH	28	=	VA	31		MT	33			
IA			U.S.	30		RI				
KY	27		NY	30		MN	22			
NE			RI			OH	32			
U.S.	27		MT	29		IA				
IN	26		ID			U.S.	31			
GA	-		WY	28		NY				
WY	23		OK	27		VA	31			
TN			MO	26		UT* TN				
CA				26	=		30			
UR ID	-		GA KV	25		ID				
HI	22	<	MI		·	GA	29			
MO			TN	24		NC	2)			
MT*		=	ND			AR				
KS	21		CA	22		MI	27			
UT	20		UT	23		ND				
ND	20		WV	21		KY		—		
MI			SC			MO	26			
WV	19		AZ	20		OK	20			
AR			AL	10		TX				
NV	18		AR	19		CA	25			
NM	17	<	NM LA	10	<	AL	24			
	1/		LA HI	18		AL	23			
	10		NV	16		WV	22			
AZ	15		MS	13		NV	21			
LA	14		DC	10		HI	20			
MS	13		AK	10		LA		<		
DC	11		СО	1		NM	17			
AK			IA	]		MS	15			
CO			IL			AK				
IL			MN	n.a.	n.a.	DC				
NH	n.a.	n.a.	NH			MD	n.a.	n.a.		
NJ			NJ			NE		<i>u</i> .		
SD	4		SD	4		OR				
WI			WI			SD				

### Table A.9 (corresponds to Table 2.13)Percentage of Students At or Above Proficient, NAEP Writing, 2002 and 2007

Notes: \*Statistical significance can differ for states with the same average score because statistical tests use unrounded percentages and take into account each state's sample size and variation in scores. Grade 4 writing was not assessed in 2007. > indicates states scoring significantly better than Kentucky, = indicates states that are not significantly different, and < indicates states scoring significantly worse than Kentucky, based on statistical testing with a 95% significance level.

Source: U.S. Dept. of Ed. Inst. Natl. NAEP Data.

	Grad	e 4	Grade 8						
State	%	Sig.	State	%	Sig.				
MA	58	8	MA	51	8				
NH			MN	43					
NJ	52		ND		1				
KS			VT	41					
MN	51		KS						
VT	/19	-	NI	40					
	ر <del>ب</del>	-	SD	30	-				
WI	47		MT	39	-				
IN		-	NH	38					
	16			30					
OU	40		FA CO		4				
ОП	4.5	4							
	45	4	MD	37					
MI			VA	-					
WA	44		WI		-				
WY		4	WA	36					
IA	43		WY	50	>				
NY	15		CT						
ME	42		IA						
VA	72		IN						
CO			NE	35					
NC	41		OH						
SD			OR						
DE			TX						
FL			ID						
ID	40		ME	34					
MD			NC	-					
TX			AK						
US	39		SC	32					
UT	39		UT						
AK		1	US	31					
MO	38		DE	51					
NE	50		II	31					
AP		-	MO		-				
M	37		NV	30					
		4	MI	20	-				
IL SC	36		DI	29	4				
	25		KI FI	28	=				
OK DI	35	4	FL	27					
KI	34	4	KY AZ	24	4				
HI			AZ	26	-				
OK	33		GA	25	-				
WV		4	AR	24					
GA	32	=	CA		4				
AZ	31		NV	23					
KY	51		TN		4				
CA	30		HI	21					
NV	50		OK	<i>2</i> 1	]				
TN	29		LA	10	<				
AL	26		WV	17	]				
LA	24		AL	18	]				
NM	24	<	NM	17	]				
MS	21	1	MS	14	1				
DC	1.4	1	DC	0	1				

### Table A.10 (corresponds to Table 2.15)Percentage of Students At or Above Proficient, NAEP Math, 2007

Notes: > indicates states scoring significantly better than Kentucky, = indicates states that are not significantly different, and < indicates states scoring significantly worse than Kentucky, based on statistical testing with a 95% significance level.

Source: U.S. Dept. of Ed. Inst. Natl. NAEP Data.

	2000							2005						
	Grad	e 4		Grade	8		Grad	e 4	Grade 8					
State	%	Sig.	State	%	Sig.	State	%	Sig.	State	%	Sig.			
MA	42		MT	44		VA	40		ND	43				
VT	38		MN	41		MA	20		MT	42				
ME	37		MA			VT	38		MA					
IA			OH	39		MT	27		NH	41				
MT	36	>	VT			NH	57		SD	41				
ND			ND	38		KY			VT		>			
CT	35		NE	50		ME	36		MN	39	-			
MN	34		ID	37	>	MO	50		WI					
MO			CT			ND		=	WY	37				
IN			ME	35		OH			ID	36				
MI	32		MI			SD	35		MI					
VA II			UT	24		WI CT			OH CO*	35				
			WV	54		UI MN	22		VA*		=			
	31		IN				33		ME*	34	>			
WY			MO	33		CO			CT	54	-			
ID	29		IL			WY	32		MO					
KY	28	=	VA	29		NJ*			NJ	33				
OR	27		U.S.	29		MI	30		UT					
NE	20		KY	20		ID	29		WA					
OK	20		NY	28	_	WA	28		OR	32	=			
U.S.	26		MD	27	_	DE			KY	31				
RI	25		RI	21		IL	27		DE	20				
MD			NC	25		IN	27		IN	2)				
NY	24		OK	20		MD			IL	27				
TN			TN	24		U.S.	27		U.S.	27	<			
WV	24		WV	24	-	FL			MD	26				
AK			AL	-		OK TN	26		KI CA					
UA NC	23		AZ GA	23		GA			OK	25				
TX			TX			NC			TN	25				
AL			AR			OK	25	<	AR					
AZ	22		NV	22	<	SC			SC					
SC	20	<	NM	20		TX			TX	23				
NV	19		SC	20		AR	24		WV					
LA	18		LA	18		WV	24		NC	22	<			
NM	17		MS	15		RI	23		FL	21				
HI	16		CA	14		AL	21		AZ	20				
CA	13		HI			LA	20		AL	10				
MS			AK			HI	19		LA	19				
AK				1		AZ NM	18							
DC			DE	1			<u> </u>		NM NM	18				
DE			FL	1		NV	17		HI	15				
FL	1		IA	1		MS	12		MS	14				
KS	n.a.	n.a.	KS	n.a.	n.a.	AK			AK					
NH			NH	1		DC	1		DC					
NJ	]		NJ	]		IA			IA		nc			
PA			PA	K	KS	n.a.	n.a.	KS	n.a.	11.a.				
SD			SD			NY	]		NY					
WA			WA	I	1	PA	1		PA					

Table A.11 (corresponds to Table 2.17)Percentage of Students At or Above Proficient, NAEP Science, 2005

Notes: \*Statistical significance can differ for states with the same average score because statistical tests use unrounded percentages and take into account each state's sample size and variation in scores. > indicates states scoring significantly better than Kentucky, = indicates states that are not significantly different, and < indicates states scoring significantly worse than Kentucky, based on statistical testing with a 95% significance level. Source: U.S. Dept. of Ed. Inst. Natl. *NAEP Data*.

Participation		on	Composite			English			Math			Reading			Science		
Rank	State	%	Rank	State	Score	Rank	State	Score	Rank	State	Score	Rank	State	Score	Rank	State	Score
1	СО	100	1	MA	23.5	1	MA	23.5	1	MA	23.6	1	MA	23.9	1	NY	22.7
1	IL	100	2	CT	23.2	2	CT	23.2	2	CT	23.2	2	WA	23.7	2	MA	22.6
2	MS	06	3	WA	23.1	2	NH	22.7	3	NY	23.1	3	СТ	23.6	2	WA	22.0
3	TN	90	4	NH	22.0	3	WA	22.1	4	WA	23.0	4	NH	23.3	4	MN	22.5
5	ND	82	-	NY	22.)	5	VT	22.6	5	HI	22.9	-	VT	25.5	5	CT	22.4
6	AL	81	6	VT	22.8	6	ME	22.4	6	NH	22.7	6	NY	23.1	5	WI	22.1
7	LA	79	7	ME	22.5	7	NY	22.0	7	CA	22.6	7	ME	22.9	7	IA	22.3
8	W Y VV	/8		MN		8	NJ	21.9	0	MIN	22.5	8	MIN	22.8	0	V I NILI	22.2
9	NE	77	0		22.3	9	NE	21.8	0	NJ VT	22.3	9	IA	22.0	9	IND HI	22.2
	KS		7	WI	22.5		CA			ME			MT		10	NE	21.9
11	SD	76	12	NI	22.2		HI		11	WI	22.2	10	OR	22.5	10	SD	21.7
13	AR	75	12	CA	22.2	11	IA	21.6	13	OR	22.1		RI			ME	
14	MO	74	13	NE	22.1		RI		14	IN	22.0		KS		13	MT	21.8
15	OK	71		IN			WI			AZ			NE			OR	
	MI		15	OR	22.0		IN		15	IA	21.9	14	NJ	22.4		IN	
16	MN	70		PA		16	MO	21.5		PA			PA		16	KS	21.7
10	UT	70		KS			PA		18	NE	21.8		WI			MI	
	WI		18	MT	21.9	19	KS	21.4	19	MT	21.7		AZ			ND	
20	OH	68		SD		•	MD		.,	SD	21.7	10	CA		19	OH	21.6
21	IA	66	21	AZ	21.8	20	SD	21.3	21	DE	21.6	19	HI	22.2		UT	
22	W V NIM	60		RI			DE			KS			UI WV		22	MO	21.5
23	ID	60	23	DE	21.7	22	DE	21.2	22	MD	21.5		W Y		22	INJ DA	21.5
24	MT	59		MD		25	OR	21.2	25	RI	21.3		MD				
2.6	FL	54		MO		26	AZ	21.1		NV		24	MO	22.1	25	DE	21.4
27	SC	43	25	ND	21.6	20	OH	21.1	26	NC	21.4		SD		20	WY	2
	U.S.	42		OH		27	VA	21.0		AK		20	NV	22.0	28	ID	21.3
28	GA	34		MI			NV		28	MI	21.3	28	OH	22.0		CA	
29	DC	31	29	NV	21.5	20	ND	20.8		OH		20	DE	21.0	20	MD	21.2
30	TX	30		WY		29	TN	20.8	31	ID	21.2	30	ND	21.9	29	NV	21.2
31	NV	29	32	ID	21.4		WV		51	VA	21.2	32	AK	21.8		RI	
32	AK	27	52	VA	21.1		U.S.	20.7	33	UT	21.1	52	MI	21.0	33	VA	21.1
33	VT	22		U.S.	21.2		ID			WY	21.0	34	VA	21.7		U.S.	21.0
34	IN	21	34	AK	21.2	33	MI	20.7		U.S.	21.0		U.S.	21.5	34	AK	21.0
26		20	35	NC VV	21.0				35	TV	21.0	35	NC OK	21.4	35	NC VV	20.7
30	47	20	36	OK	20.7	36	OK	20.5	30	II	20.8	30	KY	21.3	30	OK	20.0
37	OR	18	50	TN	20.7		AL		38	GA	20.3	37	WV	21.2	37	WV	20.5
5,	VA	10	39	WV	20.6	38	KY	20.3	39	CO	20.1	39	TN	21.1		CO	
	СТ			AR			LA		40	FL	20.0	40	AR	20.0	20	IL	20.4
40	NC	16	40	IL	20.5	4.1	IL	20.2	40	KY	20.0	40	NM	20.9	39	TN	20.4
	WA			TX		41	NC	20.2	42	AR	10.0	42	CO	20.8		TX	
	CA		43	CO	20.4	43	AK	20.1	42	TN	19.9	43	AL	20.7	13	AR	20.2
43	MA	15	44	AL	20.3	44	GA	19.9	44	OK	19.8	44	GA	20.6	CT	NM	20.2
	NH			GA	20.5	45	CO	19.7		SC	17.0		TX	20.0	45	AL	20.1
46	MD	14	46	NM	20.2	46	NM	19.6	46	NM	19.7	46	FL	20.5	47	GA	10.0
47	ME	11	47	LA	20.1	47	TX	19.5	47	AL	10.5	40	IL TA	20.2	47	LA	19.9
4/	NJ DA	11	48	FL	19.9	48	FL MS	19.1	4/	LA	19.5	48	LA	20.2	48	FL	19.5
	PA DE		49 50	MS	19.0	49	NIS SC	19.0	50		18.8	49 50	DC	19.8	50	MS	187
50	RI	9	51	DC	18.7	51	DC	18.1	51	MS	18.0	51	MS	19.1	51	DC	18.3

# Table A.12 (corresponds to Table 2.21)ACT Participation Rates and Average Scores for High School Graduates Tested, 2007

Note: Because participation rates are negatively correlated with average scores, use caution when comparing across states.

Source: ACT. 2007 Average.

# Table A.13 (corresponds to Table 2.22)Percentage of High School Graduates Participating in SAT and Average SAT Scores, 2007

Participation in SAT		Critical Reading				Math		Writing			
Rank	State	%	Rank	State	Score	Rank	State	Score	Rank	State	Score
1	ME	100%	1	IA	608	1	IA	613	1	IL	588
2	NY	89%	2	MN	596	2	IL	611	2	MO	587
3	MA	85%	2	IL	504	3	MN	603	3	IA	586
4	CT	84%	3	MO	594	4	SD	602	4	MN	577
5	NH	83%	5	SD	589	5	WI	598	5	WI	575
6	NJ	82%	6	WI	587	6	ND	596	6	KS	569
7	DC	78%	7	ND	584	7	MO	594	7	TN	568
8	PA	75%	8	KS	583	8	KS	590	8	SD	567
9	VA	73%	9	NE	579	9	NE	585	9	AR	565
10	DE	72%	10	OK	578	10	MI	579	10	LA	563
11	NC	71%	10	AR	570	11	WY	571	11	NE	562
12	MD	70%	12	TN	574		OK	0,11		ND	0.02
13	GA	69%	13	LA	569	13	TN	569	13	MS	560
14	RI	68%	14	MI	568	14	LA	567	14	OK	559
15	VT	67%		MS		15	AR	566	15	AL	554
16	FL	65%	16	KY	567	16	CO	565	16	KY	553
17	IN	62%	17	WY	565		KY		10	MI	540
10	SC	(10/	18	AL	563	18	AL	556	18		549
19	HI	61%	19	LIT	560	20	UI	540	19	W Y LIT	544
20	WA	520/	20	UI NM	555	20	IMS NM	549	21	UI NM	540
21	WA TV	520/	21	INM	541	21	MT	540	21	MT	340
22		32%	22	ID MT	538	22	OH	545	22	OH	522
23		49/0	23		536	23	ID	520	24	ID	510
24	IIS	48%	25	WA	526	24	WA	531	24	NH	512
25	NV	41%	25	OR	522	26	OR	526	25	MA	512
26	AZ	32%	20	NH	521	20	AZ	525	26	CT	511
27	MT	28%		AK		28	MA	522	28	WA	510
28	OH	27%	28	AZ	519	29	NH	521	29	VT	508
29	CO	24%	20	VT	51.6	30	VT	518	30	WV	505
30	WV	20%	30	WV	516	31	AK	517	21	OR	502
31	ID	19%	32	MA	513	32	CA	516	31	AZ	502
32	TN	13%	33	VA	511		U.S.	515	22	VA	40.9
33	NM	12%	34	CT	510	33	CT	512	33	CA	498
34	KY	10%		U.S.	502	34	VA	511	35	MD	496
	AL		35	MD	500	35	NJ	510	36	NJ	494
35	MI	9%	55	NV	500	36	NC	509		U.S.	494
	MN		37	CA	499		IN		37	RI	492
2.2	IL	0.01	2.2	DE	10-	37	TX	507	38	AK	491
38	KS	8%	38	FL	497		WV		39	DE	486
41	WY	70/	41	IN	40.6	40	HI	506	40	GA	483
41	LA	1%	41	RI	496	40	NV	505		IN	
	MO		42	NJ	495	42	NY	505		NY	
42	NE	(0/	4.4	NC CA	40.4	43	MD DA	502	42	PA	482
42		0%	44	UA DA	494	44	PA DI	499		NU TV	
			45	rA TV	495	43	DE	498	16	1 X NV	180
17		50/	40	1 A NV	492	16	FI	106	40	EI	400
4/		570	47	SC	491	40	SC SC	470	47	SC SC	475 475
48	MS	4%	49	HI	484	49	GA	495	40	HI	473
10	ND	170	50	DC	478	50	ME	465	50	DC	471
51	SD	3%	51	ME	466	51	DC	462	51	ME	457
÷ •	~~	- / 0									,

Note: Average scores are negatively correlated with participation rates. The College Board strongly discourages the comparison or ranking of states on the basis of SAT scores alone. The denominators for participation rates are Western Interstate Commission for Higher Education estimates of high school graduates.

Source: College Board. *College-Bound Seniors 2007* Table 3. Copyright (c) 2007-2008 The College Board, www.collegeboard.com. Reproduced with permission.

	Hi	gh School	Class of 2	000		High School Class of 2007							
Took	an AP E	xam in	Score	ed 3 or F	ligher	Took	an AP E	xam in	Score	ed 3 or F	ligher		
Н	ligh Sch	ool			8 -	H	ligh Sch	ool			8		
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%		
1	NY	27.3	1	NY	17.9	1	DC	39.7	1	NY	23.4		
2	VA	25.0	2	UT	17.4	2	FL	38.0	2	MD	22.4		
3	UT	24.5	3	VA	15.9	3	NY	35.5	3	VA	21.5		
4	FL	22.7	4	CA	15.0	4	MD	35.3	4	FL	20.3		
5	CA	22.2	5	MA	14.5	5	VA	34.4	7	MA	20.5		
6	MD	20.2	6	MD	14.1	6	AR	32.2	6	CT	20.1		
7	NC	19.7	7	CT	13.6	7	NC	31.9	7	VT	19.9		
8	MA	19.6	8	FL	13.5	8	CO	30.8	8	CA	19.7		
9	CT	19.1	9	NJ	12.9	9	CA	30.1	9	UT	19.5		
10	CO	18.6	10	CO	12.2	10	VT	29.1	10	CO	19.2		
11	NJ	17.9	11	VT	11.5	11	UT	28.9	11	NC	18.5		
12	SC	17.7	12	NC	11.3	12	ME	28.8	12	ME	17.8		
13	DC	17.3	13	WI	10.5	13	GA	28.6	13	NJ	17.1		
14	GA	17.2		U.S.	10.2	14	MA	28.1	14	WI	16.5		
15	TX	16.6	14	ME	10.1	15	CT	27.7	15	GA	15.3		
10	VT	10.0		AK	10.1	16	DE	27.4	10	NH	10.0		
	U.S.	15.9	16	SC	10.0	17	TX	27.3		U.S.	15.2		
17	AK	15.4	17	IL	99		U.S.	24.9	17	IL	14.9		
18	WI	15.2		TX	7.7	18	WA	24.3	18	WA	14.7		
19	NV	15.1	19	GA	9.7	19	NV	24.0	19	DE	14.5		
20	ME	14.8	20	NH	9.2	20	WI	23.9		TX	1 1.0		
21	MI	13.9	21	NV	9.1	21	NJ	23.7	21	MN	13.4		
22	IL	13.4	22	MI	8.8	22	SC	22.7	22	NV	13.3		
	MN		23	PA	8.3	23	IL	22.0		SC			
24	DE	13.3	24	MN	8.1	24	MN	21.6	24	AK	12.8		
26	NH	10.1	25	DE	7.6	25	NH	21.3	24	MI	11.0		
26	PA	12.4		WA		26	AK	20.0	26	OR	11.9		
27	IN	11.9	27	AZ	7.2	27	OK	19.8	27	PA	11.7		
28	WA	11.5	28	OH	7.1	28	MI	19.7	28	OH	11.0		
29	AZ	11.3	20	OK DI	( )	29		19.6	29	MI	10.6		
21	OH	11.1	30	KI MT	6.9	30	NM OD	19.2	30	ID AZ	10.1		
31	NM DI	11.1	31	MI DC	0.8	22	OK DI	19.1	31	AZ	10.0		
32		10.7	32	DC ID	6.5	32	IIN TN	19.0		IN			
33		10.6	24	TN	6.3	24	OU	18.0	22	IN	0.7		
35	OP	10.5	34	1 IN NM	6.1	34	РА	18.0	33	SD SD	9.7		
36	TN	10.3	36	INIVI	6.0	35		16.8	36		9.6		
37	MT	10.4	30	SD	5.0	37	HI	16.3	37	OK	9.0		
51	ID	10.1	38	HI	5.9	38	MT	15.5	38	NM	9.5		
38	SD	9.6	30	KV	5.5	30	ID	15.0	30	RI	87		
40	OK	9.5	40	OK	5.0	40	WV	15.0	40	WV	8.4		
40	WV	8.4	40	IA	4.9	40	SD	15.7	40	HI	83		
42	AR	8.1	42	WV	4.6	42	WV	15.5	42	DC	8.1		
43	AI	7.2	72	KS	т.0	43	RI	14.0	43	IA	8.0		
44	KS	7.0	43	ND	4.4	44	KS	12.5	44	KS	7.8		
45	IA	6.9	45	AR	43	45	IA	12.5	45	ND	7.4		
46	WY	6.1	46	AL	3.9	46	MS	11.5	46	WV	7.0		
47	ND	5.9	47	WY	3.8	47	AL	11.4	47	MO	6.7		
48	MS	5.6	48	MO	3.7	48	MO	10.6	48	AL	6.4		
49	MO	5.5	49	NE	3.2	49	ND	10.5	49	NE	5.9		
50	NE	5.0	50	MS	2.3	50	NE	10.0	50	MS	3.7		
51	LA	3.2	51	LA	19	51	LA	57	51	LA	2.7		

# Table A.14 (corresponds to Table 2.23)Advanced Placement Exam, 2000 and 2007

Source: College Board. Advanced Placement Report to the Nation 2005 48 and 2008 52. Copyright (c) 2008 The College Board, www.collegeboard.com. Reproduced with permission.

Rank	State	Total Score	Grade
1	MA	95.2	р
1	MD	63.2	D
3	NJ	80.6	B-
4	VT	79.2	C+
5	PA	77.2	C+
6	VA	76.2	С
7	FL	75.2	С
8	MN	74.3	С
9	NH	73.9	С
10	ME	73.6	С
11	ND	73.4	С
12	KS	73.3	С
13	TX	72.6	С
14	OH	72.4	C-
15	MT	72.3	C-
16	DE	72.2	C-
17	ID	72.1	C-
18	NY	71.9	C-
10	СО	71.5	C
19	IA	/1.5	C-
21	WI	71.4	C-
22	WA	70.9	C-
23	SD	70.5	C-
24	WY	70.4	C-
25	IN	70.0	C-
26	UT	69.9	C-
	U.S.	69.4	D+
27	IL	69.1	D+
28	GA	68.1	D+
29	TN	67.0	D+
30	AK	66.9	D+
21	NE	667	D
51	NC	00./	$D^+$
33	KY	66.5	D+
34	AR	66.3	D
35	СТ	66.2	D
36	OK	66.0	D
37	HI	65.1	D
30	CA	64.7	n
38	MO	04./	D
40	OR	64.5	n
40	SC	04.5	D
42	RI	63.8	D
43	MI	63.5	D
44	AZ	62.6	D
45	NV	62.2	D-
16	NM	(0.2	D
46	LA	60.3	D-
48	AL	59.1	F
49	WV	58.1	F
50	DC	57.7	F
51	MS	55.9	F

# Table A.15 (corresponds to Table 2.24)Education Week's Quality Counts Achievement Index, 2008

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.



### Figure A.A (corresponds to Figure 3.A) Total P-12 Student Enrollment (in thousands), FY 2008

Source: U.S. Dept. of Ed. Inst. Natl. Projections 44-45.

	2000-2008		2008-2016					
Rank	State	%	Rank	State	%			
1	NV	30.8	1	NV	22.6			
2	AZ	28.6	2	TX	18.6			
3	UT	19.1	3	AZ	18.3			
4	TX	17.1	4	ID	17.1			
5	GA	16.3	5	UT	16.4			
6	FL	14.7	6	FL	16.1			
7	NC	13.2	7	GA	15.3			
8	ID	12.2	8	HI	13.8			
9	СО	10.3	9	NC	11.9			
10	VA	8.9	10	AK	11.5			
11	DE	8.7	11	СО	10.4			
12	NJ	7.8	12	VA	9.1			
13	TN	6.8	13	TN	8.9			
14	AR	6.2	14	DE	8.8			
15	SC	6.1	15	MN	8.3			
	U.S.	5.5	16	NE	7.9			
16	IN	5.5	17	MT	7.6			
17	CA	5.2	18	AR	7.3			
18	DC	4.3	19	OR	7.2			
19	KY	3.9	20	WY	7.1			
20	IL	3.8		U.S.	7.0			
21	OK	2.7	21	OK	6.4			
22	HI	2.2	22	MD	6.1			
23	NE	2.1	23	IA	5.9			
24	NM	1.6	24	NM	5.8			
25	MD	1.4	25	KS	5.7			
26	СТ	1.2	26	MO	5.1			
27	OR	1.1	27	CA	4.9			
28	MS	0.8	28	WA	4.8			
29	MO	0.2	29	SC	4.7			
30	WA	0.1	30	NH	4.5			
31	MI	0.0	31	LA	4.1			
32	KS	-0.2	32	KY	3.9			
33	PA	-0.6		SD				
34	OH	-0.7	33	WI	3.3			
35	AL	-0.9	25	DC	2.0			
36	MN	-1.1	35	NJ	2.8			
37	IA	-1.2	37	ME	2.1			
38	WV	-1.4	38	MS	1.8			
20	AK	1.5	39	IN	1.5			
39	MA	-1.5	40	WV	0.7			
41	LA	-2.3	40	IL	0.7			
42	NH	-2.9	42	AL	0.5			
43	NY	-3.2	43	PA	0.4			
44	WI	-4.3	44	OH	0.3			
45	RI	-4.5	45	VT	0.0			
46	WY	-6.7	46	MA	-0.5			
47	SD	-7.0	47	MI	-0.6			
48	MT	-7.1	48	СТ	-0.9			
49	ME	-8.2	49	NY	-1.2			
50	VT	-9.8	50	ND	-3.2			
51	ND	-13.8	51	RI	-3.3			

# Table A.16 (corresponds to Table 3.1)Projected Percent Changes in P-12 Enrollment, Fall 2000–Fall 2016

Source: U.S. Dept. of Ed. Inst. Natl. Projections 44-45.

	3-Year-Olds		4-Year-Olds					
Rank	State	%	Rank	State	%			
1	IL	18.5	1	OK	68.4			
2	VT	15.6	2	FL	56.7			
3	NJ	15.1	3	GA	53.3			
4	AR	10.8	4	WV	45.8			
5	KY	10.7	5	TX	45.2			
6	MA	9.2	6	VT	44.9			
7	WV	5.1	7	SC	37.8			
8	CA	5.0	8	WI	36.1			
9	CT	4.5	9	NY	34.6			
10	TX	4.4	10	MD	34.0			
	U.S.	3.2	11	KY	29.3			
11	СО	3.1	12	IL	26.7			
12	OR	2.6	13	NJ	25.3			
12	PA	2.2	14	LA	24.4			
15	MO	2.2		U.S.	21.8			
15	NE	2.0	15	AR	21.4			
16	WA	1.4	16	MI	16.9			
17	IA	1.4	17	ME	16.3			
18	OH	13		СТ				
10	MN	1.5	18	KS	15.6			
20	MD	1.2		TN				
21	TN	1.0	21	NC	14.8			
22	NM	0.9	22	CO	14.6			
23	WI	0.8	23	VA	12.5			
24	SC	0.6	24	CA	10.8			
25	NY	0.5	25	MA	10.3			
26	NV	0.4	26	NM	8.9			
	OK		27	DE	7.6			
	FL		28	PA	7.2			
	GA		29	WA	5.8			
	LA		30	AZ	5.5			
	MI		31	OR MO	4.8			
27	NIE VS	0.0	32	MO	4.5			
	NC		33	IA	4.0			
	VA		35	OH	3.4			
	DE		36	NV	2.2			
	AZ		37	MN	1.9			
	AL		38	AL	1.9			
	AK		50	AK	1.0			
	HI			HI				
	ID			ID				
	IN			IN				
	MS			MS				
	MT			MT				
n.a.	NH	No program	n.a.	NH	No program			
	ND			ND				
	RI			RI				
	SD			SD				
	UT			UT				
	WY			WY				

### Table A.17 (corresponds to Table 3.2) Enrollment of 3- and 4-Year-Olds in State-Funded Preschool, 2007

Note: Nationwide, an additional 17,440 children of other ages are enrolled in state prekindergarten, for a total enrollment of 1,026,037.

Source: Natl. Inst. The State 15.

American		Asian/Pacific									White, Non-				
Indian/	Alaska N	Native	I	slander		F	lispanic	;	Black,	Non-Hi	spanic	E	lispanic	c	
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	
1	AK	26.6	1	HI	72.8	1	NM	54.0	1	DC	83.3	1	VT	95.5	
2	OK	18.9	2	CA	11.7	2	CA	48.5	2	MS	51.2	2	ME	95.1	
3	MT	11.3	3	WA	8.1	3	TX	45.3	3	LA	44.4	3	WV	93.6	
4	NM	11.1	4	NJ	7.5	4	AZ	39.0	4	SC	40.3	4	NH	93.3	
5	SD	10.5	5	NV	7.3	5	NV	33.6	5	GA	39.2	5	ND	87.2	
6	ND	8.6		AK		6	СО	27.1	6	MD	38.1	6	IA	86.6	
7	AZ	6.2	6	NY	6.9	7	FL	23.9	7	AL	36.0	7	KY	86.3	
8	WY	3.5	8	MN	5.7	8	NY	20.1	8	DE	32.5	8	SD	85.0	
9	WA	2.7		MD			U.S.	19.8	9	NC	31.5	9	WY	84.9	
10	OR	2.4	9	VA	5.2	9	IL	19.0	10	VA	27.0	10	MT	84.3	
11	MN	2.1	11	OR	4.9	10	NJ	18.2	11	TN	25.1	11	ID	83.0	
12	NE	1.7	12	MA	4.7	11	RI	17.3	12	FL	23.9	12	UT	81.8	
10	ID			U.S.	4.6	12	OR	15.9	13	AR	23.0	13	IN	80.3	
13	NV	1.6	13	IL	3.8	13	СТ	15.4	14	IL	20.6	14	OH	79.0	
	KS			СТ	2.6	14	WA	13.6	15	MI	20.3	15	MN	78.3	
15	UT	1.5	14	WI	3.6	15	MA	13.1	16	NY	19.8	16	WI	77.8	
	WI		16	CO	3.3	16	ID	12.8	17	MO	18.2	17	NE	77.5	
18	NC	1.4		RI		17	UT	12.3	18	NJ	17.6	18	MO	76.6	
	U.S.	1.2	17	TX	3.1	18	KS	12.1		U.S.	17.2	19	KS	75.4	
19	CO	1.2		UT		19	NE	11.5	19	OH	17.1	20	PA	74.8	
20	MI	1.0	2.0	DE	• •	20	DC	10.6	20	PA	16.2	21	OR	73.6	
	AL		20	GA	2.8	21	DE	9.2	21	TX	14.7	22	MA	73.5	
21	CA	0.8	22	AZ	2.5	22	WY	9.0	22	СТ	13.7	23	MI	71.9	
	LA		22	PA	2.5	23	OK	8.9	23	IN	12.5	24	RI	70.4	
24	AR	0.7	24	KS	2.4	24	GA	8.7	24	NV	11.1	25	WA	69.8	
	HI		24	MI	2.4	25	NC	8.4	25	OK	10.9	26	TN	69.5	
25	IA	0.6	26	FL	2.2	26	VA	7.7	26	KY	10.6	27	AR	68.2	
	RI		27	NC	2.1	27	MD	7.6	27	WI	10.5	28	СТ	67.0	
20	ME	0.5	20	IA	1.0	28	AR	6.8	20	KS	9.6	29	CO	62.5	
28	NY	0.5	28	NH	1.9	29	WI	6.7	- 28	RI	8.6	30	VA	59.8	
	СТ		30	NE	1.8	30	PA	6.4	30	MN	8.5	31	OK	59.6	
20	MD	0.4	31	OK	1.7	31	IA	5.8	31	MA	8.4	32	AL	59.4	
50	MO	0.4		ID		32	IN	5.7	32	CA	8.0	33	AK	57.7	
	VT		32	MO	1.6	33	MN	5.4	33	NE	7.6		U.S.	57.1	
	DE			VT		34	HI	4.5	34	CO	6.0	34	NC	56.6	
	FL			AR		35	MI	4.4	35	WA	5.7	35	NJ	56.5	
	IN			DC		36	AK	4.2	36	AZ	5.2	36	IL	56.4	
34	MA	03	35	ME	1.4	37	SC	4.0	37	IA	5.1	37	DE	55.1	
54	NH	0.5		OH		38	TN	3.8	38	WV	5.0	38	SC	54.0	
	SC			TN		39	MO	3.2	39	AK	4.6	39	NY	52.7	
	TX			LA		40	AL	28	40	OR	3.2	40	LA	51.5	
	VA		40	NM	1.3	40	NH	2.0	41	NM	2.5	41	FL	49.6	
	IL			SC		42	MT	24	42	HI	2.4	42	GA	49.2	
	KY		43	IN	1.2	14	OH	2.7	43	ME	2.0	43	MD	48.6	
42	MS	0.2	44	MT	11	44	KY	2.1	44	NH	1.7	44	AZ	47.2	
	NJ			WY	1.1		LA	2.1	45	SD	1.6	45	MS	46.5	
	TN		46	AL	1.0	46	SD	2.0	Į	ND		46	NV	46.4	
	DC			SD		47	ND	1.7	46	VT	1.5	47	TX	36.5	
	GA		48	KY	0.9	48	MS	1.4	<u> </u>	WY		48	NM	31.1	
47	OH	0.1		ND	5.5	49	VT	1.0	49	UT	1.3	49	CA	31.0	
	PA		50	MS	0.8	50	ME	0.9	50	ID	1.0	50	HI	19.8	
	WV		51	WV	0.6	51	WV	0.7	51	MT	0.9	51	DC	4.5	

Table A.18 (corresponds to Table 3.3)Racial and Ethnic Composition of Students, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Public 9.

1989				1999		2006					
Rank	State	\$	Rank	State	\$	Rank	State	\$	Sig.		
1	СТ	49,199	1	СТ	65,521	1	СТ	78,154	0		
2	NJ	47,589	2	NJ	65,370	2	NJ	77,875			
3	AK	46,581	3	MD	61,876	3	MD	77,839			
4	MD	45,034	4	MA	61,664	4	MA	74,463			
5	MA	44,367	5	AK	59,036	5	NH	71,176			
6	HI	43,176	6	NH	57,575	6	HI	70,277			
7	NH	41,628	7	HI	56,961	7	AK	69,872			
8	CA	40,559	8	MN	56,874	8	VA	66,886			
9	DE	40,252	9	CO	55,883	9	MN	66,809			
10	NY	39,741	10	IL	55,545	10	RI	64,733			
11	RI	39,172	11	DE	55,257	11	CO	64,614			
12	IL	38,664	12	VA	54,169	12	CA	64,563			
13	VA	38,213	13	WA	53,760	13	WA	63,705			
14	MN	36,916	14	MI	53,457	14	IL	63,121			
15	WA	36,795	15	CA	53,025	15	DE	62,623			
16	MI	36,652	16	WI	52,911	16	NY	62,138			
17	DC	36,256	17	RI	52,781	17	NV	61,466			
18	СО	35,930	18	NY	51,691	18	DC	61,105			
19	NV	35,837	19	UT	51,022	19	WI	60,634			
	U.S.	35,225	20	NV	50,849		U.S.	58,526			
20	WI	35,082	21	IN	50,261	20	VT	58,163			
21	PA	34,856		U.S.	50,046	21	PA	58,148	~		
22	VT	34,780	22	OH	50,037	22	UT	58,141	-		
23	OH	34,351	23	KS	49,624	23	MI	57,996			
24	IN	34,082	24	GA	49,280	24	WY	57,505			
25	GA	33,529	25	PA	49,184	25	NE	56,940			
26	UT	33,246	26	OR	48,680	26	KS	56,857			
27	KS	32,966	27	VT	48,625	27	OH	56,148			
28	ME	32,422	28	NE	48,032	28	GA	56,112			
29	OR	32,336	29	IA	48,005	29	OR	55,923			
30	WY	32,216	30	AZ	46,723	30	IN	55,781			
31	FL	32,212	31	NC	46,335	31	IA	55,735			
32	AZ	32,178	32	DC	46,283	32	AZ	55,709			
33	MO	31,838	33	MO	46,044	33	ND	55,385			
34	IA	31,659	34	TX	45,861	34	FL	54,445			
35	NE	31,634	35	WY	45,685	35	SD	53,806			
36	TX	31,553	36	FL	45,625	36	MO	53,026			
37	NC	31,548	37	ME	45,179	37	ME	52,793			
38	SC	30,797	38	SC	44,227	38	TX	52,355			
39	TN	29,546	39	ND	43,654	39	NC	52,336			
40	ID	29,472	40	TN	43,517	40	ID	51,640			
41	ND	28,707	41	ID	43,490	41	MT	51,006			
42	AL	28,688	42	SD	43,237	42	SC	50,334			
43	OK	28,554	43	AL	41,657	43	TN	49,804			
44	MT	28,044	44	KY	40,939	44	AL	49,207			
45	NM	27,623	45	OK	40,709	45	KY	48,726			
46	SD	27,602	46	MT	40,487	46	LA	48,261	=		
47	KY	27,028	47	LA	39,774	47	NM	48,199			
48	LA	26,313	48	NM	39,425	48	OK	47,955			
49	WV	25,602	49	AR	38,663	49	AR	45,093			
50	AR	25,395	50	MS	37,406	50	WV	44,012	<		
51	MS	24,448	51	WV	36,484	51	MS	42,805			

# Table A.19 (corresponds to Table 3.4)Median Family Income in Nominal Dollars, 1989, 1999, and 2006

Notes: Incomes for 1989 and 1999 are from decennial censuses. Incomes for 2006 are from the 2006 American Community Survey, which is subject to sampling error. Each difference between Kentucky and another state was tested for statistical significance with a 95 percent confidence level; > indicates states with significantly higher incomes than Kentucky, = indicates states not significantly different, and < indicates states with significantly lower incomes than Kentucky. Statistical tests used unrounded percentages and took into account each state's sample size and variance; therefore, states with the same percentages can have different levels of significance. Sources: U.S. Dept. of Commerce. Census. "American Community Survey" and "Decennial Census."

	1989			1999		2005					
Rank	State	%t	Rank	State	%	Rank	State	%	Sig.		
1	MS	25.2	1	DC	20.2	1	DC	32.6	0		
2	LA	23.6	2	MS	19.9	2	MS	29.5			
3	NM	20.6	3	LA	19.6	3	LA	27.8			
4	WV	19.7	4	NM	18.4	4	NM	25.6	>		
5	AR	19.1	5	WV	17.9	5	WV	25.2			
6	KY	19.0	6	AL	16.1		AR				
7	AL	18.3	_	AR		6	OK	24.3			
8	TX	18.1	7	KY	15.8	8	TX	23.9			
9	DC	16.9	9	ТХ	15.4	9	AL	23.0			
10	OK	16.7	10	OK	14.7	10	KY	22.8	=		
11	MT	16.1		MT		11	TN	22.7			
12	SD	15.9	11	NY	14.6	12	SC	22.1			
12	AZ	10.5	13	CA	14.2		GA	22.1			
13	TN	15.7	14	SC	14.1	13	NC	20.2			
15	SC	15.4	15	AZ	13.9	15	NY	20.0			
16	GA	14.7	16	TN	13.5	16	AZ	19.5			
17	ND	14.4	17	SD	13.2	17	OH	18.7			
17	MO	14.4	18	GA	13.0	18	MO	18.6			
18	ID	13.3	19	FI	12.5	10	US	18.3			
	US	13.1	17	US	12.5	19	MI	18.3			
20	MI	13.1	20	NC	12.4	20		18.1			
20	NV	13.1	20	RI	12.5	20	IN	17.9			
21	NC	13.0	21	ND	11.9	21	ME	17.5			
23	FL	12.7	23	ID	11.8	22	FI	17.0			
25	OH	12.7	23	MO	11.3	23	MT	17.3			
24		12.5	24	OP	11.7	24	II	17.5			
26	OP	12.4	25	WV	11.0	25	DA	17.1			
20	UK	12.4	20		11.4	20	OP	10.9			
27	IL WV	11.9	27	PA ME	11.0	27	OR SD	16.8			
20		11.7	20		10.9	20	DE	15.9			
29		11.7	29	п	10.7	29	DE CO	15.0			
30		11.5				30	VS VS	15./			
22	K5 UT	11.4	31	WA OU	10.6	31	K5 WA	15.0	<		
32	NE	11.4		M		32	WA AV	13.4			
33	DA NE	11.1	33	NIV	10.5	22		15 1			
25	PA WA	10.0	25		0.0			13.1			
33	WA	10.9	33	NE	9.9	26	KI WI	14.0			
30	WE	10.8	30	INE VA	9.7	30	WI NE	14.9			
37	W1 DI	10.7	29	VA	9.6	29	INE	14.4			
	IIN VA		30	IIN VT	9.5	38		13.9			
20	VA	10.2	20		0.4	39	IA	13./			
39	MIN	10.2	39		9.4	40	V I	13.2			
40		0.0		AK		41	ND	13.0			
42		9.9	42	MA	9.3	42	MA	12.4			
43		9.0	4.4	DE	0.2	43	IVIIN V A	12.2			
44	AK	9.0	44	DE	9.2	15	VA WV	12.0			
43	DE	0.9 0.7	43	IA WI	9.1	43		12.0			
40	MD	0./	40	WI NI	0./	40	NI	11.9			
47		8.3	47	INJ MD	8.5	4/	INJ	11.8			
40		7.6		MN		40		11.4			
49 50		6.0	49	CT	7.9	49 50	MD	07			
51	NU	6.4	51	NU	65	51		9.7			
51		0.4	51	INT	0.5	51	INT	7.0			

#### Table A.20 (corresponds to Table 3.5) Children Living Below the Federal Poverty Line 1989, 1999, and 2005

Notes: > indicates states scoring with significantly higher percentages than Kentucky, = indicates states not significantly different, and < indicates states with significantly lower percentages than Kentucky. Statistical tests used unrounded percentages and took into account each state's sample size and variance; therefore, states with the same percentages can have different levels of significance.

Source: U.S. Dept. of Commerce. Census. "American Community Survey" and "Decennial Census."

	<i>a</i>	Average CWI for All	Range of CWI Across All Labor Markets in State							
Rank	State	Labor Markets in State	Rank	State	Range	Lowest	Highest			
1	DC	1.554		U.S.	0.835	0.833	1.669			
2	NJ	1.429	1	CA	0.829	0.840	1.669			
3	NY	1.415	2	WV	0.574	0.980	1.554			
4	СТ	1.391	3	NY	0.554	1.004	1.558			
5	CA	1.387	4	TX	0.534	0.855	1.389			
6	MA	1.376	5	MD	0.491	1.063	1.554			
7	VA	1.369	6	VA	0.488	1.066	1.554			
8	MD	1.368	7	PA	0.483	0.984	1.467			
9	WA	1.314	8	СТ	0.476	1.106	1.583			
10	IL	1.306	9	FL	0.432	0.833	1.265			
11	DE	1.286	10	MA	0.430	1.004	1.435			
12	RI	1.278	11	IL	0.424	0.964	1.387			
	U.S.	1.265	12	NM	0.411	0.902	1.313			
13	NV	1.258	13	TN	0.395	0.894	1.290			
14	TX	1.251	14	AR	0.380	0.910	1.290			
15	GA	1.242	15	KY	0.379	0.913	1.293			
16	MI	1.235	16	NC	0.365	0.942	1.308			
17	MN	1.223	17	MS	0.360	0.930	1.290			
18	CO	1.213	18	MO	0.349	0.897	1.246			
19	OH	1.211	19	CO	0.347	0.931	1.278			
20	PA	1.205	20	WI	0.345	1.027	1.371			
21	AK	1.201	21	NJ	0.339	1.220	1.558			
22	WI	1.200	22	MN	0.323	0.983	1.305			
23	HI	1.198	23	GA	0.314	0.991	1.305			
24	NC	1.194	24	IN	0.313	0.979	1.293			
25	NH	1.177	25	DE	0.308	1.031	1.338			
26	FL	1.171	26	MI	0.307	1.026	1.333			
27	TN	1.163	20	KS	0.307	0.903	1.209			
28	UT	1 1 50	28	WA	0.302	1.085	1.387			
20	AZ	1.139	29	OH	0.300	0.992	1.293			
30	OR	1.156	30	LA	0.290	0.913	1.203			
31	MO	1.144	31	IA	0.257	0.923	1.179			
32	SC	1.140	32	OR	0.254	0.980	1.234			
33	IN	1.123	33	NE	0.252	0.922	1.174			
34	KY	1.117	34	AZ	0.236	0.970	1.205			
35	NM	1.109	35	SC	0.229	1.079	1.308			
36	AL	1.108	36	OK	0.223	0.905	1.128			
37	LA	1.097	37	ID	0.213	0.854	1.068			
38	WV	1.071	38	NV	0.211	1.095	1.306			
	KS		39	ME	0.210	0.931	1.142			
39	NE	1.070	40	AL	0.205	0.970	1.174			
	VT		41	NH	0.204	1.060	1.264			
42	OK	1.064	42	UT	0.186	1.009	1.194			
43	IA	1.059	43	SD	0.180	0.878	1.058			
44	ME	1.056	44	MT	0.137	0.883	1.020			
45	MS	1.051	45	WY	0.113	0.983	1.096			
46	AR	1.041	46	AK	0.111	1.145	1.256			
47	WY	1.024	47	ND	0.107	0.967	1.074			
48	ID	1.018	48	VT	0.080	1.030	1.110			
49	ND	1.014		RI		1.283	1.283			
50	SD	0.962	49	HI	0.000	1.198	1.198			
51	MT	0.936		DC		1.554	1.554			

### Table A.21 (corresponds to Table 3.6) NCES Comparable Wage Index, 2005

Source: U.S. Dept. of Ed. Inst. Natl. "NCES Comparable."
		Title I	School	ls with				Eli	igible fo	r	Eligibl	e for F	ree or	
			Sc	hoolwi	de	Eligible for Free			Red	uced-pr	ice	Reduced-price		
All Ti	tle I Scl	hools	P	rogram	15	8-	Lunch			Lunch		Lunch		
Rank	State	%	Rank	State	%	Rank	State	0/0	Rank	State	0/0	Rank	State	0/0
1	OR	99.9	1	DC	81.3	1	MS	62.3	1	HI	11.3	1	MS	69.5
2	IN	96.1	2	MS	61.6	2	LA	53.9	2	OK	10.4	2	LA	61.2
3	DC	85.4	3	TX	60.4	3	DC	49.1	3	WY	10.1	3	NM	55.7
4	MT	79.3	4	HI	53.6	4	NM	47.9	4	ID	10.0	4	OK	54.5
5	ID	67.9	5	KY	52.8	5	OK	44.1	5	KS	9.9	5	DC	53.4
6	AR	67.1	6	LA	51.0	6	AR	44.0	6	WV	9.8	6	AR	52.9
7	HI	65.9	7	NM	47.3	7	KY	43.8	7	CA	0.2	7	KY	52.4
8	OK	65.8	8	OK	46.1	8	SC	43.7	/	NE	9.2	8	AL	51.7
9	MS	65.6	9	AR	45.8	9	AL	43.0	9	UT	9.1	9	SC	51.5
10	TX	64.7	10	AL	45.1	10	GA	41.5	10	SD	9.0	10	GA	49.8
11	OH	63.8	11	TN	38.6	11	TX	40.0	11	AR	8.8	11	WV	49.1
10	ME	(2.5	12	CA	37.9	12	CA	39.3	12	FL	8.7	12	CA	48.5
13	N Y DA	63.5	13	GA	37.5	14	W V EI	27.1	13	KY AI	8.6	13	1X TN	48.2
14	PA	60.6	14	SC EL	34.4	14	FL AZ	37.1	15	AL	05	14	T IN EI	4/.1
15	NM	50.2	15	FL AZ	22.8	15	AZ NV	26.9	15	MT	8.3	15		45.0
10	WI	58.2	10	AZ NV	31.7	10	NC	30.8	10	GA	0.4	10	AZ NV	43.0
17	CA	58.0	17	US	31.7	17	OR	34.6	17	TX	8.2	17	OR	44.4
19	LA	57.7	18	DE	29.8		US	32.4		WA		19	NC	42.6
20	VT	56.0	19	NC	29.7		IL.	52.1	19	AZ	8.1		US	41.6
21	AL	55.5	20	VA	27.4	19	MO	31.5	21	ND	8.0	20	NV	41.3
22	ND	55.2		VT		21	DE	30.7		IN		21	HI	40.5
23	IL	54.4	21	MI	26.3	22	HI	29.2	22	SC	7.9	22	MO	39.1
	U.S.	52.3	23	WV	26.2	23	MI	29.1	24	NM	7.8	23	KS	38.8
24	AZ	51.6	24	IL	24.8	24	KS	28.9		IA		24	ID	37.8
24	WA	51.0	25	MA	22.0	25	RI	28.4	25	MO	7.7	25	IL	37.2
26	MA	50.8	26	OH	21.6	26	WA	28.3		NY		26	WA	36.5
27	WY	47.8	27	OR	21.5	27	IN	28.2		U.S.	7.4	27	DE	36.1
28	CO	46.1	28	WA	21.0	28	ID	27.8	28	NC	7.4		IN	50.1
29	GA	45.9	29	MT	19.8	29	CO	26.9		ME		29	MI	35.6
30	TN	45.3	30	KS	19.6	30	ME	26.4	20	MN		30	RI	35.3
31	SD	45.1	31	KI DA	19.3	22	OH	26.1	30	LA	1.3	31	NE	34.7
32	DE	44.6	32	PA	18.5	32	MI	26.1	22	MD	7.2	32	ME	34.5
24	MU	42.7	24	NE	18.0	24	DA NE	25.5	24	MD VT	1.2	24	ME	22.1
34	KS	41.9 30 /	54	INE SD	1/.0	34	rA AV	23.0	34	V I RI	0.9	34	0H	32.5
36	RI	393	35	AK	17.5	55	MD	27.1	35	VA	6.8	36	UT	32.3
37	MN	38.4	37	WY	17.2	36	IA	24.4	37	AK	67	37	IA	32.5
38	NH	38.0	38	MD	16.8	38	VA	24.3	38	MI	6.5	38	SD	32.0
39	IA	37.4	39	NV	16.5	39	UT	23.2	39	PA	6.4		WY	22.0
40	NC	37.3	40	MO	15.8	40	MA	23.1	40	СО	()	- 39	MD	31.6
41	SC	36.9	41	WI	15 (	41	SD	22.0	40	OH	6.2	41	PA	21.4
42	WV	36.7	41	UT	15.6	41	MN	23.0	42	IL	5.7	41	AK	51.4
43	FL	34.9	43	CT	12.4	43	ND	21.6	12	CT	56	43	VA	31.1
44	NE	34.5	44	ID	12.2	44	WY	21.5	43	NJ	5.0	44	MN	30.3
45	AK	33.8	45	ND	11.4	45	NJ	21.2	45	DE	5.4	45	ND	29.6
46	VA	27.4	46	MN	9.9	46	CT	20.9	46	MA	5.2	46	WI	29.3
47	MI	26.3	47	IA	8.6	47	VT	19.5	47	NH	5.1	47	MA	28.2
48	MD	20.1	48	IN	6.8	48	NH	11.9	48	DC	4.3	48	NJ	26.8
49	UT	19.6	49	NH	5.6		NV			NV		49	CT	26.5
50	NV	16.5	50	ME	4.5	n.a.	TN	n.a.	n.a.	TN	n.a.	50	VT	26.4
n.a.	NJ	n.a.	n.a.	NJ	n.a.		WI		1	WI		51	NH	17.1

## Table A.22 (corresponds to Table 3.7)Students in Title I Schools or Eligible for National School Lunch Program, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Numbers and Types of Public Elem. and Secondary Schools 4-5 and Common.

Rank	State	%		
1	CA	24.9		
2	NM	19.2		
3	AZ	16.0		
4	TX	15.7		
5	AK	15.6		
6	NV	15.5		
7	CO	12.8		
8	OR	12.0		
9	HI	99		
10	UT	9.8		
	US	8.6		
11	FL	83		
12	OK	7.5		
13	WA	7.3		
15	ID	1.5		
14	NY	69		
11	MN	0.9		
17	DC	6.5		
17	NE	6.1		
10	VA	6.0		
20	IN	5.5		
20	GA	5.5		
21	MA	5.4		
22	WIA VS	5.3		
	NC			
24		5.2		
	DE			
26	DE	4.9		
20	KI	1.6		
28		4.0		
29		4.4		
30	SD	4.2		
31	MI	3.8		
32	MD	3./		
33	NJ	3.6		
25	WY	2.4		
35	WI	3.4		
36	IA	3.1		
3/	AL	2.2		
38	SC	2.1		
39	MO	2.0		
40		1.8		
	LA			
42	ME	1.7		
43	OH	1.6		
44	KY	1.5		
45	WV	0.7		
46	MS	0.6		
		4		
	ND	4		
n.a.	NH	n.a.		
	PA	4		
I	TN	1		

#### Table A.23 (corresponds to Table 3.8)Students Receiving English Language Learner Services, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Common.

Rank	State	%
1	NJ	26.7
2	NM	19.7
3	RI	18.0
4	AZ	18.0
5	WV	17.6
6	DC	17.4
7	IN	17.1
8	ME	16.9
9	AL	16.8
10	NE	16.2
11	KY	16.0
12	SC	15.6
13	MA	15.4
14	IL	15.3
15	OK	15.2
16	SD	15.1
17	FL	14.9
	WI	
18	NH	14.8
	IA	
21	DE	14.7
22	OR	14.6
22	PA	14.6
24	OH	14.5
25	VA	14.4
26	MI	14.1
26	ND	14.1
28	KS	14.0
29	MN	13.8
30	MS	13.7
	U.S.	13.6
31	NC	13.6
22	AK	12.5
52	WY	13.3
34	TN	13.3
	NY	
35	UT	13.2
	MT	
38	LA	13.0
39	MD	12.8
40	GA	12.4
41	AR	12.3
42	WA	12.0
<u>۲</u>	HI	12.0
44	СТ	11.6
45	TX	11.3
1.7	VT	11.5
47	NV	11.1
48	ID	11.0
49	CA	10.7
50	СО	10.1
51	MO	0.0

#### Table A.24 (corresponds to Table 3.9)Students With Individualized Education Programs, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Common.

Overall Rank	State
1	Minnesota
2	New Hampshire
3	Connecticut
4	Utah
5	Massachusetts
6	Vermont
7	Iowa
8	North Dakota
9	New Jersev
10	Nebraska
11	Hawaii
12	Wisconsin
13	Washington
14	Virginia
15	Maine
16	Kansas
17	Oregon
18	New York
19	California
20	Rhode Island
21	Pennsylvania
22	Idaho
23	Colorado
24	Maryland
25	Wyoming
26	Illinois
27	Michigan
28	Ohio
29	Montana
30	South Dakota
31	Indiana
32	Florida
33	Nevada
34	Missouri
35	Delaware
36	Arizona
37	Texas
38	Alaska
39	North Carolina
40	Kentucky
41	Georgia
42	Oklahoma
43	Tennessee
44	West Virginia
45	Arkansas
46	South Carolina
47	New Mexico
48	Alabama
49	Louisiana
50	Mississippi

#### Table A.25 (corresponds to Table 3.11)Overall Child Well-being, 2007 KIDS COUNT Data Book

Source: Annie E. Casey.

Rank	State	Score	Grade		
1	Massachusetts	94.1	А		
2	New Jersey	91.0			
3	New Hampshire	90.1	A-		
4	Connecticut	89.8			
5	Vermont	89.3			
6	Maryland	88.2	B+		
7	Minnesota	87.2	-		
8	Virginia	85.9			
9	North Dakota	84.8	-		
10	Pennsylvania	83.7			
11	Iowa	83.6	В		
12	South Dakota	82.9			
13	New York, Delaware, Kansas	82.6			
16	Colorado	82.5			
17	Wisconsin	82.4			
18	Utah	82.2			
19	Illinois	81.3			
20	Rhode Island, Nebraska	81.1	B-		
22	Montana	81.0			
23	Maine	80.1			
24	Washington, Ohio	79.6			
26	Wyoming	79.0			
27	Michigan	78.6			
	U.S. <sup>1</sup>	78.4			
28	Missouri	77.5	C+		
29	Indiana, Hawaii	77.4			
31	North Carolina	77.1			
32	Florida	76.6			
33	District of Columbia	76.4			
34	Idaho	76.2			
35	Oregon	75.6			
36	Georgia	75.2			
37	Alaska	74.8	С		
38	Kentucky	74.7			
39	South Carolina	74.3			
40	California	73.9			
41	Texas	73.3			
42	Alabama	72.1			
43	Tennessee, Oklahoma	71.9			
45	Arkansas	71.7	C-		
46	Arizona	71.5			
47	West Virginia	70.8			
48	Nevada	68.5			
49	New Mexico	68.0	D+		
50	Louisiana	67.9			
51	Mississippi	67.7			

#### Table A.26 (corresponds to Table 3.12)Education Week's Quality Counts Chance-for-Success-Index

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

Rank         State         Rank         State         Rank         State         Rank         State         Rank         State         Rank           1         R1         10.7         1         VT         11.4         1         RI         11.0         1         DC         10.5           2         VT         10.9         2         RI         11.5         2         VT         11.1         2         VT         11.8           3         ME         11.7         3         ME         11.9         3         ME         12.0         3         RI         11.8           4         ND         12.4         5         WY         12.9         5         DC         12.7         5         ND         12.7           6         WY         12.9         9         NJ         13.2         9         NA         13.2         9         NA         13.2         13         NB         13.2         13         NB         13.2         13         NY         13.4         13.8         NY         13.1         NY         14.0         16         14.0         16         14.0         16         14.0         16         14.1         1	Total		Ele	mentary	7	Mid	ldle Scho	ol	High School			
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Rank	State	Ratio	Rank	State	Ratio	Rank	State	Ratio	Rank	State	Ratio
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1	RI	10.7	1	VT	11.4	1	RI	11.0	1	DC	10.5
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	2	VT	10.9	2	RI	11.5	2	VT	11.1	2	VT	11.6
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	3	ME	11.7	3	ME	11.9	3	ME	12.0	3	RI	11.8
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	4	ND	12.3	4	ND	12.4	4	NJ	12.2	4	CT	12.3
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	5	NJ	12.4	5	WY	12.9	5	DC	12.7	5	ND	12.7
	6	VA	12.6		NH		6	MA	12.8	6	AR	12.9
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	0	WY	12.0	6	SD	13.1	0	NH	12.0	7	ME	13.0
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	8	AL	12.8		VA		8	WY	12.9	8	NJ	13.1
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	9	NY	12.9	9	NJ	13.2	9	ND	13.2	0	WY	15.1
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	10	MA	13.2	10	IA	13.4	,	VA	13.2		KS	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	10	NH	15.2	11	NE	13.5	11	KS	13.4	10	MA	13.8
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	12	NE	13.4	12	MA	13.6	12	IA	13.5		NH	15.0
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		SD	10.1	13	MO	13.7	13	NE	13.6		SD	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	14	IA	13.7	14	DC	13.8		NY		14	MT	13.9
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	17	MO	10.0		NY	110	15	SD	13.7	15	NE	14.0
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	16	KS	13.9	16	KS	14.0	16	MT	14.0	16	IA	14.2
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	17	DC	14.0	17	MT	14.2	17	AR	14.1	17	VA	14.5
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	10	MI	1.4.1	18	GA	14.3	18		14.2	18	OK	14.6
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	19	WV	14.1	20	WI	14.4	20	WI	14.2	19	TX	15.0
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	20	AK	14.4	20	AL	14.4	20	W V	14.3	20	NY	15.2
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	21		14.5	21	NM	14.5	21	MO	14.7	21	WI TA	15.5
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	22	SC	14.6	22	SC	14.6	22	INIM	14.0	22	LA	15.0
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		WI CA		24	W V NC	147	23	1A NC	14.8	23	WW	15.7
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	24	UA LA	14.7	24	IA	14.7	24	DA	14.0	25	SC SC	15.9
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		NC		23	LA KV	14.9	24	FA SC	14.9	23	NC	13.0
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	26	NM	14.8	26	MD	15.1	27	GA	15.0	26	PA	15.9
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		PA		28	TN	15.3	28	LA	15.0	28	NM	16.0
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	28	TX	15.0	20	DE	15.5	29	MD	15.2	2.9	IL	16.4
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	30	DE	15.1	29	TX	15.4		MS			AL	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		MD		31	FL	15.5	30	OK	15.5	30	DE	16.5
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	31	OK	15.2	32	AR	15.6	22	IL	1.5.0	32	GA	16.7
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	33	OH	15.6	33	PA	15.8	32	OH	15.8	33	MS	16.8
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		U.S.	15.7		U.S.	15.9		U.S.	16.0		U.S.	17.0
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	34	MS	15.7	34	AK	16.1	34	HI	16.0	34	HI	17.0
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	35	IL	15.8	35	OK	16.2	35	KY	16.3		KY	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	26	KY	16.0	36	HI	16.3	36	AK	16.4	35	MD	17.2
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	30	TN	10.0	37	MN	16.7	37	CO	16.5		OH	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	38	HI	16.3	38	CO	16.8	38	DE	16.7	38	CO	17.9
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	39	MN	16.4	39	IL	17.0	39	TN	17.0	39	ID	18.1
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	40	AK	16.8	40	MS	17.1	40	MN	17.2		MN	10.1
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		FL	10.0	41	СТ	17.2	41	AL	17.4	41	IN	18.8
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	42	CO	17.0	42	NV	17.5	42	IN	17.6	42	AK	18.9
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	43	IN	17.1		OH		43	FL	17.7	43	TN	19.1
45         1D         18.0         45         MI         18.0         MI         18.0         MI         19.9           46         NV         19.0         46         ID         18.7         46         OR         19.4         46         UT         20.1           47         WA         19.3         47         WA         18.9         47         WA         19.6         47         OR         20.9           48         OR         19.5         48         OR         19.7         48         UT         20.9         48         WA         21.3           49         CA         20.8         49         CA         20.1         49         NV         22.8         49         NV         23.1           50         AZ         21.3         50         UT         20.6         50         CA         23.3         50         CA         23.7           51         UT         22.1         na         AZ         na         na         AZ         na         na	44	MI	17.4	44	IN	17.6	44	ID	17.8	44	FL	19.8
46         NV         19.0         46         ID         18./         46         OR         19.4         46         UT         20.1           47         WA         19.3         47         WA         18.9         47         WA         19.6         47         OR         20.9           48         OR         19.5         48         OR         19.7         48         UT         20.9         48         WA         21.3           49         CA         20.8         49         CA         20.1         49         NV         22.8         49         NV         23.1           50         AZ         21.3         50         UT         20.6         50         CA         23.3         50         CA         23.7           51         UT         22.1         na         AZ         na         na         AZ         na         na         AZ         na         na	45	ID NU	18.0	45	MI	18.0	44	MI	10.4	45	MI	19.9
4/       WA       19.5       4/       WA       18.9       4/       WA       19.6       4/       OR       20.9         48       OR       19.5       48       OR       19.7       48       UT       20.9       48       WA       21.3         49       CA       20.8       49       CA       20.1       49       NV       22.8       49       NV       23.1         50       AZ       21.3       50       UT       20.6       50       CA       23.3       50       CA       23.7         51       UT       22.1       na       na       na       AZ       na       na       AZ       na	46	NV	19.0	46	ID W	18.7	46	OR	19.4	46	UT	20.1
48         0K         19.5         48         0K         19.7         48         01         20.9         48         WA         21.3           49         CA         20.8         49         CA         20.1         49         NV         22.8         49         NV         23.1           50         AZ         21.3         50         UT         20.6         50         CA         23.3         50         CA         23.7           51         UT         22.1         na         na         na         AZ         na         na         AZ         na	4/	WA	19.3	4/	WA	18.9	4/	WA	19.6	47	OR	20.9
49     CA     20.8     49     CA     20.1     49     NV     22.8     49     NV     23.1       50     AZ     21.3     50     UT     20.6     50     CA     23.3     50     CA     23.7       51     UT     22.1     na     na     na     AZ     na     na     AZ	48	OK	19.5	48	OK CA	19./	48		20.9	48	WA	21.5
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	49	LA AZ	20.8	49	LIT	20.1	49		22.8	49		23.1
	51		21.5	 	A7	20.0 n.a	- 50 n a	A7	123.5 na	- 50 n a	Δ7	2J.1

## Table A.27 (corresponds to Table 4.1)Student-Teacher Ratio, Total and by Grade Level, FY 2006

Notes: Student-teacher ratio uses student membership and full-time equivalent counts of teachers. Sources: U.S. Dept. of Ed. Inst. Natl. *Common Core* and *Numbers and Types of Public Elem. and Secondary Schools* 8-9.

.

	1985			1995			2005	
Rank	State	%	Rank	State	%	Rank	State	%
1	ID	63.9	1	RI	63.5	1	Scare	71.5
2	RI	63.3	2	MN	62.7	2	NV	67.2
3	ME	61.2	3	HI	62.3	3	NY	58.6
4	WI	59.7	4	ID	58.6	4	RI	58.2
5	NH	59.6	5	NV	58.5	5	WI	57.0
6	DC	58.6	6	WI	57.9	6	ID	55.8
7	MN	58.4	7	DC	56.4	7	AL	55.7
8	UT	58.1	8	MA	55.4	8	CA	53.4
9	MA	57.7	9	OH	55.2	9	HI	53.3
10	KS	57.3		СТ		10	MT	53.1
10	SD	57.5	10	WV	54.5	11	ND	52.9
12	NE	57.2		DE		12	MA	52.7
13	ND	57.0	13	MD	54.4	13	NC	52.5
14	SC	56.7		ND		14	VA	523
15	IL	56.6	14	VA	54.3	14	WV	52.5
16	WA	56.4		IL		16	TN	52.2
17	PA	55.6	17	MT	54.2	17	MO	52.1
18	DE	55.4	18	TN	54.0	18	NJ	52.0
19	WV	55.0	19	AR	53.8	19	NE	51.9
20	VA	54.3	20	KS	53.7	20	DE	51.7
21	IA	54.0	21	UT	53.6	21	AZ	513
21	OH	54.0	22	NH	53.3	21	KS	51.5
23	MD	53.9		SC	00.0		U.S.	51.2
	NJ		24	SD	53.2	23	OK	51.1
25	OK	53.5	24	NJ	52.0	24	MD	51.0
26	AR	53.4	26	PA	53.0	25	IA	50.9
27	MO	53.3	27	AL	52.9		PA	
28	NM	53.2	20	NE	52.5	27	FL	50.6
29		53.1	29	CO ME	52.5	20		50.2
	U.S.	53.0	30	ME	52.3	29		50.2
30	GA	52.0	31	IA	52.2	30	0H OH	49.0
51		52.8	32	IA	52.0	31	CO	49.4
32	NC	52.4		CA	52.0	32	MN	49.2
34	VT	52.3	33	TX	52.0	34	TX	48.7
35	AZ	52.0	35	OR	51.8	35	NH	48.5
36	FL	51.6	36	WA	51.4	36	MI	48.3
37	CO	51.4	37	WY	51.2	37	LA	48.2
	KY		38	NY	51.0	38	SD	48.0
38	TN	51.1	39	LA	50.5	39	ME	47.3
40	AK	51.0	40	AZ	50.1		OR	17.0
41	IN	50.0	41	AK	40.1	40	WA	47.0
41	WY	50.8	41	VT	49.1	42	CT	46.9
43	TX	50.7	42	FL	40.2	43	AR	46.7
44	MI	48.7	43	NM	48.5	4.4	VT	16.5
45	CA	48.6	45	GA	48.2	44	MS	40.3
46	LA	48.2	16	IN	48.0	46	WY	46.2
47	MS	47.0	40	MO	40.0	47	NM	45.9
48	HI	42.5	48	MS	47.6	48	IN	45.5
	СТ		49	OK	47.0	49	DC	44.3
n.a.	MT	n.a.	50	MI	46.9	50	AK	44.1
	NV		51	KY	46.3	51	KY	43.3

#### Table A.28 (corresponds to Table 4.2) Teachers as a Percentage of All Staff Fall 1985, 1995, and 2005

Notes: For 1985, ranks are out of 48 states; CT, MT, and NV are not shown because they underreported support staff. The U.S. value includes estimates for these three states.

Sources: Staff compilation of data from U.S. Dept. of Ed. Inst. Natl. Digest, various years.

A	All Staf	f	Teachers		Ins	tructio Aides	nal	Instructional Coord./Supervisors			Guidance Counselors			
Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate
1	VT	196 9	<b>1</b>	RI	93 2		VT	44 1		VT	33	<b>1</b>	RI	16.6
2	ME	180.3	2	VT	91.6	2	NH	32.6	-	Н	5.5	2	WY	4 7
3	WY	172.1	3	ME	85.3	3	ME	30.9	2	SD	3.1	_	LA	,
4	DC	160.9	4	ND	81.4	4	SD	28.1	4	LA	2.7	3	VT	4.5
5	RI	160.1	5	NJ	80.7	5	WY	24.5	5	DE	2.4	5	NH	4.0
6	SD	155.9	6	WY	79.4	6	СТ	21.7	6	CO	2.0	6	HI	3.7
7	NH	155.6	7	VA	79.2	7	MA	21.2	7	MI	19	7	ME	3.2
8	NJ	155.3	8	AL	77.9	8	KY	20.7	'	NJ	1.7	8	AR	3.0
9	ND	153.9	9	NY	77.8	9	NC	20.3	9	ME	1.8		MT	0.0
10	VA	151.4	10	MA	75.7	10	IA	20.1		MN		10	MO	2.9
11	AR	149.0	11	NH	75.5	11	IN	19.2	11	NE	1.7	11	ND	2.8
12	CI NM	14/.2	12	SD NE	74.8	12	ND	19.1	10	IN	16	12	MD	2.7
13	INIM KV	140.7	13	MO	73.1	13	MN	18.9	12	UT	1.0		NC NC	
14	NE	144.1	14	IA	72.8	14	DC	17.9	15	MS	1.5	14	SD SD	2.6
15	MA	143.6	16	KS	71.9	16	OR	17.7	15	AR	1.5		OK	
17	IA	142.9	10	DC	71.9	17	MS	17.5		DC		16	SC	2.5
18	LA	141.5	17	MT	71.3	18	AK	16.8	16	WI	1.4	-	WV	
10	MO	140.2	19	WV	71.0	19	NM	16.7	1	VA			AL	
19	KS	140.5	20	AR	69.6		IL			AK			СТ	
21	AL	139.9	21	СТ	69.0	20	NE	16.6		AL			IA	
22	TX	137.1	22	SC	68 7		RI		20	KY	1.3	19	KS	24
23	GA	137.0	22	WI	00.7	23	KS	16.1		RI		17	NM	2.7
24	MS	136.7	24	LA	68.2	24	GA	16.0		WV			NY	
25	WV	135.8	25	GA	67.9	25	LA	15.8	25	MT	1.2		OR	
26	AK	134.5	26	NC	67.5	26	AR	15.6	27	ND	1.1		PA	
27	MI NV	134.2	27	INIM DA	66.0	27	VA DA	15.5	27	UK	1.1	27	DE ID	2.2
20	ΡΔ	132.7	20	TX TX	66.8	20	rA MI	14.7		0.S.	1.0	27		2.3
30	OH	129.9	30	DE	66.1	2)	TN	17.5		IA			GA	
31	MD	129.3	50	MD	00.1	30	UT	14.3		ID			MA	
32	OK	129.0	31	OK	65.9		U.S.	14.2	28	MA	1.0	30	WI	2.2
22	IN	120 (	33	OH	64.1	32	DE	14.1		MO			VA	
55	NC	128.6		U.S.	63.9		CO		1	NH			U.S.	2.1
35	DE	127.9	34	MS	63.5	33	MT	13.5		SC			AK	
36	IL	125.3	35	IL	63.4		TX			NY			FL	
	U.S.	124.7	36	TN	62.5	36	MO	13.4	35	OK	0.8	34	KY	2.1
37	MN	124.5	37	KY	62.4	37	AZ	13.3		PA		-	MS	
38	WI	120.6	38	HI	61.4	38	NY	12.9	20		0.7		OH	
39		119.7	39	MN	60.9	39	OK MD	12.4	38	NC	0.7		IN	
40	MI	119.4	40	AK FI	59.4	40	WI	12.2	41	INIM	0.6	40	WA NV	1.9
41	FI	117.5	42	CO	58.8	41	HI	11.9	41	NV	0.0	42		1.8
43	HI	115.2	43	IN	58.5	43	WV	11.0	42	TN	0.5	72	IN	1.0
44	WA	110.3	44	MI	57.3	44	FL	10.9		TX	. ·	43	NJ	1.7
45	OR	108.8	45	ID	55.4	45	ID	10.8	44	WA	0.4	45	MI	1.6
46	ID	99.3	46	NV	52.7	46	CA	10.4		FL		46	IL	1.5
47	SC	96.1	47	WA	51.8	47	WA	9.9	16	GA	0.2		AZ	
48	AZ	91.5	48	OR	51.2	48	OH	9.7	40	KS	0.5	47	DC	1.3
49	UT	90.1	49	CA	48.0	49	NV	9.2		OH			UT	
50	CA	89.9	50	AZ	46.9	50	AL	9.1	50	AZ	0.1	50	MN	1.2
51	NV	78.4	51	UT	45.2	51	SC	5.2	51	WY	1.8	51	CA	1.1

# Table A.29 (corresponds to Table 4.3)Staffing RatesFull Time Equivalent Staff Members Per 1,000 Students, FY 2006

Continued on next page.

Librarians		Student Support/Other			Sch	School Admin.		School District Admin			Admin. Support Staff				
			Sup	port S	taff					Aumm	•		Stall		
Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	Rank	State	Rate	
1	MT	2.6	1	DC	47.2	1	RI	9.2	1	ND	4.9	1	AK	17.7	
2	VT	2.3	2	AR	45.5	2	TX	7.0	2	OH	4.3	2	OH	16.3	
2	AR	2.1	3	NM	41.7	3	AK	5.8	2	SD	27	3	DC	14.3	
5	RI	2.1	4	WY	40.3	4	DC	5.2	5	WY	5.7	4	WY	12.0	
_	KS		5	KY	39.3	5	ME	4.9	5	ME	3.4	5	KY	11.9	
5	MS	2.0	6	ME	39.0	6	SC	4.8	6	AK	3.2	6	FL	11.5	
	ND		/		37.6	0		4.6	7	DE	27	/	AL	11.2	
8	AL NE	1.9	0	GA	36.9	0		4.3	/	NH	2.1	8	INIVI VA	11.1	
	LA		10	WA	36.6		CT		10	MN	2.5	10	VA	11.0	
10	MO	1.8	11	NJ	36.4		GA		11	CT	2.4	11	NJ	10.7	
10	NC		12	MS	36.2	9	MA	4.0		IA		12	MA	10.6	
12	VA	1.7	13	LA	35.9		ND		12	MS	2.0	13	ME	10.5	
	HI		14	VA	35.7		WY		12	NE	2.0	14	OK	10.4	
	KY		15	WV	35.1	15	LA	3.0		NM		15	MO	9.6	
14	OK	16	16	IN	35.0	15	MD	5.7	16	MI	1.9	16	PA	9.2	
	SC	1.0	17	KS	34.8	17	KS	3.8	17	IL	1.8	17	CO	8.7	
	TN		18	MD	34.4	10	NM	2.7		TX			OR	0.7	
20	W Y NIL	1.5	19	V1 MO	34.1	18	IN	3.7	19	DC	1.7		U.S.	8./	
20		1.3	20	ND	33.6		MS		21	WV	1.6	19	UA II	8.0	
	CT		21	NE	33.3	21	MT	3.6	21	MO	1.0	20	MT	84	
	GA		23	SD	32.8	21	NE	5.0	22	OR	1.5	20	SC	0.1	
21	MD	1.4	24	PA	32.4		AR			AR			IA		
	WI		25	AL	31.6	24	NC	3.5	24	CO	1.4	23	MS	8.3	
	WV		26	OK	30.9		U.S.	3.4	24	GA	1.4		LA		
27	ME	1.3	27	IA	30.7		MO			VT		25	MI	8.2	
	NY		28	DE	29.8	26	OK	3.4		U.S.	1.3		NY		
28	PA	1.2	29	OH	29.7		VA		28	KY	1.3	28	HI	8.0	
-	SD	-	30	MI	29.6	29	KY	3.3		VA		29	CT	7.9	
	WA	1 1	22	NH	20.4		SD		30	NC	1.2	20	IN	76	
	<u> </u>	1.1	32	MN	29.4	31	DE	3.2		пі MD		30	WIN	7.0	
	DF		34	CO	29.0		II			MT		33	NC	74	
32	IA	1.1		U.S.	29.0	33	NY	3.1	32	NY	1.1	34	WI	7.3	
	TX		35	IL	28.9		OR		-	PA	-	35	NE	7.2	
	FL		36	FL	28.8		IN			WI		36	TX	7.1	
	IL		37	TN	28.3	36	MI	2.9		IN		37	GA	6.9	
36	MA	1.0	38	AK	26.9		NJ		37	OK	1.0	38	AR	6.8	
	MN		39	MA	26.2	39	WI	2.8		NJ		39	KS	6.4	
	NJ		40	NY	25.2		FL		40	RI	0.9		SD		
41	IN	0.0	41	WI NC	23.8	40	HI	2.7	42	WA	0.8	41	DE	6.3	
41	NV	0.9	42	OP	23.7		WA		42	EI EI	0.8		MD		
			43	HI	22.4		NH		43	NV	0.7	43	NH	6.1	
	MI		45	AZ	20.8	44	OH	2.6		ID		45	AZ	5.8	
44	44 <u>MI</u> (	0.8	46	ID	20.7		PA		45	LA	0.5	46	NV	5.7	
	OR		47	UT	18.6	47	NV	2.4		AZ		47	UT	5.6	
48	ID	0.6	48	CA	18.0	4/	MN	2.4	47	CA	0.4	48	RI	5.4	
49	DC	0.5	49	RI	14.9	49	CA	2.2		SC		49	ID	5.2	
-17	UT	0.5	50	NV	4.4	50	UT	2.1	50	AL	03	50	ND	4.8	
51	CA	0.2	51	SC	3.4	20	AZ	£.1	20	TN	0.5	51	WA	4.7	

#### Table A.29 (continued)

Source: U.S. Dept. of Ed. Inst. Natl. Public 5-6, 15-16.

Rank	State	%
1	NY	78.0
2	СТ	74.2
3	KY	70.6
4	IN	61.9
5	WV	61.1
6	AL	60.6
7	MA	60.2
8	OR	58.0
0	MD	5( )
9	WA	30.5
11	NV	55.8
12	MI	55.5
12	HI	55.5
14	СО	54.0
15	IL	53.5
16	DE	53.4
17	OH	52.7
1 /	GA	52.7
19	TN	52.3
20	RI	51.6
21	DC	51.3
22	SC	51.0
23	MO	50.8
24	PA	50.3
25	MN	50.2
26	AZ	49.2
	U.S.	48.1
27	VT	45.4
28	WI	45.1
29	KS	44.5
30	CA	43.1
31	NH	42.5
32	NJ	42.2
33	AK	41.3
34	NM	41.0
35	NE	39.5
36	VA	39.4
37	AR	38.4
38	WY	37.2
39	FL	36.6
40	MS	35.5
41	ME	34.3
41	IA	34.3
43	LA	33.9
44	MT	33.6
45	OK	33.4
46	UT	32.6
47	NC	31.7
48	ND	27.4
49	TX	27.2
50	ID	27.0
51	SD	26.2

### Table A.30 (corresponds to Table 4.4)Percentage of Teachers With Advanced Degrees, FY 2004

Source: U.S. Dept. of Ed. Inst. Natl. Digest, 2007, Table 64.

Rank	State	Score	Grade
1	SC	91.9	A-
2	AR	88.9	B+
3	NC	84.8	
4	FL	83.2	р
5	GA	82.0	Б
5	LA	85.0	
7	VA	82.0	
8	AL	81.1	
9	KY	80.9	B-
10	OK	80.1	
11	IA	79.9	
12	NY	79.2	
13	WV	78.9	
14	OH	78.1	
15	PA	78.0	C+
16	HI	77.8	
17	NM	77.1	
18	WI	77.0	
19	DE	76.8	
20	MA	75.9	
21	TN	75.1	
22	MO	75.0	C
23	TX	73.1	, č
24	NJ	73.0	
	WA		
	U.S.	73.0	С
26	MD	72.0	
20	NV	71.0	
28	IN	/1.9	C
29		/1.8	C-
30	CA	70.8	
22	VT	70.2	
32	V I NE	/0.1	
33	ND	69.1	
35	II	68.1	
36		68.0	
37	MI	67.1	D+
51	KS	57.1	
38	MN	67.0	
	AZ		
40	MT	66.9	
42	NH	66.1	
43	ME	65.9	
44	MS	65.1	
45	ID	64.2	D
46	SD	63.9	
47	RI	62.9	1
48	DC	62.0	
49	WY	61.9	D-
50	AK	61.0	
51	OR	57.0	F
	•	•	

## Table A.31 (corresponds to Table 4.5)Quality Counts Teaching Profession Index, 2008

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

All T	Title I-Eligible Scho	ols	Title I Schools with Schoolwide Programs					
Rank	State	%	Rank	State	%			
1	OR	99.5	1	DC	80.8			
2	IN	90.1	2	MS	59.9			
3	DC	84.3	3	TX	58.4			
4	MT	81.2	4	HI	57.2			
	AR		5	KY	55.0			
5	ME	73.6	3	OK	55.9			
	OK		7	LA	55.5			
8	ID	71.1	8	AR	53.3			
9	HI	70.5	9	NM	52.9			
10	OH	68.8	10	AL	46.4			
11	NY	68.3	11	TN	45.1			
12	PA	67.6	12	SC	42.4			
13	NM	67.4	13	GA	39.9			
14	ND	64.4	14	NC	39.2			
15	WI	63.6	15	VA	37.1			
16	MS	63.5	16	FL	36.4			
17	KY	62.9	17	CA	35.7			
17	LA	02.7	18	WV	34.8			
19	TX	62.7		U.S.	31.4			
20	CA	57.4	19	DE	30.6			
21	WY	56.5	20	NY	30.5			
22	VT	56.1	21	AZ	27.4			
	MA	00.1	22	MI	27.0			
24	AK	55.4	23	OR	26.3			
	U.S.	55.3	24	MA	25.6			
25	AL	54.8	25	OH	24.6			
26	WA	54.2	26	VT	24.2			
27	CO	53.8	27	WA	22.9			
28	TN H	53.4	20	MD	22.7			
29	IL	52.5	29	AK	22.7			
30	MO	49.3	30		21.9			
31	KS	48.5	31	INV.	21.5			
32	GA	48.4	32	IL SD	21.4			
24	AL	48.0	24		20.8			
25	NU	47.8	25	VS	20.0			
35	SD NI	47.5	35	MT	19.8			
37	WV	47.2 45.7	30	WV	19.0			
38	DF	45.0	38	RI	18.9			
30	SC	44.9	30	PΔ	18.8			
40	IA	44 2	40	MO	18.0			
41	СТ	44.1	41	NE	16.7			
42	RI	43.2	42	WI	15.2			
	NE		43	ID	13.6			
43	FL	37.6	44	ND	13.0			
45	VA	37.1	45	CT	12.6			
46	MN	35.9	46	MN	10.4			
47	MI	25.0	47	IA	8.9			
47	MD	27.0	48	IN	8.7			
49	UT	24.8	10	NH	7.1			
50	NV	21.5	49	ME	7.1			
n.a.	NJ	n.a.	n.a.	NJ	n.a.			

#### Table A.32 (corresponds to Table 4.9)Title I-Eligible Schools and Schoolwide Programs As a Percentage of All Schools, FY 2006

Notes: Title I-eligible schools are encouraged to create schoolwide programs if 40 percent or more of their students are eligible for free or reduced-price lunch. Otherwise, assistance is targeted to individual students who are eligible for free or reduced-price lunch.

Source: Staff compilation of data from U.S. Dept. of Ed. Numbers and Types of Public Elem. and Secondary Schools 4-5.

	Over	all		Acc	ess		Us	se	Capacity			
Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)	
1	WV	A (95.3)	1	SD	A (100 0)		AZ		1	GA	A (100.0)	
2	SD	A- (92.0)	1	WI	A (100.0)	1	GA	A (100 0)	1	WV	A (100.0)	
3	GA	A- (91.2)	3	WV	A (96.3)	1	NC	A (100.0)	3	KY	A (93.2)	
4	KY	B+ (88.5)	4	WY	A (93.8)		UT			AK		
5	VA	B+ (88.7)	5	ND	A (92.5)		AR			AR		
6	FL	B (85.4)	6	ME	A- (91.3)		FL			CA		
7	ND	B (86.1)	7	VA	A- (90.0)		ID			IA		
8	LA	B- (82.0)	8	KS	B+(87.5)		KY			MD		
9	PA	B- (81.8)	10	NE DA	D (9( 2)		LA		4	NY	B- (79.5)	
10	NC	B- (81.5)	10	PA	B (86.3)	5	MD	A- (89.8)		OH		
11	UK WI	B- (80.9)	11	MT	B (85.0) B (82.5)		MO					
12		B- (80.0) B- (80.1)	12	INI	B (85.5)		OK			FA SC		
13	KS	B- (79.9)	13	KV	B (82.5)		SD			VT		
	SC	B (19.9)	15	CT	B- (81 3)		VA			WA		
15	WY	B- (79.7)	10	FL	B (01.5)		WV			CT		
17	MO	C+(77.0)	16	SC	B- (80.0)		AL			FL		
18	AK	C+ (77.5)		VT	()		AK			IL		
19	MD	C+ (77.9)	19	NC	C+(78.8)		CO			LA		
20	AZ	$C \downarrow (72.2)$	20	ID	C+ (78.5)		HI		16	NH	B (86.3)	
20	IN	C+(78.2)	21	IA	C + (77.5)		IL			ND		
22	СТ	C + (78.0)	21	MN	$C^{+}(77.5)$		IN			SD		
22	ME	C+(78.9)		MA			KS			TX		
24	IL	C+(79.4)	23	NJ	C (75.0)		ME			VA		
21	TX	0. ((). ()		OH			MA			AL		
26	NE	C (76.5)		AK		17	MN	B- (79.5)		AZ		
27	VT	C (76.3)	26	GA	C (73.5)		MS			CO		
28	MI	C (76.0)		OK			NJ			DE		
29		C (75.8)	20		C( <b>72</b> , <b>5</b> )		ND			IN		
30	MA	C (75.7)	29	MI	C (72.5)		DA		25	KS	C(72.7)	
32	INJ	C (75.4)	32		$C_{(71.0)}$		PA SC		23	MA	C (72.7)	
32	NH	C(73.4)	52		C-(/1.0)		TN			MO		
34	OH	C(74.6)	33	TN	C- (70.0)		TX			NE		
35	MN	C (74.3)		MO			WY			NJ		
36	TN	C (74.1)	35	NH	D+ (68.5)		CA			TN		
37	UT	C (74.0)	37	WA	D+(67.0)		СТ			WI		
38	NM	C (73.4)	38	NY	D (66.0)		DE			HI		
39	AL	C (72.6)	39	AL	D (65.5)		IA			ME		
40	CO	C- (72.4)	40	CO	D (65.0)		MT			MI		
41	WA	C- (71.9)	41	MD	D (64 5)		NE		38	MN	D (65.8)	
42	NY	C- (71.6)	71	NV	D (04.5)		NV		50	NM	D (05.0)	
43	MT	C- (70.6)	43	UT	D (63.0)	36	NH	D+ (69.3)		NC		
44	MS	C- (70.4)	44	AZ	D-(62.0)		NM			RI		
45	CA	D+(69.3)	4	DC	D ((0.5)	-	NY			WY		
46		D+(68.1)	46	DE	D- (60.5)	-	OH			DC		
4/	DE OP	D+(6/.5)		CA LII			KI VT			ID MT		
48	DI	D(03.8)	17	пі MS	F (50 0)				46	IVI I NV	F (59.0)	
49 50	NV	D(04.7)	4/	OR	F (39.0)		WI			OR		
51	DC	D- (60 0)		RI		51	DC	F (59 0)		UT		

## Table A.33 (corresponds to Table 4.10)Education Technology Overall Grade, Access, Use, and Capacity, 2008

 51
 DC
 D-(60.0)
 RI
 51
 DC
 F (59.0)

 Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

Access Gra	Access to Technology Grade and Score Rank State Grade			Percentage of 4 <sup>th</sup> Grade Students With Access to Computers 2007			Percentage of 8 <sup>th</sup> Grade Students With Access to Computers 2007			udents structio Comput 2006	Per onal ter	Students Per High-speed Internet-connected Computer 2006			
Rank	State	Grade (Score)	Rank	State	%	Rank	State	%	Rank	State	Ratio	Rank	State	Ratio	
1	SD	A (100 0)	1	WV	100%	1	ME	100%	1	UT	5.4	1	UT	5.3	
1	WI	A (100.0)	2	SD	99%	2	WV	97%	2	DE	5.2	2	CA	5.0	
3	WV	A (96.3)		KY		3	VA	95%	3	CA	5.1	2	MS	5.0	
4	WY	A (93.8)	3	NC	98%	4	WY	94%	4	MS	5.0	4	DE	4.9	
5	ND ME	A (92.5)		ND		5	PA	93%	(	KI AI	4.0	5	AL	4.8	
6	ME VA	A- (91.3)					SD CT		6	AL AZ	4.8	6	N V P I	4.6	
/	VA KS	A- (90.0)		FL			KV		7	AZ NV	4.7	8	HI	15	
8	NE	B+ (87.5)	7	ID	97%	7	ND	92%	9	NH	4.6	0	MD	т.5	
10	PA	B (86.3)	,	MI	2170		WI		,	HI	1.0	9	OR	4.4	
11	NM	B (85.0)		SC			GA		10	MD	4.5	1.1	AZ	4.2	
12	MT	B (83.5)		IN		1.1	IN	0.00/		OR		11	LA	4.3	
13	IN	B (82.5)		KS		11	NJ	90%	13	DC	13	13	DC	4.2	
13	KY	D (02.5)		LA			SC		15	NY	ч.5		CO		
15	CT	B- (81.3)		NJ		15	MN	89%		CO		14	NH	4.1	
16	FL	D (00.0)	12	NM	96%	-	VT		15	LA	4.1		NY		
16	SC	B- (80.0)		OK		17	IL	88%		MI		10	IN	2.0	
10		$C \pm (78.5)$		TN TV			MA		10	IN	4.0	18		3.9	
20	ID NC	C+(78.8)							19	IL KV	4.0		AK KV		
20	IA	C+(70.0)		WY		19	NE	87%	20	NJ	39	19	MI	3.8	
21	MN	C+ (77.5)		AR			NH	0,7,0	20	NC	5.7		NC		
	MA			СТ			NC			AR			GA	2.7	
23	NJ	C (75.0)		IL		24	MD	9/0/		СТ		23	MN	3.7	
	OH			IA		24         OH           26         MT           27         UT	OH	80%	23	GA	3.8		NJ		
	AK		22	MN	95%		85%		MA	<u> </u>	25	SC	3.6		
26	GA	C (73.5)		NE			UT	84%		SC			WA		
	OK			NV		•	MI	0.00	28	MN	3.7	28	CT	3.5	
20	IL M	O(72.5)		NH		28	NY	83%		MO			MO		
29	MI	C (72.5)		PA			IN		20	OH	25		MA		
32		$C_{-}(71.0)$		ME		31	IA NV	82%	29	TX	3.5	30	OK	3.4	
52	LA	C=(/1.0)		MD		51	TX	0270		WA			TX		
33	TN	C- (70.0)	31	MA	94%		AK		34	PA	3.4		AK		
25	MO	$D_{\pm}$ (69.5)		OH		34	DE	81%		FL		24	ID	2.2	
33	NH	ע⊤ (08.5)		VT			FL		35	ID	3.3	54	IN	5.5	
37	WA	$D+(\overline{67.0})$		CA		37	CO	80%		IN			IA		
38	NY	D (66.0)		GA			AZ			AK		38	FL	3.2	
39	AL	D (65.5)	27	MS	020/	38	MO	79%	38	IA	3.2		PA		
39	CO MD	D (65.0)	57	MO	93%		NM ID			W V ND			NM		
41	NIV	D (64.5)		NV		41	ID MS	78%	41	ND VT	2 1	40	ND VT	3.1	
43		D (63 0)		WA					41	V I V A	3.1		WI		
	AZ	5 (05.0)	44	UT	92%	43	OK	76%		MT			VA		
44	DC	D- (62.0)	45	DE	010/		OR		4.4	NE	2.0	44	WV	3.0	
46	DE	D-(60.5)	45	OR	91%	16	AR	750/	44	NM	3.0	46	MT	2.9	
	CA		17	AK	80%	40	RI	13%0		WI		47	NE	2.8	
	HI		т/	AZ	07/0	48	HI	74%	48	KS	2.6	48	KS	2.6	
47	MS	F (59.0)	49	DC	88%	49	WA	73%	49	WY	2.5		WY		
	OR		<u> </u>	RI	070/	50	CA	72%	50	ME	2.1	50	ME	1.9	
	KI		51	HI	8/%	51	DC	61%	51	SD	2.0		SD		

#### Table A.34 (corresponds to Table 4.11)Access to Technology, 2006 and 2007

Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

Use of	of Tech	nology		State Standards			
Gra	de and	Score		For Students	State Tests	State Has	State Offers
D 1	<b>G</b> ( )	Grade	Total	Include	<b>Students On</b>	Established a	<b>Computer-based</b>
Rank	State	(Score)	Implemented	Technology	Technology	Virtual School	Assessments
	AZ	(	P	✓	✓	✓	✓
	GA			√	✓	✓	✓
1	NC	A (100.0)	4	√	√	✓	√
	UT			√	✓	✓	✓
	AR			✓		✓	✓
	FL			✓		✓	✓
	ID			√		✓	✓
	KY			✓		✓	✓
	LA			✓		✓	<i>√</i>
5	MD	A- (89.8)	3	∕		✓ ✓	<b>√</b>
	MI			✓		<b>v</b>	✓ ✓
	MO			<b>√</b>		V	<b>√</b>
	OK SD			· ·		• •	V 
	SD VA					• •	<b>v</b>
	WV			· · ·		· ·	· ·
				· · ·			•
	AL			✓		· · · · · · · · · · · · · · · · · · ·	
	CO			√		✓	
	HI			√		1	
	IL			√		✓	
	IN			√			√
	KS			√			✓
	MA			✓			✓
	ME			✓			✓
17	MN	B- (79.5)	2	✓			✓
	MS					✓	✓
	ND			✓		✓	
	NJ						<i>✓</i>
	OR			∕			✓
	PA			↓ 	~	/	
	SC TN			<b>√</b>		v	
	TN TV			• •			V
	1 A WV						<b>v</b>
	CA			· •			
	CT			·			
	DE			✓		<u> </u>	
	IA					✓	
	MT			✓			
	NE			✓			
	NH			✓			
36	NM	D+(69.3)	1	√			
	NV	. /		✓			
	NY			✓			
	OH			✓			
	RI			✓			
	VT			✓			
	WA			✓			
L	WI			✓			
51	DC	F (59.0)	0				
Total	Numh	er With I	ndicated Policies	48	5	25	27

#### Table A.35 (corresponds to Table 4.12) Use of Technology, FY 2008

 I of all Number with Indicated Policies
 48
 5

 Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

~	Canacity To Use					Requirem	ents for an	State Requires			
	pacity '	To Use		<u> </u>		Initial Lice	ense Include	Technology PD or			
C	lechno	logy	<b>T</b> ( ) // C	State St	andards	Tech	nology	I raining	or Testing		
Gra	ade and	1 Score	I otal # of	Include 1	ecnnology	Coursewol	rk or a lest	for Rece	runcation		
Rank	State	(Score)	Implemented	Teachers	Admin.	Teachers	Admin.	Teachers	Admin.		
1	GA	A (100.0)	6	<ul> <li>✓</li> </ul>	<b>√</b>	<i>✓</i>	✓	✓			
3	WV KV	A (03.2)	5	✓ ✓	✓ ✓	 ✓	∕	✓ ✓	×		
3	CT	A (95.2)	5	· ·	· ✓	· ✓	•	· ·			
	FL			~	✓	~	✓				
	IL			✓	✓	√	✓				
	LA			✓	✓	✓			√		
4	NH	B (86.3)	4	✓	✓			✓	✓		
	ND			✓	✓	~	✓				
	SD			<ul> <li>✓</li> </ul>	<b>√</b>	✓	✓				
	TX			✓	∕		✓				
	VA			<b>↓</b>	∕	✓	<b>√</b>				
	AK			<b>V</b>	✓			×			
	AK			×				×	*		
			• •		• •			v			
	MD			· ·	· ·	· ·					
	NY			· ·	· •	· ·					
13	OH	B- (79.5)	3	✓ ✓	~	~					
	OK			✓	✓	✓					
	PA			~	✓	✓					
	SC			✓	✓			✓			
	VT			✓	~	~					
	WA			✓	✓			✓			
	AL			✓	✓						
	AZ			✓							
	CO			∕	<b>v</b>						
	DE			<b>√</b>	<b>√</b>						
	IN			V	*						
25	K5 MA	C(72.7)	2	• •	• ·						
23	MS	C(12.1)	2	· ·	· ·						
	MO			· ·	· ·						
	NE			✓	✓						
	NJ			✓	✓						
	TN			✓	✓						
	WI			✓	✓						
	HI										
	ME			<ul> <li>✓</li> </ul>							
	MI			✓							
38	MN	D (65.8)	1	✓							
	NM	. /						×			
	NC DI			×							
	WV										
	DC										
	ID										
	MT	F (50.0)	_								
46	NV	F (59.0)	0								
	OR										
	UT										
Total	States	With Indi	cated Policies	44	35	19	9	10	6		

#### Table A.36 (corresponds to Table 4.13)Capacity To Use Technology, FY 2008

Note: PD = professional development.

Source: Editorial Projects in Education. Education Week's Technology Counts 2008.

Rank	State	%
1	ME	53.4
2	VT	52.1
3	MS	51.3
4	WV	48.6
5	NC	47.0
6	SD	43.6
7	ND	40.3
8	KY	40.1
9	AR	38.4
10	IA	37.6
11	МТ	37.1
12	NH	33.5
13	AL	31.5
14	OK	31.1
15	VA	29.6
16	SC	29.2
17	LA	28.1
18	KS	28.0
10	NF	27.6
20	IN	27.0
20		27.2
21		20.8
22		20.7
23	WV	23.3
24	W I MO	24.3
25	MO	24.2
20	MIN	23.0
27	UD UD	23.0
28		22.7
29	MI	20.8
30	PA	19.6
51		17.7
	0.5. NV	17.4
32		12.9
24		12.4
25	NM	12.2
33		11.9
36		11.8
20	U	11.2
38		11.2
39	WA MD	10.0
40	MD	9.1
41	INJ A 7	8.0
42	AL	/./
43	FL PI	/.4
44	KI	5.3
45		4.8
46		4.3
4/	MA	4.1
48	NV	2.4
49	DC	0.0
	HI	
n.a.	WI	n.a.

## Table A.37 (corresponds to Table 4.14)Students in Rural Areas, FY 2006

Source: U.S. Dept. of Ed. Inst. Natl. Numbers and Types of Public Elem. and Secondary Agencies 12.

	FY 1994			FY 2004		Percent Change				
Rank	State	%	Rank	State	%	Rank	State	%		
1	DC	24.4	1	DC	18.0	1	СО	-71.0		
2	FL	20.1	2	MD	13.5	2	WY*	-65.9		
3	MD	19.9	3	FL	11.2	3	KS	65.7		
4	DE	18.7	4	NY	10.5	3	RI*	-05.7		
5	NC	17.1	5	LA	9.9	5	AR	-65.3		
6	LA	17.0	6	PA	9.5	6	VT	-60.5		
7	NY	16.2	7	MI	9.3	7	OH	-59.3		
8	SC	15.3	8	HI	9.1	8	MS	-58.9		
9	OH	15.2	9	AK	8.9	9	DE	-58.8		
10	VA	14.9	10	NC	8.7	10	WY	-57.8		
11	GA	14.0	11	SC	8.6	11	VA	-56.3		
	KY		12	MO	8.3	12	AL	-54.3		
	AR		13	MN	8.2	13	GA	-54.2		
13	IN	13.8	14	IL	8.0	14	UT	-53.5		
	WI		15	KY	7.9	15	OR	-52.0		
16	AK	13.7	16	NM	7.8	16	NC	-49.2		
17	MS	13.4	17	TX	7.7	17	IA	-47.9		
10	RI	10.0		DE		10	NH	15.0		
19	AL	13.3		U.S.	7.5	19	IN	-47.8		
20	NV	13.2	19	NE	7.5	20	TN	-47.2		
21	<u> </u>	13.1	20	NV	7.3	21	AZ	-46.9		
22	AZ	13.0	21	IN	7.2	22	WA	-46.7		
	U.S.	12.8		WV		23	NJ	-45.4		
23	NM	12.8	23		6.9	24	OK ID	-44.7		
25	WA	10.7	25	AZ	( )	25		-44.6		
25		12.7	25	WA	6.8	07	NV			
26	MO	12.6	26	IN	6.6	27	FL	-44.4		
27	IN	12.5	27	VA	6.5	28	KY	-43.6		
28	VI	12.4	28	GA	6.4	29	SC ME	-43.6		
29	CI	11.9	20	MA	( )	30	ME	-42.0		
30	OR	11.5	30	OH	6.2	31		-41.9		
31	WV	11.4		CA		32	LA	-41./		
32		11.2	31	UK	6.1		U.S.	-41.4		
33	NH	11.1		INI I	-	33	MA	-40.9		
34	DA	11.0	25	AL	5 0	25	1 A NM	-39.4		
	PA II		35	ND	5.6	33	INIVI WVV	-38.9		
		4	50	MS	5.0	30		-30.7		
36	MA	10.8	37	OR	5.5	38	NV	-35.2		
	MI	4	30	ID	5 /	30	MO	-33.2		
40	NE	10.4	39		5.4	39	MD	-33.9		
40	HI	9.0	40	ME	5.5	40 <u>41</u>	NE	-32.0		
42	ID	9.9	41	LIT	5.2	42	DC	-263		
43	MN	9.6		IA		42	П	-26.3		
44	IA	9.0	43	VT	4.9	44	MT	-20.2		
<b>--</b>	ME	ד. ע	45	AR	4 8	45	SD	-183		
45	WY	9.0	46	WI	4 7	46	CA	-17.3		
47	NI	79	47	RI*	4.6	47	MN	-15.0		
48	MT	77	48	NI	4 3	48	MI	-13.9		
49	CA	7.4		CO	-	49	PA	-13.3		
50	SD	6.5	49	WY*	3.8	50	HI	-8.2		
51	ND	5.5	51	KS	3.7	51	ND	1.0		

#### Table A.38 (corresponds to Table 4.15) Public School Teachers Threatened With Injury by a Student at School in Past 12 Months FY 1994 and FY 2004

Notes: \*Interpret with caution due to low frequency (an estimated 300 teachers in Wyoming and 600 in Rhode Island). Staff calculated the percent change using unrounded percentages for FY 1994 and FY 2004.

Source: Staff compilation of data from U.S. Dept. of Ed. Inst. Natl. Indicators 79.

Table A.39 (corresponds to Table 4.16)
Public School Students Involved in Violence or Substance Abuse in Previous 12 Months
Average of 2003 and 2005 Surveys

On School Property													Anywhere			
			Were '	Threater	ed or		V						v			
Were	in a Ph	ysical	Ini	ured Wi	th	Had D	riigs Avs	ailable	Use	d Alcoh	പ	Used Alcohol				
	Fight		inj	Woonon	UII	mau D	1 455 1110	mabic	USC	u / meon	UI	Use	u meoi	101		
Dank	State	0/	Dank	State	0/	Damle	State	0/	Dault	State	0/	Dank	State	0/		
	State	<b>70</b>		State	<b>70</b>	капк	State	70		State	<b>70</b>		State	<b>70</b>		
2	NM	15.8	2	MD	12.4	1	NIVI	33.5	2	ПI NM	0.0	2	MT	31.0 40.1		
2	MD	13.7	2	NIM	11./	2	IN V	22.4	2	NIVI	7.0	2		49.1		
3	MD	14.9	3	NM SC	10.4	3	AZ	22.0	3		7.1	3	AZ	49.0		
4		14.5	4	SC AZ	10.1	4	ME	33.0	4	AZ	7.0	4	SD	48.4		
5	AK	13.9	5	AZ	9.9	5	HI	32.7	5		6.7	5	WI CO	48.2		
0	AL	13./	0	AK	9.0	0	NJ CA	32.0	0	M I WV	0.5	0		47.4		
/	IN I NIV	13.3	7	INE TV	9.3	/	CT UA	21.5	/	W I SC	6.2	/		47.5		
8	INV	13.4	0		0.2	8		31.5	8	SC	6.0 5.0	8	W Y	47.2		
	0.5.	13.2	9	MI CT	9.2	9	OH	31.0	9		5.9	9	MA	40.8		
9		12.7	10		9.1	10	MA	30.9	10		5.7	10	NJ	40.5		
10	W Y	12.5	11	ID AT	8.9	11		30.7	11	W V	5.5	11	NH	45.0		
11	FL VT	12.4	12	AL	07	12	MI	30.1	12	AK	5.2	12	U MO	45.3		
12	V1 CO	12.2	13	WY	8./	13	NC AD	29.0		DE		13	MO	45.0		
13		12.1	14	UI	8.6	14	AR	29.2	14	VI	5.1	14	NE	44.7		
14	ID WI	11.9		U.S.	8.0	15	SC	29.1		KS		15	DE	44.2		
16	WI	11.0	15	KI	8.4	16	MD	28.9	16	KI	10		UK	4 4 1		
16	MI	11.8	16	W V	8.3	17	IN	28.6	16	MS	4.9		U.S.	44.1		
17	OK	11.7	10	MO	0.0	18	AK	28.4		AK		17	KS	43.9		
18	GA	11.6	18	GA	8.2	19	NH	27.5	19	FL	4.8	18	IA	43.8		
19	IN	11.5	10	NH		20	DE	27.0		MA		20	NY	12.6		
20	KY	11.4	19	AK	8.1		U.S.	27.0		U.S.	4.7	20	KI GG	43.6		
	AZ			FL		21	VI	26.2	21	DC	4.7	21	SC	43.2		
21	RI	11.3	22	OH	8.0	22	AL	26.1		SD		22	IN	42.1		
	IA			NJ	-		MT	22.6	23	NY	4.6	23	AR	43.1		
24	NH	11.2	24	TN	7.9	24	WV	25.6		IA		24	WV	43.0		
	WV			IN		25	TN	25.5	25	NC	4.5	25	VT	42.6		
	IN		25	ME	7.8	26	DC	25.3	26	ND	4.4		ME			
26	NC	11.1	•	IA	- /	27	KY	25.1	27	AL	4.3	27	NV	42.4		
• •	UT		28	CO	7.6	28	RI	25.0		KY		28	NM	42.3		
29	OH	10.8	29	NC	7.5	29	FL	24.4	28	NE	4.1		OH			
30	MT	10.6		MT		30	WI	24.0	-	MI		30	MS	41.8		
	DE		31	KS	7.4	31	NY	23.3		ID		31	TN	41.4		
32	CT	10.5	32	SD	7.3	32	UT	22.7	32	NH	4.0	32	KY	41.3		
33	MS	10.3	33	NY	7.2	33	NE	22.6		GA		33	FL	41.2		
34	MA	10.2	34	NV	7.0	34	MS	22.3	34	TN	3.9	34	MI	41.0		
35	NJ	10.1	35	HI	6.9	35	ID	22.2	35	ME	3.8	35	NC	40.9		
	KS			DE		36	SD	21.5	36	NJ	3.7	36	AL	39.8		
	NE		37	VT	6.8	37	CO	21.2	37	IN	3.6	36	MD	39.8		
37	MO	10.0	38	OK	6.7	38	WY	20.4		OH		38	GA	38.8		
	HI		39	KY	6.6		ND		39	OK	3.5	39	AK	38.7		
40	ME	96		MS		40	OK	20.3	40	MD	3.2	40	ID	37.3		
	ND	2.0	41	WI	6.5	41	MO	19.9	41	MO	2.9	41	HI	34.8		
42	SD	8.7	42	ND	6.2	42	KS	16.7		UT	2.9	42	DC	28.5		
43	AK	8.6	43	MA	5.9	43	IA	15.5		CA		43	UT	18.5		
	CA			CA			CA			IL			CA			
	IL			IL			IL			LA			IL			
	LA			LA			LA			MN			LA			
n.a	MN	n.a	n.a	MN	n.a	n.a	MN	n.a	n.a.	OR	n.a.	n.a	MN	n.a		
	OR			OR			OR			PA			OR			
	PA			PA			PA			VA			PA	PA		
	VA			VA			VA			WA			VA			
1	WA		1	WA			WA			WI	1	1	WA	1		

Note: Because some states reported only in 2003 and some reported only in 2005, staff averaged the 2003 and 2005 data; states that failed to report in both 2003 and 2005 are indicated by n.a. Source: Staff compilation using data from U.S. Dept. of Ed. Inst. Natl. *Indicators* 76, 91, 101, and 105.

Rank	State	Unadjusted \$	Rank	State	Cost-adjusted \$
1	DC	18,332	1	VT	16,943
2	NY	16,800	2	WY	16,466
3	NJ	16,743	3	HI	15,625
4	СТ	14,893	4	NY	15,019
5	HI	14,799	5	DC	14,923
6	MA	14,782	6	NJ	14,824
7	VT	14,329	7	ME	14,026
8	WY	13,329	8	MA	13,586
9	RI	13,279	9	PA	13,582
10	DE	13,143	10	СТ	13,543
11	PA	12,942	11	RI	13,144
12	MD	12,430	12	DE	12,924
13	AK	12,229	13	AK	12,885
14	NH	11,753	14	MT	12,706
15	ME	11,709	15	NH	12,630
16	OH	11,606	16	NE	12,466
17	WI	11,160	17	IN	12,418
18	IN	11,028	18	ND	12,243
19	MN	11,010	19	OH	12,120
20	MI	10,900	20	LA	12,060
	U.S.	10,771	21	WV	11,850
21	VA	10,672	22	KS	11,789
22	NE	10,543	23	WI	11,760
23	IL	10,506	24	SD	11,708
24	LA	10,456	25	IA	11,669
25	CA	10,264	26	MD	11,496
26	GA	10,113	27	MN	11,393
27	WV	10,032	28	MI	11,165
28	KS	9,973	29	AR	10,886
29	ND	9,815		U.S.	10,771
30	IA	9,771	30	NM	10,763
31	OR	9,668	31	SC	10,700
32	SC	9,643	32	MO	10,597
33	MO	9,585	33	OR	10,582
34	FL	9,542	34	MS	10,408
35	NM	9,438	35	FL	10,307
36	MT	9,399	36	GA	10,301
37	WA	9,359	37	IL	10,179
38	CO	9,285	38	KY	9,995
39	TX	9,210	39	VA	9,860
40	AR	8,960	40	AL	9,771
41	NV	8,937	41	CO	9,681
42	SD	8,904	42	OK	9,597
43	KY	8,828	43	CA	9,362
44	MS	8,644	44	TX	9,311
45	AL	8,560	45	ID	9,015
46	NC	8,434	46	WA	9,014
47	OK	8,069	47	NV	8,984
48	AZ	8,025	48	NC	8,937
49	TN	7,512	49	AZ	8,762
50	ID	7,257	50	TN	8,174
51	UT	6,802	51	UT	7,425

#### Table A.40 (corresponds to Table 5.1)Revenues Per Pupil, FY 2006

Source: U.S. Dept. of Commerce. Census. Public 2006 11.

$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	%           88.3           59.1           58.6           57.3           57.1           55.5           55.3           54.6           54.5           53.7           52.5           50.8           50.7           50.1           49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	88.3           59.1           58.6           57.3           57.1           55.5           55.3           54.6           54.5           53.7           52.5           50.8           50.7           50.4           49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	59.1           58.6           57.3           57.1           55.5           55.3           54.6           54.5           53.7           52.5           50.8           50.4           50.1           49.8
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{r} 58.6 \\ 57.3 \\ 57.1 \\ 55.5 \\ 55.3 \\ 54.6 \\ 54.5 \\ 53.7 \\ 52.5 \\ 50.8 \\ 50.7 \\ 50.4 \\ - \\ 50.1 \\ 49.8 \end{array}$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	57.3           57.1           55.5           55.3           54.6           54.5           53.7           52.5           50.8           50.4           50.1           49.8
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	57.1           55.5           55.3           54.6           54.5           53.7           52.5           50.8           50.7           50.4           49.8
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	55.5           55.3           54.6           54.5           53.7           52.5           50.8           50.7           50.4           49.8
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	55.3           54.6           54.5           53.7           52.5           50.8           50.7           50.4           -           50.1           49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	54.6           54.5           53.7           52.5           50.8           50.7           50.4           -           50.1           49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	54.5 53.7 52.5 50.8 50.7 50.4 50.1 49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	53.7 52.5 50.8 50.7 50.4 - 50.1 49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	52.5 50.8 50.7 50.4 - 50.1 49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	50.8 50.7 50.4 - 50.1 49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	50.7 50.4 
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	50.4 
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	49.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	49.7
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	48.1
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	47.9
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	46.8
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	45.9
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	45.7
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	45.3
27         ME         8.8         27         SC         44.8         27         AZ           28         IL         8.5         28         WY         44.2         28         WI           29         MO         8.4         30         MA         44.0         30         IA           HI         31         MO         43.6         31         OP	44.4
28         IL         8.5         28         WY         44.2         28         WI           29         MO         8.4         30         MA         44.0         30         IA           HI         31         MO         43.6         31         OP	42.9
29         MO         28         GA         44.2         29         MT           29         IA         8.4         30         MA         44.0         30         IA           HI         31         MO         43.6         31         OP	41.8
29         IA         8.4         30         MA         44.0         30         LA           HI         31         MO         43.6         31         OP	40.1
HI 31 MO 43.6 31 OP	39.6
	39.5
31 WA 8.3 32 TN 43.2 32 OK	37.0
33 MI 81 33 NY 431 KS	
34 PA 79 34 CO 427 33 UT	35.8
35 VT 7.8 35 OH 42.3 35 NV	35.0
36 RI 7.5 LA 36 ID	33.5
36 ME 41.5 37 AL	32.9
<sup>37</sup> DE 7.3 <u>38</u> NJ 41.3 <u>38</u> MI	32.6
20 OH 22 39 FL 40.2 39 NC	31.4
<sup>39</sup> CO 7.2 40 RI 40.0 40 KY	31.3
41 NY 7.1 41 VA 39.6 41 WA	30.6
42 NV 7.0 to MD 2000 42 MS	30.5
43 VA 6.7 42 NH 39.2 43 CA	30.1
44 IN 6.5 44 CT 38.0 44 WV	28.9
45 MD (2) 45 ND 36.2 45 DE	20.7
45 MN 6.2 46 PA 35.0 46 AK	28.2
47 WI 6.0 47 TX 32.9 47 MN	28.2 25.6
48 NH 5.5 48 SD 32.8 48 AR	28.2 25.6 23.1
49 MA 5.3 49 IL 32.3 49 NM	28.2 28.2 25.6 23.1 15.2
50 CT 4.7 50 NE 31.4 50 VT	28.2 28.2 25.6 23.1 15.2 14.3
51 NJ 4.3 n.a. DC n.a. 51 HI	28.2 25.6 23.1 15.2 14.3 5.1

### Table A.41 (corresponds to Table 5.2)Revenues by Source, FY 2006

Source: U.S. Dept. of Commerce. Census. Public, 2006 5.

	Unadjusted		Cost-adjusted						
Rank	State	\$	Rank	State	\$				
1	NY	14,884	1	VT	14,915				
2	NJ	14,630	2	WY	13,832				
3	DC	13,446	3	NY	13,306				
4	VT	12,614	4	NJ	12,953				
5	CT	12,323	5	ME	12,682				
6	MA	11,981	6	AK	12,075				
7	RI	11,769	7	RI	11,650				
8	DE	11,633	8	MT	11,601				
9	AK	11,460	9	PA	11,574				
10	WY	11,197	10	DE	11,440				
11	PA	11,028	11	CT	11,206				
12	MD	10,670	12	WV	11,047				
13	ME	10,586	13	MA	11,011				
14	NH	10,079	14	DC	10,946				
15	WI	9,970	15	NH	10,831				
16	HI	9,876	16	ND	10,732				
17	OH	9,598	17	WI	10,506				
18	MI	9,572	18	HI	10,428				
19	VA	9,447	19	NE	10,330				
20	WV	9,352	20	SD	10,061				
21	IL	9,149	21	OH	10,023				
	U.S.	9,138	22	IA	9,984				
22	MN	9,138	23	KS	9,920				
23	IN	8,793	24	IN	9,901				
24	NE	8,736	25	MD	9,869				
25	ND	8,603	26	MI	9,805				
26	MT	8,581	27	LA	9,692				
27	GA	8,565	28	AR	9,632				
28	OR	8,545	29	MN	9,456				
29	CA	8,486	30	OR	9,353				
30	LA	8,402	31	NM	9,221				
31	KS	8,392		U.S.	9,138				
32	IA	8,360	32	<u>SC</u>	8,978				
33	MO	8,107	33	MO	8,964				
34	SC	8,091	34	IL	8,864				
35		8,080	35	VA	8,728				
30		0,007	27	AL	0 724				
30	WA	7 830	30	MS	0,724				
30	FI	7 750	30	KV	8 675				
40	KY	7,662	40	CO	8 401				
40	SD	7,651	40	FL	8 381				
42	AL	7,646	42	OK	8 280				
43	TX	7,561	43	ID	8,000				
44	NC	7,388	44	NC	7,828				
45	NV	7,345	45	CA	7,741				
46	MS	7,221	46	TX	7,644				
47	OK	6,961	47	WA	7,541				
48	TN	6,883	48	TN	7,489				
49	AZ	6,472	49	NV	7,383				
50	ID	6,440	50	AZ	7,066				
51	UT	5,437	51	UT	5,935				

### Table A.42 (corresponds to Table 5.3)Current Spending Per Pupil in P-12 Fall Enrollment, FY 2006

Sources: Unadjusted numbers are from U.S. Dept. of Commerce. Census. *Public* 8. Staff calculated adjusted numbers using the Comparable Wage Index from U.S. Dept. of Ed. Inst. Natl. "NCES Comparable Wage."

Tota	al Curi	rent				Ins	tructio	on									
S	pendin	g	Total	Instru	ction	Sala	aries O	nlv	Ben	efits O	nlv	Gene	ral Ad	min.	Scho	ol Adm	in.
Rank	State	<u> </u>	Rank	State	S	Rank	State	S	Rank	State	Ś	Rank	State	\$	Rank	State	\$
1	AK	62.68	1	VT	37 57	1	VT	24 19	1	WV	10.34	1	ND	1 90	1	VT	3 98
2	VT	59.44	2	NY	37.44	2	NY	23.98	2	NY	943	2	NE	1.62	2	AK	3.72
3	NI	55 31	3	AK	35.86	3	AK	22.64	3	AK	9.15	3	VT	1 47	3	NM	3 11
4	WV	54 56	4	ME	33.87	4	ME	21.94	4	IN	8 90	4	NH	1 46	5	GA	5.11
5	NY	53.66	5	WV	32.73	5	GA	21.51	5	WI	8.76	5	AR	1.45	4	MI	3.00
6	ME	52.03	6	NJ	32.65	6	NI	21.13	6	MI	8 30	6	IL	1 41	6	WV	2.93
7	WY	51.02	7	AR	30.28	7	AR	20.74	7	VT	8.07	7	WV	1.39	7	WY	2.85
8	AR	50.10	8	GA	30.23	8	ΤХ	20.15	8	ME	7.95		MS		8	ME	2.80
9	MI	50.08	9	WY	30.21	9	WY	20.04	9	NJ	7.19	8	MT	1.38	9	MD	2.75
10	NM	48.92	10	RI	29.20	10	WV	20.01	10	DE	7.18	10	PA	1.35	10	HI	2.73
11	LA	48.77	11	WI	29.07	11	MS	19.87	11	RI	7.15	11	SD	1.31	11	SC	2.70
12	RI	48.41	12	MI	28.45	12	NM	19.58	12	MA	7.11	12	OH	1.29	12	AR	2.68
13	GA	48.21	13	LA	28.36	13	SC	19.52	13	WY	7.10	13	NJ	1.26	13	CA	2.67
14	MS	48.18	14	MS	28.31	14	RI	19.22	14	LA	6.94		WI	1.05	14	NJ	2.66
15	OH	47.72	15	NM	27.78	15	LA	18.99	15	OR	6.78	- 14	KS	1.25	15	IN	2.65
16	WI	47.64	16	IN	27.64	16	KY	18.65	16	UT	6.58	16	OK	1.00	16	AL	2.64
17	SC	47.07	17	MT	27.59	17	WI	18.60	17	СТ	6.39	16	MO	1.23		OR	a. (a.
18	IN	46.63	18	OH	27.51	18	IA	18.56	18	NH	6.23	18	IA	1.21	17	MS	2.63
19	MT	45.64	19	SC	27.42	19	OH	18.43	19	GA	6.19	19	NM	1.19	19	OH	2.60
20	PA	45.03		NH		20	MT	18.35	20	PA	6.10	20	MN	1.17	20	KS	2.59
21	TX	44.26	20	NE	27.21		ND		21	KY	5.99	21	LA	1.16	21	IA	2.57
22	KY	44.02	22	MA	27.14	21	NH	18.06	22	OH	5.92	- 21	AL	1.16		LA	
23	KS	43.57	23	PA	27.04		MI		23	AL	5.90	23	WY	1.13	22	MT	2.52
	U.S.	43.34	24	TX	26.70	24	PA	18.03	24	NE	5.87	24	ME	1.09	24	NC	2.49
2.4	IA	42.20	25	KS	26.49		U.S.	17.93	25	MD	5.81	25	KY	1.06	25	WI	2.44
24	NE	43.30		U.S.	26.43	25	MO	17.89	26	ID	5.72	26	MI	1.00	24	DE	0.40
26	DE	42.83	26	СТ	26.39	24	IN	17.07	27	NM	5.70	27	AK	0.95	26	TX	2.42
27	AL	42.75	27	DE	26.20	26	MN	17.87		U.S.	5.68	28	NY	0.91	20	RI	2.40
28	NH	42.69	28	KY	26.05	28	ID	17.81	28	MS	5.63	29	ID	0.89	28	KY	2.40
29	MA	42.63	29	IA	25.99	29	NE	17.78	29	IA	5.62	30	IN	0.86	30	СТ	2.39
30	CT	42.00	30	ID	25.44	30	KS	17.76	30	HI	5.51	31	CT	0.84		U.S.	2.38
31	OR	41.94	31	MN	25.29	31	NC	17.65	31	AR	5.35		U.S.	0.82	31	NH	2.37
32	ND	41.52	32	ND	25.23	32	CT	17.58	32	SC	5.33	32	TN	0.75	32	ID	2.36
33	OK	41.47	33	AL	25.00	33	IL	17.57	33	MN	5.25	33	TX	0.67	33	CO	2.35
34	IL	41.45	34	OR	24.72	34	VA	17.49	34	VA	5.21	34	NC	0.64	34	NE	2.34
35	ID	41.17	35	MO	24.64	35	MA	16.74	35	MT	5.17	35	GA	0.63	25	MO	2.22
36	MO	40.81	36	VA	24.45	36	DE	16.65	36	CA	5.03	36	DC	0.62	35	UT	2.33
37	HI	40.73	37	IL	24.40	37	MD	16.55	37	ND	4.99	37	VA	0.60	37	VA	2.32
38	VA	39.93	38	UT	24.36	38	AL	16.33	38	IL	4.62	38	NV	0.58	38	NV	2.28
39	CA	39.62	39	HI	24.30	39	CA	16.15	39	NV	4.41	20	OR	0.57	39	OK	2.24
40	MD	39.48	40	CA	24.03	40	UT	16.07	40	OK	4.35	- 39	SC	0.37	40	WA	2.18
41	MN	39.23	41	MD	23.81	41	TN	15.91	41	MO	4.33	41	RI	0.56	41	IL	2.11
42	UT	38.47	42	NC	23.76	42	OK	15.85	42	TN	4.30	42	MA	0.55	42	NY	2.08
43	NC	38.42	43	TN	23.01	43	SD	15.66	43	WA	4.19	43	CO	0.54	42	TN	1.07
44	SD	37.79	44	OK	22.89	44	HI	15.61	44	KS	4.12	44	AZ	0.50	43	ND	1.97
45	AZ	35.94	45	SD	22.40	45	WA	15.24	45	NC	4.08	45	WA	0.47	45	FL	1.90
46	WA	35.93	46	WA	21.47	46	AZ	14.00	46	SD	4.06	43	DE	0.47	46	PA	1.87
47	CO	35.76	47	AZ	20.64	40	OR	14.99	47	AZ	3.74	47	UT	0.43	47	MA	1.92
48	TN	35.55	48	NV	20.39	48	CO	14.48	48	FL	3.56	48	MD	0.42	4/	AZ	1.03
49	NV	33.58	49	CO	20.27	49	FL	12.87	49	CO	3.24	49	CA	0.35	49	SD	1.81
50	FL	33.51	50	FL	19.80	50	NV	12.86	50	TX	3.22	50	FL	0.34	50	MN	1.65
51	DC	30.09	51	DC	15.35	51	DC	9.00	51	DC	0.83	51	HI	0.31	50	DC	1.03

Table A.43 (corresponds to Table 5.4)Current Spending Per \$1,000 in Personal Income, FY 2006

Source: U.S. Dept. of Commerce. Census. Public 2006.

	Instruction		Su	pport Servi	ces	All Other			
Rank	State	%	Rank	State	%	Rank	State	%	
1	NY	69.2	1	DC	45.6	1	OK	10.7	
2	ME	64.3	2	CO	39.5	2	UT	9.2	
3	TN	64.0	3	AK	39.3	3	MN	8.9	
4	NH	63.5	4	MI	39.2	4	AL	8.4	
5	MA	63.4	5	OH	38.4	5	ND	8.1	
6	VT	63.0	6	NM	38.1	6	HI	7.7	
7	NE	62.7	7	NJ	37.5	7	WV	7.0	
0	GA	62.5	0	OR	27.4		FL		
0	CT	02.5	0	IL	57.4	8	MS	6.8	
10	ID	61.6	10	AZ	37.1		KY		
11	NC	61.5	11	RI	36.5	11	SC	6.4	
11	MN	01.5	11	IN	50.5	12	LA	62	
13	UT	61.4	13	OK	36.4	12	MO	0.2	
14	VA	60.9	14	SC	36.0		NC		
15	KS	60.7	15	LA	35.9	14	AZ	6.0	
16	DE	60.6	16	SD	35.8	14	TN	0.0	
17	WI	60.4	17	PA	257		CA		
17	NV	00.4	17	NV	55.7	18	TX	5.8	
19	ND	60.3	19	FL	35.4	19	NM	5.7	
	U.S.	60.2	20	MT	35.3	20	AR	5.6	
20	MD	60.2	21	WA	25.1	21	WA	5.4	
20	MT	00.2	21	IA	55.1	22	IA	5.3	
22	AR	60.0	22	WI	25.0		U.S.	5.2	
23	TX	59.8	23	MD	55.0		SD		
24	IA	59.6	25	CA	34.9	23	MI	5.2	
25	WA	50.4	26	KY	34.7		GA		
23	PA	39.4	27	MS	34.7	26	NE	5.1	
27	RI	59.3	28	DE	34.6	27	ID	5.0	
	MO		29	MO	34.6		PA		
28	WV	59.2		U.S.	34.6	28	KS	4.9	
	CA		30	HI	34.5		MD		
21	SD	50.0		VA			OH		
51	WY	59.0	21	TX	24.4	31	DE	4.8	
33	IN	58.9	51	AR	54.4		ME		
34	OR	58.7		KS			CO		
	NJ		35	AL	34.2	34	IN	4.7	
35	MS	58.5	26	VT	22.0		VA		
	KY		30	WV	55.9	37	WI	4.6	
38	IL	58.4	38	ID	33.4	38	MT	4.5	
39	LA	57.9	38	CT	55.4	30	RI	4.2	
40	HI	57 0	40	MA	22.2	39	IL	4.2	
40	FL	57.8	40	NH	35.2	41	DC	4.1	
42	SC	57.6	42	NC	32.5	41	CT	4.1	
43	AL	57.4	43	GA	32.3	43	NJ	4.0	
44	AK	56.0	44	NE	32.2	11	OR	3.0	
	AZ	50.9	45	ND	31.6		NV	5.7	
46	OH	56.8	46	ME	30.9	46	AK	3.8	
47	NM	56.2	47	TN	30.1	47	WY	3.5	
48	CO	55.8	48	MN	29.6	48	MA	3.4	
49	MI	55.6	49	UT	29.5	49	NH	3.3	
50	OK	52.9	50	NY	27.8	50	VT	3.1	
51	DC	50.3	51	WY	37.5	51	NY	3.0	

#### Table A.44 (corresponds to Table 5.5) Spending on Instruction, Support Services, and Other Functions As Percentages of Current Spending, FY 2006

Source: U.S. Dept. of Commerce. Census. Public 2006.

#### Table A.45 (corresponds to Table 5.6)

Spending on Instruction and Instruction-related, Student Support Services, Administration, and Operations Functions, As Percentages of Current Spending, FY 2006

Instruction &			Stud	lent Sup	port	Ad	ministro	tion	Onorations			
Instr	uction-r	elated		Services	8	Au	mmstra	luon	U	peratio	115	
Rank	State	%	Rank	State	%	Rank	State	%	Rank	State	%	
1	NY	72.0	1	HI	12.0	1	CO	17.7	1	WV	23.1	
2	MA	60.0	2	RI	11.9	2	NV	15.1	2	LA	22.4	
2	TN	09.9	3	NM	9.6	3	OR	13.8	3	DC	22.1	
4	ME	69.2	4	NJ	9.0	4	DE	13.2	4	OK	21.6	
5	MN	69.0	5	VT	7.4	5	OH	13.1	5	MS	21.2	
6	GA	68.0	6	MI	7.3	6	MI	12.7	6	KY	20.9	
7	UT	67.9	7	OR	7.1	7	WI	12.6	-	ND		
8	VA	67.7	8	NH	6.9	8	DC	12.4	8	IN	20.7	
9	NH	67.6	10	SC	(5	9	SD	12.3	9	AL	20.6	
10		67.1	10	WA	6.0	10	U U	12.0	10	DE	20.4	
11	NE	66.7	11	AK AK	0.4	11	IL IA	11.9	11	SD	20.1	
12	AR	00.7	12	DC	63	13	IN	11.7	12		19.9	
13	CT	66.6	12	II.	0.5	14	KS	11.7	14	MO	19.6	
15	MD	00.0		CT		15	WA	11.5		AZ	10.5	
16	WI	66.1	15	OH	6.0	16	ND	11.4	15	TX	19.5	
	U.S.	65.9	17	IA	5.0	17	VT	11.2	17	MT	19.2	
	FL		1/	WY	5.9	1/	WY	11.3	10	PA	10.0	
17	ID	65.9		AZ		19	OK	11.2	18	UT	19.0	
	NC		19	ID	5.6	20	AK	11.1	20	NJ	18.9	
20	MO	65.2		KS		20	NC	11.1		ID		
20	PA	05.2	22	MA	5 5		MT		21	MD	18.7	
22	KY	65.1		SD	5.5	22	NM	11.0		VA		
23	RI	65.0	24	MT	5.4		PA	10.0	24	MI	18.4	
25	TX	(1.0		NC	5.0	25	AL	10.9	26	NM	10.2	
25	SC	64.9		U.S.	5.2		U.S.	10.8	26	KS	18.3	
20	W Y I A	04./	26	AL DA	5.1	26	AK	10.8		IL NE		
27	IA	64.5	27	PA TV	4.9				27	NE SC	18.2	
29	MT	64.4		DF		28	TX	10.6		WA		
30	AZ	64.3	29	GA	48		GA			US	18.1	
50	MS	01.5		VA			MD			AR	10.1	
31	NV	63.9	22	FL	4.7	30	MN	10.5	31	WY	18.1	
	WA		32	MO	4.7		MO		33	ME	17.9	
34	WV	63.8		AR		34	MS	10.4	34	MN	17.8	
35	ND	63.7	34	CA	4.6	35	HI	10.3	35	IA	17.7	
55	OH	05.7		WI		36	LA	10.2		CT		
37	IL	63.5	37	MS	4.5	37	KY	10.0	36	NC	17.6	
38	AL	63.4	38	CO	4.4	38	SC	9.9		TN		
39	HI	63.3		IN		39	CT	9.8	39	NV	17.2	
41	LA	(2.2	40	MD	4.2		ID NU			OH		
41	1IN OP	62.7		NE KV		41	NH	9.7	41	UA WI	16.7	
42		62.6	42		4.1	13	WV	0.5	13		16.6	
44	NI	62.0	44	ND	4.0	44	UT	9.5	ر ד	NY	10.0	
45	SD	62.3	45	NV	3.8	45	FL	9.3	44	OR	16.4	
46	DE	61.7		ME	2.0	46	ME	9.2	46	CA	16.3	
47	MI	61.6	46	UT	3.7	47	TN	9.1	47	MA	16.0	
48	СО	61.4	48	WV	3.6	48	VA	8.9	48	NH	15.8	
49	NM	60.9	49	TN	3.4	49	MA	8.6	49	RI	14.6	
50	OK	60.7	50	NY	3.2	50	RI	8.5	50	HI	14.4	
51	DC	59.2	51	MN	2.7	51	NY	8.4	51	VT	14.0	

Source: U.S. Dept. of Ed. Inst. Natl. Revenues and Expenditures 15.

#### **Differences, FY 2006 Unadjusted \$** Rank Rank State State Adjusted \$ California 59.825 Illinois 56,861 1 2 59,304 2 Pennsylvania 56,698 Connecticut 3 District of Columbia 59,000 3 Alaska 56,426 58,686 4 Illinois 4 56,073 Michigan 5 New Jersey 58,156 5 Vermont 55,129 57,354 54,772 6 New York 6 Oregon 7 56,369 7 54.571 Massachusetts California 8 54,739 8 Rhode Island 54,173 Michigan 9 Rhode Island 54,730 9 Connecticut 53,928 10 54,333 10 53,850 Maryland Montana 54,264 11 Delaware 11 Wyoming 53,435 12 Pennsylvania 54,027 12 Delaware 53,361 13 Alaska 53 553 13 53 211 Indiana 14 50,314 14 52,540 Ohio Ohio 15 Oregon 50,044 15 Hawaii 52,044 49,292 51,966 16 Hawaii 16 Arkansas United States 49,026 17 Massachusetts 51,807 17 Minnesota 48,489 18 New Jersey 51,489 51.274 18 48 300 19 New York Georgia 47,255 20 19 Indiana Idaho 51,114 20 Vermont 46,622 21 Maryland 50,253 21 Wisconsin 46,390 22 Minnesota 50,175 46,326 23 22 Washington Georgia 49,198 23 24 New Hampshire 45,263 Iowa 49,065 24 United States 44,672 49,026 Arizona --25 25 Colorado 44,439 Kansas 49,015 26 44,426 26 Wisconsin 48,886 Nevada North Carolina 43,922 27 27 48.857 Mississippi 28 43,823 28 48,800 Virginia Maine 29 Florida 43,302 29 Arizona 48,770 43,255 30 Wyoming 30 New Hampshire 48 639 43,011 48,222 31 South Carolina 31 Kentucky 32 42,768 32 District of Columbia 48,031 Arkansas 33 42,592 33 47,750 Kentucky Nebraska 34 Tennessee 42,537 34 South Carolina 47,723 35 41,744 35 New Mexico 47,481 Texas 36 36 New Mexico 41,637 47,107 North Dakota 37 41,467 37 Florida 46,774 Kansas 38 Idaho 41,150 38 North Carolina 46,538 39 39 Iowa 41,083 Colorado 46,336 40 Maine 40,737 40 46,284 Tennessee 41 Mississippi 40,576 41 46,172 Louisiana 42 Missouri 40,462 42 Oklahoma 46,118 43 Nebraska 40,382 43 Alabama 46,056 44 40,347 44 South Dakota 45,641 Alabama 45 45 Louisiana 40,029 West Virginia 45,223 46 Utah 40,007 46 Missouri 44,738 47 Montana 39,832 47 Nevada 44,659 48 Oklahoma 38,772 48 Washington 44,615 49 West Virginia 38,284 49 Utah 43,674 50 North Dakota 37,764 50 Texas 42,198 34,709 Virginia 51 South Dakota 51 40,488

# Table A.46 (corresponds to Table 5.7)Average Public School Teacher Salaries, Unadjusted and Adjusted for Geographic CostDifferences, FY 2006

Note: Staff calculated adjusted salaries using NCES Comparable Wage Index.

Source: Natl. Ed. Assoc. *Rankings & Estimates 2006-2007*, Rankings, Table C-11. Data used with permission of the National Education Association © 2007. All rights reserved.

#### Table A.47 (corresponds to Table 5.8)Education Week's Equity/Disparity Measures, FY 2005

Wealth-neutrality Score			Μ	cLoone In	dex	Coeffic	ient of Va	ariation	Restricted Range			
		U	Act	ual Spendi	ing as						8	
Relati	onshin Be	etween	Perce	ntage of A	mount	Amour	nt of Disn	arity in	Differ	ence in P	er-nunil	
Distri	et Fundin	a and	Neer	led To Rri	ng All	Sne	nding Ac	ross	Spend	ling I eve	of pupi	
Local I	Ct Fullull Duonoutu '	ig anu Waalth	Distria	ta ta Madi	an Laval	Diatria	ta (Larvar	i USS Moluo	o <sup>5th</sup> or	ning Leve	at the	
Local I	roperty	wealth Dettern	Distric	ts to Mean	an Level	Distric	ts (Lowel	r value	95 al	na 5 Pel	r centiles	
(Lower	Value is	Better)	(High	er Value 1s	Better)	= G	reater Eq	uity)	(Lowe	r Value 1	s Better)	
Rank	State	Score	Rank	State	Index	Rank	State	CV	Rank	State	\$ Range	
1	AK	-0.185	1	NV	100.0	1	WV	0.058	1	WV	\$1,626	
2	NV	-0.123	2	UT	98.1	2	FL	0.074	2	KY	1,920	
3	NE	-0.108	3	NM	97.7	3	KY	0.091	3	AL	1,980	
4	SD	-0.030	4	NE	95.7	4	DE	0.092	4		2,092	
5	UI NM	-0.029	5	SD NC	95.6	6	AL	0.004	5	AK	2,355	
7	WV	-0.011		KS		0	LA WI	0.094	7	INIS	2,390	
8	τ <u>Δ</u>	-0.008	7	I A	95.3	7	TN	0.095	8	OH	2,414	
9	KS	-0.004	9	MD	95.2		IA		9	FL	2,044	
10	NJ	0.014	10	IA	95.1	9	NC	0.098	10	MI	2,996	
11	MN	0.015	11	WI	94.7	11	GA	0.099	11	SC	3,060	
12	OK	0.016	12	FL	94.4	12	AR	0.103	12	NC	3,090	
13	IN	0.017	13	WV	94.3	13	MD	0.105	13	LA	3,335	
14	AR	0.033	14	WY	04.1	14	SC	0.112	14	PA	3,435	
15	CA	0.036	14	GA	94.1	14	MS	0.112	15	DE	3,521	
16	LA	0.045	16	AL	94.0	16	NV	0.115	16	GA	3,530	
17	WI	0.053	10	AK	24.0	17	RI	0.119	17	WI	3,588	
18	MA	0.058	18	AR	93.9	18	CT	0.120	18	MO	3,659	
19	KY	0.070	•••	TX		19	CO	0.128	19	MD	3,696	
20	WV MO	0.075	20	SC	93.7	20	NY	0.122	20	IN	3,785	
21	MU 47	0.077	21	OP	93.0	20	PA	0.133	21	MIN OP	3,899	
22	WA	0.078	22	CO	93.4	23	UA II	0.135	22	OK	4.062	
23	NY	0.080	24	TN	93.3	25	MI	0.155	23	NE	4,002	
	U.S.	0.085	25	AZ	93.2	24	MN	0.136	25	ID	4.121	
25	ND	0.086	26	PA	00.0	26	VA	0.137	26	VA	4,163	
26	CT	0.089	26	VA	92.8	27	WA	0.139	27	KS	4,176	
27	MT	0.091	28	CA	92.7	28	OR	0.141	28	ND	4,418	
28	TN	0.093		U.S.	92.6	29	TX	0.143	29	SD	4,510	
29	OH	0.094	29	MS	92.5	30	OH	0.144	30	CA	4,633	
30	ME	0.113	30	MT	92.5	31	IN	0.145		U.S.	4,725	
31	GA	0.127	31	СТ	92.3	32	ME	0.146	31	IL	4,743	
22	MS	0.126	32	WA	92.2		U.S.	0.147	32		4,756	
33		0.130	32	DE KV	02.1	33	MO	0.159	33		4,605	
35	MI	0.142	55	OH	92.1	35	UT	0.162	35	NM	5 233	
36	FL	0.148	36	MI	91.8	36	NI	0.177	36	CT	5,391	
37	CO	0.149	37	ND	91.1	37	NE	0.178	37	ME	5,605	
38	TX	0.156	38	NJ	91.0	38	OK	0.180	38	AZ	5,775	
39	DE	0.159	39	OK	90.9	39	WY	0.181	39	WA	5,839	
40	RI	0.170	40	IL	90.7	40	NM	0.187	40	NH	6,138	
40	PA	0.160	41	MA	89.9	41	MA	0.101	41	NJ	6,173	
42	NH	0.169	42	RI	89.7	41	NH	0.191	42	UT	6,343	
43	IL	0.170	- <b>T</b> 2	MO	07.1	43	ID	0.198	43	MA	6,399	
44	AL	0.171	44	IN	89.5	44	SD	0.205	44	MT	6,505	
45	NC	0.201	45	ID	89.2	45	AZ	0.206	45	VT	7,092	
46	SC	0.212	46	ME	87.6	46	VT	0.225	46	NY	7,313	
47	ID MD	0.276	47	NY	85.5	47	ND MT	0.259	47	W Y NV	9,910	
48	MD VA	0.285	48	NH VT	83.0	48		0.299	48		13,541	
49	DC	0.200	49	DC	03.2	49	DC	0.334	47	DC	14,/04	
n.a.	HI	n.a	n.a.	HI	n.a.	n.a.	HI	n.a.	n.a.	HI		

Source: Editorial Projects in Education. Education Week's Quality Counts.

Gap, Low-po	overty Minus High	-poverty Districts	Gap, Low-minority Minus High-minority Districts						
Rank	State	Gap (\$)	Rank	State	<b>Gap (\$)</b>				
1	AK	5,720	1	AK	3,952				
2	NJ	1,918	2	NJ	1.840				
3	MN	1.134	3	MA	1.116				
4	UT	523	4	OH	1.032				
5	MA	513	5	MN	833				
6	NM	491	6	AR	547				
7	KY	462	7	MO	535				
8	OR	381	8	IN	428				
9	AR	230	9	OR	277				
10	WY	174	10	WV	238				
11	TN	155	11	LA	229				
12	OH	73	12	KY	152				
13	SC	-19	13	GA	134				
14	MD	-23	14	SC	81				
15	ND	-40	15	TN	-81				
16	OK	-72	16	UT	-122				
17	IA	-157	17	NM	-126				
18	СТ	-162	18	FL	-133				
19	FL	-199	19	VT	-135				
20	SD	-228	20	MS	-166				
20	NE	-237	20	WA	-167				
22	IN	-238	22	VA	-250				
23	MS	-267	23	OK	-294				
23	WA	-292	23	MI	-358				
25	ID	-371	25	AL	-500				
26	WV	-383	26	AZ	-539				
	CA		27	СТ	-563				
27	GA	-436	28	IA	-574				
29	СО	-518	29	CA	-661				
30	VA	-526	30	RI	-716				
31	LA	-560	31	NC	-738				
32	KS	-632	32	MD	-803				
22	AZ	652	33	ID	-824				
22	RI	-033	34	ME	-864				
35	AL	-692	35	DE	-933				
36	ME	-718	36	SD	-939				
37	NV	-737	37	ND	-951				
38	TX	-796	38	PA	-1,030				
39	VT	-798	39	NV	-1,094				
40	MO	-803	40	CO	-1,206				
41	MT	-810	41	WI	-1,221				
42	NC	-825		U.S.	-1,275				
43	WI	-990	42	TX	-1,385				
44	DE	-1,126	43	MT	-1,540				
45	MI	-1,388	44	KS	-1,594				
	U.S.	-1,532	45	NE	-1,763				
46	NH	-1,662	46	IL	-2,021				
47	PA	-1,708	47	WY	-2,034				
48	IL	-2,827	48	NH	-2,332				
49	NY	-3,972	49	NY	-3,544				
ne	DC	ne	ne	DC	ne				
11.ä.	HI	11.ä.	11.ä.	HI	11.a.				

#### Table A.48 (corresponds to Table 5.9)Education Trust Poverty and Minority Funding Gaps, 2005

Notes: Rank out of 49; DC and HI have only one district. A negative dollar amount indicates that fewer dollars were provided to high-poverty or high-minority districts. A positive amount indicates more dollars for high-poverty or high-minority districts. Education Trust made a 40 percent adjustment for low-income students. Source: Education Trust. *The Funding Gap* 9.

1999 Gap High	), Low-povert -poverty Dist	y Minus ricts	2005 Ga Hig	ip, Low-pov h-poverty D	erty Minus istricts	1999- C	1999-2005 Poverty Gap, Change in Dollars			
Rank	State	Gap (\$)	Rank	State	<b>Gap (\$)</b>	Rank	State	Change (\$)		
1	VT	2.193	1	AK	6.523	1	AK	5.839		
2	MA	1.435	2	NJ	2.712	2	NJ	2,145		
3	MN	1.368	3	MN	1.629	3	MD	1.376		
4	KY	801	4	MA	1,396	4	OH	910		
5	UT	799	5	NM	923	5	WY	527		
6	TN	729	6	KY	878	6	AR	523		
7	AK	684	7	OH	833	7	NM	427		
8	OR	659	8	СТ	825	8	NY	358		
9	СТ	615	9	UT	739	9	MN	261		
10	NJ	568	10	OR	647	10	СТ	210		
11	NM	495	11	AR	541	11	IN	197		
12	MO	480	12	WY	468	12	LA	180		
13	ID	457	13	TN	454	13	PA	163		
14	KS	388	14	MD	395	14	CA	143		
15	NE	384	15	IN	322	15	SC	136		
16	FL	350	16	SC	302	16	IA	112		
17	NC	337	17	OK	271	17	DE	98		
18	OK	312	18	RI	266	18	KY	77		
19	TX	280	19	ND	159	19	RI	68		
20	GA	245	20	CA	154	20	AZ	55		
21	SD	240	21	MS	151	21	WA	23		
22	VA	234	22	IA	108	22	CO	7		
23	RI	197	23	WA	87	23	MT	-5		
24	MS	192	24	GA	82	24	OR	-13		
25	ND	183	25	NE	66	25	AL	-19		
26	SC	166	26	FL	-18	26	ND	-23		
27	IN	126	27	WV MO	-19	27	MA	-39		
28	WA	64	28	MO	-104	28	W V	-40		
29	W V	22	29	VA	-122	29	MS	-41		
30	AR	18	30	47	-120	30	UK	-42		
31		11	31	AZ	-145	31	UI MI	-60		
32	IA	-4	32		-103	52		-/8		
33	WIE	-9	33	ID SD	-105	22	0.3. GA	-90		
35	WY	-28	35	I A	-228	33	TN	-102		
36	OH	-77	36	VT	-241	35	NE	-318		
37	CO	-133	37	KS	-284	36	ME	-321		
38	NV	-189	38	AL	-328	37	VA	-356		
39	AZ	-198	39	ME	-331	38	FL	-368		
40	AL	-309	40	WI	-468	39	WI	-439		
41	LA	-421	41	MT	-505	40	TX	-445		
42	MT	-500	42	NC	-603	41	SD	-468		
43	MI	-682	43	NV	-680	42	NV	-491		
44	NH	-723	44	MI	-759	43	MO	-584		
	U.S.	-848		U.S.	-938	44	NH	-617		
45	MD	-981	45	DE	-954	45	ID	-642		
46	DE	-1,052	46	PA	-1,055	46	IL	-668		
47	PA	-1,218	47	NH	-1,340	47	KS	-672		
48	IL	-1,568	48	IL	-2,235	48	NC	-939		
49	NY	-3,426	49	NY	-3,068	49	VT	-2,457		
n.a.	DC HI	n.a.	n.a.	DC HI	n.a.	n.a.	DC HI	n.a.		

#### Table A.49 (corresponds to Table 5.10)Poverty Funding Gap Trends, 1999 to 2005

Notes: DC and HI each has only one district. A negative dollar amount indicates that fewer dollars were provided to high-poverty districts. A positive amount indicates more dollars for high-poverty districts. Education Trust made no adjustment to these numbers for low-income students.

Source: Education Trust. The Funding Gap 6.

1999 Gap	, Low-mina	ority Minus	2005 Gap,	Low-mino	rity Minus	1999-2005 Minority Gap,				
High-	minority D	istricts	High-n	ninority Di	stricts	0	Change in D	ollars		
Rank	State	Gap (\$)	Rank	State	Gap (\$)	Rank	State	Change (\$)		
1	AK	2,746	1	AK	4 442	1	NI	2.215		
2	MA	1.865	2	NJ	2.633	2	AK	1.696		
3	MO	1 446	3	MA	1 891	3	OH	800		
4	MN	1,027	4	OH	1,520	4	LA	578		
5	GA	957	5	MN	1,020	5	NY	549		
6	VT	823	6	MO	788	6	OR	515		
7	IN	766	7	IN	721	7	SC	445		
8	OH	720	8	AR	707	8	CT	440		
9	VA	521	9	GA	496	9	AR	400		
10	ID	484	10	OR	377	10	RI	227		
11	NJ	418	11	СТ	367	11	PA	207		
12	AR	308	12	SC	297	12	KY	206		
13	MS	215	13	LA	293	13	CA	201		
	WA		14	WV	200	14	MI	119		
14	WV	91	15	MS	163	15	WV	109		
	NM		16	RI	129	16	MN	86		
16	UT	50	17	NM	103		U.S.	73		
18	FL	31	18	WA	81	17	AL	67		
	ME	-	19	MI	55	18	NM	53		
19	OK	17	20	KY	44	19	TN	51		
21	AZ	-34	21	TN	10	20	IA	38		
22	TN	-41	22	UT	-12	21	MA	26		
23	MI	-64	23	FL	-15	22	WA	-10		
24	CT	-74	24	OK	-56	23	IN	-45		
25	NC	-75	25	VA	-57	24	FL	-46		
26	RI	-99	26	VT	-63	25	MS	-52		
27	OR	-138	27	AZ	-100	26	UT	-62		
28	SC	-148	28	CA	-216	27	AZ	-65		
29	KY	-162	29	AL	-280	28	OK	-73		
30	LA	-285	30	IA	-474	29	MD	-157		
31	AL	-347	31	MD	-578	30	KS	-168		
32	DE	-385	32	PA	-662	31	ND	-176		
33	CA	-417	33	NC	-663	32	CO	-276		
34	MD	-421	34	ME	-757	33	TX	-327		
35	WI	-430	35	ID	-814	34	SD	-341		
36	NV	-496		U.S.	-877	35	WI	-450		
37	IA	-512	36	WI	-880	36	GA	-461		
38	SD	-542	37	SD	-883	37	DE	-507		
39	TX	-586	38	ND	-890	38	MT	-509		
40	CO	-652	39	DE	-892	39	NV	-574		
41	ND	-713	40	TX	-912	40	VA	-579		
42	IL	-864	41	CO	-928	41	NC	-587		
43	PA	-869	42	NV	-1,070	42	MO	-658		
44	WY	-921	43	KS	-1,417	43	NE	-674		
	U.S.	-950	44	MT	-1,467	44	NH	-710		
45	MT	-958	45	IL	-1,623	45	IL	-758		
46	NE	-982	46	NE	-1,656	46	ME	-774		
47	KS	-1,249	47	WY	-2,015	47	VT	-887		
48	NH	-1,557	48	NH	-2,267	48	WY	-1,094		
49	NY	-3,450	49	NY	-2,902	49	ID	-1,298		
na	DC	na	na	DC	na	na	DC	na		
11.a.	HI	11.a.	11.a.	HI	11.a.	11.a.	HI	11.a.		

#### Table A.50 (corresponds to Table 5.11)Minority Funding Gap Trends, 1999 to 2005

Notes: DC and HI each has only one district. A negative dollar amount indicates that fewer dollars were provided to high-minority districts. A positive amount indicates more dollars for high-minority districts. Education Trust made no adjustment to these numbers for low-income students.

Source: Education Trust. The Funding Gap 7.

Rank State		<b>Total Score</b>	<b>Overall Grade</b>			
1	WV	92.5	А			
2	NJ	91.0	A-			
3	WI	89.5				
4	NY	88.7				
5	CT	87.9	B+			
6	VT	86.9				
7	ME	86.6				
8	WY	85.9				
9	RI	85.0				
10	MD	84.9	В			
11	IN	84.3				
12	DE	83.5				
13	OH	82.0				
14	MA	81.9				
15	PA	81.5				
16	AR	81.1	B-			
17	MI	80.9				
18	NE	80.2				
19	IA	80.1				
20	KS	78.9				
21	GA	78.7				
21	MN	70.7				
23	NH	78.2	C+			
24	VA	77.6				
	U.S.	77.6				
25	LA	77.2				
26	SC	76.1				
27	ND	75.1				
28	SD	74.4				
29	KY	74.0	С			
_,	NM					
31	AL	73.8				
32	IL	73.2	-			
33	MO	72.9				
34	MS	72.3				
-	MT		-			
36	OR	72.1	-			
37	<u> </u>	72.0	C-			
38	FL	70.6				
39	NC	70.3				
40	TX	69.9				
41		69.7				
42	OK	69.5				
43	CA	69.2				
44	WA	68.5	D+			
45	AK	68.2	4			
46	NV	67.6				
47	ID IT	65.3				
48	UT	64.9	U			
49	AZ	64.8				
n.a.	DC	n.a.	n.a.			
	HI					

#### Table A.51 (corresponds to Table 5.13)Education Week's Overall School Finance Index, 2008

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

Rank	State	Overall State Grade	Total score	Chance for Success	K-12 Achieve ment	Standards, Assessment, and Accountability	Transitions and Alignment	Teaching Profession	School Finance
1	NY		84 9	В	C-	A	А	C+	B+
2	MA	в	84.0	A	B	A-	C	C	B-
3	MD		83.5	B+	B	B	B+	C-	B
4	NJ		82.4	A-	B-	B-	C+	C	A-
5	VA		82.3	B	C	A	C+	B-	C+
6	WV		80.8	C-	F	A	B+	 C+	A
7	OH		80.6	B-	C-	A	C+	C+	B-
8	AR	_	80.5	 C-	D	B+	B	B+	B-
9	SC	В-	80.4	C	D	A	B-	A-	C
10	PA		80.3	В	C+	C+	B-	C+	B-
11	GA		80.2	C	D+	A-	В	B	C+
12	IN		79.7	C+	C-	А	C+	C-	В
13	VT		79.5	B+	C+	B-	C-	C-	B+
14	FL		79.2	C+	C	A-	C+	В	C-
15	WI		78.3	B-	C-	C+	C-	C+	B+
	TN		-0.0	C-	D+	A-	А	С	C-
16	DE	C+	78.0	В	C-	B+	D	C+	В
18	MI		77.8	C+	D	A-	В	D+	B-
19	CT		77.6	A-	D	С	С	C-	B+
20	ME		77.5	B-	С	C+	В-	D	B+
21	LA		76.5	D+	D-	А	С	В	C+
22	KY		76.4	С	D+	B+	С	B-	С
23	TX		75.9	С	С	B+	C+	С	C-
	U.S.		75.9	C+	D+	В	С	С	C+
24	NC		75.0	C+	D+	B+	D+	В	C-
24	MN	1	/5.8	B+	С	С	C-	D+	C+
26	AL		75.7	C-	F	A-	C+	B-	С
27	NH	1	75.6	A-	С	С	C-	D	C+
28	OK		75.5	C-	D	A-	С	B-	D+
29	IA	C	75.3	В	C-	D+	D+	B-	B-
20	NM	C	75.2	D+	D-	A-	В-	C+	С
30	RI		13.2	B-	D	B+	C-	D	В
32	HI		74.8	C+	D	B-	C-	C+	NA
33	ND		74.2	В	С	С	D+	D+	С
34	WA		73.9	B-	C-	B-	C-	C	D+
35	IL		73.4	B-	D+	C+	C-	D+	C
55	CA		75.4	C	D	A-	C-	C-	D+
37	KS		73.3	В	C	C+	D-	D+	C+
38	CO		73.1	В	C-	B-	D	D+	C-
39	WY		72.8	C+	C-	C+	D-	D-	В
40	MO		72.1	C+	D	С	D+	С	С
41	SD		71.8	В	C-	C+	D-	D	С
42	AZ	C-	71.6	C-	D	A-	C-	D+	D
43	AK	-	70.9	C	D+	B	<u>C-</u>	D-	D+
44	UT		70.7	В-	<u>C-</u>	C+	F	<u>C-</u>	D
	MT			B-	<u>C-</u>	C-	D-	D+	C-
46	NV		69.4	D+	D-	<u>C+</u>	D+	<u>C-</u>	D+
47	NE		69.3	B-	D+	D	F	D+	B-
48	MS	D+	68.9	D+	F	B	D+	D	<u>C-</u>
49	OR	1	68.6		D	<u>C+</u>	D	F	<u>C-</u>
50	ID D ~		68.4	C	<u> </u>	C	F	D	D
51	DC	1	68.1	L C	F	C+	D+	I D-	NA

### Table A.52 (corresponds to Table 6.1) Education Week's Quality Counts Overall Grade

Note: Because DC and HI are single-district jurisdictions, there is no measure of financial equity among districts. Therefore, neither DC nor HI has a grade for school finance.

Source: Editorial Projects in Education. Education Week's Quality Counts 2008.

Table A.53 (corresponds to Table 6.3)	
Index of NAEP Proficiency Purchasing Power Relative to Obstacles, 2008	3

Per Punil Spending			NAFD	NAEP Proficiency in			D Drofi	aionay				NAFI	D Drofi	aionay
	upn sp	Cout of	NALI			D		D			Cast	NALF Fronciency		
Adjust	ed for	Cost-oi-	G	ades 4	& 8	Purci	nasing	Power	Ob m	stacles to	o Cost-	Purch	lasing	Power
Livin	g Diffe	erences	Rea	ding &	Math	(% p	roficie	nt per	effec	tive Edu	cational	R	elative	to
	FY 200	)5	20	<u>05 &amp; 2</u>	<u>.007</u>	\$1,00	)0 spen	ding)	Spending Index			C	<u>)bstacl</u>	es
Rank	State	\$	Rank	State	%	Rank	State	#	Rank	State	Index	Rank	State	%
1	NJ	12,252	1	MA	47.5	1	UT	6.07	1	WV	0.81392	1	UT	137.5
2	NY	12,218	2	MN	41.6	2	ID	5.03	2	NM	0.77485	2	ID	129.2
3	VT	12,105	2	VT	41.2	3	WA	5.01	3	MS	0.75334	3	WA	128.5
4	WY	11,126	3	NJ	41.5	4	MA	4.79	4	KY	0.71686	4	TX	125.3
5	DE	10,661	5	NH	40.7	5	MN	4.67	5	AR	0.70718	5	NC	121.5
6	СТ	10,652	6	KS	38.9	6	СО	4.54	6	AL	0.68972	6	MA	120.0
7	RI	10,581	7	СТ	38.3	7	KS	4.38	7	OK	0.68328	7	AZ	119.6
8	ME	10,539	/	PA	56.5	8	NH	4.37	8	LA	0.67382	8	KY	118.2
9	WI	10,199	9	MT	38.2	9	NC	4.33	9	TN	0.66423	9	KS	117.7
10	WV	10,073	10	ND	37.6	10	MT	4.27	10	SC	0.66394	10	FL	116.1
11	PA	9,985	11	OH	37.4	11	SD	4.23	11	AK	0.62483	11	OK	112.3
12	MA	9 930	12	WA	37.2	12	TX	4.19	12	TX	0.60727	12	SC	110.2
12	NE	),)30	13	WI	37.1	13	AZ	4.13	13	NV	0.59638	13	MT	109.6
14	MD	9,829	14	VA	37.0	14	ND	4 10	14	IN	0.59493	14	CO	108.2
15	IN	9,542	15	SD	36.9	14	FL	4.10	15	AZ	0.57045	15	MN	107.8
16	OH	9,441	16	WY	36.5	16	VA	4.04	16	ME	0.56630	16	OR	107.7
17	NH	9,323	17	ME	36.3	17	IA	3.99	17	FL	0.54216	17	TN	107.3
18	MI	9,197	18	CO	36.1	18	OR	3 96	18	CA	0.53230	18	AK	107.0
19	ND	9,181	19	IA	36.0	10	OH	5.70	19	NC	0.53064	19	IN	105.7
20	VA	9,169	20	NE	35.1	20	PA	3.83	20	IL	0.52025	20	SD	105.6
21	IA	9,026	21	ID	34.5	21	MO	3.80	21	GA	0.50806	21	AR	105.1
22	HI	9,022	22	MD	34.4		U.S.	3.78	22	NY	0.50277	22	NH	104.4
23	MT	8,951		NY		22	WI	3.64		U.S.	0.50000		U.S.	103.6
24	MN	8,891	24	IN	34.1	23	IL	3.62	23	OR	0.48878	23	OH	103.2
25	KS	8,862	25	DE	33.2	24	СТ	3.60	24	MO	0.48487	24	MO	103.1
26	AR	8,790		UT	22.1	25	KY	3.59	25	KS	0.47175	25	IL	100.8
27	SD	8,736	27	OR	33.1	26	IN	3.57	26	MI	0.46959	26	VA	99.7
	0.8.	8,701		U.S.	32.9	27	NE	3.54	27	DE	0.46491	27	ME	99.6
28	GA	8,658	28	NC	32.6	28	AK	3.53	28	WY	0.43536	28	PA	99.2
29		8,621	29		32.2	29	OK	3.52	29	OH	0.43336	29	NV	98.3
30	LA	8,582	30	MI	31.7	30	SC	3.51	30	NJ D4	0.42932	30	IA	97.7
31	AK	8,562	31	MO	31.5	31	MD	3.50	31	PA	0.42231	31	ND	96.3
32	NM	8,431	32	IL FI	31.2	32	MI	3.45	32	KI	0.419/4	32	CA	94.2
33	OK	8,353	33	FL	30.9	24	ME	2.42	33	NE	0.41868	33	MI	92.4
34	SC MO	8,339	34	AK	30.2	34	IN VT	3.42	34	ID WA	0.41323	34	NE	91.3
26	WU VV	0,270	26	DI DI	29.5	26	V I NT	2.41	26	WA MT	0.41123	33		09.2
27	KI CO	7,978	27		29.1	27		2.5/	27	MA	0.4111/	27	AL	00.3
20		7,939	20		20.7	20	CA NW	2.22	20	MA SD	0.37839	20	INJ WW	0/./
20	AL TV	7,924	20		26.5	20	IN V WVV	2.32	20	SD VA	0.37410	20	WV	80.5 85.6
39		7,087	39	OK	20.8	39		3.28	39	VA WI	0.33730	<u> </u>		85.0
40	TL NC	7,539	40		23.0	40	DE	3.22	40	TA VVI	0.34093	40	CT	03.5
41	MS	7,525	41	TN	25.7	41	GA DE	3.12	41	HI	0.34310	41	DE	04.0 83.2
42	TN	7,515	12	WW	24.0	42	NV	2.09	42	NU	0.32310	42	DE VT	03.2 91.1
43	M A	7,300	43		24.0	43	DI	2.02	43		0.30878	43	V I NM	01.1 80.2
44	0K	7 3 2 1	44	NV	23.7	44		2.75	44	VT	0.30280	44	MD	80.5
46	NV	7 1/1	46	HI	23.4	46	HI	2 50	46	СТ	0.28778	45	MS	79.6
40	CA	7 081	47		21.4	47	WV	2.39	47	ND	0.28776	40	NV	77.4
47		6 867	47	I A	21.0	4/	T A	2.39	47	MN	0.26233	47	INI I A	75.4
49	Δ7	6 232	49	NM	19.5	49	MS	2.30	49	MD	0.23437	40	RI	71.0
50	LIT	5 163	50	MS	17.5	50	NM	2.34	50	UT	0.24172	50	Ш	62.6
50		5,405	50	1410	1/.0	50	1 N IVI	4.31	50	01	0.22001	50	111	04.0

Notes: Kentucky Long-Term Policy Research Center assigned an index value of 0.50000 to the average of all states. OEA used spending and proficiency data from NCES to calculate purchasing power for the U.S.

Source: Staff compilation using data from Commonwealth. Legislative. Kentucky Long-Term. *Reducing*; U.S. Dept. of Ed. Inst. Natl. *NAEP Data* and *Common*.

Preparation		Participation		Affordability			Completion			Benefits				
Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)	Rank	State	Grade (Score)
1	MA	A (100)	1	NM	A (100)	1	CA	C- (71)	1	NH	A (100)	1	MA	A (100)
2	NJ	A (94)	1	RI			UT	0 (/1)	2	WA	A (99)	2	MD	A (99)
	UT		3	ND	A (98)	3	HI	D (65)	3	PA	A (98)	3	NJ	A (97)
4	CT	A- (92)	4	MN	A (97)	4	ID NOV	D (64)	6	RI	A (07)	6	VA	1.000
5	MD	A- (91)	5	IL VS	A (96)	6	MIN	D (62)	5	W Y	A(97)	3		A (96)
7	IN I VA	$A_{-}(90)$	7	MD	A (95)	7	WA	$D_{-}(60)$	0	GA	A (90)	6	MO	A (95)
,	CO	A-(70)	/	CA	A(JJ)	8	II.	F(59)	7	IA	A (95)	0	NH	$\Lambda(00)$
8	WI	B+ (88)	8	NE	A (94)	9	WI	F (58)	,	WI	11(55)		IL	
	IA	B+(87)	10	MA	. (22)	-	IN	()		FL		9	OR	A (93)
10	MT	B+ (87)	10	SD	A (93)		NC		10	MN	A (94)		CO	
10	NC	B+ (87)		CT		10	NM	F (57)		VT		11	UT	A- (92)
	NH	B+ (87)	12	MI	A- (92)		TX		13	NY	A- (92)		VT	
14	PA	B (86)		NJ			VA		14	DE	A- (90)		MI	
15	IL	B (85)	15	CO	A- (91)	15	CO	F (55)	15	MO	B+ (89)	14	PA	A- (91)
	MN	B (84)		IA			OK	- ()		CT		15	WA	. (00)
16	NE	B (84)	17	WI	A- (90)		AR			IL N		17	HI	A- (90)
	SD ME	B (84)	18	W Y	B+(89)	17	DE	E (54)	16	IN	B+ (88)	10	AZ	$\mathbf{D} \vdash (00)$
19	WIE	D (03)	20	DE	$B^{+}(86)$	17	NV	г (34)		NC SC		10	NV	<b>Б</b> + (89)
	AK	B <sub>-</sub> (81)	20	MO	Б (80)		PA			VA			KS	
	KS	B- (81)	21	VA	B (84)		MD			KS		21	OH	B+(87)
21	OH	B-(81)	23	U.S.	B (83)	22	NE	F (53)	22	NE	B+ (87)	23	RI	B (86)
	VT	B- (81)	22	PA	D (02)		U.S.	F (52)		SD	· · ·	24	FL	B (84)
25	ND	D (90)	23	UT	B (83)	24	VT	E (52)	25	MD	B (86)		U.S.	B (84)
23	TX	Б- (80)	25	ME	B- (82)	24	WY	г (32)		CO			AL	
	U.S.	C+(79)	26	KY	B- (81)		GA		26	ME	B (85)	25	NC	B (83)
27	RI	C+(78)	20	NY	B (01)	26	KY	F (51)	20	OH	B (05)		NE	
28	GA	C+(77)	28	NC	B- (80)		MI			UT		28	DE	B- (82)
	SC			OH	. ,		AK			U.S.	B (85)		GA	· · /
	DE		20	AK	$C^{+}(70)$	20		F (50)	30	MS	B (84)	30	UK WI	B- (81)
30	ГL IN	C (75)	30		C+(79)	29	IA	r (30)						
	MO			IN			MS			CA		32	ME	B- (80)
34	CA	C (74)	33	OR	C+(78)		FL		32	MI	B (83)		TX	2 (00)
35	ID	C (73)		TX	- ()	34	NV	F (49)	-	ND	()	35	ND	C+(79)
26	HI	C (72)		AL			AZ			TN			KY	
30	OR	C-(72)	36	AR	C (76)		MA		37	OR	B- (82)	36	MT	C + (78)
38	KY	$C_{-}(71)$		HI		36	MO	F (47)	38	AL	B- (81)	50	SD	C+(78)
50	WY	0 (/1)		FL			ND		50	MT	B (01)		TN	
	MI	C- (70)	39	NV	C (75)		TN	-	40	HI	B- (80)	40	MS	C (76)
40	NV	C- (70)	10	VT	Q (71)	41	WV	F (46)	41	KY	C+(78)		AR	
	IN	C - (70)	42	MI	C-(/I)	42	AL	E (42)		WV		41	IA NM	C (75)
44		D + (60)		LA TN		42	SU SD	г (43)	43		C+(77)		SC	
44	OK	D + (67)	43	W A	C-(70)		ME			AR		45	IN	C (74)
46	AZ	D (66)		WV		45	OH	F (42)	45	OK	C (76)	46	NV	C - (72)
47	MS	D- (62)		ID			OR	- ()	47	LA	C-(72)	47	ID	C- (71)
48	AL	D- (61)	47	SC	D+ (69)	48	RI	F (40)	48	NM	D (66)	48	WY	C- (70)
49	NM	F (57)	49	GA	D+(67)	40	MT	E (20)	49	NV	F (59)	40	LA	$D \perp (60)$
50	LA	F (56)	50	MS	D (66)	49	NH	г (39)	50	AK	F (49)	49	WV	D+ (08)

#### Table A.54 (corresponds to Table 6.5) Measuring Up Grades and Scores, 2006

Note: Staff calculated U.S. average scores and grades. Source: Natl. Ctr. for Public Policy. "Compare."

State	Overall Rank	Economy	Health	Crime	Education	Society	Government
NH	1	7	6	2	8	4	9
ND	2	21	8	1	9	9	1
SD	3	23	18	2	3	9	4
IA	4	26	3	16	1	8	13
MN	5	8	1	22	6	3	33
NE	6	17	10	21	4	12	11
WY	7	3	26	11	6	2	30
UT	8	11	12	17	15	5	21
VT	9	22	14	4	12	14	17
ID	10	30	20	6	16	1	12
СО	11	5	19	26	25	6	7
СТ	12	2	4	9	18	14	42
MA	13	6	2	19	14	24	32
VA	14	18	29	11	26	13	2
NJ	15	9	15	14	22	19	27
WI	16	32	13	18	1	7	41
MT	17	27	36	10	10	11	22
ME	18	40	17	5	10	16	39
WA	19	1	7	38	19	26	38
KS	20	33	24	30	5	22	17
OR	21	16	23	24	23	17	34
MD	22	14	26	37	34	20	7
AK	23	4	21	40	29	26	30
MO	24	28	31	39	13	28	16
NY	24	13	9	15	19	50	49
PA	26	35	22	20	28	22	35
KY	27	45	33	8	37	28	13
HI	28	14	4	30	40	31	46
DE	29	12	39	36	45	33	2
RI	30	36	10	7	30	46	48
IL	31	24	28	27	38	20	44
MI	32	47	25	32	33	18	36
AZ	33	37	37	45	27	44	6
CA	35	19	16	29	43	42	47
IN	35	42	38	25	31	36	25
FL	36	20	34	46	49	35	15
OK	37	39	48	32	21	33	28
WV	38	50	43	11	32	25	43
NC	39	28	42	41	46	38	10
GA	40	31	47	35	47	42	5
OH	10	46	31	27	35	32	36
TX	42	25	40	42	44	40	19
AR	43	49	44	43	17	38	23
NV	44	10	34	48	47	41	39
AL	45	44	46	34	39	30	29
TN	46	43	41	49	24	49	20
NM	47	41	30	44	42	48	23
SC	48	38	45	50	50	37	25
MS	49	48	49	23	41	44	45
LA	50	34	49	46	36	47	50

## Table A.55 (corresponds to Table 6.7)Camelot Index, 2008

Source: Federal. "The 2008" 18.

FY 2007			<b>Rankings in Previous Years</b>			
Rank	State	Score	FY 2003	FY 2004	FY 2005	FY 2006
1	VT	18.57	2	2	3	1
2	MA	16.09	7	1	1	3
3	СТ	14.46	1	3	2	2
4	NJ	14.35	4	5	4	4
5	ME	10.79	5	6	11	5
6	VA	10.07	37	17	12	7
7	MT	9.55	3	4	10	9
8	WI	9.04	6	8	5	8
9	IA	8.82	11	8	8	14
10	PA	8.69	15	7	9	11
11	NE	6.40	13	11	13	12
12	NH	5.90	19	26	14	15
13	MN	5.33	12	12	7	6
14	RI	4.31	10	16	23	16
15	KS	4.27	14	15	15	13
16	NY	3.66	26	10	6	10
17	SD	3.25	34	31	22	18
18	MD	2.27	30	18	18	19
19	WY	1.35	8	14	16	17
20	ID	1.29	22	30	29	28
21	ND	0.95	21	24	19	20
22	MO	0.94	31	28	26	21
23	NC	0.84	24	21	25	22
24	IN	0.06	9	13	17	26
25	TX	-0.11	16	34	33	24
26	SC	-1.19	36	41	32	29
27	CO	-1.32	27	35	21	23
28	DE	-2.47	43	19	27	25
29	FL	-2.90	47	40	39	36
30	TN	-3.01	39	42	41	41
31	KY	-3.24	28	37	37	35
32	AR	-3.44	38	38	36	37
33	WA	-3.85	35	33	30	30
34	OH	-4.00	41	22	20	31
35	IL	-4.32	33	27	24	32
36	OK	-5.81	32	39	40	39
37	WV	-5.82	18	29	33	34
38	UT	-6.30	17	25	28	33
39	MI	-6.43	20	20	31	27
40	OR	-6.87	23	32	35	38
41	GA	-6.92	40	36	38	40
42	HI	-9.31	45	43	42	42
43	NM	-10.60	50	50	50	48
44	LA	-10.95	49	47	46	45
45	AL	-11.00	41	46	44	43
46	AK	-11.91	25	23	45	44
4'/	CA	-13.10	29	44	43	46
48	MS	-14.78	48	48	47/	49
49	NV	-15.81	46	49	49	47
50	AZ	-17.61	44	45	48	50

## Table A.56 (corresponds to Table 6.8)Smartest State Index, 2003-2007

Source: Congressional. Smartest.
# Appendix B

## **Information Regarding Data Sources**

## ACT, Inc.

Established in 1959 as the American College Testing Program, Inc., and later officially shortened to ACT, Inc., this independent, not-for-profit organization initially offered one high school achievement and college readiness test. This test, called the ACT, is still used today. In the past 50 years, ACT, Inc. has added a variety of other assessment, research, information, and program management services in education and workforce development (ACT. *ACT Newsroom*).

## **College Board**

The College Board creates and scores the SAT Reasoning Test and Advanced Placement tests. Founded by a group of colleges in 1900, the College Board simplified the application process by developing a common set of entrance examinations so that students could apply to several colleges without having to sit for an entrance exam at each one. Many also hailed the "democratizing benefit" of these exams, which allowed individuals to demonstrate their ability to handle college work without regard to family background and inconsistent grading system and curriculum standards throughout the nation. The College Board later developed additional tests, such as the Advanced Placement exam and the College Scholarship Service that provides financial aid information and assistance (College Board. *College Board History*).

## **Congressional Quarterly**

Established in 1945, Congressional Quarterly Inc. (CQ) provides political journalism in weekly, daily, and real-time reports in print and online. CQ seeks to "advance the quality of reporting about government, helping elected officials and citizens alike understand and improve democracy in the United States." A private, for-profit organization, CQ is a wholly owned affiliate of the Times Publishing Co., which publishes the *St. Petersburg Times* of Florida. The stock of the publishing company is owned by the Poynter Institute, a nonprofit school for journalists in St. Petersburg (Congressional. "Mission"). In 2007, CQ acquired Morgan Quitno Press, the publisher of annual state rankings and indices (Congressional. *CQ Press Acquires*).

## Editorial Projects in Education and Education Week

Editorial Projects in Education Inc. (EPE) publishes materials that cover local, state, and national news and issues pertaining to education from preschool through grade 12. A nonprofit, tax-exempt organization based in Washington, D.C., EPE describes its primary mission as helping to "raise the level of awareness and understanding among professionals and the public of important issues in American education" (Editorial. *About*).

EPE publishes *Education Week* as well as *Teacher Magazine*, edweek.org, Agent K-12 Jobs, periodic special reports on a wide range of issues, and books of special interest to educators. The

EPE Research Center provides research support. It compiles and analyzes data for annual issues of *Diplomas Count, Quality Counts*, and *Technology Counts*. The center also integrates measures from those and other EPE publications over time into an online database called *Education Counts*. The database is supported by a 4-year, \$2.5 million grant from the Bill & Melinda Gates Foundation (Editorial. *About*).

#### **Education Trust**

The American Association for Higher Education established Education Trust in 1990 to encourage colleges and universities to support K-12 reform. Today, Education Trust is an independent, nonprofit organization working for "the high achievement of all students at all levels, prekindergarten through college, and forever closing the achievement gaps that separate low-income students and students of color from other youth" (Education Trust. *What is*). Education Trust advances this mission through advocacy in policy debates; policy analysis and expert testimony; research and dissemination of data; and assistance to school districts, colleges, and community-based organizations trying to raise student achievement, especially among minority and poor students.

Major funding for Education Trust comes from

- Annie E. Casey Foundation
- Carnegie Corporation of New York
- Bill & Melinda Gates Foundation
- William and Flora Hewlett Foundation
- Ewing Marion Kauffman Foundation
- Walters Johnson Foundation
- The Joyce Foundation
- Lumina Foundation for Education
- MetLife Foundation
- State Farm Companies Foundation
- Washington Mutual Foundation

#### **Federal Funds Information for States**

Federal Funds Information for States (FFIS) is a joint subscription service of the National Governors Association and the National Conference of State Legislatures. FFIS tracks and reports on the fiscal impact of federal funding and policies on state budgets and programs. The organization maintains a database of federal grant-in-aid programs (Federal. *FFIS Services*).

#### Kentucky Long-Term Policy Research Center

In 1992, the Kentucky General Assembly established the Kentucky Long-Term Policy Research Center under the aegis of the Legislative Research Commission to

serve as a catalyst to change the way decisions are made in government by providing decision makers a broader context in which to make decisions, taking into consideration the long-term implications of policy, critical trends, and emerging issues which may have a significant impact on the state (KRS 007B).

#### National Center for Higher Education Management Systems

The National Center for Higher Education Management Systems (NCHEMS), established in 1969, is a private, nonprofit organization whose mission is "to improve strategic decision making in higher education for states and institutions in the United States and abroad." With project support from multiple organizations, such as the Ford Foundation, NCHEMS offers research, consulting, development projects, a higher education database, publications, a membership program, and training. Together with the State Higher Education Executive Officers Association and the Western Interstate Commission for Higher Education, NCHEMS formed the State Higher Education Policy Center (Natl. Ctr. for Higher. *About*).

## National Center for Public Policy and Higher Education

Established in 1998, the National Center for Public Policy and Higher Education is a nonprofit organization whose mission is to promote policies that enhance postsecondary education and training opportunities. The center conducts research and policy analyses on opportunities and achievements in higher education.

The center receives continuing support from a consortium of national foundations that includes the Pew Charitable Trusts, the Atlantic Philanthropies, and the Ford Foundation. The board of directors comprises decision makers across the political spectrum from government, business, and education. In 2000, the center began publishing *Measuring Up*, a biennial report card for each state and the nation, whose purpose is "to provide the public and policy makers with information to assess and improve postsecondary education in each state" (Natl. Ctr. for Public Policy. *About.*)

#### **National Education Association**

The National Education Association (NEA) was founded in 1857 to "elevate the character and advance the interests of the profession of teaching and to promote the cause of education in the United States." With 3.2 million members, NEA calls itself the nation's largest professional employee organization. Anyone who works for a public school district, a college or university, or any other public institution devoted primarily to education may join. NEA describes its affiliates in more than 14,000 communities as working to advance public education at every level of education, from preschool to university graduate programs, through such activities as raising funds for scholarships and conducting professional workshops. Activities of the national office and state affiliates include lobbying legislators for education resources, campaigning for professional standards, and filing legal actions to protect academic freedom and rights of school employees (Natl. Ed. Assoc. *About*).

#### National Institute for Early Education Research

The National Institute for Early Education Research (NIEER) supports early childhood education initiatives by providing information based on research. Its goal is "to produce and communicate the knowledge base required to ensure that every American child can receive a good education at ages three and four." The institute offers independent research-based advice and technical

assistance to policy makers, journalists, researchers, and educators. NIEER was established in 2002 at Rutgers University with a grant from the Pew Charitable Trusts. Past and current supporters include the Carnegie Corporation, the Fund for New Jersey, Geraldine R. Dodge Foundation, the David and Lucile Packard Foundation, the Prudential Foundation, the Schumann Fund for New Jersey, Smith Richardson Foundation, Tulsa Community Foundation, and the U.S. Department of Education's Office of Educational Research and Improvement (Natl. Institute. *About*).

#### U.S. Department of Commerce, Census Bureau

Throughout the past two centuries since the first national census in 1790, an increasing demand for information has driven steady expansion of the Census Bureau. Today, in addition to conducting the decennial census, the Census Bureau is engaged in a wide variety of data collection and analysis activities, including the annual American Community Survey of almost 1.5 million randomly selected residents and the Economic Census of businesses every 5 years (U.S. Dept. of Commerce. Census. *Economic* and "American Community").

The U.S. Census Bureau operates as a data collection agent for National Center for Education Statistics, using standardized forms, definitions, and instructions designed by NCES for comparability among states. For example, NCES commissions the U.S. Census Bureau to administer the School District Finance Survey at the time of its Annual Survey of Local Governments (U.S. Dept. of Ed. Inst. Natl. *Common*).

#### U.S. Department of Education, National Center for Education Statistics

The National Center for Education Statistics (NCES) is the primary federal entity for collecting, analyzing, and reporting data on education in the United States as well as monitoring and reporting on education in other nations (PL 103-382, 20 U.S.C 9003; U.S. Dept. of Ed. Inst. Natl. *Overview of Public*). NCES is a branch of the Institute of Education Sciences within the U.S. Department of Education.

NCES often collaborates with other federal agencies, such as the Census Bureau and the Bureau of Justice Statistics. The U.S. Census Bureau operates as a data collection agent for NCES, using standardized forms, definitions, and instructions designed by NCES to enhance the comparability of information among states. For example, NCES commissions the U.S. Census Bureau to administer the School District Finance Survey at the time of its Annual Survey of Local Governments.

The Common Core of Data (CCD) is the Department of Education's primary database on all public elementary and secondary schools, districts, students, and staff. Fiscal and nonfiscal data are comparable across all states and updated with five annual surveys.<sup>12</sup> The database includes such measures as pupil-teacher ratios, expenditures per pupil, student ethnicity, graduation rates,

<sup>&</sup>lt;sup>12</sup> Nonfiscal data are collected by the State Nonfiscal Survey of Public Elementary/Secondary Education, the Local Education Agency Universe Survey, and the Public Elementary/Secondary School Universe Survey. Fiscal data are collected by the School District Finance Survey and the National Public Education Financial Survey (U.S. Dept. of Ed. National Center. *Overview* 31).

and counts of students receiving special education or free lunch. Data collection for the CCD began with the 1981-1982 school year and was most recently reauthorized by the Education Sciences Reform Act of 2002 (PL 107-279, 20 U.S.C 9543; U.S. Dept. of Ed. Inst. Natl. *Common*).

Because of their authoritative positions and rigorous follow-up processes, NCES and the Census Bureau attain higher response rates than most surveys. Nevertheless, not all states collect and report all the data required. If information is missing for a relatively small number of schools or districts, NCES estimates those data. NCES also adjusts some values to improve comparability across states (U.S. Dept. of Ed. Inst. Natl. *Overview of Public*). For this reason, data reported in NCES publications will not always match states' reports.

# Appendix C

## **State Abbreviations and Names**

## In Order of Abbreviations

AK	Alaska	KY	Kentucky	NY	New York
AL	Alabama	LA	Louisiana	OH	Ohio
AR	Arkansas	MA	Massachusetts	OK	Oklahoma
AZ	Arizona	MD	Maryland	OR	Oregon
CA	California	ME	Maine	PA	Pennsylvania
СО	Colorado	MI	Michigan	RI	Rhode Island
СТ	Connecticut	MN	Minnesota	SC	South Carolina
DC	District of Columbia	MO	Missouri	SD	South Dakota
DE	Delaware	MS	Mississippi	TN	Tennessee
FL	Florida	MT	Montana	TX	Texas
GA	Georgia	NC	North Carolina	UT	Utah
HI	Hawaii	ND	North Dakota	VA	Virginia
IA	Iowa	NE	Nebraska	VT	Vermont
ID	Idaho	NH	New Hampshire	WA	Washington
IL	Illinois	NJ	New Jersey	WI	Wisconsin
IN	Indiana	NM	New Mexico	WV	West Virginia
KS	Kansas	NV	Nevada	WY	Wyoming

## In Order of State Names

AL	Alabama	KY	Kentucky	ND	North Dakota
AK	Alaska	LA	Louisiana	OH	Ohio
AZ	Arizona	ME	Maine	OK	Oklahoma
AR	Arkansas	MD	Maryland	OR	Oregon
CA	California	MA	Massachusetts	PA	Pennsylvania
СО	Colorado	MI	Michigan	RI	Rhode Island
СТ	Connecticut	MN	Minnesota	SC	South Carolina
DE	Delaware	MS	Mississippi	SD	South Dakota
DC	District of Columbia	MO	Missouri	TN	Tennessee
FL	Florida	MT	Montana	TX	Texas
GA	Georgia	NE	Nebraska	UT	Utah
HI	Hawaii	NV	Nevada	VT	Vermont
ID	Idaho	NH	New Hampshire	VA	Virginia
IL	Illinois	NJ	New Jersey	WA	Washington
IN	Indiana	NM	New Mexico	WV	West Virginia
IA	Iowa	NY	New York	WI	Wisconsin
KS	Kansas	NC	North Carolina	WY	Wyoming

# Appendix D

## National Assessment Of Educational Progress (NAEP) Achievement Level Definitions

Table D.1
NAEP Achievement Level Definitions for Grades 4 and 8 Reading, 2007

Achievement Level	Grade 4	Grade 8
Advanced	Generalize about topics in the reading selection and demonstrate awareness of how authors compose and use literary devices; judge texts critically and, in general, give thorough answers that indicate careful thought	Describe the more abstract themes and ideas of the overall text; analyze both meaning and form, supporting analyses explicitly with examples from the text, and extend text information by relating it to their experiences and to world events; respond thoroughly, thoughtfully, and extensively
Proficient	Demonstrate an overall understanding of the text, providing inferential as well as literal information; extend the ideas in the text by making inferences, drawing conclusions, and making connections to their own experiences; make clear the connections between the text and what the student infers	Show an overall understanding of the text, including inferential as well as literal information; extend the ideas in the text by making clear inferences from it, by drawing conclusions, and by making connections to their own experiences—including other reading experiences; identify some of the devices authors use in composing text
Basic	Demonstrate an understanding of the overall meaning; make relatively obvious connections between the text and their own experiences; extend the ideas in the text by making simple inferences	Demonstrate a literal understanding of what they read and make some interpretations; identify specific aspects of the text that reflect the overall meaning; extend the ideas in the text by making simple inferences; recognize and relate interpretations and connections among ideas in the text to personal experience; draw conclusions based on the text

Note: "Below Basic" is not defined.

Source: Staff compilation based on U.S. Dept. of Ed. Natl. Assessment. Reading 24-26.

Table D.2	
NAEP Achievement Level Definitions for Grades 4 and 8 Writing, 2	2007

Achievement Level	Grade 4	Grade 8
Advanced	Produce an effective, well-developed, clearly organized response that shows a clear understanding of the assigned task; use such techniques as consistency in topic or theme, sequencing, and a clearly marked beginning and ending; address the intended audience with precise and varied language, including details and elaboration that support and develop the main idea of the piece; show signs of analytical, evaluative, or creative thinking; use grammar, spelling, and capitalization accurately enough to communicate clearly, with mistakes made so few and so minor that a reader can easily skim over them	Produce an effective and fully developed response that shows a clear understanding of the assigned task and intended audience; show some analytical, evaluative, or creative thinking; demonstrate precise word choice and varied sentence structure; include details and elaboration that support and develop the main idea; use such strategies as analogies, illustrations, examples, anecdotes, or figurative language to clarify a point; organize clearly and consistently, with few errors in grammar, spelling, punctuation, capitalization, and sentence structure; demonstrate good control of these elements and possibly use them for stylistic effect
Proficient	Produce an organized response that shows an understanding of the assigned task; include details that support and develop the main idea of the piece; show awareness of the intended audience through form, content, and language; use grammar, spelling, and capitalization accurately enough to communicate to a reader without mistakes getting in the way of meaning	Produce an effective response that shows an understanding of the assigned task and intended audience; make organized use of such techniques as sequencing or a clearly marked beginning and ending; make use of details and some elaboration to support and develop the main idea of the piece; include precise language and some variety in sentence structure and show analytical, evaluative, or creative thinking; use grammar, spelling, punctuation, and capitalization accurately enough to communicate to a reader without errors obscuring the meaning
Basic	Produce a somewhat organized response that shows a general grasp of the assigned task; include some supporting details; use grammar, spelling, and capitalization accurately enough to communicate to a reader, although there may be mistakes that get in the way of meaning	Produce an effective response that shows a general understanding of the assigned task and intended audience; include supporting details in an organized way; use grammar, spelling, punctuation, and capitalization accurately enough to communicate to a reader, although there may be mistakes that get in the way of meaning

Notes: These achievements levels are for first drafts, not final or polished student writing, that are generated within limited time constraints in large-scale assessment environment. "Below Basic" is not defined. Source: Staff compilation based on U.S. Dept. of Ed. Natl. Assessment. *Writing Framework* 59-61.

Table D.3
NAEP Achievement Level Definitions for Grades 4 and 8 Mathematics, 200'

Achievement Level	Grade 4	Grade 8
Advanced	Solve complex nonroutine real-world problems in all areas; display mastery in the use of four-function calculators, rulers, and geometric shapes; draw logical conclusions and justify answers and solution processes by explaining why, as well as how, they were achieved; go beyond the obvious in interpretations and be able to communicate clearly and concisely	Probe examples and counterexamples in order to shape generalizations, and develop models from these; use number sense and geometric awareness to consider the reasonableness of an answer; use abstract thinking to create unique problem-solving techniques and explain the reasoning processes underlying conclusions
Proficient	Use whole numbers to estimate, compute, and determine whether results are reasonable; have a conceptual understanding of fractions and decimals; solve real-world problems in all areas; use four-function calculators, rulers, and geometric shapes appropriately; employ such problem- solving strategies as identifying and using appropriate information; organize and present written solutions with supporting information and explanations of how they were achieved	Conjecture, defend ideas, and give supporting examples; understand connections among fractions, percents, decimals, and other mathematical topics such as algebra and functions; thoroughly understand basic-level arithmetic operations for problem solving in practical situations; be familiar with quantity and spatial relationships in problem solving and reasoning; convey underlying reasoning skills beyond the level of arithmetic; compare and contrast mathematical ideas and generate examples; make inferences from data and graphs, apply properties of informal geometry, and accurately use the tools of technology; understand the process of gathering and organizing data; calculate, evaluate, and communicate results within the domain of statistics and probability
Basic	Estimate and use basic facts to perform simple computations with whole numbers; show some understanding of fractions and decimals; solve some simple real-world problems in all areas; use—though not always accurately— four-function calculators, rulers, and geometric shapes; have minimal written responses presented without supporting information	Complete problems correctly with the help of structural prompts such as diagrams, charts, and graphs; solve problems in all areas through the appropriate selection and use of strategies and technological tools, including calculators, computers, and geometric shapes; use fundamental algebraic and informal geometric concepts in problem solving; determine which of the available data are necessary and sufficient for correct solutions, and use them in problem solving; have some limited skills in communicating mathematically

Note: "Below Basic" is not defined.

Source: Staff compilation based on U.S. Dept. of Ed. Natl. Assessment. Mathematics 53-55.

	Table D.4	
NAEP	Achievement Level Definitions for Grade 4 Science,	2005

Achievement Level	Grade 4
Advanced	Demonstrate solid understanding of the Earth, physical, and life sciences and apply this understanding to practical situations at grade-appropriate levels; perform and critique simple investigations and apply fundamental concepts to practical applications; combine information, data, and knowledge from one or more sciences to reach a conclusion or make a valid prediction; recognize, design, and explain simple experimental procedures; recognize nonrenewable sources of energy; know that light and sound travel at different speeds; understand some principles of ecology; compare and contrast life cycles of various common organisms; show a developmental awareness of technology's benefits and challenges
Proficient	Demonstrate grade-level knowledge and reasoning required for understanding the Earth, physical, and life sciences; formulate solutions to familiar problems; provide an explanation of day and night when given a diagram; recognize major features of the Earth's surface and the impact of natural forces; recognize water in its various forms in the water cycle and suggest ways to conserve it; recognize that various materials possess different properties that make them useful; explain how structure and function help living things survive; show a beginning awareness of technology's benefits and challenges and recognize some human effects on the environment; make straightforward predictions and justify these positions
Basic	Demonstrate some grade-level knowledge and reasoning required for understanding the Earth, physical, and life sciences; show a beginning understanding of classification, simple relationships, and energy; follow simple procedures, manipulate simple materials, make observations, and record data; read simple graphs and diagrams and draw reasonable but limited conclusions based on the data provided; recognize appropriate experimental designs, although unable to justify decisions; identify seasons through diagrams; distinguish between day and night through diagrams; place the position of the Earth, Sun, and planets through diagrams; recognize major energy sources and simple energy change; understand the relationship between sound and vibrations; identify organisms by physical characteristics and group organisms with similar physical features; describe simple relationships among structure, function, habitat, life cycles, and different organisms

Notes: "Below Basic" is not defined.

Source: Staff compilation based on U.S. Dept. of Ed. Natl. Assessment. Science 47-48.

Table D.5	
NAEP Achievement Level Definitions for Grade 8 Science,	2005

Achievement Level	Grade 8
Advanced	Demonstrate solid grade-level understanding of the Earth, physical, and life sciences, and apply this understanding in practical situations; perform and critique the design of investigations, relate scientific concepts to each other, explain their reasoning, and discuss the impact of human activities on the environment; explain scientific results; show a modest understanding of scale, and design a controlled experiment; understand models as representations of natural systems, and describe energy transfer in living and nonliving systems; understand that such physical clues as fossils and geological formations are indications that the Earth has not always been the same and that the present is a key to understanding the past; show solid knowledge of forces and motions within the solar system and an emerging understanding of atmospheric pressure; recognize a wide range of physical and chemical properties of matter and some of their interactions, and understand some of the properties of light and sound; infer relationships between structure and function; know the differences between plant and animal cells and apply knowledge of food as a source of energy to a practical situation; explain the impact of human activities on the environment and the economy
Proficient	Demonstrate much of the grade-level knowledge and reasoning essential for understanding the Earth, physical, and life sciences; show awareness of environmental issues, especially energy and pollution; create, interpret, and make predictions from charts, diagrams, and graphs based on information provided or from own investigations; design an experiment and show an emerging understanding of variables and controls; read and interpret geographic and topographic maps; shown an emerging ability to use and understand models, partially formulate explanations of scientific phenomena, and design plans to solve problems; begin to identify forms of energy and describe the role of energy transformations in living and nonliving systems; show knowledge of organization, gravity, and motion within the solar system and identify some factors that shape the Earth's surface; show some understanding of the properties of materials and of the particulate nature of matter, especially the effect of temperature on states of matter; know that light and sound travel at different speeds, and apply knowledge of force, speed, and motion; demonstrate a developmental understanding of the flow of energy from the Sun through living systems, especially plants; know that organisms reproduce and inherit characteristics; list some effects of air and water pollution; know the environmental and economic advantages and disadvantages of different energy sources
Basic	Demonstrate some of the grade-level knowledge and reasoning required for understanding the Earth, physical, and life sciences; show a beginning understanding of cause-and-effect relationships; observe, measure, collect, record, and compute data from investigations; read simple graphs and tables and make simple data comparisons; follow directions and use basic science equipment to perform simple experiments; show an emerging ability to design experiments; show some awareness of causal relationships; recognize the position of planets and their movement around the Sun and know basic weather-related phenomena, explain changes in position and motion such as the movement of a truck in relation to that of a car; show an emerging understanding of the interrelationships among plants, animals, and the environment

Notes: "Below Basic" is not defined.

Source: Staff compilation based on U.S. Dept. of Ed. Natl. Assessment. Science 49-51.